Characterisation of Mucosal Associated Invariant T cells function and metabolism in health and obesity





Marta Pisarska

B.Sc. (Hons)

Thesis submitted to National University of Ireland Maynooth in fulfilment of the

requirements for the degree of Doctor of Philosophy

Under the supervision of Dr Andrew Hogan

National University of Ireland Maynooth and National Children Research Centre

School of Biology, National University of Ireland Maynooth

Head of School: Paul Moynagh

February 2021

Table of contents

List of Figures	v
List of Tables	ix
Acknowledgements	x
Publications arising from this thesis	xii
List of Abbreviations	xiii
Abstract	xvii

1. General introduction	1
1.1. Obesity	2
1.1.1. Obesity Overview	2
1.1.2. Obesity associated co-morbidities	4
1.2. Immune system in health and obesity	8
1.2.1. Immune system and inflammation	8
1.2.2. Adipose tissue associated inflammation and immune cell dysregulation	on in
obesity	10
1.3. Mucosal Associated Invariant T cells	14
1.3.1. Mucosal Associated Invariant T cells overview	14
1.3.2. MAIT cells in infection	17
1.3.3. MAIT cells in inflammatory and autoimmune diseases	19
1.3.4. MAIT cells in diabetes and obesity	22
1.4. Immunometabolism	25

2. Materials and methods	.31
2.1. Ethics	.32
2.2. Study design	.32
2.3. Isolation of human peripheral blood mononuclear cells	.33
2.4. Cell enumeration	.34
2.5. MAIT cell proliferation and expansion	.34
2.5.1. Cell Trace Violet Staining	.34
2.5.2. TCR only induced MAIT cell expansion & proliferation over 5 days	.35
2.5.3. TCR and IL-2 induced MAIT cell expansion & proliferation over 5 days	.35

2.5.4. TCR & IL-2 induced MAIT cell expansion over 7 days	37
2.5.5. TCR & IL-2 induced MAIT cell expansion – 28 days	37
2.5.6. MAIT cell expansion for Seahorse, ELISA and real time RT-qPCR	37
2.6. Preparation of expanded MAIT cells for experiments	38
2.6.1. MAIT cell magnetic labelling	38
2.6.2. MAIT magnetic separation	38
2.7. MAIT cell molecular analysis	39
2.7.1. mRNA expression experimental set up	39
2.7.2. mRNA isolation	40
2.7.3. mRNA quantification	41
2.7.4. cDNA synthesis	41
2.7.5. Realtime RT-qPCR	42
2.8. ELISA	43
2.8.1. ELISA Experimental set up	43
2.8.2. ELISA testing	44
2.9. Seahorse	45
2.9.1. Seahorse experimental set up	45
2.9.2. Seahorse protocol	45
2.10. Flow Cytometry	47
2.10.1. MAIT cell glycolytic metabolism	47
2.10.2. MAIT cell amino acid requirements for activation	47
2.10.3. Expression of CD25	48
2.10.4. Antibodies and panels used in this thesis	48
2.10.5. Extracellular and intracellular staining in tubes	49
2.10.6. Extracellular and intracellular staining in plates	50
2.10.7. Gating of MAIT cells	51
2.10.8. Viability staining	52
2.10.9. Kynurenine Assay	52

3. Results- Functional Characterisation of MAIT cells	54
3.1. Introduction	55
3.1.1. MAIT cells	55
3.1.2. MAIT cells in obesity	60
3.2. Specific aims of this chapter	62

3.3. Results
3.3.1. MAIT cell identification
3.3.2. People with obesity display lower MAIT cell frequencies in the peripheral
blood65
3.3.3. TCR stimulation induces MAIT cell proliferation67
3.3.4. TCR triggering upregulates expression of IL-2 receptor on MAIT cell
surface69
3.3.5. IL-2 supplementation enhances MAIT cell proliferation and
expansion72
3.3.6. MAIT cells from people with obesity display limited expansion capacity in
response to 5-ARU-MG and IL-276
3.3.7. MAIT cells produce IFNy80
3.3.8. MAIT cells produce IL-17A82
3.3.9. MAIT cell produce less IFN γ and more IL-17 cytokines in obesity84
3.3.10. MAIT cells produce an expanded repertoire of cytokines87
3.3.11. Individuals with obesity display no significant difference in the expression
of other immune cytokines89
3.4. Discussion

4.3.5. MAIT cells express amino acid transporters LAT1- LAT4114
4.3.6. MAIT cells in obesity express lower level of LAT1, but show no significant
difference in transcription of other amino acid transporters116
4.3.7. Stimulation of MAIT cells increases their size119
4.3.8. MAIT cells upregulate glycolysis upon stimulation121
4.3.9. Stimulation of MAIT cells modulates expression of enzymes of
glycolysis123
4.3.10. MAIT cells from people with obesity display impaired glycolysis125
4.3.11. MAIT cells actively use LAT1 amino acid transporters128
4.3.12. MAIT cells from people with obesity display defective LAT1 mediated
transport130
4.3.13. MAIT cells require amino acid influx via LAT1 to activate mTORC1 and
glycolysis132
4.4. Discussion

5. Results- Metabolic requirements for MAIT cell function140
5.1. Introduction141
5.1.1. Immunometabolism and cell function141
5.2. Specific aims of this chapter144
5.3. Results145
5.3.1. MAIT cells use glucose to support proliferation145
5.3.2. MAIT cells require glycolysis for proliferation149
5.3.3. MAIT cells require an influx of amino acids via LAT1 for proliferation150
5.3.4. MAIT cells require mTORC1 for proliferation154
5.3.5. MAIT cells require glutamine metabolism and OxPhos for
proliferation156
5.3.6. Glycolysis is required for IFNy production161
5.3.7. Amino acid transport is required for IFNy production163
5.3.8. mTORC1 is required for IFNγ production166
5.3.9. Glutamine metabolism supports IFNy production166
5.4. Discussion
6. General discussion174

6.1. Discussion1	7	Ę	5	
------------------	---	---	---	--

7. Bibliography	
-----------------	--

Appendix	232
----------	-----

List of figures

Figure 1.1 Obesity associated co-morbidities5
Figure 1.2 Response of the immune system to disruption in homeostasis8
Figure 1.3 Cells of the immune system9
Figure 1.4 Changes in adipose tissue in obesity11
Figure 1.5 MAIT cells in inflammatory and autoimmune diseases20
Figure 2.1 Isolation of peripheral blood mononuclear cells
Figure 2.2 PBMC counting
Figure 2.3 Plan of MAIT cell proliferation experiment
Figure 2.4 Diagram showing experimental plan for analysis of MAIT cell mRNA
expression40
Figure 2.5 Isolation and quantification of RNA41
Figure 2.6 ELISA steps45
Figure 2.7 MAIT cell identification using flow cytometric antibodies51
Figure 2.8 MAIT cell identification using flow cytometric antibodies and MAIT cell
tetramer52
Figure 3.1 MAIT cell activation57
Figure 3.2 Phenotype of MAIT cells59
Figure 3.3 MAIT cell identification using flow cytometry64
Figure 3.4 MAIT cell frequencies in control subjects and individuals with
obesity66
Figure 3.5 MAIT cells proliferate in response to 5-ARU-MG68

Figure 3.6 TCR stimulation of MAIT cells increases IL-2 receptor (CD25)
expression70
Figure 3.7 TCR triggering in combination with IL-12 and IL-18 cytokine stimulation
enhances expression of IL-2 receptor (CD25) by MAIT cells71
Figure 3.8 MAIT cells expand in response to 5-ARU-MG and IL-2 over 28
days73-74
Figure 3.9 MAIT cells proliferate in response to 5-ARU-MG and IL-275
Figure 3.10 MAIT cells from people with obesity show impaired expansion in
response to 5-ARU-MG and IL-277-78
Figure 3.11 MAIT cells from people with obesity show similar proliferation profile
on in response to 5-ARU-MG and IL-279
Figure 3.12 MAIT cell produce IFNγ upon stimulation81
Figure 3.13 MAIT cell produce IL-17A upon stimulation83
Figure 3.14 MAIT cell production of IFNy upon stimulation is reduced in
obesity85
Figure 3.15 MAIT cells from individuals with obesity, display enhanced IL-17
protein production, but no difference in the transcription of IL17A & RORC upon
stimulation in comparison to the controls86
Figure 3.16 Stimulation of MAIT cells induces the production of IL-17F, IL-22 and
IL-32 cytokines
Figure 3.17 Stimulation of MAIT cells reveals no difference in <i>IL17F</i> and <i>IL22</i>
mRNA transcription by the cells from healthy and obese individuals but shows
impairment in <i>IL32</i> transcription90
Figure 3.18 Summary of MAIT cell function in control and people with obesity94
Figure 4.1 Glycolysis98
Figure 4.2 Regulation of mTORC1 activity100

Figure 4.3 MAIT cells express mRNA and protein for glucose transporter GLUT1
and mRNA for GLUT3&4107
Figure 4.4 MAIT cell from people with obesity shown no difference in mRNA
expression for GLUT glucose transporters109
Figure 4.5 MAIT cells express mRNA for glutamine transporters ASCT2 and
SNAT1&2111
Figure 4.6 MAIT cell from donors with obesity shown no difference in mRNA
expression for SNAT glutamine transporters113
Figure 4.7 MAIT cells express mRNA and protein for CD98 subunit of LAT1 amino
acid transporter and mRNA for LAT2-4115
Figure 4.8 MAIT cell from obese donors show lower expression of LAT1
transporter117
Figure 4.9 MAIT cells from obese donors express similar levels of mRNA
transcription for LAT2-4 to the control samples118
Figure 4.10 MAIT size increases upon extended stimulation
Figure 4.11 MAIT cells upregulate glycolysis upon stimulation122
Figure 4.12 MAIT cells upregulate expression HK2 and LDHA, but not PKM2
protein expression upon stimulation124
Figure 4.13 MAIT cells from people with obesity don't upregulate glycolysis upon
stimulation126
Figure 4.14 MAIT cells from people with obesity display a non-significant
alteration mRNA expression of HK2 and PKM enzymes
Figure 4.15 MAIT cells import kynurenine via LAT1 amino acid transporter129
Figure 4.16 Obese MAIT cells have impaired uptake of kynurenine via LAT1
amino acid transporter131
Figure 4.17 Amino acid transport via LAT1 is important for mTOR activation133

Figure 4.18 Amino acid transport via LAT1 is necessary for upregulation of
glycolysis134
Figure 4.19 Summary of findings of chapter four139
Figure 5.1 MAIT cell proliferation is restricted by glucose
availability146
Figure 5.2 Restriction of glucose negatively impacts CD25 surface expression by
MAIT cells147
Figure 5.3 MAIT cell expansion is restricted by substitution of glucose with
galactose148
Figure 5.4 MAIT cells require glycolysis to proliferate in response to 5-ARU-MG
and IL-2151
Figure 5.5 MAIT cells require glycolysis to express IL-2 receptor on cell
surface152
Figure 5.6 MAIT cells require amino acid influx to proliferate in response to 5-
ARU-MG and IL-2153
Figure 5.7 MAIT cells require mTORC1 activation to proliferate in response to 5-
ARU-MG and IL-2155
Figure 5.8 MAIT cells require active ATP synthase to proliferate in response to 5-
ARU-MG and IL-2158
Figure 5.9 MAIT cells require oxidative phosphorylation complex III to proliferate
in response to 5-ARU-MG and IL-2159
Figure 5.10 MAIT cells require glutamine metabolism to proliferate in response to
5-ARU-MG and IL-2160
Figure 5.11 Blocking glycolysis with 2DG reduces IFNy production by MAIT
cells

Figure 5.12 Blocking amino acid transport by BCH reduces IFNy production by
MAIT cells164
Figure 5.13 Blocking amino acid transport using BCH doesn't affect the
transcription of HK2 or PKM by MAIT cells at early stage during
activation165
Figure 5.14 Inhibition of mTORC1 with rapamycin reduces IFN γ production by
MAIT cells167
Figure 5.15 Inhibition of glutaminase with CB-839 reduces IFNy production by
MAIT cells168
Figure 5.16 Summary of results of chapter five173

List of tables

Table 2.1 Sequences of forward and reverse primers	.43
Table 2.2. Seahorse inhibitors injection	.47
Table 2.3. Flow cytometry panels for studying of MAIT cells	.48
Table 2.4. Clones of antibodies used in this thesis	.49

Acknowledgements

I'd like to take this opportunity to thank people who have greatly contributed to this project. There is a saying "It takes a village to raise a child", but my God, people wouldn't believe how many people it takes to complete one PhD thesis.

Sincere thanks to Dr Andrew Hogan for his supervision and guidance throughout this PhD. Professor Donal O'Shea for his valuable insights and constructive criticism. The assessor and advisor of this PhD, Dr Karen English and Professor Paul Moynagh for reviewing this project, offering their great knowledge and pointing out the areas for improvement. All medical and administrative staff involved in the recruitment of lean individuals and people with obesity and collection of their blood samples for this project. Thank you to all the patients who voluntarily agreed to take part in this project.

My colleagues from the Obesity Immunology research group, who provided the knowledge and training throughout this project. In particular I'd like to thank Ms Nicole Wood, who has spent endless hours training me in multiple laboratory techniques, discussing details of the experimental set ups and data analysis. She was always available to answer any of my questions at any time of the day (or night) and provided me with emotional (and chocolate) support when I needed it the most.

Dr Laura Tobin, for everything really. Thanks to Laura, I became fascinated by research during our first encounter in 2014. Laura has taught me the very basics of research methods, which were enhanced and polished in the years ahead. She

Х

is my gold standard I always go back to. Whenever I am stuck, I think – what would Laura do? Thank you for helping me to find my feet when I was starting this project and for always being there to support me in so many ways! I will be forever grateful.

I'd like to thank my family, especially my parents and grandparents. Without their personal sacrifices I wouldn't be where I am today. Being raised in such a safe, loving and supporting environment was a great privilege, not experienced by many. Thank you for all the hard decisions you had to make, to provide me with the best possible future and making me believe that I can achieve anything I want if I work hard and put my heart into it.

Thank you to my dear friends, who were always there to talk and cheer up when experiments just weren't coming together.

Last but not least I'd like to thank my partner. I have no words to describe his support for the past three and a half years. Thank you for never stopping to believe in me, pushing me forward when I thought I cannot go any further. Thank you for listening, when I need to talk out my frustrations and for about two million and fifty-four hugs I needed to get through these years. If it wasn't for you, this thesis wouldn't have been completed.

Thank you to everyone who I met during this project, who offered their help, guidance and support in multiple different ways.

Publications arising from this thesis

Obesity reduces mTORC1 activity in mucosal-associated invariant T cells, driving defective metabolic and functional responses

Aisling O'Brien*, Roisin M. Loftus*, **Marta M. Pisarska***, Laura M. Tobin, Ronan Bergin, Nicole A.W. Wood, Cathriona Foley, Arimin Mat, Frances C. Tinley, Ciaran Bannan, Gary Sommerville, Natacha Veerapen, Gurdyal S. Bersa, Linda V Sinclair, Paul N. Moynagh, Lydia Lynch, David K. Finlay, Donal O'Shea and Andrew E. Hogan. The Journal of Immunology. 2019: 202(12):3404-3411. *Joint first authorship

Interleukin-17 producing mucosal-associated invariant T cells- emerging players in chronic inflammatory diseases?

Marta M. Pisarska, Margaret R. Dunne, Donal O'Shea and Andrew E. Hogan. European Journal of Immunology. 2020; 50(8):1098-1108. Review.

Manuscripts in preparation

Glycolysis supports proliferation and expansion of mucosal-associated invariant T cells

Marta M. Pisarska, Nicole A.W. Wood, Nidhi Kedia-Mehta and Andrew E. Hogan.

Manuscript in preparation

List of Abbreviations

2DG	2-Deoxy-D-Glucose
5-ARU	5-Amino-6-D-Ribityaminouracil
5-OE-RU	5-(20oxoethyludeneamino)-6-D-Ribitylaminouracil
5-OP-RU	5-(2-Oxopropulideneamino)-6-D-Ribitylaminouracil
6-FP	6-Formyl Pterin
Acetyl Co-A	Acetyl Coenzyme-A
АМРК	AMP-activated protein kinase
AP-1	Activator Protein-1
APC	Antigen Presenting Cell
ASCT2	Alanine, Serine, Cysteine Transporter 2
AT	Adipose Tissue
АТМ	Adipose Tissue Macrophages
ATP	Adenosine Triphosphate
ВСН	2-amino-2-norbornanecarboxylic acid
Blimp-1	B-Lymphocyte-Induced Maturation Protein -1
BMI	Body Max Index
BSA	Bovine Serum Albumin
C/EBΡδ	CAAT/Enhancer-Binding Protein Gamma
CCL	CC Chemokine Ligand
CCR	CC Chemokine Receptor
CD	Cluster of Differentiation
CFSE	Carboxyfluorescein Succinimidyl Ester
сти	Cell Trace Violet
CVD	Cardiovascular Disease

DEPTOR DEP Domain Containing mTOR Interacting Protein

DMSO	Dimethyl Sulfoxide
EAE	Experimental Autoimmune Encephalomyelitis
ECAR	Extracellular Acidification Rate
ELISA	Enzyme-linked immunosorbent assay
Eomes	Eomesodermin
ЕТС	Electron Transport Chain
FCS	Foetal Calf Serum
FFA	Free Fatty Acids
FMO	Fluorescence minus one
FSC	Forward Scatter
FTO	Fat Mass and Obesity Associated (gene)
FVD	Fixable Viability Dye
GAPDH	Glyceraldehyde 3-Phosphate Dehydrogenase
GLS	Glutaminase
GLUT	Glucose Transporter
HBSS	Hank's Balanced Salt Solution
нсv	Hepatitis C Virus
HFD	High Fat Diet
HIF-1α	Hypoxia Inducible Factor-1α
HIV	Human Immunodeficiency Virus
нкіі	Hexokinase II
HLA-DR	Human Leukocyte Antigen- DR
IBD	Inflammatory Bowel Disease
IFNγ/α/β	Interferon $\gamma \alpha / \beta$
IGRP	Islet-Specific Glucose-6-Phosphate Catalytic Subunit-Related
	Protein

IL	Interleukin
ILC2	Innate Lymphoid Cell 2
iNKT	Invariant natural killer T cell
JNK	C-Jun N-Terminal Kinase
LAT	L-Type Amino Acid Transporter
LDHA	Lactate dehydrogenase A
MAIT	Mucosal Associated Invariant T cell
MCP-1	Monocyte Chemoattractant Protein-1
MG	Methylglyoxal
МНС	Major Histocompatibility Complex
mLST8	Mammalian Lethal with Sec13 Protein 8
MR1	MHC class I related protein 1
mROS	Mitochondrial Reactive Oxygen Species
MS	Multiple Sclerosis
mTOR	Mammalian Target of Rapamycin
mTORC1	Mammalian Target of Rapamycin Complex 1
mTORC2	Mammalian Target of Rapamycin Complex 2
NFAT	Nuclear Factor of Activated T Cells
ΝϜκΒ	Nuclear Factor-Kappa Light Chain Enhanced of Activated B Cells
NK	Natural Killer (cells)
NOD	Non-Obese Diabetic
OxPhos	Oxidative Phosphorylation
PBMC	Peripheral Blood Mononuclear Cells
PBS	Phosphate Buffered Saline
PD-1	Programmed Cell Death Protein-1
PEP	Phosphoenolpyruvate

PFA Paraformaldehyde PKM2 Pyruvate kinase 2 PLZF Promyelocytic Leukaemia Zinc Finger PPP Pentose Phosphate Pathway PRSA40 Proline-Rich AKT Substrate 40kda RA **Rheumatoid Arthritis** Rheb Ras Homolog Enriched in Brain RORyT RAR-Related Orphan Receptor Gamma yT RT-qPCR Real Time Quantitative Polymerase Chain Reaction SERCA Sarcoendoplasmic Reticulum Calcium ATPase SGLT Sodium-Glucose Linked Transporters siRNA Small Interfering RNA SLC Solute Carrier SLE Systemic Lupus Erythematosus SNAT Sodium Coupled Neutral Amino Acid Transporter T-bet T-Box Protein Expressed in T Cells T1DM Type 1 Diabetes Mellitus T2DM Type 2 Diabetes Mellitus TCR T Cell Receptor TNF-α **Tumour Necrosis Factor Alpha** TSC2 **Tuberous Sclerosis Complex 2** WAT White Adipose Tissue WHO World Health Organisation WΤ Wild Type α-KG α-Ketoglutarate

Abstract

Obesity has developed into a worldwide pandemic, affecting individuals regardless of their socio-economic status, gender or age. It is associated with development of multiple co-morbidities including type 2 diabetes mellitus, which have been shown to be underpinned by inflammation. Indeed, excessive adiposity has been shown to be associated with a large dysregulation of immune cell function, including altered phenotype of mucosal-associated invariant T (MAIT) cells. MAIT cells from people with obesity were shown to have enhanced IL-17 production, accompanied by impaired IFNy expression. As of now, very little is known about the control of MAIT cell function, therefore more research is required to understand their biology in order to identify the altered process, which leads to their dysfunction in obesity. Immunometabolism of MAIT cells became the focus of this thesis as previous data indicated it as a key biological process that controls immune cell function. Here we confirmed previously published data, showing that MAIT cells are potent producers of IFNy and IL-17 cytokines and that MAIT cells can proliferate and expand upon stimulation. In addition, we reported that MAIT cells enhance glycolysis upon activation, which in turn is controlled by the influx of amino acids into the cells via LAT1 amino acid transporters. Glycolytic metabolism was required for successful IFNy production as well as proliferation and expansion of MAIT cells. People with obesity have impaired IFNy expression and MAIT cell expansion that was accompanied by impaired glycolytic metabolism. Collectively our data indicates that MAIT cells effector functions are in part controlled by the intrinsic metabolic pathways including the glycolytic metabolism. Defect in glycolysis or glycolysis-associated pathways as observed in obesity, leads to their dysfunction and altered immune cells responses.

Chapter One

General Introduction

1.1 Obesity

1.1.1. Obesity Overview

Obesity is a chronic progressive disease, which has evolved into a global pandemic (Bray et al., 2017; WHO, 2020a). Between 1975 to 2016 the number of individuals living with obesity has almost tripled. As of 2016 it is estimated that 650 million adults are living with obesity worldwide (WHO, 2020a). Initially, obesity affected only high-income countries, but it has quickly spread to the middle-income and low-income countries with the number of cases of obesity having increased in the vast majority of countries worldwide between 1975 and 2014 (Collaboration(NCD-RisC), 2016).

Obesity is described as an excessive accumulation of adipose tissue (AT) which is disproportionate to the height of an individual (CDC, 2020). Diagnosis of the disease in adults is most commonly based on the calculation of Body Mass Index (BMI), due to the lack of more precise and freely available diagnostic tools (Rothman, 2008; Shah and Braverman, 2012; WHO, 2020a). BMI is calculated by dividing the weight of the individual in kilograms by their height in meters squared. This equation allows for a crude estimation of an individual's nutritional status, proportion of AT and classification into one of the following categories as set out by World Health Organisation (WHO): underweight BMI of <18.5 kg/m², normal weight as 18.5-24.9 kg/m², pre-obesity/overweight as 25-29.9 kg/m² and obesity as a BMI >30 kg/m² (WHO, 2020a). This equation is inherently flawed, as it doesn't take into consideration other factors, which would aid the assessment of the severity of obesity and its impact on the individual's health, such as localisation of the AT deposits, muscle density or sex (Bray et al., 2018; Rothman, 2008). BMI measurement can be supported by the measurement of the waist circumference, which correlates with the amount of visceral AT and increased risk of development of cardiovascular disease (CVD) risk, type 2 diabetes mellitus (T2DM) and overall mortality (Cerhan et al., 2014; Jeon et al., 2019; Mamtani et al., 2013; Nazare et al., 2015; Siren et al., 2012). Initially, the primary cause of obesity was identified as an energy imbalance, where calorie intake exceeded the calorie expenditure (WHO, 2020a). Since then, multiple studies have highlighted the complexity and heterogenicity of this disease, depicting how multiple factors affect disease development (González-Muniesa et al., 2017; Swinburn et al., 2011). Susceptibility of an individual to weight gain is largely

influenced by their environment, behaviour, socio-economic status and biological factors. Calorie intake is greatly affected by food availability and accessibility, as well as the marketing strategies employed by the food companies, which promote snacking and consumption of highly processed, calorie dense foods (González-Muniesa et al., 2017; Swinburn et al., 2011). On the other hand, calorie expenditure has been reduced as a result of improvement of the public transport systems and sedentary lifestyle, resulting in a much-decreased amount of time which is dedicated to physical activity (González-Muniesa et al., 2017; Swinburn et al., 2011). Weight gain is also affected by biological factors such as sleep deprivation, location of AT or genes (Goodarzi, 2018; Swinburn et al., 2011; Vishvanath and Gupta, 2019). In recent years, mutations in fat mass and obesity associated (FTO) gene and genes coding for proteins involved in the leptinmelanocortin pathway such as leptin, leptin receptor and melanocortin 4 receptor, have been identified in people with obesity (Pigeyre et al., 2016). The contribution of the genetic variation to the development of obesity yet remains to be fully elucidated.

The rate of obesity in Ireland is growing at an alarming rate. Pineda et al. have estimated that without appropriate interventions, by 2025 Ireland will become the country with the highest rate of obesity in Europe (Pineda et al., 2018). This is in line with an earlier study by Keaver *et al.*, which estimated that by 2030 as many as 89% of Irish men and 85% of Irish women will be living with overweight or obesity (Keaver et al., 2013). Healthy Ireland Survey revealed that 37% of Irish adults are overweight and a further 23% are living with obesity (Department of Health and MRBI IPSOS, 2015). It also revealed bad behavioural habits in regard to calorie intake by adults. Out of those surveyed, 65% reported to consume snack food or sugar drinks daily, whereas little under half of the participants met the minimum level of recommended physical activity of 150 minutes per week (Department of Health and MRBI IPSOS, 2015). Although obesity has significant implications on a person's health, the economic consequences cannot be overlooked. It has been estimated that the direct and indirect costs of obesity in Ireland in 2009 accounted for €1.13 billion (Perry et al., 2012). Keaver et al. projected that by 2030 this bill will be greatly exceeded, as the direct healthcare cost of obesity alone in Ireland will reach staggering €5.4 billion (Keaver et al., 2013). Although obesity was initially considered an adult disease, a substantial

increase of cases in the paediatric population has been observed worldwide (WHO, 2020a). As reported by the Childhood Obesity Surveillance Initiative, at least 1 in 5 Irish primary school children were living with overweight or obesity (Bel-Serrat et al., 2017). The Children's Sport participation and Physical Activity study showed only 17% of primary and 10% of post-primary school children met the national guidelines for physical activity (Woods et al., 2018). If paired with bad eating habits, this may contribute to the increasing rates of obesity in children. It is now well established that childhood obesity strongly tracks into adulthood (Evensen et al., 2016; Gordon-Larsen et al., 2004; Wang et al., 2008; Ward et al., 2017). Study by Simmonds et al. estimated that 55% of children with obesity will become obese adolescents, and 80% of obese adolescents will become obese adults, highlighting the need for early interventions (Simmonds et al., 2016). As obesity is affecting children at a very young age, obesity-associated co-morbidities such as T2DM, begin to manifest much earlier than previously observed, often still in childhood or adolescence (Mayer-Davis et al., 2017). Unfortunately, treatment options of obesity are still limited and include lifestyle interventions, pharmacological treatments and bariatric surgery (Bray et al., 2018). None of these methods are fully successful, as studies have shown that following each of these treatment regimes, weight can be regained (Felsenreich et al., 2016; Franz et al., 2007). Undisputedly, prevention of obesity starting in early childhood years and development of safe and non-invasive treatments are desperately needed.

1.1.2. Obesity associated co-morbidities

Obesity is associated with an extensive list of co-morbidities, ranging from T2DM, CVD, development of multiple types of cancers and mental health problems (Figure 1.1) (Avgerinos et al., 2019; Daousi et al., 2006; Mannan et al., 2016; Pantalone et al., 2017).



Figure 1.1. Obesity associated co-morbidities. Obesity contributes to development of multiple diseases such as T2DM, CVD and multiple types of cancers (Doheny, 2015).

Individuals who are living with overweight or obesity are at a major risk of developing T2DM, in fact up to 90% of cases are weight related (Daousi et al., 2006; Iglay et al., 2016; Public Health England, 2014). T2DM is characterised by a defect in insulin secretion and/or impaired insulin sensitivity of tissues, which ultimately leads to chronic hyperglycaemia (DeFronzo et al., 2015). Although pathogenesis of T2DM in obesity is still not fully elucidated, it is known to be underpinned by low-grade inflammation. White adipose tissue (WAT) has been identified as the main source of pro-inflammatory mediators in obesity. It was shown to release high amounts of molecules such as tumour necrosis factor (TNF)- α , interleukin (IL)-1 β , IL-6, leptin and monocyte chemoattractant protein-1 (MCP-1). This leads to the recruitment of inflammatory environment and insulin resistance in the AT (Longo et al., 2019; Mantzoros et al., 2011; Zatterale et al., 2020; Zorena et al., 2020).

TNF- α is one of the best studied cytokines in the establishment of T2DM and was shown to be implicated in the onset of insulin resistance by inducing a defect in

insulin signalling (Aguirre et al., 2000; Hotamisligil et al., 1995, 1993). Despite the promising data, clinical trials which focused on the neutralisation of TNF- α in individuals with T2DM, no improvement in insulin sensitivity after treatment was reported (Bernstein et al., 2006; Dominguez et al., 2005; Ofei et al., 1996). Although other inflammatory cytokines have been also implicated in the development of T2DM and insulin resistance (McArdle et al., 2013), the role of immune cells and immune derived factors in this disease is yet to be fully understood.

Obesity is also a significant risk factor associated with the development of CVD (Khan et al., 2018). Excessive weight was shown to alter not only the function of the cardiovascular system, but also the structure of the heart. Individuals with obesity displayed a larger left ventricular mass and worse systolic function, both of which are likely to contribute to the development of hypertension (Bello et al., 2016; Kotchen, 2010; Wu et al., 2019). Failure to regulate blood pressure leads to further complications, such as coronary heart disease, heart failure, stroke and renal disease (Landsberg et al., 2013). Dyslipidaemia, often diagnosed alongside obesity, is strongly linked to atherosclerosis (Duncan et al., 2019; Klop et al., 2013). Excessive exposure to high levels of low-density lipoprotein cholesterol is the main risk factor associated with development of atherosclerosis in humans, as it has been shown to initiate atherogenesis. Over time the plaques deposited on the arterial wall, increase in size, accumulating more lipids. This eventually leads to a substantial restriction of the blood flow and increases a risk of plaque rupture, which can cause acute thrombosis and myocardial infarction. Development of atherosclerosis is accompanied by an inflammatory response in immune cells (Libby et al., 2019). This highlights the common denominator between the pathogenesis of CVD and T2DM in the context of obesity – altered immune response and pro-inflammatory environment.

It is acknowledged that obesity is positively correlated with the development of certain cancers such as colorectal, ovarian, gallbladder, kidney, postmenopausal breast and liver (Bhaskaran et al., 2014; Lauby-Secretan et al., 2016). Recently, it was reported that excessive body weight is attributed to 3.9% of all cancers worldwide (Sung et al., 2018). The contribution of excess adiposity to cancer development is complex, as multiple types of cancer and multiple factors are involved (Bhaskaran et al., 2014; Sung et al., 2018). Chronic inflammation, which persists in obesity, favours tumour development, through the disruption of tissue homeostasis and dysregulation of immune cell function (Y. S. Lee et al., 2018; Quail and Dannenberg, 2019). Although inappropriate activation of the immune system is closely related to obesity, excessive adiposity is also associated with a loss of immune cell function. People with obesity have a defect in natural killer cell mediated killing of the tumour cells *in vitro* (Michelet et al., 2018; Tobin et al., 2017). In addition, chronic stimulation of T cells in obese AT induces expression of exhaustion markers such as programmed cell death protein-1 (PD-1) and leads to dampened anti-tumour responses. However, using PD-1 checkpoint inhibitors, anti-tumour activity in subjects with obesity can be restored (Canter et al., 2018; Z. Wang et al., 2019). Once again, this illustrates how the immune system dysregulation caused by obesity underpins the development of another co-morbidity.

Although obesity has a detrimental effect on the body, the impact of obesity on mental health is often overlooked. It has been reported that more than half of adults with obesity admitted to experience stigmatisation by their co-workers, whereas children with obesity have over 60% higher chance of being bullied (WHO, 2017). Mistreatment of individuals due to their weight can lead to the development of mental health diseases. A study by van Vuuren et al. found that adolescents who are overweight displayed more suicidal thought tendencies than their lean counterparts (Van Vuuren et al., 2019). A meta-analysis which looked at the relationship between weight stigma and mental health, reported a medium or large correlation between weight-related stigma and symptoms of depression, anxiety and psychological distress (Emmer et al., 2020; Luppino et al., 2010). Discrimination and prejudice of the individuals with excessive weight is undisputable, although in 2016 almost 4 in 10 adults were living with overweight and 1 in 8 were living with obesity, indicating a high prevalence of this condition in society (WHO, 2020a). Hence, appropriate actions must be taken to end the stigma of obesity.

1.2 Immune system in health and obesity

1.2.1 Immune system and inflammation

The aim of the immune system is to provide the host with protection from invading microorganisms and injury. Upon pathogen entry or tissue damage, an inflammatory response is initiated, led by the innate immune system which include cells like monocytes, neutrophils and dendritic cells (Figure 1.2) (Delves et al., 2011). This results in the production of pro-inflammatory cytokines and chemokines and recruitment of other leukocytes in order to contain the infection (Coico and Sunshine, 2015; Delves et al., 2011). The innate immune response is supported by the more specialised adaptive arm of the immune system, which assists in the eradication of the infection and aims to provide long-term protection from the pathogen via memory (Figure 1.2) (Coico and Sunshine, 2015; Delves et al., 2011). The adaptive immune system is composed of T lymphocytes, which are classified as cytotoxic CD8⁺ cells and cytokine producing helper CD4⁺ cells, and B lymphocytes, which support the immune response by producing antigen specific antibodies.



Time after infection

Figure 1.2 Response of the immune system to infection with pathogen. Upon infection, the innate immune system is activated upon recognition of pathogen associated molecular patterns and damage associated molecular patterns by antigen presenting cells (APC). This is followed by the engagement of the adaptive immune system in order to successfully restore homeostasis and provide long-term protection to the host thorough formation of immunological memory(Khader et al., 2019).

Activation of the adaptive arm of the immune system leads to clonal expansion of the cell population and the formation of effector cells, providing a more specific memory response upon re-challenge with the pathogen and therefore a better protection to the host (Coico and Sunshine, 2015; Delves et al., 2011). As research in the area of immunology is quickly advancing, it becomes clear that the initial classification of cells into the innate and adaptive immune system requires reconsideration. In recent years more subtypes of immune cells have been discovered and termed as unconventional T cells, which include cells such as mucosal associated invariant T (MAIT) cells, $\gamma\delta$ cells and invariant natural killer T (iNKT) cells (Figure 1.3) (Godfrey et al., 2015). These T cells have been classified as "unconventional" as they are not restricted by major histocompatibility complex (MHC) molecules like classical T cells. However, they still in part rely on the antigen presentation and recognition by their T cell receptor (TCR), but in addition these cells can be activated through cytokine stimuli alone (Godfrey et al., 2015). Furthermore, they don't recognise protein antigens. MAIT cells recognise vitamin B metabolites in the context of MHC class I related protein 1 (MR1) complex, $\gamma\delta$ cells recognise lipid antigens and phosphoantigens presented by CD1c & CD1d molecules, whereas iNKT cells respond to lipid antigens in the context of CD1d (Godfrey et al., 2015). This provides the immune system with another rapid response to pathogens, bridging the innate and adaptive systems.



Figure 1.3 Cells of the immune system. Immune cells can be divided into innate and adaptive immune systems, however some cells such as MAIT cells, don't fit into either of these categories possess feature of both systems. (Adapted from Bonam *et al.*, 2017).

Inflammation is deemed to be beneficial to the host in many circumstances. Acute inflammation is a short-lived response, which is followed by resolution to achieve tissue homeostasis (Coico and Sunshine, 2015). Acute inflammation can become chronic if the agent, which stimulates the immune response cannot be removed (Chen et al., 2018). Sustained low-grade inflammation underpins the

development of multiple inflammatory diseases such as T2DM, rheumatoid arthritis (RA) and systemic lupus erythematosus (SLE) (Guo et al., 2018; Moulton et al., 2017; Wu and Ballantyne, 2020). However, the cause of many of these diseases remains to be elucidated. Identifying the primary triggers of an inappropriate immune response would aid the development of successful treatments. Currently, our understanding of inflammatory diseases is insufficient, and we are limited to the treatment of symptoms. Current therapies aim to dampen down the immune response by using immunomodulatory drugs such as anti-TNF- α and anti-IL-17A, however they don't eradicate the triggers of such diseases (Bissonnette et al., 2018; Kimball et al., 2016; Thaçi et al., 2015; Wang et al., 2016). Advancing research to understand the pathophysiology of the immune response during inflammation and identifying the antigens, which stimulate the cells, will aid in developing successful treatments.

1.2.2 Adipose tissue associated inflammation and immune cell dysregulation in obesity

Obesity is associated with a systemic, low-grade chronic inflammation (Hotamisligil, 2017; Y. S. Lee et al., 2018). It has been well established that this inappropriate activation of the immune system strongly contributes to the development of insulin resistance and T2DM (Zatterale et al., 2020). The origin and cause of this inflammation is still to be fully elucidated and researchers are currently investigating multiple sites including liver, pancreas, muscles. However, studies have strongly focused on the AT as the primary site of inflammation (Y. S. Lee et al., 2018).

For decades, AT has been considered merely as an energy storing tissue. As research progressed, it has been demonstrated that AT is capable of producing and releasing a variety of factors such as adipokines and hormones and is enriched in immune cells, which ultimately lead to the re-classification of AT as an endocrine organ (Ouchi et al., 2011; Rodríguez et al., 2015). Nonetheless, AT plays a very important role in energy homeostasis in a lean individual. It provides long-term storage of excess nutrients, which are released during demand for energy (Ouchi et al., 2011). From the point of view of evolution, this is a very good process of saving nutrients for periods of starvation, where they can be easily accessed and used for survival. Today in the westernised world, this process is

not as crucial, due to the wide availability of food, which often requires very little effort to obtain. Instead, this very process is now leading to over-storing of energy, contributing to the development of diseases. AT adapts to the excess of nutrients by undergoing structural remodelling and expansion. Hypertrophy and hyperplasia of the adipocytes occurs. This leads to increased oxygen demand and therefore the induction of angiogenesis which was shown to be supported by the resident adipose tissue macrophages (ATM) (Figure 1.4) (Cho et al., 2007; Sun et al., 2011; White and Ravussin, 2019). Expansion of the AT is associated with an increase in immune cell infiltration, release of pro-inflammatory factors from the adipocytes and leukocytes leading to inflammation (Figure 1.4). Chronic over-nutrition causes those changes to be sustained for a long-time causing hypoxia, fibrosis and sustained inflammation, leading to AT dysfunction (Zatterale et al., 2020).



Figure 1.4. Changes in adipose tissue in obesity. Obesity leads to hypertrophy and hyperplasia of adipocytes and recruitment of immune cells, resulting in chronic inflammation (Wu and Ballantyne, 2020).

In obesity, adipocytes alter their secretory profile and produce higher levels of leptin but downregulate the production of adiponectin (Considine et al., 1996; Hanley et al., 2007; Ryan et al., 2003). In addition to the role of these hormones in maintaining energy homeostasis in humans, both impact the function of immune cells. Leptin was shown to have pro-inflammatory properties. It can stimulate macrophages to produce pro-inflammatory cytokines such as IL-6 and TNF- α (Acedo et al., 2013), promote chemotaxis of neutrophils (Montecucco et al., 2013), promote chemotaxis provide philos (Montecucco et al., 2013), promote chemotaxis ph

al., 2006), and enhance the differentiation of T cells into Th1 and Th17 subsets (Gerriets et al., 2016; Reis et al., 2015; Rodríguez et al., 2007). On the other hand, adiponectin can upregulate the secretion of IL-10 from macrophages, suppress M1 type in favour of M2 polarisation and decrease production of interferon γ (IFN γ) and IL-17A by CD4⁺ T cells (Ohashi et al., 2010; Surendar et al., 2019; Wolf et al., 2004). Taking into consideration the opposing effects of leptin and adiponectin, it can be hypothesised that the imbalance of these hormones observed in obesity, contributes to the inflammatory state. In addition, production of AT tissue derived TNF- α was found to be increased in obesity (Cao et al., 2008; Hotamisligil et al., 1993). As TNF- α was shown to have a negative impact on insulin sensitivity, this further suggests that changes in the AT and the pro-inflammatory environment induced by obesity is contributing to development of T2DM (Aguirre et al., 2000; Hotamisligil et al., 1993).

Alterations in the function of AT is not the only change observed in obesity. Infiltration of immune cells plays one of the key roles in initiating and maintaining low-grade inflammation. Both arms of the immune system are implicated, however macrophages have received a lot of attention as they are enriched in the AT of people and mice with obesity (Harman-Boehm et al., 2007; Weisberg et al., 2003). In mice, macrophages were shown to make up 40-50% of immune cells in obese AT vs 5-10% observed in the AT of lean mice (Weisberg et al., 2003). Macrophages can be broadly classified as pro-inflammatory M1 or antiinflammatory M2. In obesity, AT is largely enriched in the M1 subtype of macrophages which produce pro-inflammatory cytokines such as IL-6, TNF- α , IL-1β (Figure 1.4) (Lumeng et al., 2007). Their recruitment is likely driven by secretion of chemokines by AT such as of MCP-1, and death of adipocytes, which must be removed to maintain tissue homeostasis and resolve inflammation. (Guilherme et al., 2008; Vishvanath and Gupta, 2019). The association between inflammatory macrophages and insulin resistance in obesity is well established (Catrysse and van Loo, 2018). They are primary producers of TNF- α , and it has been shown that the depletion of these cells normalizes insulin sensitivity (Patsouris et al., 2008). As more evidence is mounting, it is now becoming clearer that macrophages in the AT cannot be strictly identified as M1 or M2 (Kratz et al., 2014), but rather their phenotype is more flexible and lies on a spectrum, which changes our approach in investigating the role of macrophages in the AT. The

adaptive arm of the immune system is also implicated in creating this proinflammatory environment. CD3 expressing lymphocytes are the second most abundant cells in obese AT after the macrophages. Of those lymphocytes, an increase in both CD8 and CD4 expressing T cells is observed (Y. S. Lee et al., 2018). Although the sequence in which the cells arrive in the AT is still to be elucidated, some evidence suggests that the accumulation of CD8⁺ T cells precedes the increase in ATM (Nishimura et al., 2009). T cells contribute to the AT inflammation by releasing Th1 and Th17 type cytokines such as IFNy and IL-17, which can induce recruitment of M1 macrophages to the tissue (Fabbrini et al., 2013; Winer et al., 2009). These changes were found to be accompanied by a decrease in T reg cells, which are known for their anti-inflammatory functions, suggesting further loss of control over inflammation (Winer et al., 2009). Other immune cells have been also identified to be altered in obese AT. MAIT cells were shown to accumulate in human AT in obesity and display a type 17 phenotype characterised by predominant production of IL-17 cytokines rather type 1 phenotype which in turn is identified by IFNy production. This suggests that MAIT cells are involved in maintaining the inflammatory environment in AT (Carolan et al., 2015; Magalhaes et al., 2015). Similarly, IL-17 producing $\gamma\delta$ T cells were also found to be enriched in AT of obese mice and were shown to regulate T reg populations in this tissue (Kohlgruber et al., 2018). On the other hand, populations of innate lymphoid cells 2 (ILC2) and iNKT cells in obese AT is diminished (Brestoff et al., 2015; Lynch et al., 2012). As ILC2 and iNKT cells support M2 profile of macrophages and contribute to the anti-inflammatory profile of AT, it suggests that a reduction in their frequencies may favour proinflammatory environment in obese AT (Lynch et al., 2015; Molofsky et al., 2013).

The primary trigger of inflammation in obese AT is yet to be discovered, however some hypotheses have been put forward. Increased circulation of free fatty acids (FFA) in obesity is likely caused by the eruption of the adipocytes during AT expansion (Arner and Rydén, 2015; Ni et al., 2015). FFA have been shown to initiate the pro-inflammatory immune response by binding to toll like receptor 4 on adipocytes and macrophages and activate nuclear factor-kappa light chain enhanced of activated B cells (NF κ B) and c-Jun N-terminal kinase (JNK) signalling (Shi et al., 2006). In addition, FFA were shown to contribute to hypoxia, by uncoupling of mitochondrial oxidative metabolism in adipocytes. This leads to

deprivation of oxygen in the cell, enhancement in hypoxia inducible factor-1 α (HIF-1 α) expression and activation of the pro-inflammatory pathways within the adipocyte (Lee et al., 2014). Increased gut permeability has been also reported in obesity, which has been suggested to be one of the multiple triggers of systemic chronic inflammation in this disease (Toubal et al., 2020). In fact, an increase in lipopolysaccharide circulation in individuals with obesity has been reported and correlated with development of T2DM (De La Serre et al., 2010; Trøseid et al., 2013). The onset of obesity is multifactorial therefore, the trigger of its associated inflammation is hypothesised to be affected by multiple events. Identification of the triggers of the disease as well as the sequence of immune cell infiltration into the AT and factors involved in initiation of the immune response will aid in further examination of the causes of development of diabetes and other inflammation associated diseases in obesity, hopefully leading to a successful treatment.

1.3 Mucosal Associated Invariant T cells

1.3.1 Mucosal Associated Invariant T cells overview

MAIT cells are a type of unconventional adaptive T cells, which display innate properties and they don't strictly belong to the innate or adaptive category. Initially they have been identified to be enriched at mucosal tissues, hence their name, however it is now well established they are abundant at multiple sites including liver, AT and lungs (Dusseaux et al., 2011; Lezmi et al., 2019; Tobin et al., 2017). They can be also readily identified in the peripheral blood, where they compromise 1-10% of T cells (Le Bourhis et al., 2010).

MAIT cells are defined by the semi-invariant, highly conserved, $\alpha\beta$ TCR, which is accompanied by high expression of CD161 molecules. MAIT cell TCR is composed of TRAV1-2 (V α 7.2) and TRAJ33/12/20 (J α 33/12/20) alpha chains paired with a limited number of beta chains (Lepore et al., 2014; Porcelli et al., 1993; Tilloy et al., 1999). In contrast to classical T cells, MAIT cells were shown to be restricted by MR1, rather than MHC-I or MHC-II antigen presenting complexes (Treiner et al., 2003). The antigen presented by MR1 to MAIT cells has been unidentified for many years. The first clue of the potential origin of the antigens was uncovered by Gold *et al.* and Le Bourhis *et al.* who showed that MAIT cells respond to a wide array of gram- positive and gram-negative bacteria as well as some strains of yeast (Gold et al., 2010; Le Bourhis et al., 2010). It wasn't until 2012, when the landmark study Kjer-Nielsen *et al.* identified the antigen of MAIT cells as metabolites of vitamin B₂ and B₉, which are produced by multiple strains of microorganisms, utilising the riboflavin synthesis pathway (Kjer-Nielsen et al., 2012). This antigen is very unique to microorganisms, as humans don't synthesise vitamin B, providing the host with an advantage in detecting these pathogens. The most potent antigen which activates MAIT cells is synthesised from a 5-amino-6-D-ribityaminouracil (5-ARU), an intermediate produced during riboflavin synthesis and methylglyoxal (MG), a metabolite produced during glycolysis, which together form 5-(2-oxopropulideneamino)-6-D-ribitylaminouracil (5-OP-RU) (Corbett et al., 2014).

MAIT cells development begins in the thymus, where they undergo thymic selection. In contrast to T cells, which are selected by binding to thymic epithelial cells, MAIT cells bind to double positive (CD4⁺/CD8⁺) cortical thymocytes, which express MR1 molecule (Seach et al., 2013). The process of thymic selection of MAIT cells still remains largely unexplored. It is yet to be established whether MR1 is loaded with any antigen, and if so whether it is low or high affinity and where it originates from. Future studies should also investigate whether thymic negative selection applies to MAIT cells and how and if overreactive MAIT cells are removed. Following the successful positive selection, MAIT cells undergo a three-step maturation process which begins in the thymus. A study by Koay *et al.* has significantly contributed to our current understanding of MAIT cell maturation in human and mice. They classified the different stages of development based on surface markers. Stage 1 of MAIT cell were defined as CD27⁻ CD161⁻, stage 2 as CD27⁺ CD161⁻ and stage 3 by CD27^{lo/-}, CD161⁺ and co-expression of IL-18R (Koay et al., 2016).

The mouse study showed that progression through the three stages is depended on the MR1 expression, as addition of anti-MR1 antibody successfully inhibited this process. The progress from stage 2 to stage 3 was described to be of particular importance in functional development of MAIT cells. Promyelocytic leukaemia zinc finger (PLZF), a transcription factor shared by the cells of the innate immune system, was found to be absolutely necessary for MAIT cell maturation from stage 2 to stage 3 and development of their functional competence. PLZF-null mice completely lacked stage 3 MAIT cell population and residual stage 1 and stage 2 cells failed to produce cytokine upon stimulation in contrast to stage 3 MAIT cells from wild type (WT) mouse (Koay et al., 2016). Expression of PLZF, T-box protein expressed in T cells (T-bet) and RAR-related orphan receptor gamma vT (RORvT) transcription factors increased upon MAIT cell maturation to stage 3. In human MAIT cells, T-bet and RORyT were coexpressed in contrast to murine MAIT cell development, where these two transcription factors were mutually exclusive (Koay et al., 2016). To assess the functional capacity of stage 3 MAIT cells, Koay et al. stimulated samples obtained from matched blood and thymus and measured their cytokine production. More than 80% of stage 3 MAIT cells from peripheral blood produced IFNy and TNF- α , in comparisons to <25% of stage 3 MAIT cells derived from the thymus. The superiority of peripheral blood stage 3 MAIT cells in cytokine expression, indicates that MAIT cells leaving the thymus are somewhat functionally competent, with further maturation of the cells occur extrathymically (Koay et al., 2016). A more recent study by Koay et al. focused on the transcriptional profile of MAIT cells during the three stages of development and maturation, using single-cell RNA sequencing in mice and humans (Koay et al., 2019). This study confirmed previous data that murine MAIT cells at stage three cells show a more mature prolife and express more Zbtb16 (PLZF), Rorc (RORyT), and IL18r (IL-18 receptor) but less Ccr7 than stage 1 cells. In addition, upregulation of other genes was identified at stage 3 such as GzmB (Granzyme B), IL7R (CD127) and chemokine receptors Ccr6, Ccr2, Cxcr6 which bind C-C Motif Chemokine Ligand (CCL) 20, CCL2 and CCL16 respectively. The increase in the chemokine receptors was confirmed on the protein level by flow cytometry (Koay et al., 2019). Upon closer investigation of the stage 3 MAIT cells, this study was able to differentiate between type 1 and type 17 MAIT cells and found a significant difference in the expression of genes between those two subtypes of cells.

Mature MAIT cells can be identified in the peripheral blood and display a memory phenotype identified by the lack of the expression of CC-chemokine receptor (CCR)7, CD62L and CD45RA (Dusseaux et al., 2011; Gold et al., 2013). They are largely CD8⁺, with a small proportion being double negative of CD4⁺ (Gherardin et al., 2018b; Kurioka et al., 2017; Walker et al., 2012). As discussed in more detail in chapter four, MAIT cells can be activated via engagement of their TCR with MR1 bound to an antigen or via cytokine stimulation alone (Le Bourhis

et al., 2010; Leng et al., 2019; Van Wilgenburg et al., 2016). MAIT cells express a wide range of cytokine receptors including IL-2R, IL-12R, IL-15R or IL-18R, suggesting their activation can be induced by a various cocktails of cytokines in the absence of TCR antigens (Sattler et al., 2015; Wallington et al., 2018; Wilgenburg et al., 2018). Indeed, MAIT cells were shown to respond with enhanced IFNy production when stimulated with a combination of cytokines namely IL-12 and IL-18 or IL12 and IL-15, whereas stimulation with single cytokines did not induce cytokine production or led to their very weak expression (Ussher et al., 2014; Van Wilgenburg et al., 2016). This shows plasticity of MAIT cells in responding to a potential threat, as cytokine stimulation alone, allows the cells to become activated in the absence of TCR antigens. This response may be very beneficial during viral infections or infections with bacteria which do not produce riboflavin; however, the non-specific MAIT cell activation may be harmful and contribute to the development of autoimmune and inflammatory diseases. Activation of MAIT cells induces rapid secretion of pro-inflammatory cytokines such as IFNy, TNF- α and IL-17 and chemokines CCL3 and CCL4 which support recruitment of other immune cells to the site of inflammation and their activation (Hinks et al., 2019; Lamichhane et al., 2019). In addition, MAIT cells were shown to produce other effector molecules such as granzyme B and perforin, indicating their role in direct killing of cells infected with a pathogen (Dusseaux et al., 2011; Lamichhane et al., 2019; Wilgenburg et al., 2018). Indeed, MAIT cells were shown to kill bacterially infected cells in *in vitro* studies (Kurioka et al., 2015; Le Bourhis et al., 2010), however their role seems to extend beyond infections. MAIT cells exposed to MR1 expressing cancer cells lines were shown to induce killing of the cancer cells, indicating their role in cancer surveillance (Gherardin et al., 2018a; Won et al., 2016).

Although MAIT cells have been discovered relatively recently, their pleotropic functions and highly conserved TCR indicate a key role in the immune protection of the host. More research into MAIT cell biology and function is required to fully understand the role they play in bridging the innate and adaptive immune system.

1.3.2 MAIT cells in infection

MAIT cells mount an immune response to bacterial and yeast riboflavin biosynthesis metabolites in the context of MR1 molecules expressed by antigen

presenting cells. Study by Le Bourhis et al. in 2010 was one of the first to highlight the important contribution of MAIT cells to the immune response against microbial infections (Le Bourhis et al., 2010). This study showed MAIT cells respond to APC infected with *E. coli* in an MR-1 dependent manner (Le Bourhis et al., 2010). The response of MAIT cells to multiple bacteria strains has since been investigated in much more detail. Meierovics et al. reported MAIT cell activation and cytokine production during infection with *Francisella Tularensis*, which was dependent on MR1 and IL-12 signals (Meierovics et al., 2013). MAIT cells were also found to be implicated during infections with bacteria from the mycobacterium family. Initially, MAIT cells were shown to be reduced in the peripheral blood of patients with active infection with *Mycobacterium tuberculosis* and exhibited defective IFNy production in comparison to the healthy controls (Jiang et al., 2016; Kwon et al., 2015). In contrast, Suliman et al. and Paguin-Proulx et al. reported no significant difference in the frequencies of MAIT cells or their IFNy production between individuals with active infection versus healthy controls (Paquin-Proulx et al., 2018; Suliman et al., 2020). This indicates that the role of MAIT cells in tuberculosis infection may differ between the stages of infection, and more research is required to clearly establish their involvement in the response against this pathogen. Additionally, MAIT cells are capable of responding to multiple other bacteria via TCR-MR1 engagement such as Klebsiella pneumoniae and Staphylococcus aureus as well as yeast strains including Candida albicans (Le Bourhis et al., 2010). Although the majority of these studies identified MAIT cells as a major subtype of T cells mediating beneficial immune response, the opposite has been observed with Helicobacter *pylori* in mice. During *H. pylori* infection in WT mice, MAIT cells accumulated in the mucosa of the stomach and contributed to gastritis, whereas changes in gastric pathology were ameliorated in MR1 deficient mice, suggesting a pathogenic role of MAIT cells in this infection (D'Souza et al., 2018).

While the involvement of MAIT cells in mounting an effective immune response against bacterial infection wasn't surprising, as their primary antigen was identified to be a bacterial metabolite, MAIT cells were also shown to respond during viral infections. Early reduction in MAIT cell frequency has been shown in the peripheral blood of individuals infected with human immunodeficiency virus (HIV) and Hepatitis C virus (HCV) and this decrease was reported to be
irreversible in adults despite successful antiviral treatment during the chronic stage of the disease (Barathan et al., 2016; Cosgrove et al., 2013; Eberhard et al., 2014; Spaan et al., 2016). A study of HIV by Lal et al. showed that MAIT cells were highly activated at the peak of viremia, which was followed by an expansion of the population and enhancement of its effector functions. Decreases in the MAIT cell frequencies and impairment of function were observed at about oneyear post infection, suggesting that early treatment of HIV patients, could help to rescue MAIT cell population and preserve their effector functions for a longer time (Lal et al., 2020). On the other hand, MAIT cells from peripheral blood of individuals with HCV infection displayed an activated phenotype measured by the CD69 expression and a significant increase in the expression of markers such as human leukocyte antigen- DR (HLA-DR) and PD-1, suggesting their exhausted phenotype (Barathan et al., 2016; Hengst et al., 2016). MAIT cells are also involved during influenza virus infection. Studies by Van Wilgenburg et al. and Loh et al. have demonstrated a decrease in the frequency of circulating MAIT cells and upregulation of IFNy and granzyme B during the infection (Loh et al., 2016; Van Wilgenburg et al., 2016). The direct role of MAIT cells in response to influenza was demonstrated in MR1 deficient mice, which upon infection with H1N1 showed a higher mortality rate than controls. Adoptive transfer of MAIT cells into the MR1 deficient mice, extended the lifespan of infected mice, suggesting a non-redundant role of MAIT cells in influenza infections (Wilgenburg et al., 2018). All these studies demonstrate that MAIT cells are implicated in multiple bacterial and viral infections, making them an interesting target for vaccine development. However, more studies are required to establish their exact role, as discussed, they have a beneficial role in multiple infection, but like in the case of H. pylori they might cause excessive inflammation and contribute to development of pathologies.

1.3.3 MAIT cells in inflammatory and autoimmune diseases

The role of MAIT cells in autoimmune and inflammatory diseases is still under investigation, however an alteration in peripheral blood MAIT cell frequencies and function has been observed in a range of diseases including, RA or inflammatory bowel disease (IBD) (Figure 1.5) (Hayashi et al., 2016; Tominaga et al., 2017).



Figure 1.5 MAIT cells in inflammatory and autoimmune diseases. MAIT cell phenotype and function is altered in multiple inflammatory and autoimmune diseases, where they largely present with an IL-17 phenotype (Pisarska et al., 2020).

Despite multiple studies, the role of MAIT cells in multiple sclerosis (MS) remains controversial. One of the primary T cells implicated in MS are Th17 cells, however, MAIT cells were shown to contribute to IL-17 production in MS (Abrahamsson et al., 2013; Willing et al., 2018). In addition, they were also shown to overexpress migratory molecules, suggesting their ability to migrate across the blood brain barrier (Salou et al., 2016). Indeed, MAIT cells were found to accumulate in central nervous system lesions supporting the hypothesis that they may be implicated in the pathology of the disease (Illés et al., 2004; Salou et al., 2016; Willing et al., 2014). It is not clear whether the frequency of MAIT cells in MS is altered due to conflicting reports. Willing et al. and Miyazaki et al. have shown a decrease in the frequencies of MAIT cells in the peripheral blood of MS subjects (Miyazaki et al., 2011; Willing et al., 2014), whereas other studies have shown an increase or no difference in MAIT frequencies in the peripheral blood of MS individuals (Ammitzbøll et al., 2020; Annibali et al., 2011; Salou et al., 2016; Sugimoto et al., 2016). In search for the cause of RA, the role of MAIT cells has been also evaluated. In comparison to the healthy controls, MAIT cell frequencies in the peripheral blood of RA patients vary between studies, with some reporting a decrease or no change (Cho et al., 2014; Gracey et al., 2016; Koppejan et al., 2019). However, an increase in MAIT cell frequencies has been identified in the synovial fluid versus blood of individuals with RA, suggesting their migration to the inflamed tissue (Cho et al., 2014; Gracey et al., 2016). Indeed, TNF- α and IL-1 β found at high concentrations in the synovial fluid of RA patients were shown to trigger MAIT cell migration into the tissue through the induction of migration-associated molecules on the endothelium (Kim et al., 2017). The negative effect of MAIT cells on the disease development was confirmed in a mouse model of arthritis, in which MR1 deficient mice developed less severe disease. Repopulation of MR1 deficient mice with MAIT cells caused an exacerbation of the disease (Chiba et al., 2012).

Similarly to RA, MAIT cell frequency is reduced in the peripheral blood of SLE patients (Chiba et al., 2017; Cho et al., 2014), which display impaired IFNγ production (Cho et al., 2014). A direct relationship between MAIT cells and development of SLE has been depicted in a mouse study of SLE, which reported that MR1 knockout mice had lower levels of autoantibodies than controls, whereas co-incubation of MAIT cells & B cells from a murine lupus mouse model, showed an enhancement in the production of antibodies which are a hallmark of the disease (Murayama et al., 2019).

MAIT cells are known to be involved in multiple inflammatory diseases such as IBD. The pathophysiology of the disease is not fully understood; however, immune system dysfunction strongly underpins the development of this disorder (Neurath, 2019). IBD individuals with ulcerative colitis or Crohn's Disease displayed a very similar phenotype of MAIT cells observed in RA. Patients with IBD had reduced frequency of peripheral blood MAIT cells which largely displayed IL-17 phenotype, with only Hiejima et al. reporting no difference in IL-17A production (Haga et al., 2016; Hiejima et al., 2015; Serriari et al., 2014; Tominaga et al., 2017). In addition, IBD patients presented with enhanced accumulation of MAIT cells in the inflamed mucosa (Haga et al., 2016; Serriari et al., 2014; Tominaga et al., 2017). Together with MS, RA and IBD reports, this further suggests that MAIT cells may be actively recruited from the blood to the site of inflammation. Studies of immune cells in the skin revealed that MAIT cells form a small population of the resident immune cells in the normal healthy skin (Li et al., 2017; Teunissen et al., 2014). However, investigation of the skin of individuals with psoriasis or dermatitis herpetiformis reported the presence of IL-

17 producing MAIT cells in the inflamed skin areas (Teunissen et al., 2014), suggesting that during inflammation, MAIT cells are likely to contribute to the proinflammatory environment.

MAIT cell frequency in the peripheral blood, lungs tissue and bronchioalveolar lavage of adults with asthma was shown to be reduced in comparison to the healthy controls and correlated with the disease severity and administration of inhaled corticosteroids (Hinks et al., 2015). It is unknown whether the reduction of MAIT cell frequency was directly related to the disease pathology or inhaled corticosteroids treatment, however, this reduction in MAIT cell frequency may contribute to an increased susceptibility of infection in this cohort. A study of children with asthma by Lezmi et al. identified a higher frequency of IL-17 producing MAIT cells, in children who had one or more severe exacerbations during a twelve-month period versus children who did not. In addition, the frequency of IL-17 producing MAIT cells positively correlated with the number of exacerbations of the disease (Lezmi et al., 2018). More recently, Lezmi et al. obtained similar results in a study of children with severe asthma, showing an enhancement in the frequency of IL-17A producing MAIT cells in their bronchoalveolar lavage samples, which positively correlated with the number of exacerbation (Lezmi et al., 2019). Recently it was identified that chronic stimulation of MAIT cells can result in IL-13 production (Kelly et al., 2019). This suggests Th2 cells in asthma may not be the only source of the allergy associated cytokines which contribute to the severity of disease.

Majority of the diseases discussed here can be characterised by a reduction in the frequencies of MAIT cells in the peripheral blood and their enhanced IL-17 phenotype, suggesting their migration to the site of inflammation and aggravation of the pro-inflammatory response. More research is required to identify the trigger which alters the phenotype of MAIT cells and to find targets to ameliorate their pro-inflammatory response.

1.3.4 MAIT cells in diabetes and obesity

Type 1 and type 2 diabetes mellitus are considered chronic inflammatory diseases, which involve an array of immune cells, primarily B and T lymphocytes (DeFronzo et al., 2015; Katsarou et al., 2017). Like in the vast number of

autoimmune diseases, the primary factors which contribute to the onset of the disease are yet to be identified. The microbiome was shown to play an important role in development of type 1 diabetes mellitus (T1DM) (Alkanani et al., 2015; Wen et al., 2008). As MAIT cells are associated with gut mucosal sites and are in close interactions with the microbiota, this suggests their possible involvement in the pathogenesis of T1DM.

Analysis of juvenile T1DM revealed no difference in the proportion of MAIT cells in peripheral blood between diabetic or control subjects, however T1DM individuals had a significantly larger subset of CD27⁻MAIT cells in comparison to the controls. In addition, CD27⁻ MAIT cells from healthy donors produced more IL-17A than CD27⁺ MAIT cells (Harms et al., 2015). The same group also reported a reduction in MAIT cell frequency in seroconverted patients (without T1DM yet) in comparison to the control group (Harms et al., 2018). Rouxel et al. reported a similar finding whereby MAIT cell frequencies were reduced in the peripheral blood of children with recent-onset T1DM versus healthy children. MAIT cells from children who recently developed T1DM expressed exhaustion markers of inflammation such as PD-1 and they produced less IFNy, but more IL-17 and granzyme B upon challenge in vitro in comparison to controls (Rouxel et al., 2017). In addition, experiments performed with purified donor MAIT cells and β-cell lines demonstrated the ability of MAIT cells to kill them (Rouxel et al., 2017). In a mouse model of T1DM, non-obese diabetic (NOD) mice had an increased migration of MAIT cells into the pancreas during development of T1DM, accompanied by an increase in the production of IFNy and Granzyme B. Interestingly, NOD mice showed changes in the gut during the progression of diabetes. MR1^{-/-} NOD mice displayed a deterioration of gut integrity, favouring translocation of bacteria and possibly ultimately promoting T1DM development (Rouxel et al., 2017). On the other hand, Kuric et al. noted no difference in MAIT cell frequency in areas of insulitis or its proximity in patients with early onset of T1DM as compared to pancreas of healthy donors. They reported low expression of V α 7.2 MAIT cell TCR mRNA levels in the pancreas of both healthy and diseased donors (Kuric et al., 2018). These results contradict the previous findings, which suggested a strong role of MAIT cell in the development of T1DM. More research is required to assess the level of involvement of MAIT cells in the development of T1DM.

MAIT cells may also be involved in the development of T2DM, however this area is not very well researched. A reduction in MAIT cell frequencies have been reported in individuals with T2DM (Magalhaes et al., 2015; Zhang et al., 2019). Resting peripheral blood MAIT cells from individuals with T2DM exhibited a more activated phenotype as assessed per CD25 expression and upon stimulation expressed higher levels of IL-2, granzyme B and IL-17. In addition, MAIT cells in the AT of individuals with T2DM displayed enhanced IL-17 production, suggesting their contribution to AT inflammation (Magalhaes et al., 2015). More recent study, by Zhang et al. assessed the expression of OX40 by MAIT cells in T2DM. OX40 is a co-stimulatory molecule expressed by T cells, whose engagement leads to their activation and differentiation. This study reported significantly higher frequency of OX40⁺ MAIT cells from individuals with T2DM, which was negatively correlated with MAIT cell frequencies. In addition, engagement of OX40 leads to the induction of apoptosis by MAIT cells, suggesting that over-expression of OX40 may be leading to the diminished frequencies observed in subjects with T2DM (Zhang et al., 2019). The knowledge of MAIT cell activation and function in T2DM remains very limited, therefore more studies are required to identify their role in the development of the disease.

Development of T2DM is closely associated with obesity, and in fact MAIT cells from people with obesity displayed very similar phenotype. A significant reduction in MAIT cell frequencies has been recorded in peripheral blood of people with obesity or obesity and T2DM (Carolan et al., 2015; Magalhaes et al., 2015). As described in detail in the introduction to chapter 4, these MAIT cells displayed an activated phenotype, with an enhanced IL-17 production and a defect in expression of IFN γ in comparison to the lean controls (Carolan et al., 2015; Magalhaes et al., 2015). Although it has been established that MAIT cells are highly dysfunctional in obesity, currently it remains unknown what causes such profound alterations in their frequencies and function. As MAIT cells have been discovered relatively recently, more basic research is required to establish what mechanisms control their activation. This will establish which of the pathways are dysfunctional in MAIT cells from individuals with obesity and hence identify targets which would aim to eradicate those flaws.

1.4 Immunometabolism

For a long time, the function and metabolism of immune cells have been investigated separately, overlooking the strong connection between these two areas. Immunologists have gained more insight into how metabolism of an organism can influence immune cells during the early studies of nutrient deprivation, obesity and obesity related diseases such as T2DM. Although it may seem obvious now, the disruption to nutrient availability and metabolism of such nutrients by immune cells is critical to their function, it hasn't been that evident for many years. One of the first clues that indicated the importance of metabolism in immune cell functions was reported as early as 1950s (Oren et al., 1963; Pachman, 1967; Sbarra AJ and Karnovsky M. L., 1959) however, research in the area didn't escalate until about 10 years ago. This is when immunometabolism emerged as a new exciting area of immunology.

Cell metabolism is very complex and involves multiple pathways, which are closely linked. A defect in one of those pathways may have further downstream consequences on whole cell metabolism and function. The aim of these pathways is to provide energy and biosynthetic intermediates which are required for the cell survival, activation and proliferation (Dimeloe et al., 2017; Munford and Dimeloe, 2019; O'Neill et al., 2016). It has been described that the metabolism of T cells varies greatly depending on their activation state. In naïve T cells in the quiescent state or during memory formation, their energy production is largely supported by the Kreb's cycle and mitochondrial oxidative phosphorylation (OxPhos), which are known to generate large amounts energy through adenosine triphosphate (ATP) production (Munford and Dimeloe, 2019; O'Neill et al., 2016).

Upon stimulation, T cells undergo metabolic reprogramming and rapid upregulation of glycolysis. This is known as the Warburg effect, which was first discovered by Otto Warburg in 1920s in cancer cells (Warburg, 1925). Although OxPhos is upregulated in activated T cells, glycolysis becomes the dominant pathway (Palmer et al., 2015; Sena et al., 2013). Glucose is one of the primary fuels required for cell function. It is obtained from the extracellular environment of immune cells and its uptake is facilitated by glucose transporters such as GLUT1 (Jacobs et al., 2008; Macintyre et al., 2014). Glycolysis breaks down a molecule

of glucose into pyruvate in a series of enzymatically controlled steps. It is considered to be an inefficient source of energy production as it only yields two net ATP energy molecules per one molecule of glucose, however it provides multiple intermediates, which can feed into other metabolic pathways. In this way, glycolysis contributes to the synthesis of nucleotides by feeding into the pentose phosphate pathway (PPP) as well as the synthesis of fatty acids by feeding into the Krebs's cycle or synthesis of amino acids by fuelling the serine synthesis pathway (Munford and Dimeloe, 2019; O'Neill et al., 2016). This shows that the upregulation of glycolysis upon activation is more beneficial to the cell, as it provides it with enough energy, but also yields biosynthetic precursors required for its proliferation and exertion of its functions. Upregulation of glycolysis can also happen more swiftly than the upregulation of OxPhos, providing the cell with the advantage of faster response to stimuli (O'Neill et al., 2016). The importance of glucose and glycolysis in T cell activation was demonstrated by multiple studies. Mice deficient in the GLUT1 receptor on their T cells demonstrated reduced proliferation capacity and impaired glycolysis upon stimulation (Macintyre et al., 2014). Incubation of T cells in media in which glucose was substituted for galactose, resulted in a defect in their IFNy production (Chang et al., 2013). Stimulation of human T cells demonstrated an increase in glycolysis during the period of 24-72 hours as measured by extracellular acidification rate (ECAR), whereas incubation of T cells with 2-deoxy-d-glucose (2DG), inhibitor of hexokinase 2, the first enzyme of glycolysis, significantly reduced IFNy production (Renner et al., 2015).

Glycolysis is dependent on mammalian target of rapamycin complex 1 (mTORC1) and complex 2 (mTORC2) activation, although the action of mTORC2 is less understood (Saxton and Sabatini, 2017). Mammalian target of rapamycin (mTOR) plays a central role in T cell metabolism and has multiple functions. It is upregulated upon T cell activation and acts as a nutrient sensor in the cells. It integrates the signals of nutrient status and antigen signalling to orchestrate the activation of appropriate pathways. It detects levels of amino acids such as leucine, promotes lipid synthesis, as well as RNA transcription (Salmond, 2018; Saxton and Sabatini, 2017). Multiple studies have demonstrated the role of mTOR activation in glycolytic metabolism. Inhibition of mTOR activation with rapamycin was shown to impair their expression of glucose transporters GLUT1

and GLUT3 by cytotoxic T lymphocytes and impair glucose uptake (Hukelmann et al., 2016). mTORC1 signalling implicated in aerobic glycolysis is still not very clear, however it is thought to be controlled by the regulation of Myc and HIF1 α transcription factors (Finlay et al., 2012; Wang et al., 2011). As mentioned earlier, mTOR is sensitive to amino acid levels. In the past few years, studies have focused on elucidating the role of amino acids in T cell metabolism. It was found that upon activation, cells not only increase the flux of glucose, but the uptake of amino acids is higher (Carr et al., 2010; Nakaya et al., 2014). Stimulation of the cells leads to an increase in the expression of amino acid transporters (Carr et al., 2010; Sinclair et al., 2013). The most well studied amino acid transporter in lymphocytes to date is the L-type amino acid transporter 1 (LAT1). LAT1 is a heterodimer of CD98 and solute carrier (SLC)7A5. It is an antiporter which transports extracellular leucine into the cell in exchange of intracellular glutamine (Hsu and Dzhagalov, 2019). It is known to be upregulated on T cells upon stimulation via TCR/CD28 and is controlled by the expression of Myc (Hayashi et al., 2013; Sinclair et al., 2013). It can transport several amino acids, however it plays a major role in transporting leucine, as leucine is required for mTORC1 activity (Saxton et al., 2016; Sinclair et al., 2013). The role of LAT1 was strongly highlighted in a study by Cantrell lab, which showed the deletion of SLC7A5 in mouse T cells leads to the impairment of cell proliferation and differentiation into CD4 and CD8 cells (Sinclair et al., 2013). Similar results were obtained by Hayashi et al. where LAT1 expression on human T cells was disrupted by small interfering RNA (siRNA). They have reported a reduction in the uptake of amino acids and impaired cytokine production by T cells (Hayashi et al., 2013). There is mounting evidence highlighting a role for glutamine in T cell activation and proliferation (Carr et al., 2010; Johnson et al., 2018; Sener et al., 2016). Glutamine is involved in multiple metabolic aspects of the cell. It is transported into the cell via multiple transporters belonging to the solute carrier family including SLC1A5, SLC3A2, SLC7A5 and SLC38A1, however many more glutamine transporters exist (Carr et al., 2010; Nakaya et al., 2014; Wang et al., 2011). Upon entry into the cell, glutamine can be converted to α -ketoglutarate and fed into the Kreb's cycle or it can fuel acetyl Coenzyme A (Acetyl Co-A) production. It can also be used in the generation of glutathione, a natural antioxidative or enter the serine biosynthesis pathway (Altman et al., 2016). Glutamine was shown to be utilised by cells upon activation and proliferation

(Carr et al., 2010; Wang et al., 2011). Deprivation of this amino acid during stimulation of naïve CD4 T cells resulted in T reg differentiation, despite Th1 cytokine signal. It was shown that α -ketoglutarate, a metabolite derived from glutamine metabolism, acts as a CD4 T cell differentiation controller, as addition of α -ketoglutarate analogue to the cultured lead to Th1 differentiation (Klysz et al., 2015). This was further confirmed by Johnson *et al.* who found that the inhibition of glutaminase, an enzyme which converts glutamine to glutamate, promoted T cell differentiation into Th1 type, while impairing Th17 differentiation, indicating Th17 cells are dependent on glutaminase and glutamine influx during cell activation (Johnson et al., 2018). These studies highlight the importance of amino acids, not only as building blocks required for cell function, but as a signalling molecule which together with the external stimulation signals largely impact the cell function.

Identifying this close relationship between immune cell function and metabolism opened a new avenue in targeting immune cells in inflammatory diseases. As mentioned before, inhibition of glycolysis with 2DG has profound effects on T cell function and leads to the impairment of its activation. The effects of the inhibitor have been further explored and studied by Yin et al. in a mouse model of SLE. Combined treatment of the SLE mice with 2DG, the glycolysis inhibitor and metformin which was shown to impair OxPhos by inhibiting the complex I of electron transport chain, resulted in the downregulation of metabolism in T effector cells, with no effect in the control mice (Owen et al., 2000; Wheaton et al., 2014; Yin et al., 2015). However, since then, the mode of action of metformin has been under review (Wang et al., 2019). Yin et. al also found a decrease in the production of anti-nuclear antibodies and anti-double stranded DNA IgG, leading to remittance of the SLE phenotype (Yin et al., 2015). Liu et al. investigated the effect of 2DG in a mouse model of Guillain-Barre syndrome, an inflammatory demyelinating disease of the peripheral nervous system. They found that the administration of 2DG prevented progression of the disease and inhibited its initiation (Liu et al., 2018). Treatment with 2DG was also applied to the NOD mouse model of an autoinflammatory diabetes and resulted in an increase in β cell granularity and a reduction in T cells reactive to diabetes antigen, the islet-specific glucose-6-phosphate catalytic subunit-related protein (IGRP) (Garyu et al., 2016). As GLUT1 plays a very important role in immune cell

function, inhibition of GLUT1 mediated transport of glucose is another potential target in treatment of inflammatory diseases. Inhibition of GLUT1 with WZB117 inhibitor in early-stage memory T cells derived from patients with T1DM inhibited the expansion of this cell subset (Vignali et al., 2018). The investigations of using glycolysis inhibitors to treat autoimmune diseases are in very early days. mTOR inhibitors such as Sirolimus are currently approved for use in the clinics to prevent organ transplant rejection (Nguyen et al., 2019). A recent clinical trial of Sirolimus in active SLE over 12 months found an expansion in Treg and CD8 memory T cell populations in the treated patients and improved disease activity (Lai et al., 2018). Administration of rapamycin in a murine model of EAE has also shown promising results. The treatment increased the percentage of Treg cells in the rapamycin treated group versus untreated. The percentage of Th17 cells, which are considered to be highly pathogenic in MS and EAE, were found to be decreased in the rapamycin treated group as well as infiltration of IL-17 expressing cells in the spinal cord of the EAE mice was reduced in comparison to the controls (Li et al., 2019). These studies show a potential in controlling inappropriate immune cell activation, as is observed in diabetes and obesity. Although these results are very promising, these approaches must be considered very carefully, not only immune cells use glycolysis and mTOR to exert their functions. The side effects of these drugs could be very widespread, causing increased susceptibility to infections and cancer development, due to the dampening of the immune system. The effect of these inhibitors must also be considered on cells in other tissues and organs, as widespread inhibition of such vital life pathways like glycolysis, may impair the function of whole organs.

Metabolism has been shown to be an integral part of the immune cell function; however, upon undertaking this thesis no data has been published on the metabolism of MAIT cells. In this thesis we aimed to identify the metabolic requirements of these cells by investigating the role of glycolysis and glycolysis related pathways in MAIT cell activation. We further examined how these pathways are involved in supporting cytokine production, proliferation and expansion of these cells. It has been established in the literature that people with obesity present with a severe impairment of MAIT cell function in comparison to the control population (Carolan et al., 2015; Magalhaes et al., 2015). Therefore, this study also aimed to investigate the metabolism of MAIT cells from people

with obesity and compare it to control individuals in order to determine whether there are any differences, which may be negatively affecting MAIT cell function. Identification of any alterations could open new avenues of targeted treatments to restore MAIT cell function in individuals with obesity.

Chapter Two

Materials and Methods

2.1. Ethics

Ethical approval for this study and collection of samples from people with obesity and healthy controls was granted by Medical Research Ethics Committee from St. Vincent's University Hospital Dublin. Local ethics were also obtained from Maynooth University Research Ethics committee.

2.2. Study Design

This was a cross-sectional investigative comparative study. Control blood samples were obtained from St. Vincent's University Hospital Dublin and Irish Blood Transfusion Service. People with obesity were recruited from the Weight Management Clinic in Loughlinstown (part of the St. Vincent's Healthcare group). All recruited control individuals (apart from blood samples obtained from IBTS) and people with obesity were recruited into the study after giving informed and written consent. The variation in the number of samples obtained from the control individuals and people in obesity occurs due to the variability in sample availability and sample quality for each of the groups. As control samples were more accessible and of better quality, than samples from people with obesity, more individuals have been included in the control groups for each experiment.

2.3. Isolation of human peripheral blood mononuclear cells

Peripheral blood mononuclear cells (PBMC) were isolated using gradient density centrifugation from venous blood samples collected from control lean individuals or people with obesity into lithium heparin coated tubes (vacutainer, BD Bioscience); or from donated buffy coats from the Irish Blood Transfusion Service. The blood was diluted 1:1 with sterile phosphate buffered saline (PBS, Sigma Aldrich) for venous blood samples and 1:5 for buffy coats and carefully layered on top of 15mls of Lymphoprep[™] medium (StemCell Technologies) in a 50mls tube (Sarstedt) (Figure 2.1). The blood preparation was centrifugated at 1800 rpm for 20 minutes, with low acceleration and break turned off as shown in Figure 2.1. After centrifugation, the buffy layer was removed into a new sterile tube and washed twice in 20mls of PBS at 1800 rpm for 8 minutes with high acceleration and deacceleration. Depending on the pellet size, cells were resuspended in a volume of complete RPMI (cRPMI) 1640 1X (GIBCO, ThermoFisher Scientific) containing 2%v/v HEPES buffer solution 1M (GIBCO), 10% v/v heat inactivated foetal calf serum (FCS) (GIBCO), 1X antibioticantimycotic (100untis/mL of penicillin, 100µg/mL of streptomycin, 250ng/mL Gibco Amphotericin B) (GIBCO) 1mM Sodium Pyruvate and non-essential amino acids diluted to 1X(GIBCO)) and the cells were counted as per section 2.4 and Figure 2.2.



Figure 2.1 Isolation of peripheral blood mononuclear cells. A diagram showing layering of the blood and compartmentalisation of the blood cells after centrifugation. Image created with BioRender.

2.4. Cell enumeration

The cells were stained using trypan blue solution (GE Healthcare) using 1:10 or 1:20 dilution and 10μ l of the stained cell suspension was applied to a haemocytometer counting chamber. Using a dye exclusion test, cells which didn't take up the dye (white in colour) were considered to be viable and only those cells were counted in the middle square of the haemocytometer as shown in figure 2.2. To calculate the number of the cells in suspension per 1mL the following formula was used:

$Cells/mL = Cell \ count \ per \ middle \ square \times \ dilution \ factor \ \times 10^4$





Figure 2.2 PBMC counting. Trypan blue stained PBMC were applied to the haemocytometer and counted as per diagram. Image created with BioRender.

2.5. MAIT cell proliferation and expansion

2.5.1. Cell Trace Violet Staining

MAIT cell proliferation was measured with CellTraceTM Violet (CTV) cell proliferation kit for use on flow cytometer (ThermoFisher Scientific). The required number of freshly isolated PBMC was pelleted by centrifugation at 1800rpm for 8 minutes and the supernatant was completely removed. CTV solution was prepared immediately prior to use, by adding 20μ L of dimethyl sulfoxide (DMSO) to the supplied vial with the stain, to obtain 5mM stock. Next, the dye was diluted to 5μ M working concentration by adding appropriate amount of the stock solution

into pre-warmed PBS. The cells were gently resuspended and stained at 10⁶ cells per mL of the PBS-dye solution. Cells were incubated for 20 min at room temperature, protected from light with circular agitation. Next, five times the original staining volume of cRPMI was added to the tubes and incubated for 5 minutes at room temperature, which allowed for the removal of any unbound dye. The cells were pelleted at 1800rpm for 8 minutes and resuspended in warm cRPMI culture media at 2x10⁶ cells per mI and re-counted. Cells were incubated for at least 10 minutes to allow acetate hydrolysis of the dye, before proceeding with the experimental set up.

2.5.2. TCR only induced MAIT cell expansion & proliferation over 5 days

PBMC were stained with CTV as per section 2.5.1 for assessment of MAIT cell proliferation, or PBMC remained unstained for measurement of MAIT cells expansion. A sample of 0.25-0.5x10⁶ cells was set aside to stain for flow cytometry to measure MAIT cell frequency or CTV uptake prior to proliferation or expansion. Next, PBMC were plated at 1×10^{6} /mL in a 48 well flat bottom plate, in a total volume of 1mL of cRPMI. Cells were stimulated with 1µg/mL of 5-ARU (synthesised by and received from Professor Gurdynal Besra) and 100µM of MG (Sigma) for 18 hours. On day 1 (18 hours after stimulation), 50µL of the media was removed from all the wells and replaced with 50µL of fresh complete media. Replacement of 50µL of the media was carried out, to ensure consistency between these experiments and experiments of MAIT cell expansion and proliferation, which were supplemented with IL-2 on day 1 in 50μ L volume (please see section 2.5.4). On day 4, 500µL of media was removed and replaced with 500µL of fresh complete media. On day five, each well was resuspended by pipetting and 100µL of cells were removed and stained for flow cytometry as per section 2.10.5 MAIT cell frequencies were determined as the percentage of CD3⁺ cells whereas CTV was on MAIT the cell population using geometric mean.

2.5.3. TCR and IL-2 induced MAIT cell expansion & proliferation over 5 days

CTV stained (proliferation) or unstained (expansion) PBMC were plated at 1×10^{6} /mL in a 48 well flat bottom plate, in a total volume of 1mL of cRPMI. Cells were stimulated with 1µg/mL of 5-ARU and 100µM of MG for 18 hours in the presence or absence of the following inhibitors: 2DG (10mM) (Sigma), rapamycin

(20nM) (Sigma), oligomycin (2µM) (Sigma), antimycin A (4µM) (Sigma) or CB-839 (10µM) (Sigma). To assess the effect of 2-amino-2-norbornanecarboxylic acid (BCH) (50mM) (Sigma) on MAIT cell proliferation and expansion, the same experimental procedure was followed, except the complete media was diluted with Hanks balanced salt solution (HBSS)(Gibco) in 1:2 ratio. On day 1 (18 hours after stimulation), 750 µL of the media was removed from all the wells and replaced with 750µL of fresh complete media supplemented with low dose of IL-2 (6.75ng/mL). On day 4, 500µL of media was removed and replaced with 500µL of fresh complete media supplemented with high dose of IL-2 (33.33ng/mL). On day 5, 18 hours post high dose IL-2, 100µL was removed from each well and stained for flow cytometry as per section 2.10.6 (Figure 2.3). MAIT cell frequencies and CTV uptake and dilution were assessed prior to expansion and on day 5 as described in section 2.5.2. To assess the effect of nutrient restriction MAIT cell proliferation and expansion, the cells were stained on (proliferation)/unstained (expansion) and the same protocol was followed as described above, with the exception that for the duration of the experiment cells were kept in their respective nutrient restricted media. i.e., 10mM, 5mM and 1mM glucose or 10mM galactose.



Figure 2.3 Plan of MAIT cell proliferation experiment. Schematic showing the experimental set up to investigate the effect of metabolic inhibitors on MAIT cell proliferation. Image created with BioRender.

2.5.4. TCR & IL-2 induced MAIT cell expansion over 7 days

PBMC were plated at 1×10^6 /mL in a 48 well flat bottom plate, in a total volume of 1mL of cRPMI. A sample of 0.25-0.5x10⁶ cells was stained for flow cytometry prior to expansion of MAIT cells. Cells were stimulated with 1µg/mL of 5-ARU and 100µM of MG for 18 hours. On day 1 (18 hours after stimulation), 50µL of the media was removed from all the wells and replaced with 50µL of fresh cRPMI supplemented with low dose IL-2 media (6.75ng/mL). On day 4, 500µL of media was removed and replaced with 500µL of fresh cRPMI supplemented with high dose IL-2 media (33.3ng/mL). On day 7, each well was resuspended by pipetting and a sample of cells was removed for counting. Next 100µL of cells were removed and stained for flow cytometry as per section 2.10.6. Based on cell counts and MAIT cell percentage in samples pre- and post-expansion, absolute numbers of MAIT cells in the culture were calculated. Fold expansion was calculated by dividing the number of MAIT cells present in the culture on day 7 by the number of MAIT cells prior to stimulation.

2.5.5. TCR & IL-2 induced MAIT cell expansion – 28 days

PBMC were plated at $2x10^6$ cells in a 24 flat well plate in a total volume of 2mL of cRPMI. Cells were stimulated with 1µg/mL of 5-ARU and 100µM of MG for 18 hours. Following the stimulation, on day 1, low dose of IL-2 at 6.75ng/mL was added into each well, followed by a high dose of IL-2 at 33.3ng/mL on day 4,7,11,14,18 and 21. During the high dosage feeding, 1mL of supernatant was removed and replaced with fresh media containing the high dose of IL-2. Expansion of MAIT cells was checked prior to expansion and on day 4,7,11,14,18,21 and 28 by flow cytometry as per table 2.10.5.

2.5.6. MAIT cell expansion for Seahorse, ELISA and real time RT-qPCR

PBMC were plated at $2x10^6$ cells in a 24 flat well plate in a total volume of 2mL cRPMI. Cells were stimulated with 1µg/mL of 5-ARU and 100µM of MG for 18 hours. Following the stimulation, on day 1 low dose of IL-2 at 6.75ng/mL was added into each well, followed by a high dose of IL-2 at 33.3ng/mL on day 4,7,11 and 14. Expanded MAIT cells were harvested between day 13 and 18 and the culture was purified using magnetic isolation as per section 2.6.1 and 2.6.2

2.6. Preparation of expanded MAIT cells for experiments

2.6.1. MAIT cell magnetic labelling

Expanded MAIT cells were removed from the culture between day 13 and 18 for purification. The cells were counted as previously described in section 2.2 and the desired number of cells were removed for isolation. The cells were centrifuged at 300g for 10 min (as all centrifugation steps in this protocol) to remove all media and then spun in Auto MACS running buffer (BSA/EDTA/0.9%Azide/PBS) (Miltenyi Biotec) using 2mL of the buffer per 10⁷ of cells. The supernatant was completely removed, and the cells were resuspended in 100µL of MACS buffer per 10⁷ of cells. 10µL of APC anti-human TCR V α 7.2 conjugated antibody (BioLegend) was added per 10⁷ of cells, mixed well and incubated for 10 min in the fridge. The cells were resuspended in 80µL of MACS buffer per 10⁷ of cells were washed with MACS buffer as above and after removing the supernatant, the cells were resuspended in 80µL of MACS buffer per 10⁷ of cells. Cells were incubated for further 15min in the fridge, washed as above and resuspended in 500µL of MACS buffer.

2.6.2. MAIT magnetic separation

To isolate the labelled MAIT cells, LS columns (Miltenyi Biotec) and a magnetic separator (Miltenyi Biotec) were used. The LS columns were carefully placed into the magnet holders and prepared by rinsing with 3mL of MACS buffer. Next 500µL of the labelled cell suspension was applied to the column, which was then rinsed three times with 3mL of MACS buffer. The fraction of unlabelled cells was collected and discarded. To recover the labelled cells of interest, the column was removed from the separator and placed over a suitable tube. 5mL of MACS buffer as added into the column reservoir and immediately flushed out with the plunger supplied with the column. The cells were spun out of the MACS buffer via centrifugation at 300g for 10mins, resuspended in full media and counted. The purity of the isolated cells was determined by extracellular staining as per section 2.4.3 and run on the Attune NxT flow cytometer. Purity for purifications exceeded 95%.

2.7. MAIT cell molecular analysis

2.7.1. mRNA expression experimental set up

MAIT cells were expanded, harvested and purified as per sections 2.5.5, 2.6.1 and 2.6.2. Isolated MAIT cells were plated at $3x10^6$ per well in a 24 flat bottom plates in a total volume of 1mL of cRPMI and rested overnight. Next, cells were stimulated for 4 hours with TCR microbeads (DynabeadsTM Human T activator CD3/CD28, prepared as per manufacturer's instructions) (ThermoFisher Scientific) in 1:1 ratio of beads to the number of cells and with IL-12/IL-18 at 50ng/mL each (BioLegend) (Figure 2.4) in a total volume of 1.5mL. At the end of the stimulation period, the cells were removed into 2ml sterile tubes (Sarstedt) and centrifuged at 400g for 10 minutes. The supernatant was removed, the pellets were resuspended in 500µL of RNA LaterTM Stabilisation solution (Invitrogen) and stored at 4°C overnight, before freezing at -20°C for long term storage.

For assessment of the effect of BCH inhibition on MAIT cell activation on the molecular level, expanded and purified MAIT cells rested overnight in a flask. The following day, cells were recounted, resuspended in cRPMI and plated at 3x10⁶ per well in a 24 flat bottom plates. BCH (50mM) or HBSS was added to appropriate wells for 1 hour prior to MAIT cell activation, ensuring that the media is diluted 1:1 with HBSS for the duration of the experiment. After 1 hour, MAIT cells were stimulated with TCR Dynabeads and cytokines (IL12/18 at 50ng/ml) in a total volume of 1.5mL for further 4 hours and harvested as described above.



Figure 2.4 Diagram showing experimental plan for analysis of MAIT cell mRNA expression. Experimental set up used to investigate the expression of mRNA by MAIT cells upon stimulation. Image created with BioRender.

2.7.2. mRNA isolation

Isolation of mRNA from MAIT cells was performed using E.Z.N.A Total RNA kit I (Omegabio-tek). Manufacturer's protocol was followed. Briefly, HiBind RNA Mini Column were inserted into the provided 2mL collection tube and labelled. MAIT cells in RNA Later[™] were pelleted at 10,000g for 5 minutes and the supernatant was removed. Next 350µL of TRK Lysis buffer was added to the pellet and the cells were homogenised by pipetting and vortexing, followed by the addition of 350µL of 70% ethanol (Sigma) (Figure 2.5). Suspension was vortexed thoroughly and the total volume of 700 µL of the sample was transferred to the HiBind RNA Mini Column, which was centrifuged at 10,000g for 1 minute. The filtrate was discarded, and the membrane was washed with 500µL of RNA Wash Buffer I, followed by centrifugation at 10,000g for 30 seconds. Next, 500µL of RNA Wash Buffer II was added to the columns. The tubes were centrifuged for 10,000g for 1 minute and the filtrate was discarded. Wash with RNA Wash Buffer II was repeated once more, as above. Then the columns were spun on maximum speed for 4 minutes to dry the HiBind RNA Mini Column membrane. Next the columns were transferred into labelled 1.5mL RNAse free tubes and 40µL of DEPC water

was applied straight to the middle of the membrane. The tubes were centrifuged at maximum speed for 2 minutes to elute mRNA (Figure 2.5).



Figure 2.5 Isolation and quantification of RNA. Diagram depicting steps involved in RNA isolation using E.Z.N.A Total RNA kit I and RNA quantification with Nanodrop2000. Adapted from E.Z.N.A. Total RNA kit I manual.

2.7.3. mRNA quantification

After mRNA isolation, mRNA was quantified using NanodropTM 2000 spectrophotometer (ThermoFisher Scientific). Nanodrop software settings were set up to measure RNA nucleic acids and the instrument was blanked using 1µL of DEPC water. All samples were measured and concentration of RNA (µg/µL) and 260:230 ratios were noted. RNA was diluted to the desired concentration, uniform for each batch of the samples between 35ng-100ng/µL, depending on the initial concentration of RNA in the samples.

2.7.4. cDNA synthesis

Synthesis of cDNA was performed using qScript cDNA Synthesis kit (QuantaBio). Each reaction contained 500ng of RNA. Total volume of each reaction was equal to 20μ L, composed of up to 15μ L of RNA (if RNA was very concentrated this volume was made up to 15μ L with molecular grade nuclease free water), 4μ L of concentrated master mix (tittered primer blend, qPCR optimised dNTP blend, flexible magnesium titration) and 1μ L of qScript reverse transcriptase. The cDNA

reaction was prepared in 0.2mL Corning Thermowell GOLD PCR tubes (Sigma) and ran on G- Storm Thermal Cycler (Gene Technologies Ltd). First the samples were heated up to 22°C for 5minutes, next the temperature was increased to 42°C for 30 minutes and 85°C for further 5 minutes, before the temperature was dropped to 0°C. After cDNA synthesis, samples were diluted 1:5 with molecular grade water (Sigma) and kept at -20°C.

2.7.5. Realtime RT-qPCR

Real time RT-qPCR was performed using PerfeCTa SYBR Green FastMix Reaction Mix (Green Fastmix, ROX[™]) (QuantaBio) and primers as per table 2.1 below (Sigma). Primers were resuspended in molecular grade water (Sigma) as per manufacturer's instructions to obtain a stock of 100µM, and then further diluted with molecular grade water to 4µM working solution. A master mix of the SYBR Green and primers was prepared, which contained 5µL of SYBR Green Enzyme Mix, 1.25µL of forward primer, 1.25µL of reverse primer, 0.5µL of molecular grade water per sample. 8µL of the master mix was added into appropriate wells in the 96 well reaction plate (MicroAmp Fast 96-well reaction plate (0.1mL) (Life Technologies) and 2µL of the cDNA template was added. After addition of the master mix and cDNA template, the plate was covered with an adhesive coverslip (MicroAmp Optical Adhesive film) (ThermoFisher Scientific) and spun for 5 seconds at 1800rpm. The plate was loaded into a PCR instrument (Applied Biosystems) and appropriate program was used as follows. The temperature was increased to 95°C for 30 seconds, this was followed by the cycle of 95°C for 10 seconds, 57°C for 15 seconds and 70°C for 10 seconds. This cycle was repeated 45 times for all primers and tests. Samples were run in duplicate for each treatment and target. A melt curve was also performed from 60°C to 95°C at 2°C steps to assess primer integrity. Controls genes used included IPO8 (coding for nuclear import protein, which transports miRNA), TBP (coding for TATA-box binding protein; transcription factor) and RPL13A (coding for a ribosomal component), based on a study by Ledderose et al. and recommendations from Doreen Cantrell's laboratory, Dundee (Ledderose et al., 2011). Results were analysed using $\Delta\Delta$ CT method. Average of the three control genes was used to normalise the expression of the gene of interest to the housekeeping genes. Results of the control subjects and people with obesity were further normalised to their own basal expression.

Name	Forward sequence 5'-3'	Reverse sequence 5'-3'
ТВР	GCCAAGAGTGAAGAACAG	GAAGTCCAAGAACTTAGCTG
IPO8	CCACTTGACCTTGACAATAG	CAATCTTCTTCTTTGCCTCTG
RPL13A	GTCTGAAGCCTACAAGAAAG	TGTCAATTTTCTTCTCCACG
SLC2A1	AGTTCTACAACCAGACATGG	CAGGTTCATCATCAGCATTG
SLC2A3	TCACTATTTAGGAGCCTACG	CTATGAAATGCTCATCCTTGG
SLC2A4	CTATGAAATGCTCATCCTTGG	TCAGAATGCCGATAACAATG
SLC3A2	CACTGATTATGACTGTGGAC	CCAGAGCATCCTATTTTAGAG
SLC7A5	ACATTGTGCTGGCATTATAC	CTGTAGGGGTTGATCATTTC
SLC7A8	GTTCAAGACATCTTCACAGC	CCTTTGCATATCTGTACAATCC
SLC43A1	CATGTGTCTGTGTATGTGTG	TTTTCAGTCTCTCACTTCCC
SLC43A2	CTCAACTGCTTCTTTAACTGG	CAGCTGAACTTGATCTTCAC
SLC1A5	GTAAAGATCATCACCATCCTG	CCAGGATCAAGGAGATATGG
SLC38A1	CTTGTTGGTGATCTTCATACC	AGCCCAAATTCTTTGAGTTC
SLC38A2	TGTTCTTTCTGATTGTGGTC	TGCAAGAGTCATTTTCAGTC
НК2	GAAAGCAACTFTTTGAGAAG	CAATGTCTGAGATGTCTTTGG
РКМ	ATGGTTGATATGGTGTTTGCG	ATGTTGATATGGTGTTTCGC
IFNY	CATTCAGATGTAGCGGATA	ATTCATGTCTTCCTTGATGG
IL17A	CATTGGTGTCACTGCTAC	TCGGTTGTAGTAATCTGAGG
IL17F	TGGAATTACACTGTCACTTG	GAACGGAATTCATGGAGAGT
IL22	CTGTGAGCTCTTTCCTTATG	AACGTCTGTGTTGTTATCAG
IL32	AATGCAAAATGCAGAATCAG	GTAGAGGAGTGAGCTCTG
TBX21	CTTTCCAAGAAACCCAGTTC	GTCAACAGATGTGTACATGG
RORC	TTTTGAAGGCAAATACGGTG	AGTGGGAGAAGTCAAAGATG

Table 2.1. Sequences of forward and reverse primers

2.8. ELISA

2.8.1. ELISA Experimental set up

MAIT cells were expanded and purified as per section 2.5.5, 2.6.1 and 2.6.2. Next MAIT cells were counted and plated at $1x10^6$ per well, in 48 flat bottom well plate (Sarstedt) in a total volume of 1mL. To assess the metabolic requirements for MAIT cell function, cells were stimulated with TCR Dynabeads 1:1 bead to cell ratio (DynabeadsTM Human T activator CD3/CD28) (ThermoFisher Scientific) and with IL-12/IL-18 at 50ng/mL each (BioLegend) in the presence or absence of 2DG (10mM), Rapamycin (20nM) and CB-839 (10µM) in cRPMI for 24 hours. To investigate the requirement of amino acid transport via LAT1, cells were stimulated with TCR DynabeadsTM Human T activator CD3/CD28)

(ThermoFisher Scientific) and with IL-12/IL-18 at 50ng/mL each (BioLegend) in the presence or absence of BCH (50mM) for 24 hours. For this experimental set up the 1mL volume of cells consisted of complete RPMI diluted 1:2 with HBSS. Following the 24-hour stimulation, the plate was spun at 300g for 10 minutes and 900μ L of the supernatant was collected and frozen.

2.8.2. ELISA testing

Prior to the testing all the reagents were resuspended as per manufacturer's instructions for IFNy, IL-17, IL-17F, IL-22, IL-32 ELISA kits (DuoSet ELISA development systems; R&D). Capture antibody was diluted to the required working concentration in PBS. NUNC-Immuno 96 well plate was immediately coated with 50µL of the diluted antibody per well (ThermScientific). The plate was sealed and incubated at room temperature overnight. As per figure 2.6, the following day the plate was washed with a wash buffer (PBS & 0.05% Tween20). Each well was filled with approximately 300µL of the wash solution. This process was repeated three times and was followed by gentle tapping of the plate against a clean towel to blot any residual fluid out of the wells. Every wash step in this protocol was performed as described here. Next the plate was blocked using 150μL of reagent diluent (1% bovine serum albumin (BSA) in PBS; filtered) per well and incubated at room temperature for an hour. Next the standards and samples were prepared. Standard curve samples were prepared by adding appropriate amount of the reconstituted standard into a total volume of 1mL of reagent diluent to achieve the recommended concertation of top standard for each separate analyte measured. Next, a six-point serial dilution of the top standard was performed in reagent diluent and for the last standard point reagent diluent was used. If required, samples were diluted in reagent diluent in a 96 well plate. Following the blocking step, the plate was washed and blotted as descried above and 50µL of standards or samples was added to the appropriate wells. The plate was incubated in the fridge at 4°C overnight. Next day, the plate was washed, and the detection antibody was diluted to the recommended concentration. 50μ L of the solution was added into all wells and the plate was incubated for two hours at room temperature and then washed. HRP-Avidin was diluted to the recommended concentration and 50µL of the solution was added into the wells. The plate was incubated at room temperature, protected from light for 20 minutes, which was followed by the wash step. Next TMB substrate

solution was prepared by adding equal volumes of Substrate A and Substrate B (BioLegend) and 50μ L of the solution was added to the wells. The plate was incubated at room temperature protected from light up to 20 minutes and the reaction was stopped using 50μ L of stop solution (1M Sulphuric acid) (Sigma-Aldrich). The plate was lightly tapped to ensure mixing of the reagents. Optical density was determined using a microplate reader (ThermoFisher) and measurements were taken at 450nm and 540nm. Raw data was analysed in Prism 9 Software.



Figure 2.6 ELISA steps. Diagram depicting steps involved in ELISA testing using R&D Duoset ELISA kit. Image created with BioRender.

2.9. Seahorse

2.9.1. Seahorse experimental set up

MAIT cells were expanded and purified as per section 2.5.5, 2.6.1 and 2.6.2. Once successfully expanded, the cells were removed from the culture and plated in a 24 well plate at 2x10⁶ per well in a total volume of 1mL. To investigate the glycolysis levels in control individuals and people with obesity, cells were plated in complete RPMI and stimulated with IL-12/IL-18 each at 50ng/mL and TCR microbeads at 25ng/mL (T cell Activation/Expansion kit, prepared as per manufacturer's instructions) (Miltenyi Biotec) for 18 hours. To investigate the impact of inhibition of amino acid influx, the cells were plated in complete media which was diluted 1:2 with HBSS. To the inhibitor well, BCH (50mM) was added and the cells were stimulated as above.

2.9.2. Seahorse protocol

Seahorse XFe96 sensor cartridge (Agilent) was hydrated with 200μ L per well with Seahorse XF Calibrant Solution (Agilent) and left at 37° C in a non-CO₂ incubator for 18 hours prior to use. Seahorse XF phenol-free DMEM media, pH 7.4 (Agilent) was prepared by adding glucose and glutamine to final concentration of 10mM and 2mM respectively. This media has limited buffering capacity, therefore is sensitive to any changes in the pH, which is used to calculate ECAR. Media was kept at 37°C. Next Cell-Tak (Agilent) for 96 well plate was prepared at a final concentration of $3.5\mu g/cm^2$. 2979 μ L of sodium bicarbonate (Sigma) at 0.1M pH8, 16.7µL of cell-tak at 1.85mg/ml (Agilent), and 4.65µL of 1M sodium hydroxide (Sigma-Aldrich) were mixed and 25µL of the solution was added to each well of the XF96 cell culture microplates plate (Agilent). The plate was incubated for 20 minutes at room temp. Next Oligomycin (Sigma), FCCP (Sigma), Rotenone (Sigma), 2DG (Sigma) inhibitors were prepared as per table 2.2 and kept on ice. Then, cells were harvested, counted and the exact number of cells was removed from each sample treatment. The cells were pelleted and washed twice in 500µL of pre-warmed seahorse DMEM media at pH 7.4 (Agilent) at 1800rpm for 8 minutes and resuspended at $2x10^5$ cells per 180μ L of pre-warmed seahorse media. The cells-tak was washed with sterile deionised water (Baxter), by adding 170-200µL of water into each well and aspirating water with p200 multichannel pipette. Ensure no droplets of water remain in the wells before proceeding. The samples were plated in quadruplicates at 2x10⁵ cells per 180µL per well. 180µL of seahorse media was also added to all unused wells. The plate was spun at 300g without break for 3 minutes and the wells were check under the microscope to ensure even monolayer of the cells in the wells. The plate was left in a non-CO₂ incubator, while the inhibitors were loaded into the injection ports of the sensor cartridge in the following order - oligomycin, FCCP, Rotenone, 2DG as per table 2.2, using template guide. The plate was calibrated on the Seahorse XF instrument (Agilent), the location of the sample wells was defined, and protocol was selected. The protocol used to run was defined as adding 4 injections of the inhibitors into the cells in the order of oligomycin, FCCP, Rotenone and 2DG. Once the calibration was completed, the plate with the cells was loaded onto the instrument. The results were exported into prism file and analysed in prism.

Table 2.2. Seahorse inhibitors injection. A table showing injection ports sequence and final concentrations of the inhibitors in the seahorse wells.

Ports	Compounds	Working	Starting well	Injecting	Final
		stock	volume	Volume	concertation
Port A	Oligomycin	20µM	180	20	2µM
Port B	FCCP	10µM	200	22	1µM
Port C	Rotenone	1.21µM	222	20	0.1µM
Port D	2DG	393mM	240	20	30mM

2.10. Flow Cytometry

2.10.1. MAIT cell glycolytic metabolism

Isolated PBMC plated at 0.5-1x10⁶ (depending on the experiment) in a 96 round bottom well plate (Sarstedt) in a total of 250µL of complete RPMI per well for analysis of MAIT cell metabolism associated proteins. To assess expression of GLUT1, CD98, HKII, LDHA and PKM2 and cells were incubated in the media alone (control) or with IL-12/IL-18 each at 50ng/mL and TCR microbeads at 25ng/mL (T cell Activation/Expansion kit, prepared as per manufacturer's instructions) (Miltenyi Biotec) or TCR Dynabeads 1:1 bead to cell ratio (Dynabeads[™] Human T activator CD3/CD28) for 18 hours at 37°C. After 18 hours, the cells were harvested and stained for flow cytometric analysis as per table 2.3 and section 2.10.5 or 2.10.6 using extracellular or extracellular and intracellular staining protocols.

2.10.2. MAIT cell amino acid requirements for activation

To assess MAIT cell cytokine production, PBMC were plated at $0.5-1x10^6$ in a 96 round bottom well plate (Sarstedt) in a total volume of 250μ L. To investigate IFN γ production, PBMC were incubated in cRPMI with 12/IL-18 each at 50ng/mL and TCR Dynabeads 1:1 bead to cell ratio (DynabeadsTM Human T activator CD3/CD28) for 18 hours at 37°C in the presence of 1x of protein transport inhibitor cocktail (Brefeldin A and Monensin) (Invitrogen). To investigate the role of amino acid influx on MAIT cell metabolism and function, PBMC were plated as above in full media diluted 1:2 with HBSS for investigating the role of amino acid influx on MAIT cell metabolism and function. Cells were incubated in cRPMI:HBSS alone (control) or with IL-12/IL-18 (each 50ng/mL) and TCR microbeads at 25ng/mL (T cell Activation/Expansion kit, prepared as per manufacturer's instructions) (MiltenyiBiotec) in the presence or absence of BCH (50mM) for 18 hours at 37°C in the presence of the 1x of protein transport inhibitor

cocktail (Brefeldin A and Monensin) (Invitrogen). After 18 hours, the cells were harvested for extracellular and intracellular flow cytometric staining.

2.10.3. Expression of CD25

To investigate the effect of MAIT cell activation on the high affinity IL-2 receptor (CD25) expression PBMC were plated at 0.5x10⁶ in a 96 round bottom well plate (Sarstedt) in a total of 250µL of complete RPMI per well. Cells were incubated in the media alone (control) or were stimulated with TCR Dynabeads 1:1 bead to cell ratio (Dynabeads[™] Human T activator CD3/CD28) or in combination with IL-12/IL-18 (each 50ng/mL) and TCR Dynabeads in the presence or absence of 2DG (10mM) for 18 hours at 37°C. After stimulation, the cells were harvested and stained for flow cytometric analysis as per table 2.3 and section 2.10.5 using extracellular staining protocols.

2.10.4. Antibodies and panels used in this thesis

Staining used in this thesis is detailed as per table 2.3. Whereas details of the clones of flow cytometric antibodies are detailed in table 2.4.

	BL1	YL1	YL4	RL1	RL3	VL1	VL2	VL4
Excitation laser (nm)	480	516	516	638	638	405	405	405
Emission filter (nm)	530/30	585/16	780/60	670/14	780/60	440/50	512/25	530/30
MAIT Purification				Va7.2				
MAIT Frequencies			Va7.2	CD161*	CD161*		CD3	
сти		CD3	Va7.2	CD161	CD8	CTV		
Viability		CD3	Va7.2	CD161	CD8		Viability dye	
LDHA, GLUT1,HKII	LDHA	GLUT1	Va7.2	нкіі	CD161		CD3	
PKM2		PKM2	Va7.2	CD161			CD3	
CD98	CD98		Va7.2	CD161			CD3	
pS6		pS6		MAIT Tetramer			CD3	
IFNγ	IFNγ		Va7.2	CD161			CD3	
Kynurenine			CD3	MAIT Tetramer		Kynurenine		
CD25		CD3	Va7.2	CD161				CD25

Table 2.3. Flow cy	ytometry pane	els for studying	of MAIT cells
--------------------	---------------	------------------	---------------

* CD161 APCCy7 used for only for IL-2 expansion experiments

Antibody	Fluorochrome	Clone	Manufacturer
Vα7.2	PE-Vio770	REA179	Miltenyi Biotec
Vα7.2	PE/Cyanine7	3C10	BioLegend
Vα7.2	APC	3C10	BioLegend
CD161	APC	REA631	Miltenyi Biotec
CD161	APC/Cyanine7	NKR-P1A	BioLegend
CD8	APC-Vio770	REA734	Miltenyi Biotec
CD3	PE	REA613	Miltenyi Biotec
CD3	VioGreen	REA613	Miltenyi Biotec
CD3	PE-Vio770	REA613	Miltenyi Biotec
IFNγ	FITC	45-15	Miltenyi Biotec
CD98	FITC Bright	REA387	Miltenyi Biotec
GLUT1	PE	EPR3915	Abcam
нкіі	AlexaFluor647	EPR20839	Abcam
LDHA	AlexaFluor488	EP1563Y	Abcam
PKM2	PE	EPR10138(B)	Abcam
pS6	PE	REA454	Miltenyi Biotec
CD25	BV711	M-A251	BioLegend

Table 2.4. Clones of antibodies used in this thesis

2.10.5. Extracellular and intracellular staining in tubes

Cells were harvested from the culture (between 0.5-1x10⁶), transferred into labelled 5mls FACS tubes (BD Falcon) and washed in 1ml of stain buffer (PBS with 1% (v/v) FCS) at 1800 rpm for 8 minutes. The supernatant was removed, and the cell pellets were stained with the fluorochrome conjugated surface antibodies (as per table 2.3) as per recommendations. Cells were vortexed and incubated in a total volume of 100µL of stain buffer per test, at 4°C for 20-30 minutes in darkness. Next, cells were washed as described above and fixed in a 300µL of a 1X fix working solution (True-Nuclear Transcription Factor Buffer set) (BioLegend) vortexed and incubated for 30 minutes at 4°C. Then, 1mL of 1X Perm Buffer was added to the tubes (True-Nuclear Transcription Factor Buffer set) (BioLegend) and centrifuged at 1800 rpm for 10 minutes. If no further staining was required, the cells were resuspended in 300µL of staining buffer and ran on flow cytometer. If cells required intracellular staining, intracellular antibodies conjugated to the fluorochromes were added to the washed cells (as per table 2.3), as per previous lab optimisation protocol, vortexed and incubated in a total volume of 100µL of 1X Perm Buffer (True-Nuclear Transcription Factor Buffer

set) (BioLegend) per test, at 4°C for 30 minutes in darkness. Next, 1mL of stain buffer was added and the tubes were centrifugated at 1800rpm for 8 minutes. The pellet was resuspended in 300µL of stain buffer for flow cytometry analysis. Unstained and fluorescence minus one (FMO) controls were stained and ran with every experiment. The instruments used for analysis included AttuneNxT (Thermofisher). Data was analysed with FlowJo software.

2.10.6. Extracellular and intracellular staining in plates

Cells were removed from the culture plate (between $0.5-1 \times 10^6$ cells) and placed in a new 96 round bottom well plate for staining. The cells were centrifuged for 5min at 1500rpm to remove media. Next 250µL of staining buffer was added to the wells and the cells were spun as detailed above. The fluorochrome conjugated surface antibodies were added to each well as per recommendations, in a total volume of 50-100µL staining volume, mixed well. The plate was incubated at 4°C for 20-30 minutes in darkness and was washed by adding 150- 200μ L of staining buffer to the wells by spinning at 1500rpm for 5 minutes. The supernatant was discarded. The cells were fixed in 150µL of 1X fix working solution (True-Nuclear Transcription Factor Buffer set) (BioLegend) for 30 minutes, at 4°C in darkness, and then 100µL of 1X Perm Buffer was added to the wells (True-Nuclear Transcription Factor Buffer set) (BioLegend) and centrifuged at 1500 rpm for 5 minutes and the supernatant was discarded. If no intracellular staining was required, cells were resuspended in 200µL of staining buffer, transferred to tubes, topped up to 300µL total volume and ran on flow cytometer. If staining for intracellular targets, after discarding the supernatant intracellular antibodies conjugated to the fluorochromes were added to the cells in a master mix of 100µL per well, which was made up in 1X Perm Buffer (True-Nuclear Transcription Factor Buffer set) (BioLegend). The plate was incubated at 4°C for 30 minutes in darkness. Next 150µL of staining buffer was added into each well and the cells were spun at 1500rpm for 5minutes. The supernatant was removed, and the cell were resuspended in 200µL of staining buffer, next they were transferred to tubes, topped up to 300µL of staining buffer and ran on included AttuneNxT (ThermoFisher Scientific) with unstained and FMO controls. Data was analysed with FlowJo software.

2.10.7. Gating of MAIT cells

PBMC isolated as per section 2.3 were stained for extracellular MAIT cell surface markers as per section 2.10.5 or 2.10.6 and table 2.3. First lymphocytes were identified and gated using forward and side scatter. Next doublets were excluded using FSC-area and FSC-height parameters (Figure 2.7). This allowed for further characterisation and identification of MAIT cells among the lymphocyte population using the specific fluorochrome bound antibodies and/or MAIT cell tetramer. MAIT cells were first identified by the positive expression of CD3, a surface T cell marker. From the CD3 positive population, MAIT cells were identified by their high expression of CD161 molect84.3 nd the co-expression o $V\alpha7.2$ TCR on their cell surface (Figure 2.7) or by binding to MAIT ell tetramer

(Figure 2.8).



Figure 2.7 MAIT cell identification using flow cytometric antibodies. Identification of MAIT cells using extracellular antibodies specific to CD3, CD161 and V α 7.2 conjugated to appropriate fluorochromes.



2.10.8. Viability staining

Cells were stained with a fixable viability dye eFluorTM506 (FVD) (eBioscience, Invitrogen) prior to extracellular or intracellular staining. Cells (between 0.5- 1×10^6) were moved to a new 96 well round bottom plate for staining, spun at 1500rpm for 5 minutes. The supernatant was removed, the cells were washed twice with 250µL of protein free and azide free PBS by centrifugation at 1500 rpm for 5 minutes and the supernatant was decanted. FVD dye was diluted 1:1000 in PBS and the cells were quickly resuspended in 100µL of the working solution. The plate was incubated in the refrigerator, protected from light for 30 minutes. Next the cells were washed twice with 150µL of flow staining buffer, centrifuging at 1800rpm for 5 minutes. After this step, other antibody-based flow cytometry staining was performed as per section 2.10.6.

2.10.9. Kynurenine Assay

Kynurenine (Sigma) stock of 800 μ M was made up fresh in HBSS. 2 aliquots of the stock were prepared where 1 was kept at 37°C whereas the other was kept at 4°C. Also, 2 aliquots of HBSS were prepared and kept at 37°C or 4°C prior to the beginning of the assay. Cryopreserved cells or expanded MAIT cells were counted as per section 2.4. First, 1x10⁶ cells per condition were stained with extracellular flow cytometry antibodies (without the fixation and permeabilisation step) as per section 2.10.5. After staining, cells were resuspended in 200 μ L per treatment of pre-warmed HBSS and kept in the water bath at 37°C or in 200 μ L of

cold HBSS and kept on ice. For the first control, cells and the reagents were kept at 37°C. To the 200µL of cells, 200µL of HBSS was added and kept at 37°C. For the second control, cell and reagents were kept at 4°C. To the 200µL of cells, 100µL of HBSS and 100µL of kynurenine was added and kept at 4°C. For kynurenine uptake, cells and reagents were kept at 37°C. 100µL of Kynurenine and 100µL of HBSS were added to the 200µL of the cells and kept at 37°C. To investigate the effect of BCH on kynurenine uptake, cells and reagents were kept at 37°C. 100µL of HBSS, 100µL of Kynurenine and 100µL of BCH at 25mM was added to the 200µL of the cells. The addition of the warm reagents was performed sequentially in the water bath, whereas the addition of the cold reagents was performed on ice at the same time. The cells and reagents were incubated for 4 minutes and the reaction was stopped by adding 125µL of 4% paraformaldehyde (PFA) (Sigma) and incubated for further 30 minutes at room temperature protected from light. Next the cells were washed twice in PBS buffer with 0.5% of BSA and resuspended in 300µL of the buffer for flow cytometry. Kynurenine was read of the violet laser at 405nm with 440/50nm filter.

Chapter Three

Results

Functional Characterisation of MAIT cells
3.1 Introduction

3.1.1. MAIT cells

MAIT cells are a non-classical subset of T lymphocytes in humans. To date, MAIT cells have been identified at multiple sites, including the liver, where they comprised up to 45% of total T cells (Dusseaux et al., 2011), and peripheral blood where they represent up to 10% of circulating T cells (Gherardin et al., 2018b; Le Bourhis et al., 2010). In humans, MAIT cells were also found in lungs, female reproductive tract, skin or adipose tissue (Carolan et al., 2015; Gibbs et al., 2017; Hinks et al., 2016). Peripheral blood MAIT cells are typically CD8⁺ or double negative, with a very small population expressing CD4 (Gherardin et al., 2018b; Kurioka et al., 2017; Walker et al., 2012). They also display an effector-memory phenotype, as identified by the lack of CCR7, CD62L, CD45RA (Dusseaux et al., 2011; Gold et al., 2013).

MAIT cells are defined by the expression of a highly conserved, semi-invariant $\alpha\beta$ T cell receptor, which is accompanied by high expression of CD161 (C-type lectin-like receptor) molecules (Dusseaux et al., 2011; Lepore et al., 2014; Porcelli et al., 1993; Tilloy et al., 1999). In humans, the alpha chain is composed of TRAV1-2 (V α 7.2) and TRAJ33/12/20 (J α 33/12/20) (V α 19- J α 33 in mice) and is paired with a limited number of beta chains (Lepore et al., 2014; Porcelli et al., 1993; Tilloy et al., 1999). The confined TCR repertoire indicates a narrow antigen range, suggesting that MAIT cell response can be very rapid. The alpha chain rearrangement was first identified by Porcelli et al. in 1993, during analysis of TCR expression by human peripheral blood CD4⁻ CD8⁻ cells (Porcelli et al., 1993). Later Tilloy et al. described a homologous V α 7.2-J α 33 in the murine T cell repertoire, suggesting that this TCR receptor is highly conserved among species (Tilloy et al., 1999). The MAIT cell TCR recognises MR1, another highly conserved molecule across species. MR1 is associated with B2-microglobulin and is expressed by antigen presenting cells or epithelial cells (Gold et al., 2010; Treiner et al., 2003; Yamaguchi and Hashimoto, 2002). The type of ligand presented by the MR1 molecule remained unknown for a long time, however some studies have suggested they are expressed during microbial infections (Gold et al., 2010; Le Bourhis et al., 2010). Landmark study by Kjer-Nielsen et al. in 2012 finally revealed that MR1 presents non-peptide antigens to MAIT cells, which were identified as vitamin B metabolites produced by various strains of bacteria (Kjer-Nielsen et al., 2012). During that study, the primary ligand which bound to MR1 molecule was named as 6-formyl pterin (6-FP), a derivative of folic acid. Binding of 6-FP to MR1 lead to increase in the expression of the antigen presenting complex on the cell surface, but it didn't induce MAIT cell activation (Kjer-Nielsen et al., 2012). Upon closer investigation of the riboflavin biosynthesis, it's precursor 5-A-RU was pinned as the key intermediate, which induced MAIT cell activation. Free amine of 5-A-RU binds non-enzymatically to methylglyoxal, metabolite of glycolysis, to form a more potent 5 OP-RU antigen. 5 A-RU can also bind with glyoxal to form less potent 5-(200x0ethyludeneamino)-6-D-ribitylaminouracil (5-OE-RU) (Corbett et al., 2014). Riboflavin is synthesised by certain bacteria and yeast such as Staphylococcus or Candida species, but not by humans, making it a unique antigen capable of inducing a response to variety of microbial infections (Le Bourhis et al., 2010). Up to date, a limited number of other MAIT cell antigens have been identified. Some drugs were found to bind to the MR1 molecule such as derivatives of salicylic acid or metabolites of an anti-inflammatory drug diclofenac but these have limited effects on MAIT cell activation (Keller et al., 2017). Activation of MAIT cells via their engagement with MR1 has been shown to be indispensable during infections with riboflavin synthesising microorganisms (Figure 3.1). This has been elegantly shown by Le Bourhis et al. during infection with E. coli, where addition of anti-MR1 antibodies, prevented MAIT cell activation (Le Bourhis et al., 2010).

MAIT cells are activated independently of their TCR via cytokine stimulation (Figure 3.1). Indeed, MAIT cells were shown to express a variety of cytokine receptors including IL-2R, IL-7R, IL-12R, IL-15R, IL-18R and IL-23R, suggesting they can respond to a range of stimuli (Figure 3.2) (Dusseaux et al., 2011; Gracey et al., 2016; Sattler et al., 2015; Wallington et al., 2018; H. Wang et al., 2019; Wilgenburg et al., 2018). Study by Ussher *et al.* and van Wilgenburg *et al.* reported that MAIT cells stimulated by IL-12 or IL-18 alone do not modulate expression of IFN γ , however combination of these cytokines enhanced production IFN γ (Ussher et al., 2014; Van Wilgenburg et al., 2016). Van Wilgenburg *et al.* has explored the effect of a larger number of cytokines on IFN γ production by MAIT cells. They found that incubation of the cells with a combination of IL-12 and IL-15 led to similar IFN γ production as the IL-12/IL-18 stimulation, whereas combination of IFN- α or - β with IL-12, IL-15 or IL-18 led to

variable results (Van Wilgenburg et al., 2016). MAIT cells were also shown to be activated independently of TCR engagement in a mouse study, which investigated the role of MAIT cells in influenza infections (Wilgenburg et al., 2018). MR1 deficient mice who received adoptive transfer of MAIT cells have shown no significant difference in their activation in comparison to the wild type mice, whereas deletion of receptors for IL-12, IL-18, IL-15 and IFN α cytokines significantly impacted MAIT cell activation (Wilgenburg et al., 2018). TCR independent activation of MAIT cells is very beneficial in response against pathogens, which don't produce antigens recognised by these cells. These include viruses or bacteria and yeast which don't use riboflavin synthesis pathway. Due to this mechanism, MAIT cell can provide a much greater protection to the host. Although MAIT cells can be activated in a TCR dependent or independent manner, it is very likely that in vivo they combine both signals where available, to reach their full effector capacity. Studies in vitro have shown that stimulation of MAIT cells using TCR and cytokine stimulation can enhance their activation. For example, Gracey et al. have shown that MAIT cells primed with IL-7 prior to TCR stimulation significantly enhanced their cytokine and granzyme B production (Gracey et al., 2016).



Figure 3.1. MAIT cell activation. MAIT cells can be activated via TCRdependent or independent manner. This results in MAIT cell proliferation and production of effector molecules such as cytokines and cytotoxic mediators (Pisarska et al., 2020).

PLZF is crucial transcription factor required for the innate function and cytokine production of MAIT cells, however it is co-expressed with multiple other transcription factors including RORyT governing type17 response; T-bet, Eomesodermin (Eomes) and B-lymphocyte-induced maturation protein -1 (Blimp-1) associated with Type 1 immunity and CAAT/enhancer-binding protein gamma $(C/EBP\delta)$ required for the expression of appropriate tissue homing molecules (Koay et al., 2016; C. H. Lee et al., 2018; Leeansyah et al., 2015) (Figure 3.2). In contrast to T cells, MAIT cell RORyT and T-bet were shown to be co-expressed (Kurioka et al., 2015; Leeansyah et al., 2015). Expression of T-bet implicated that MAIT cells might engage in type 1 immune response by producing IFNy. Upon stimulation MAIT cells indeed significantly upregulated their IFNy production (Dusseaux et al., 2011; Van Wilgenburg et al., 2016). Similarly, to CD8 T cells, activation of MAIT cells led to higher expression of cytotoxic proteins, which in T cells were controlled by T-bet, Eomes and Blimp-1 transcription factors. Upon stimulation MAIT cells produced granzyme B, granzyme A, granzyme K and perforin (Dusseaux et al., 2011; Kurioka et al., 2015; Van Wilgenburg et al., 2016), whereas granulysin was expressed but the levels were not modified upon activation (Le Bourhis et al., 2013; Sobkowiak et al., 2019). This indicates direct killing capabilities of MAIT cells. Indeed, MAIT cells primed with E. coli were shown to kill antigen presenting cells infected with this microorganism in an MR1 dependent manner (Kurioka et al., 2015). MAIT cells also expressed RORyT and IL-17, markers typical for type 17 phenotype (Dusseaux et al., 2011; Hinks et al., 2019; Lamichhane et al., 2019). IL-17 is a major pro-inflammatory cytokine, implicated in multiple immune mediated diseases. It was originally identified as a cytokine produced by the cells of the adaptive system, namely Th17 cells, however as new data emerged, it became clear that MAIT cells also contribute to the production of IL-17 in many inflammatory diseases (Pisarska et al., 2020). Similarly, to conventional T cells, MAIT cells express IL-23R, which is known to facilitate IL-23 signalling and differentiation of cells towards type 17 profile. It is not yet fully understood whether IL-23 may contribute to the transition of MAIT cells to type 17 cells, however IL-23 acts directly on the cells and enhances their proliferation (Raychaudhuri et al., 2020; H. Wang et al., 2019). Other less described cytokines expressed by MAIT are IL-17F and IL-22, of which both are members of the IL-17 family. Lu et al. who investigated the role of MAIT cells in pneumonia in children, have reported higher expression of IL-17F producing

MAIT cells in the samples of bronchial lavage versus blood in children with infection (Lu et al., 2020). On the other hand, Gibbs *et al.* identified IL-17 and IL-22 producing MAIT cells in mucosa of the female genital tract and suggested this phenotype may be protective against microbial infections (Gibbs et al., 2017).



Figure 3.2. Phenotype of MAIT cells. MAIT cells express V α 7.2 TCR receptor which is accompanied by a range of cytokine and chemokine receptors. In addition, they express a combination of transcription factors and cytokines which are typical for responses of Th1, Th17 and innate cells (Pisarska et al., 2020).

Activation of MAIT cells has been shown to induce their expansion and proliferation, although the data is still very limited. Studies which have investigated MAIT cell responses in bacterial infections, reported that MAIT cells are capable of expansion upon activation (Howson et al., 2018; Wong et al., 2019). Howson et al. has shown that during infection of human subjects with S. enterica, MAIT cells displayed clonal expansion which was followed by a contraction phase (Howson et al., 2018). Similar findings were reported by Wong et al. who investigated the impact of *M. Tuberculosis* on MAIT cell function and has shown an oligoclonal expansion of MAIT cells in response to the infection (Wong et al., 2019). Other studies measured MAIT cell proliferation using Ki67 marker in order to compare their activation and have shown that MAIT cells from individuals with IBD or liver cirrhosis displayed more proliferative in comparison to healthy controls (Hegde et al., 2018; Serriari et al., 2014). Recently, Gutierrez-Arcelus et al. have shown that MAIT cells can proliferate in response to CD3/CD28 bead stimulation over 5 days of culture as depicted by carboxyfluorescein succinimidyl ester (CFSE) assay (Gutierrez-Arcelus et al.,

2019). Up to date, the control of MAIT cell expansion and proliferation has not been explored in detail.

3.1.2. MAIT cells in obesity

The impact of obesity on MAIT cell phenotype and function still remains largely unexplored. As new studies emerged, it became more evident that obesity is associated with alterations in MAIT cell distribution and a switch to more proinflammatory phenotype. Magalhaes et al. and Carolan et al. who assessed MAIT cells in the peripheral blood of people with obesity, revealed a significant decrease in their frequencies in comparison to the control individuals, whereas study by Li et al. reported no difference in the percentage of MAIT cells between the two groups (Carolan et al., 2015; Li et al., 2020; Magalhaes et al., 2015). The inconsistency in the frequencies reported by these studies may be due to the difference in the BMI in the cohorts of people with obesity. Cohorts used by Magalhaes et al. and Carolan et al. had much higher BMI than the subjects used in the study by Li et al. In fact, frequencies of MAIT cells were shown to be negatively correlated with BMI, where in some people with obesity MAIT cells have not been detected (Magalhaes et al., 2015), however this requires further investigation as Carolan et al. reported no association between those two parameters (Carolan et al., 2015). People with obesity who underwent bariatric surgery, were shown to increase the population of their MAIT cells in the peripheral blood, suggesting that this defect is reversible upon weight loss (Magalhaes et al., 2015).

In addition, MAIT cells from people with obesity were shown to be largely dysfunctional and displayed dysregulation in the proportion of the produced cytokines. Carolan *et al.* reported a decrease in the IFN γ expression by peripheral blood MAIT cells from people with obesity, whereas other studies have shown no apparent defect in the production of this cytokine (Carolan et al., 2015; Magalhaes et al., 2015; Toubal et al., 2020). In contrast, peripheral blood and AT MAIT cells from people with obesity displayed strongly enhanced IL-17 production in comparison to the control individuals (Carolan et al., 2015; Magalhaes et al., 2015; O'Brien et al., 2020). Similar results were obtained in a mouse study by Toubal *et al.* where high-fat diet (HFD) mice had a higher proportion of IL-17 producing MAIT cells in the ependymal fat in comparison to

mice fed on normal diet (Toubal et al., 2020). Collectively this suggests that MAIT cells from people with obesity may have a defect in the production of IFNy, however the enhancement in IL-17 may be contributing to the inflammation observed in obesity. Indeed Toubal et al. has demonstrated a close relationship between MAIT cells and macrophages, which are present in the AT (Toubal et al., 2020). Coculture of MAIT cells with M1 macrophages, significantly enhanced MAIT cell IL-17 production, whereas activated MAIT cells promoted polarisation of macrophages into M1 phenotype (Toubal et al., 2020). This mechanism may possibly be contributing to the sustained AT inflammation observed in obesity. Although gut dysbiosis has been reported in people with obesity, very little is known about possible causes and mechanisms which are involved in the development of this condition (Boulangé et al., 2016; Muscogiuri et al., 2019). Recently, Toubal et al. has shown that MAIT cells contribute to the alteration in the gut microbiota. In HFD mice, MAIT cells were shown to increase gut inflammation and leakiness, whereas transfer of the microbiota from HFD mice induced dysfunction of the mucosa, leakiness of the gut and enhanced inflammation of the gut in the recipient mice (Toubal et al., 2020).

As immunometabolism became a focal point of many immunological studies, a recent study focused on the relationship between MAIT cell metabolism and their dysfunction in obesity. O'Brien *et al.* have shown that MAIT cells from people with obesity had elevated levels of mitochondrial reactive oxygen species (mROS) and mitochondrial membrane potential in comparison to the lean controls. Reducing mROS using MitoTEMPO and MitoQ antioxidants, was shown to reduce IL-17 production by MAIT cells from people with obesity, suggesting that metabolic alterations in MAIT cells may be contributing to their dysfunction (O'Brien et al., 2020).

These studies show that obesity significantly affects MAIT cell frequencies and function in the peripheral blood as well as other tissues including AT. The alterations in their phenotype are shown to support systemic inflammation in obesity, which may further lead to development of conditions such as T2DM. More studies are required to elucidate the causes of the altered MAIT cell function and to identify potential targets for treatment to ameliorate their pro-inflammatory responses in obesity and regain homeostasis.

3.2. Specific aims of this chapter

The specific aim of this chapter was the characterisation of MAIT cells in cohorts of adults with a healthy or obese BMI, and therefore to:

- 1. Enumerate MAIT cell frequencies in peripheral blood
- 2. Determine the proliferative capacity of MAIT cells
- 3. Characterise MAIT cell cytokine profiles

3.3. Results

3.3.1. MAIT cell identification

MAIT cells have been previously reported to represent 1-10% of peripheral blood T cells in healthy individuals (Carolan et al., 2015; Gherardin et al., 2018b; Le Bourhis et al., 2010). To confirm their frequencies in the peripheral blood of our cohorts of adults with a healthy or obese BMI we utilized multi-colour flow cytometry.

The gating strategy used to identify MAIT cells is shown in materials and methods figure 2.7. Briefly, PBMC isolated from control individuals were stained using antibodies against surface antigens conjugated to appropriate fluorochromes namely CD3 (VioGreen), CD161 (APC) and V α 7.2 (PeCy7). To identify MAIT cells, we first selected lymphocytes based on their morphology (forward and side scatter), next doublets were excluded using forward scatter (FSC)-area and FSC-height parameters. Finally, MAIT cells were first identified by the positive expression of CD3, high expression of CD161 molecule and the co-expression of V α 7.2 TCR on their cell surface (Figure 3.3 A&B). MAIT cell frequencies in the cohort of control individuals ranged from 0.53% to 13.7% and displayed a mean of 3.2% (Figure 3.3 A&B).



Figure 3.3 MAIT cell identification using flow cytometry. (A) Scatter plot of MAIT cell frequencies in the peripheral blood of control individuals (n=39). (B) Representative dot plots showing MAIT cells from 3 control individuals as percentage of CD3⁺ cells. MAIT cells identified as V α 7.2⁺ and CD161^{hi}.

3.3.2. People with obesity display lower MAIT cell frequencies in the peripheral blood

Previous work from our lab and others has shown obesity related alterations in the frequencies and functions of a range of immune cells, including MAIT cells (Carolan et al., 2015; Magalhaes et al., 2015). After establishing the range of MAIT cell frequencies in the control individuals (Figure 3.3), next we investigated the frequencies of these immune cells in people with obesity.

People with obesity had significantly lower frequencies of MAIT cells in the peripheral blood, in comparison to the control individuals (Figure 3.4). MAIT cell frequencies in people with obesity ranged from 0.11% to 4.9% and had a mean of 1.2%. This data is in line with previously published studies, which have shown a significant decrease in circulating MAIT cells in people with obesity (Carolan et al., 2015; Magalhaes et al., 2015).



**

Obese -

3.3.3. TCR stimulation induces MAIT cell proliferation

Proliferation of lymphocytes plays a crucial role in response to microbial infections. MAIT cell proliferation remains largely unexplored and to date very little evidence is recorded in the literature. Recently, Gutierrez-Arcelus et al. investigated the function and proliferation of multiple innate immune cells including MAIT cells and suggested that innate immune cells are capable of proliferation in response to TCR stimulation, albeit to a much lower extend than cells of the adaptive immune response (Gutierrez-Arcelus et al., 2019). The proliferation capacity of MAIT cells was also demonstrated in a study by Howson et al.. This group utilised an in vivo approach of controlled infection in humans with S. Paratyphi A. They found that at the peak of infection, MAIT cells from the peripheral blood of the participants were in a state of proliferation, which they measured by Ki67 (Howson et al., 2018). Although the proliferation capacity of MAIT cells has been identified in response to a general stimulation, we decided to explore MAIT cell proliferation in response to the MAIT cell specific antigen, 5-ARU in combination with MG. 5-ARU has been identified as an intermediate in the riboflavin metabolism pathway, and was shown to combine with MG, a byproduct of glycolysis to form a highly 5-OP-RU (Corbett et al., 2014).

MAIT cell expansion was assessed by investigating the proportion of MAIT cells expressed as percentage of all T cells present in the culture five days after activation. Stimulation of PBMC with 5-ARU-MG resulted in very limited, nonsignificant expansion of MAIT cells by day five (Figure 3.5 A&B). To investigate this further, MAIT cell proliferation was also assessed using CTV proliferation assay. CTV is a fluorescent dye, which allows the traceability of cell proliferation, through dye dilution and analysis using flow cytometry. CTV stained PBMC were activated with 5-ARU-MG and incubated for 5 days. Although the expansion of MAIT cells was shown to be limited (Figure 3.5 A&B), CTV proliferation assay revealed that MAIT cells were dividing (Figure 3.5 C&D). This data demonstrates that in response to a pathogen derived antigen MAIT cells display limited expansion, but enhanced proliferation However, the modest increase in the proportion of MAIT cells in the culture suggests limited capability to expand in this experimental setting, possibly due to the lack of other signals which could be originating from other immune and non-immune cells during active bacterial infection.



Figure 3.5 MAIT cells proliferate in **Fesponse to 5 ARU-MG**. (A) Scatter plot of MAIT cell proliferation from control subjects represented as a percentage of CD3⁺ population. (B) Data depicts MAIT cells prior to proliferation (pre-proliferation) and on day 5 of p <u>Sample Name</u> <u>Subset Name</u> <u>Count</u> <u>Geometric Mean</u>: <u>Comp-VL1-A</u> proliferation) or stimulated (day **Fer**) **5 CARU-MG** (100µL) **1 COUNT 1 COUNT COUNT**

3.3.4. TCR triggering upregulates expression of IL-2 receptor on MAIT cell surface

Stimulation of MAIT cells has been shown to enhance the expression of multiple cytokine and chemokine receptors such as IL-2R, IL-12R, IL-18R, CCR2, CCR4 or CCR6 (Chen et al., 2019; C. H. Lee et al., 2018; Sattler et al., 2015; Van Wilgenburg et al., 2016). Binding of the specific ligands to their receptors can shape the immune response to achieve the optimal protection of the host. IL-2 is known to be an immunomodulatory cytokine that can regulate multiple processes in immune cells such as controlling the fate of CD4⁺ T cell differentiation, modulation of CD8⁺ T cell function as well as T cell proliferation (Ross and Cantrell, 2018). In the context of MAIT cell proliferation, next we investigated whether stimulation of MAIT cells through TCR alone or in combination with IL-12/IL-18 cytokines for 18 hours can induce high affinity IL-2 receptor (IL-2R/CD25) expression on MAIT cell surface. TCR triggering alone was sufficient to significantly enhance IL-2R expression on MAIT cell surface (Figure 3.6 A&B). Addition of IL-12 and IL-18 in combination with TCR stimulation further enhanced IL-2R expression in comparison to TCR stimulation alone (Figure 3.7 A&B). This data demonstrates a TCR-mediated increase in the expression of IL-2R, which might facilitate the engagement of MAIT cells in proliferation.



Figure 3.6 TCR stimulation of MAIT cells increases IL-2 receptor (CD25) expression. (A) Scatter plot and (B) representative histograms of CD25 (IL-2R) expression by MAIT cells from control individuals, basally or upon stimulation with TCR Dynabeads (1:1 bead to cell ratio) for 18 hours. Statistical analysis performed using paired student's t-test, ***p≤0.001.



Sample Name	Subset Name	Count	Geometric Mean : Comp-VL4-A
N006 DYNA.fcs	MAIT cells	1149	3230
N006 BASAL.fcs	MAIT cells	1539	-2957



Figure 3.7 TCR triggering in combination with IL-12 and IL-18 cytokine stimulation enhances expression of IL-2 receptor (CD25) by MAIT cells. (A) Scatter plot and (B) representative histograms of CD25 expression by MAIT cells from control individuals, basally, upon stimulation with TCR Dynabeads (1:1 bead to cell ratio) with or without IL-12/IL-18 (50ng/mL each) for 18 hours. . Statistical analysis performed using paired student's t-test, **p≤0.01.

3.3.5. IL-2 supplementation enhances MAIT cell proliferation and expansion Stimulation of MAIT cells via their TCR was shown to induce MAIT cell proliferation with a limited expansion (Figure 3.5) and a significant upregulation of IL-2R (Figure 3.6). To further investigate the proliferative capabilities of MAIT cells, we investigated the effect of exogenous IL-2 supplementation in combination with 5-ARU-MG activation on MAIT cell proliferation and expansion. Activation of PBMC with 5-ARU-MG and supplementation with exogenous IL-2 over a period of 28 days resulted in the significant expansion of MAIT cells from day 7 onwards (Figure 3.8 A&B). Expansion of MAIT cells peaked around day 11 after initial stimulation, which was followed by a period of contraction, where despite IL-2 supplementation, the percentage of MAIT cells in the culture decreased (Figure 3.8 A&B). To confirm this further, MAIT cell proliferation was also assessed using CTV, where stained PBMC were stimulated with 5-ARU-MG and IL-2 for five days. It was confirmed that MAIT cells were capable of significant proliferation over a five-day period in the presence of exogenous IL-2, where between five and six generations could be identified with CTV (Figure 3.9). This suggests that additional factors such as IL-2, which may be produced during infection in vivo, are required to support and enhance MAIT cells proliferation.





Β.

Figure 3.8 MAIT cells expand in response to 5-ARU-MG and IL-2 over 28 days. (A) Representative dot plots of MAIT cell proliferation from 3 control subjects observed over 28 days of culture in response to 5-ARU-MG and IL-2. PBMCs were stimulated with 5-ARU (1µg/mL) and MG (100µL) on day 0 and fed with a low dose of IL-2 (6.75ng/mL) on day 1 followed by high doses of IL-2 (33.3ng/mL) on days 4, 7, 11, 14,18 and 21. (B) Scatter plot of MAIT cell expansion from control subjects (n=10; n=6 on day 14) over 28 days of culture. Statistical analysis performed using one-way ANOVA with Tukey's correction, *p≤0.05, ***p≤0.001.



Figure 3.9 MAIT cells proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot of MAIT cell proliferation as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. PBMCs were unstimulated (pre-proliferation) or stimulated (day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 (n=11). (B) Representative histograms from 3 control subjects of MAIT cell proliferation over 5 days of culture. Statistical analysis performed using paired student's t-test, ****p<0.0001.

3.3.6. MAIT cells from people with obesity display limited expansion capacity in response to 5-ARU-MG and IL-2

It is now well established that people with obesity have reduced frequencies of peripheral blood MAIT cells (Figure 3.4) (Carolan et al., 2015; Magalhaes et al., 2015). In addition to reduction in the frequencies, MAIT cells from people with obesity were shown to have impaired function as assessed by their cytokine production. However, up to now no one has investigated whether the proliferative capacity of MAIT cells in these individuals is affected (Carolan et al., 2015; Magalhaes et al., 2015).

To assess expansion of MAIT cells from individuals with obesity and control subjects over 28 days, PBMC were treated as described in section 3.3.5. MAIT cell expansion from individuals with obesity was significantly lower in comparison to control individuals (Figure 3.10 A&B). To confirm this further, we stimulated PBMC from people with obesity and control individuals with 5-ARU-MG and IL-2 over 7 days and determined their fold expansion. MAIT cells from people with obesity displayed significantly lower fold increase at days 7 after cell activation, further indicating a defect in MAIT cell expansion in people with obesity (Figure 3.10 C). To investigate this in more detail, MAIT cell proliferation was measured using CTV assay as described in section 3.3.5 and has shown no significant difference in proliferation rates of MAIT cells between people with obesity and control individuals (Figure 3.11 A&B). This data suggests that the expansion capacity of MAIT cells from people with obesity is impaired, but their proliferation by day five is yet unaffected.











Figure 3.10 MAIT cells from people with obesity show impaired expansion in response to 5-ARU-MG and IL-2. (A) Representative dot plots of MAIT cell expansion from 3 individuals with obesity, observed over 28 days of culture in response to 5-ARU-MG and IL-2. PBMCs were stimulated with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by high doses of IL-2 (33.3ng/µL) on days 4, 7, 11, 14, 18 and 21 (B) Scatter plots of control (n=10) and people with obesity (n=5) expansion over 28 days of culture. (C) Scatter plot of MAIT cell fold expansion of control subjects (n=8) and people with obesity (n=8) over 7 days of culture. Statistical analysis performed using one-way ANOVA with Tukey's correction or unpaired student's t-test, **p≤0.01, ***p≤0.001, ****p≤0.0001.



Figure 3.11 MAIT cells from people with obesity show similar proliferation profile in response to 5-ARU-MG and IL-2. (A) Scatter plot of MAIT cell proliferation as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture from control individuals (n=11) and people with obesity (n=5). PBMCs were unstimulated (pre-proliferation) or stimulated (day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4. (B) Representative histograms from 2 control and 2 subjects with obesity of MAIT cell proliferation over 5 days of culture. Statistical analysis performed using one-way ANOVA with Tukey's correction ***p≤0.001, ****p≤0.0001.

3.3.7. MAIT cells produce IFNy

Previous studies have reported MAIT cells are potent producers of multiple cytokines including IFN γ and can express a wide repertoire of transcription factors (Koay et al., 2019; Leeansyah et al., 2015). Expression of IFN γ has been shown to be induced with TCR dependent and independent stimulation, suggesting its production by MAIT cells can be induced in a variety of infections (Hinks et al., 2019; Ussher et al., 2014; Van Wilgenburg et al., 2016). Transcription control of MAIT cell cytokine expression has been only investigated in the past five years. Koay *et al.* reported co-expression of T-bet and ROR γ T at stage 3 of MAIT cell maturation in humans (Koay et al., 2016). Lamichhane *et al.* investigated the transcription profile of activated CD8⁺ MAIT cells activated via TCR or cytokines stimulation and reported an upregulation of T-bet and ROR γ T in MAIT cells on both a transcriptional and protein level (Lamichhane et al., 2019).

Here, we sought to confirm these findings in our cohort by assessing the cytokine and transcription factor profile of peripheral blood MAIT cells. Stimulation of PBMC using TCR Dynabeads, IL-12 and IL-18 for 18 hours, significantly increased the percentage of IFN γ producing MAIT cells (Figure 3.12 A&B). To determine whether the enhancement in IFN γ production was due to the direct effect of TCR and cytokine activation on MAIT cells, rather than indirect effect exerted by other cells in the PBMC culture, we stimulated expanded MAIT cells using TCR Dynabeads and cytokines for 4 hours for mRNA analysis or 24hours to measure protein expression. Upon activation MAIT cells significantly upregulated protein and mRNA expression of IFN γ (Figure 3.12 C&D). This was accompanied by significant enhancement in the expression of T-bet (*TBX21*) mRNA in our cohort of control individuals following stimulation (Figure 3.12 E). Overall, these results indicate that TCR and cytokine co-stimulation of MAIT cells increase their expression of T-bet which is likely to facilitate the upregulation of IFN γ synthesis (Figure 3.12 A-D).



Figure 3.12 MAIT cell produce IFNy upon stimulation. (A) Representative dot plots and (B) scatter plot showing MAIT cell IFNy production by control individuals basally or upon stimulation with TCR microbeads (Milteneyi 25ng/mL), IL-12 and IL-18 (50ng/mL each) for 18 hours (n=11). (C) Scatter plot showing IFNy production (pg/mL) by purified MAIT cells basally and upon stimulation with TCR Dynabeads (Dynabeads; 1:1 ratio), IL-12 and IL-18 (50ng/mL each) for 24 hours (n=11). (D) and (E) Scatter plots showing normalized expression of *IFNG* and *TBX21* mRNA by expanded and purified MAIT cells in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 4 hours (n=13). Statistical analysis performed using paired student's t-test, ****p≤0.0001.

3.3.8. MAIT cells produce IL-17A

As described in chapter one, MAIT cells are involved in a range of bacterial and viral infections (Lal et al., 2020; Le Bourhis et al., 2010; Spaan et al., 2016). However, in addition to their role in host protection, MAIT cells have also been implicated in a range of chronic inflammatory diseases, such as obesity or MS (Carolan et al., 2015; Willing et al., 2018). In particular, type 17 MAIT cells have been highlighted as a potentially pathogenic subset. Th17 cells were originally established as the primary producers of IL-17, however as research in this area progressed, MAIT cells were identified to be a significant contributor to the pool of this cytokine during inflammation.

In order to further explore MAIT cell function, next we investigated the ability of MAIT cells to produce IL-17 in our cohort of control individuals. Expanded MAIT cells, activated as per section 3.3.7 significantly increased IL-17A protein production (Figure 3.13 A) and mRNA expression (Figure 3.13 B). This is in line with previously published data, which presented MAIT cells as potent producers of IL-17A (Pisarska et al., 2020).

As previously discussed in section 3.3.6, Koay *et al.* and Lamichhane *et al.* reported co-expression of T-bet and ROR γ T transcription factors by MAIT cells. Having confirmed that *TBX21* (T-bet) mRNA is upregulated upon stimulation of MAIT cells from our cohort of control individuals (Figure 3.12 E), next we sought to investigate whether expression of mRNA coding for ROR γ T follows a similar pattern to *TBX21*. Indeed, stimulation of expanded MAIT cells as per section 3.3.7 significantly increased *RORC* mRNA expression (Figure 3.13 C). This data shows that TCR and cytokine stimulation is sufficient to increase mRNA expression for ROR γ T and enhance expression of IL-17A by MAIT cells.



Figure 3.13 MAIT cells produce IL-17A upon stimulation. (A) Scatter plot showing IL-17A production (pg/mL) by purified MAIT cells from control individuals basally and upon stimulation with TCR Dynabeads (Dynabeads; 1:1 ratio), IL-12 and IL-18 (50ng/mL each) for 24 hours (n=11). (B) Scatter plots showing normalized expression of *IL17A* and (C) *RORC* mRNA by expanded and purified MAIT cells in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 4 hours (n=13). Statistical analysis performed using paired student's t-test, *p≤0.05, ***p≤0.001.

3.3.9. MAIT cell produce less IFNy and more IL-17 cytokines in obesity

Dysregulation of MAIT cell cytokine production has been reported in obesity. MAIT cells were shown to have impaired production of IFNγ and enhanced IL-17 expression in comparison to their lean counterparts (Carolan et al., 2015; Magalhaes et al., 2015). Here we sought to confirm these findings in our cohort of people with obesity.

Expanded MAIT cells from people with obesity or control individuals were stimulated with TCR Dynabeads, IL-12 and IL-18 cytokines for 4 hours for mRNA analysis or 24 hours for the measurement of protein expression. MAIT cells from people with obesity displayed lower IFNy production upon activation in comparison to the controls (Figure 3.14 A). This was also reflected at the transcriptional level where stimulated MAIT cells from people with obesity had lower levels of IFNG mRNA in comparison to the control individuals (Figure 3.14 B). As we have shown that transcription of TBX21 mRNA is significantly upregulated upon MAIT cell activation (Figure 3.12 E), next we investigated whether transcription of *TBX21* is impacted in people with obesity. We showed that the expression of mRNA coding for T-bet upon activation was lower in people in obesity in comparison to the control individuals (Figure 3.14 C). Investigation of IL-17 expression by MAIT cells from individuals with obesity, demonstrated a significant increase in IL-17 protein expression upon activation in comparison to the control individuals (Figure 3.15 A), however no significant difference was observed in the expression of mRNA for IL-17 or RORyT (Figure 3.15 B&C). These results further confirm dysregulation in cytokine production by MAIT cells in obesity and provide additional information regarding the transcriptional phenotypic differences between the two cohorts.



Figure 3.14 MAIT cell production of IFN_Y upon stimulation is reduced in obesity. (A) Scatter plots of IFN_Y protein production by control individuals and individuals with obesity in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 24 hours (control n=28; obese n=21). (B) Scatter plots of *IFNG* (IFN_Y) and (C) *TBX21* (T-bet) mRNA transcription by MAIT cells from healthy and people with obesity basal or upon stimulation with TCR Dynabeads (1:1 bead to cell ratio), IL-12 and IL-18 (50ng/mL each) for 4 hours (control n=13; obese n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, *p≤0.05, ***p≤0.001, ****p≤0.0001.



Figure 3.15 MAIT cells from individuals with obesity, display enhanced IL-17 protein production, but no difference in the transcription of *IL17A* & *RORC* upon stimulation in comparison to the controls. (A) Scatter plots showing IL-17A protein production by control individuals (n=12) and individuals with obesity (n=10) in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 24 hours. (B) Scatter plots showing normalized expression of *IL17A* (controls n=12; obese n=6) and (C) *RORC* (controls n=13; obese n=6) mRNA by expanded and purified MAIT cells from controls and individuals with obesity in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 4 hours. Statistical analysis performed using one-way ANOVA with Tukey's correction, ****p≤0.0001.

3.3.10. MAIT cells produce an expanded repertoire of cytokines

At present, the majority of the studies that investigated the role of MAIT cells in the immune system, have largely focused on the production of the well-studied cytokines, IFN γ , IL-17A and TNF- α (Provine and Klenerman, 2020). Interestingly, it has been reported that MAIT cells are capable of producing other cytokines such as IL-22 and IL-17F (Gibbs et al., 2017; Lu et al., 2020; Toussirot et al., 2018). To further characterise MAIT cells in the control subjects, expression of IL-17F, IL-22 and IL-32 was measured. Expanded MAIT cells were activated as described in section 3.3.9. MAIT cells displayed an increase in their expression of protein and mRNA for IL-17F and IL-22 cytokines (Figure 3.16 A&B; D&E). A more modest increase in IL-32 was noted (Figure 3.16 C), while the transcription of *IL32* mRNA was reduced upon activation (Figure 3.16 F). Overall, these results indicate that although MAIT cells are potent producers of IFN γ , they are also capable of producing other cytokines such as IL-17F, IL-22 and IL-32. As these cytokines play a role in inflammation, the contribution of MAIT cells to the levels produced during an immune response should be further investigated.



Figure 3.16 Stimulation of MAIT cells induces the production of IL-17F, IL-22 and IL-32 cytokines. (A-C) Scatter plots showing concentration of IL-17F, IL-22 & IL-32 in the cell supernatants, produced by the expanded and purified MAIT cell upon stimulation with TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 24 hours (IL-17F n=11; IL-22 n=4; IL-32 n=7). (D-F) Scatter plots showing normalized expression of *IL17F*, *IL22*, *IL33* by expanded and purified MAIT cells in the control population in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 4 hours (IL-17F n=7; IL-22 n=12; IL-32 n=12) Statistical analysis performed using paired student's ttest, *p≤0.05, **p≤0.01, ***p≤0.001.

3.3.11. Individuals with obesity display no significant difference in the expression of other immune cytokines

Investigations around MAIT cells phenotypes in people with obesity has focused on IFN_Y and IL-17, the most studied cytokines produced by these cells. Here, we decided to examine whether alteration of MAIT cell function extends beyond the imbalances observed with IFN_Y and IL-17.

Expanded MAIT were activated as per section 3.3.9. MAIT cells from people with obesity produced IL-17F and IL-32 protein upon stimulation, but no significant difference was observed in the expression between people with obesity and control individuals (Figure 3.17 A&B). mRNA expression coding for IL-17F mirrored its protein synthesis and showed no significant alteration in transcription between the two cohorts (Figure 3.17 C). On the other hand, stimulation of MAIT cells from people with obesity revealed a defect in the downregulation of IL32 mRNA expression in comparison to the control subjects. In fact, MAIT cells from people with obesity produced higher levels of IL-32 protein upon stimulation in comparison to the control individuals, however the difference in the expression was not significant (Figure 3.17 B&D). Investigation of IL22 transcription upon stimulation of MAIT cells from people with obesity revealed no significant difference in the expression between the two cohorts (Figure 3.17 E), however this data must be confirmed on the protein level. Collectively, this data shows non-significant alterations in expression of these three cytokines by MAIT cells from people with obesity. Increasing the number of experiments on MAIT cells from people with obesity and control individuals may provide more definite answers in the future.



Figure 3.17 Stimulation of MAIT cells reveals no difference in *IL17F* and *IL22* mRNA transcription by the cells from healthy and obese individuals but shows impairment in *IL32* transcription. (A-B) Scatter plots showing concentration of IL-17F and IL-32 in the cell supernatants, produced by the expanded and purified MAIT cell upon stimulation with TCR microbeads (Dynabeads; 1:1 ratio) IL-12 and IL-18 (50ng/mL each) for 24 hours (IL-17F control =18, obese n=8; IL-32 control =21, obese n=9). (C-E) Scatter plots showing normalized expression of *IL17F*, *IL22*, *IL32* by expanded and purified MAIT cells in the control population in response to TCR microbeads (Dynabeads; 1:1 ratio) IL-18 (50ng/mL each) for 4 hours (*IL17F* control =7, obese n=4; *IL32* control =12, obese n=5; *IL22* control =12, obese n=5). Statistical analysis performed using paired student's t-test, *p≤0.05, **p≤0.01, ***p≤0.001.
3.4 Discussion

MAIT cells comprise 1-10% of circulating T cells in healthy individuals (Gherardin et al., 2018b; Le Bourhis et al., 2010). In line with these publications, we report that MAIT cell frequencies in the peripheral blood of our cohort of control individuals ranged from 0.5% to 13.7%. To further characterise MAIT cells, next we described their proliferative and expansion capacity in response to TCR activation. Stimulation of MAIT cells with their cognate antigen resulted in limited proliferation. This data aligns with previous studies which demonstrated MAIT cell proliferation in vitro in response to E. coli or 5-OP-RU (Dias et al., 2016; Kurioka et al., 2017; Rahimpour et al., 2015). In agreement with published data, TCR stimulation enhanced IL-2 receptor expression, suggesting it may play a role in MAIT cell function and proliferation (Van Wilgenburg et al., 2016; Vorkas et al., 2018). It has been well established that IL-2 supports T cell proliferation, therefore we hypothesized that the increase in IL-2R combined with antigen stimulation and exogenous IL-2 supplementation may enhance proliferation of MAIT cells. Indeed, here we showed that the combination of TCR and IL-2 signals led to significantly greater expansion and proliferation of MAIT cells in vitro. This suggested that appropriate combination of signals in vivo can potentially drive expansion of MAIT cells in response to a pathogen.

Next, we examined the repertoire of cytokines and transcription factors expressed by MAIT cells upon activation. IFNγ a cytokine which is highly implicated in supporting anti-microbial responses, has been shown to be upregulated upon TCR and/or cytokine stimulation (Dusseaux et al., 2011; Van Wilgenburg et al., 2016; Wang et al., 2018). In line with these findings, we demonstrated that the activation of MAIT cells using TCR, IL-12 and IL-18 led to the transcription of mRNA coding for IFNγ and T-bet, followed by an increase in IFNγ protein. This is consistent with the study published by Lamichhane *et al.* who reported an increase in the expression of mRNA coding for IFNγ and T-bet upon stimulation of MAIT cells (Lamichhane et al., 2019). Another cytokine produced by MAIT cells which has been of particular interest is IL-17, due to its potent pro-inflammatory properties (Pisarska et al., 2020). IL-17⁺ MAIT cells have been confirmed at the site of inflammation in multiple diseases including MS and RA (Kim et al., 2017; Willing et al., 2018). In agreement with published data, stimulation of MAIT cells upregulated IL-17A production and enhanced

transcription of mRNA coding for RORYT (Lamichhane et al., 2019). Collectively, this data showed that the activation of MAIT cells induced a very rapid alteration in the transcription events. MAIT cells respond by increasing the expression of transcription factors, most likely to facilitate the increased demand for synthesis of these cytokines in response to the activation signals. As the research progressed, the range of cytokines produced by MAIT cells expanded. Recently, other cytokines of the IL-17 cytokine family, IL-17F and IL-22 have gained more attention (Gibbs et al., 2017; Lamichhane et al., 2019; Toussirot et al., 2018). In line with Lamichhane *et al.*, we showed an enhancement in the synthesis of IL-17F and IL-22 upon MAIT cell stimulation (Lamichhane et al., 2019). In addition, these cells expressed IL-32, a cytokine which has not been previously described to be produced by MAIT cells. These results illustrate that MAIT cells can produce a much wider repertoire of pro-inflammatory cytokines, although further studies are required to explore what other effector molecules are expressed by these cells.

After the characterisation of MAIT cell functional responses in the control subjects, next we sought to describe their activity in people with obesity. In line with other studies, people with obesity presented with lower MAIT cell frequencies in the peripheral blood in comparison to the control subjects (Carolan et al., 2015; Magalhaes et al., 2015; Toubal et al., 2020) and displayed severely impaired expansion upon cell activation. Surprisingly, no difference was observed in the rate of proliferation assessed using CTV. This may be due to the limitations associated with the CTV assay. This method allowed for tracking MAIT cell proliferation only up to five days, whereas the defects in MAIT cells expansion from people in obesity has been observed seven days after activation and was sustained for up to 14 days. To date proliferation of MAIT cells in obesity was only briefly assessed in adipose tissue using Ki67 expression. MAIT cells from individuals with obesity were shown to have higher proliferative capacity in the AT in comparison to the peripheral blood, however this study didn't compare Ki67 expression between people with obesity and lean controls (Magalhaes et al., 2015). Toubal et al. has shown that MAIT cells from the adipose tissue of the control and HFD mice displayed no difference in proliferation as measured by expression of Ki67, however proliferation capacity from the peripheral blood was not assessed (Toubal et al., 2020). We propose that the impairment in MAIT cell

expansion in people with obesity may be contributing to the low frequencies of MAIT cells observed in their peripheral blood. However, other factors such as activation induced cell death or MAIT cell migration to the site of inflammation should be studied in more detail to provide a more comprehensive explanation.

As obesity had a profound impact on MAIT cell frequencies and proliferation, next we assessed their cytokine production upon cell activation. In comparison to the control individuals, MAIT cells from people with obesity displayed lower expression of IFNy which was accompanied by decreased transcription of T-bet transcription factor. This is in line with Carolan et al. who demonstrated an impairment in the frequencies of IFNy producing MAIT cells in obesity (Carolan et al., 2015). Magalhaes et al. displayed contradicting results, as in their hands people with obesity had a higher proportion of IFNy producing MAIT cells in comparison to the lean controls (Magalhaes et al., 2015). This discrepancy may be due to the fact Magalhaes used a pan activator of cells (PMA/Ionomycin), whereas our study used a more specific approach in activating MAIT cells. This data indicates that the deficiency in IFNy expression by MAIT cells from people with obesity occurs at the transcription level. This cytokine plays a central role in fighting microbial infections, therefore loss of IFNy may contribute to the greater susceptibility to infections associated with obesity. In contrast to IFNy, MAIT cells from people with obesity produced more IL-17 cytokine than the control individuals but displayed no difference in the transcription of mRNA for IL-17A or RORyT. This suggested that the defect, which leads to enhanced IL-17 production occurs at the post-transcriptional level. Similar observations were published by Carolan et al. and Magalhaes et al., who also reported an enhancement in IL-17 production by MAIT cells in obesity. There is growing evidence indicating that overexpression of IL-17 is harmful and contributes to the chronic inflammation observed in obesity and therefore may contribute to the development of T2DM (Chehimi et al., 2017; Liu and Nikolajczyk, 2019). Further studies have revealed that MAIT cells from people with obesity display lower (non-significantly) expression of IL-17F and IL-22 in comparison to the controls. In addition, MAIT cells from people with obesity failed to suppress transcription of IL32 mRNA upon stimulation, which resulted in an enhanced albeit nonsignificant expression of IL-32 cytokine in comparison to the controls. This data suggested that IL-17F and IL-22 may not play an important role in maintaining

the pro-inflammatory environment in obesity, however the role of IL-32 should be further investigated.

Collectively this study has provided more insight into MAIT cell proliferation and cytokine production. We show that MAIT cells can proliferate and express a wide range of cytokines and transcription factors, whereas MAIT cells from people with obesity display a marked deficiency in MAIT frequencies and expansion and significant alterations in the cytokine expression (Figure 3.18). This may lead to the lower frequencies, susceptibility to infections and inflammation observed in obesity.



Figure 3.18 Summary of MAIT cell function in control and people with obesity. Stimulation MAIT cells enhances expression of IL-2 receptor, induces their proliferation, transcription of T-bet and RORyT transcription factors and production of IFNy, IL-17A, IL-17F, IL-22 and IL-32 cytokines. MAIT cells from people with obesity (red) display diminished proliferation and alterations in the expression of multiple cytokines and transcription factors.

Chapter Four

Results

MAIT cell metabolism

4.1. Introduction

4.1.1. Glycolysis in immunometabolism

As the field of immunometabolism advances, it becomes clear that glycolysis is essential to support immune cell survival and function. Glucose is one of the few fundamental nutrients required for T cell function, therefore it is no surprise glycolysis has become a focal point of multiple studies of immunometabolism. In order for glucose to enter the cell, it requires glucose transporters. Glucose transporters can be put into two main categories of sodium-glucose linked transporters (SGLT) and facilitated diffusion glucose transporters (GLUT) (Navale and Paranjape, 2016). The category of glucose transporters which are expressed by immune cells and are largely studied are GLUTs. This family of transporters consists of 12 members and to date, T cells were found to express GLUT1, 3, 4, 6 and 8 (Kavanagh Williamson et al., 2018; Macintyre et al., 2014), of which GLUT1 is the major and most thoroughly studied glucose transporter in T cells (Palmer et al., 2016). Surface expression of this transporter is markedly increased upon their TCR & CD28 activation (Jacobs et al., 2008; Ricciardi et al., 2018; Wieman et al., 2007). Expression of the transporter is controlled at multiple stages including transcription and trafficking to the cell membrane, however shuffling of GLUT1 to the cell surface was shown to be the key event responsible for the rapid upregulation of its expression (Jacobs et al., 2008; Macintyre et al., 2014).

Upon uptake of glucose, the carbohydrate enters glycolysis, which takes place in the cytosol of the cell and is central to cell metabolism. This process breaks down glucose into pyruvate, which further feeds into the Kreb's cycle to contribute to the energy production for cellular function via oxidative phosphorylation (Almeida et al., 2016). Glucose and glycolysis are crucial in T cell function. One of the first scientists who identified that glycolysis is linked to T cell activity was Robson MacDonald in 1977. Using glycolysis inhibitor 2DG, MacDonald reported inhibition of the cytotoxic activity of T cells (Robson MacDonald, 1977). More recent studies have shown that indeed glycolysis plays a central role in T cells during activation. CD8 effectors cells, Th1 and Th17 cells can substantially upregulate their glycolysis upon stimulation (Gubser et al., 2013; Michalek et al., 2011), whereas the substitution of glucose with galactose in the culture media impairs IFNγ production by CD4 T cells (Chang et al., 2013). Inhibition of

glycolysis with 2DG promotes the development of Treg rather than Th17 cells, highlighting the role of glycolysis in determining the fate of naïve T cells (Shi et al., 2011). As more studies emerge, it is becoming clear that for some aspects of T cell activation, especially IFNγ production, the role of glycolysis extends beyond energy production.

A study by Chang et al. identified a key role for glyceraldehyde 3-phosphate dehydrogenase (GAPDH), which is one of the multiple enzymes involved in glycolysis, in T cell functional responses. In the resting state of T cells, GAPDH binds to IFNy mRNA and prevents its translation. Upon cell activation and triggering of glycolysis, GAPDH dissociates from the IFNy mRNA strand and reverts back to fulfil its function in the glycolysis pathway (Chang et al., 2013). IFNy is also controlled metabolically at the transcription level in T cells. Acetylation of histones affects the access of transcription enzymes to the IFNy gene. Acetyl Co-A is produced during oxidation of pyruvate, the end product of glycolysis and can be used to acetylate proteins, including histones (Peng et al., 2016). An intermediate of glycolysis, phosphoenolpyruvate (PEP), was also found to impact IFNy production. PEP inhibits the activity of endoplasmic reticulum calcium transporter sarcoendoplasmic reticulum calcium ATPase (SERCA), which shuffles calcium from the cytoplasm to endoplasmic reticulum. This leads to the activation of the calcium dependent nuclear factor of activated T cells (NFAT) signalling and transcription of IFNy (Ho et al., 2015). Although glycolysis plays a role in multiple ways to impact IFNy production, metabolic intermediates produced as a result of this process, may also be supporting lymphocytes in their metabolism and function. Pyruvate, the end product of glycolysis, can be used in synthesis of amino acids and fatty acids (Figure 4.1) (O'Neill et al., 2016), however it has been also shown to play a role in CD4 effector memory cell survival by stabilising mitochondrial membrane potential to prevent apoptosis. Dimeloe et al. have shown that independely of oxygen availability, disruption of glycolysis results in an increased expression of apoptotic proteins. This was associated with decreased mitochondrial membrane potential, which was shown to be rescued by addition of pyruvate(Dimeloe et al., 2016). However, pyruvate isn't the only product of glycolysis, which supports the cell metabolism. Metabolic intermediates produced during this process, such as glucose-6-phosphate can enter PPP to produce ribose, which is required for

nucleotides synthesis (Figure 4.1) (O'Neill et al., 2016). This suggests that the involvement of glycolysis in the control of immune cells function is more complex than first anticipated. Current findings uncover a very exciting area for further research, emphasising the extensive influence of metabolic processes on immune responses.



Figure 4.1 Glycolysis. Glucose is metabolised to pyruvate during glycolysis, which can enter Kreb's cycle. Glycolysis also leads to synthesis of multiple intermediates, which can feed into other metabolic pathways and contribute to the production of macromolecules (in green). Adapted from DeBerardinis and Chandel, 2020.

4.1.2. mTORC1

mTOR is an evolutionarily conserved serine/ threonine kinase, which regulates the activation and function of multiple proteins (Saxton and Sabatini, 2017). mTOR integrates an array of signals including growth factor signals, oxygen level, cell energy, nutrient availability and antigen receptor signalling. Based on the received signals, it controls cell metabolism, growth, differentiation and apoptosis (Salmond, 2018; Saxton and Sabatini, 2017). mTOR is the key catalytic component of two complexes it can be associated with, namely mTORC1 and mTORC2 (Figure 4.2) (Salmond, 2018). Although these complexes share three subunits- mTOR, mLST8 (mammalian lethal with sec13 protein 8) and DPETOR (DEP domain containing mTOR interacting protein). mTORC1 further associates with Raptor (regulatory protein associated with mTOR) and PRSA40 (proline-rich AKT substrate 40kDa), whereas mTORC2 is defined by the presence of Rictor (rapamycin insensitive companion of mTOR), mSIN1, or Protor-1/2 (Saxton and

Sabatini, 2017). Out of these two complexes, mTORC1 is shown to be of particular importance in immune cell function and control of glycolysis (Delgoffe et al., 2011, 2009; Finlay et al., 2012; Sinclair et al., 2013)

mTORC1 plays a substantial role in regulating cell growth, proliferation and activation. Activation of T cells via TCR or cytokine stimulation increases mTORC1 activity indicating a role for mTORC1 in T cell function (Rao et al., 2010; Ray et al., 2015). Expression of mTORC1 is shown to influence the fate of T cells, as CD4 T cells which lack mTORC1 fail to develop into Th1, Th2 or Th17, but rather favour Treg differentiation (Delgoffe et al., 2011, 2009). On the other hand, enhanced activity of mTORC1 in CD8 T cells leads to highly activated effector cells, which fail to transition into memory cells (Pollizzi et al., 2015). This indicates that tight control of mTORC1 is essential for maintaining T cell homeostasis.

As multiple factors can activate mTORC1, the regulation of this kinase is very complex. Among various points of control, changes in the cellular environment such as nutrient, energy and oxygen availability, can affect the activity of the kinase via the action of AMP-activated protein kinase (AMPK) (Saxton and Sabatini, 2017). Glucose restriction was shown to activate AMPK, leading to the phosphorylation of raptor or tuberous sclerosis complex 2 (TSC2) (regulator of mTORC1) and attenuation of the mTORC1 complex (Rolf et al., 2013; Saxton and Sabatini, 2017). Although mTORC1 is sensitive to glucose concentration, it was also shown to be absolutely required for control of glycolytic metabolism. mTORC1 is required to maintain high levels of glycolytic activity to support T cell function (Finlay et al., 2012). Treatment of T cells with rapamycin, an inhibitor of mTORC1, impairs glucose uptake induced by TCR signalling. It also prevents the upregulation of glucose transporters and glycolytic enzymes (Finlay et al., 2012; Hukelmann et al., 2016). Similar findings were obtained by Donnelly et al. in a study of natural killer (NK) cells, which demonstrated that stimulation of mTORC1 is responsible for the control of enhanced glycolysis and production of IFNy and Granzyme B (Donnelly et al., 2014).

Amino acids were also shown to affect mTORC1 signalling. Although amino acids are required for protein synthesis and their enhanced import during cell activation is important to facilitate the increased nutrient demands of the cells, it is the concentration of leucine, which directly affects the activity of mTORC1(Figure 4.2) (Saxton and Sabatini, 2017). The presence of amino acids activates Rag, a GTPase, which in turn mediates the translocation of mTORC1 to the lysosomes. This facilitates the lysosome bound Ras homolog enriched in brain (Rheb) to activate mTORC1 (Figure 4.2) (Saxton and Sabatini, 2017). Activity of Rag is controlled by the interaction between Sestrin2, a protein which controls cell metabolism and GATOR, which is a Rag activating protein (Figure 4.2) (Kim et al., 2015). Briefly, Kim et al. reported that Sestrin2 binds to GATOR2 resulting in its inhibition. GATOR2 is a negative regulator of GATOR1, therefore this interaction results in the liberation of GATOR1 from GATOR2 complex and its activation. The release of GATOR1 inhibits the activity of the Rag protein resulting in suppression of mTORC1 activity (Kim et al., 2015). Sestrin2 itself is a leucine sensor. Leucine binds to Sestrin2, leading to the freeing of GATOR2, which binds to GATOR1, allowing Rag mediated activation of mTORC1 (Chantranupong et al., 2014; Saxton et al., 2016; Wolfson et al., 2016). This method of mTORC1 control was shown to be relevant in T cells. Sinclair et al. described that deletion of amino acid transporter LAT1, which transports leucine, leading to the downregulation of mTORC1 activity in murine T cells (Sinclair et al., 2013).



Figure 4.2. Regulation of mTORC1 activity. mTORC1 is tightly controlled by cellular and environmental signals such as growth factors, nutrient availability, DNA damage or energy availability. All combined signals incorporated by mTORC1 decide on the faith of the cell (Saxton and Sabatini, 2017).

4.1.3. Amino acid metabolism

Although glucose and glycolysis are central in T cell homeostasis and function, the role of amino acid metabolism cannot be underestimated. Amino acids are essential building blocks of proteins, but their role in the cell function was reported to be far more complex. It has been established that interruption to amino acid transport or metabolism impairs immune cell function, however the mechanisms behind this altered immune response have begun to emerge in recent years (Ma et al., 2017; Rodriguez et al., 2007; Sinclair et al., 2013). The role of multiple amino acids is currently under investigation in the field. However, leucine and glutamine have been the focus of this study due to their effects on T cell activation (Nakaya et al., 2014; Sinclair et al., 2013; Wolfson et al., 2016).

Amino acids are transported into the cells by a wide range of membrane bound transporters. Leucine, one of the essential amino acids, was found to be transported into the cells by LAT1 (Hsu and Dzhagalov, 2019). LAT1 belongs to a family of four L-type amino acid transporters, which are sodium independent and transport neutral amino acids. In addition to leucine, they facilitate the influx of isoleucine, valine, phenylalanine and methionine (Hsu and Dzhagalov, 2019). This family of transporters includes LAT1 (SLC7A5), LAT2 (SLC7A8), LAT3 (SLC43A1) and LAT4 (SLC43A2), with LAT 1 is the most studied member of this family. LAT1 forms a heterodimer with CD98 (SLC3A2) and was shown to be an antiporter, which imports leucine in exchange for intracellular glutamine (Hsu and Dzhagalov, 2019). SLC7A5 was found to be upregulated in human and murine T cells upon stimulation (Hayashi et al., 2013; Sinclair et al., 2013). Study by Sinclair et al. elegantly showed that T cells require LAT1 for differentiation into the effector cells. Deletion of SLC7A5 in mice resulted in a reduction in T cell differentiation into Th1, Th17 and CTL cells, however, Treg differentiation remained unaffected (Sinclair et al., 2013). As leucine was shown to be transported via LAT1 and the activation of mTOR, the defect in T cell proliferation and differentiation in the absence of SLC7A5 expression was linked to the lack of this amino acid and failure to induce the appropriate metabolism response to support the cell function (Sinclair et al., 2013). Further investigation into the association between amino acid transport via LAT1 and T cell function revealed that SLC7A5 deficient T cells failed to reprogram their metabolism and resembled c-Myc null cells. It was shown that an influx of amino acids via LAT1 L-system

transporter was required to sustain the expression of c-Myc transcription factor in T cells, which is critical for cell metabolic reprogramming, proliferation and survival (Gnanaprakasam and Wang, 2017; Sinclair et al., 2013). This was also confirmed by another study, which investigated the role of amino acids for NK cell function and activation. They also highlighted a role for glutamine, rather than leucine in supporting the c-Myc expression (Loftus et al., 2018). Furthermore, Hayashi *et al.* confirmed the upregulation of LAT1 expression by human primary T cells upon stimulation. In line with the findings by Sinclair et al., Hayashi showed that the inhibition of LAT1 with a specific inhibitor, JPH203, or transfection with LAT1 specific siRNA, resulted in a decrease of leucine uptake and cytokine production by activated T cells. This study showed that TCR stimulation of T cells upregulates LAT1 in a NFκB and activator protein-1 (AP-1) dependent manner (Hayashi et al., 2013), whereas Sinclair at al. suggested SLC7A5 expression is controlled by ERK/MAPK and NFAT pathway (Sinclair et al., 2013). Another study by Sinclair et al., which studied the transport of kynurenine, a tryptophan metabolite, via SLC7A5 in T cells, depicted a very comprehensive method to investigate uptake of this metabolite and indirectly assess the function of the amino acid transporter. This paper formed the basis for the investigation of this amino acid metabolism and uptake by MAIT cell for this thesis (Sinclair et al., 2018). Expression patterns and function of other amino acids transporters including LAT2-4 in the cells of the immune system remains largely unexplored. Similarly, to LAT1, LAT2-4 transport large neutral amino acids (Babu et al., 2003; Bodoy et al., 2005; Yan et al., 2020). In addition, LAT2 was shown to also support transport of small neutral amino acids such as serine or alanine (Yan et al., 2020). Although the specific amino acids transported by LAT3 remain to be elucidated, it has been shown that LAT4 transports essential and branched chained amino acids such as phenylalanine, methionine and leucine or valine (Bodoy et al., 2005). Further research is required to assess the expression, function and importance of amino acid transporters other than LAT1 for the cells of the immune system. In addition, more detailed studies are vital to assess, whether all neutral amino acids are equally transported by all four transporters.

Among other amino acids, glutamine has also attracted a lot of attention from researchers in the context of immunometabolism. Glutamine is synthesised de novo by the cell or absorbed from the diet and imported into the cells through a range of glutamine transporters, such as sodium coupled neutral amino acid transporters (SNAT)1-4 (SLC38A1-SLC38A4), and the most heavily studied glutamine importer Alanine, Serine, Cysteine Transporter 2 (ASCT2/SLC1A5) (Hsu and Dzhagalov, 2019). Upon entry, glutamine can be used for multiple purposes such as generation of energy thorough feeding into the Kreb's cycle. Mitochondrial glutaminase enzyme converts glutamine to glutamate, which can be further metabolised to α -ketoglutarate and enter the Kreb's cycle to produce ATP and contribute to the fatty acid synthesis via reductive carboxylation (Altman et al., 2016). In addition, glutamine can be also directed for synthesis of nucleotides, glutathione or O-GlcNAcylation of proteins (Altman et al., 2016).

Uptake of glutamine as well as expression of the glutamine transporters and enzymes involved in glutaminolysis are upregulated upon T cell stimulation, which in turn are dependent on the c-Myc pathway (Carr et al., 2010; Nakaya et al., 2014; Wang et al., 2011). The importance of glutamine uptake was highlighted by Nakaya et al., who showed that the deletion of ASCT2 in mice led to impaired differentiation of Th1 and Th17 cells, and failure to produce their signature cytokines (Nakaya et al., 2014). ASCT2 was also required for successful mTOR signalling, suggesting that the influx of glutamine via this transporter is important for the intracellular pool of glutamine, to be exchanged for leucine by transport via LAT1. Although glutamine is required for T cell proliferation, surprisingly deficiency in ASCT2 does not negatively affect proliferation (Carr et al., 2010; Nakaya et al., 2014), suggesting other glutamine transporters must be implicated. Indeed, Carr et al. and Raposo et al. have reported that T cell stimulation leads to the upregulation of other amino acid transporters, namely SNAT1 (SLC38A1) and SNAT2 (SLC38A2) (Carr et al., 2010; Raposo et al., 2015), whereas inhibition of SNAT glutamine transporters, reduces T cell proliferation and production of IFNy and IL-2 (Raposo et al., 2015).

Research in the area of amino acid metabolism and immune response was shown to be very exciting, allowing for better understanding of the relationship between cell metabolism and immune cell function. Study of amino acids revealed their complex roles in the activity of the cell. It is now known, they are not merely substrates used for protein synthesis, but can also act as signalling molecules to support the immune cell function. Although a progress has been made in understanding the role of amino acids in T cells, up to date there were no studies published, which would investigate their importance in MAIT cell function and activation.

4.2. Specific aims of this chapter

The specific aim of this chapter was the characterisation of MAIT cell metabolism in both healthy individuals and people with obesity.

Therefore, to:

- 1. Investigate nutrient transporter expression by MAIT cells in health and obesity
- 2. Elucidate metabolic pathways used by MAIT cells
- 3. Determine metabolic regulators of MAIT cell metabolism
- 4. Compare MAIT cell metabolism rates between healthy individuals and people with obesity

4.3. Results

4.3.1. MAIT cells express glucose transporters GLUT1, GLUT3 and GLUT4 As outlined, research has provided us with substantial evidence, which shows T cells require glucose and glycolysis to successfully perform their effector functions. It has been shown that stimulation of T cells leads to substantial upregulation of glycolysis, suggesting that glucose is one of the key nutrients, which supports T cell activation (Gubser et al., 2013; Michalek et al., 2011; Wang et al., 2011). In order for glucose to be metabolised, it must be transported into the cell. Among the 14 GLUT glucose transporters, which have been identified to be expressed by humans, T cells have been shown to express GLUT1, 3, 4, 6 and 8 (Kavanagh Williamson et al., 2018; Macintyre et al., 2014). At present, the majority of studies investigating a role of GLUT1 and GLUT3 in T cell function have reported an upregulation of their expression upon T cell stimulation (Hukelmann et al., 2016; Macintyre et al., 2014; Palmer et al., 2015). To the best of our knowledge, the expression of glucose transporters by MAIT cells has not yet been described.

We first determined the expression of glucose transporters GLUT 1, 3 and 4 on expanded MAIT cells using real time RT-qPCR. Expanded MAIT cells were found to express mRNA for these three GLUT transporters (Figure 4.3 A, D&E). Stimulation of expanded MAIT cells with TCR Dynabeads, IL-12 and IL-18 cytokines for 4 hours showed no significant difference in the SLC2A1 mRNA level coding for GLUT1 in comparison to the basal expression (Figure 4.3 A) However, as GLUT1 is the best studied glucose transporter in T cells and its expression is known to be controlled at the point of trafficking of the complex from the cytosol to the cell surface, we decided to measure the expression of GLUT1 protein on the surface of MAIT cells by flow cytometry (Wieman et al., 2007). GLUT1 protein was expressed on the cell membrane of MAIT cells, and its expression was significantly decreased upon stimulation of PBMC via TCR, and cytokines after 18 hours (Figure 4.3 B&C). Analysis of mRNA expression of the other two glucose transporters by expanded MAIT cells stimulated via TCR and cytokines for 4 hours revealed significant increase in the transcription of SLC2A3 (GLUT3) and SLC2A4 (GLUT4) (Figure 4.3 D&E). This data indicates that MAIT cells express mRNA for the four major transporters tested and display a limited alteration in their expression upon stimulation at the chosen timepoints.



Figure 4.3 MAIT cells express mRNA and protein for glucose transporter GLUT1 and mRNA for GLUT3&4. (A) Scatter plot of normalized MAIT cell mRNA expression of glucose transporter GLUT1(SLC2A1) by purified MAIT cells basally and upon stimulation with TCR Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours in the control population sample(n=11). (B) Scatter plot of surface MFI GLUT1 expression by MAIT cells from a PBMC culture; unstimulated or stimulated with TCR Dynabeads (cell to bead ratio 1:1) IL-12 and IL-18 (at 50ng/mL each) for 18 hours (n=8). (C) Representative histograms of two control samples showing GLUT1 expression on MAIT cells basally or upon stimulation with TCR Dynabeads cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 18 hours. (D-E) Scatter plot of normalized MAIT cell mRNA expression of glucose transporters GLUT3 and GLUT4 (SLC2A3-4) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours in the control population sample (n=11). Statistical analysis performed using paired student's t-test, ns-not significant, * p≤0.05, ** p≤0.01, *** p≤0.001.

4.3.2. MAIT cells from individuals with obesity display no difference in glucose transporters mRNA expression

MAIT cells from the control individuals expressed GLUT1, 3 and 4 glucose transporters and displayed modulation of their expression upon activation (Figure 4.3), therefore next investigated the impact of obesity on the expression of these glucose transporters by MAIT cells. Similarly, to control subjects, MAIT cells from people with obesity expressed mRNA for all three glucose transporters (Figure 4.4 A-C). Expanded MAIT cells from people with obesity, stimulated as per section 4.3.1 displayed no significant differences in the expression of mRNA coding for GLUT1, 3 and 4, when compared to the cohort of control individuals (Figure 4.4 A-C). This investigation showed that during early stages of MAIT activation, people with obesity display no defect in transcription of mRNA for glucose transporters GLUT1, 3 and 4.



Figure 4.4 MAIT cells from people with obesity showed no difference in mRNA expression for GLUT glucose transporters. (A-C) Scatter plot of normalized MAIT cell mRNA expression of glucose transporters *SLC2A1, SLC2A3 and SLC2A4* (GLUT1,3 and 4) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours derived from the controls and people with obesity (control n=11; obese n=6. Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-not significant.

4.3.3. MAIT cells express glutamine transporters ASCT2, SNAT1 and SNAT2 Having established the expression of the major glucose transporter, we next investigated if MAIT cells express glutamine transporters. Glutamine is a non-essential amino acid, implicated in multiple cell processes including the production of ATP by feeding into Kreb's cycle, synthesis of nucleic acids by a provision of biosynthetic precursors and in protein O-GlcNAcylation (Altman et al., 2016; Hsu and Dzhagalov, 2019). Although glutamine can be synthesised de novo, it may also be obtained from the diet, requiring transport via glutamine transporters (Hsu and Dzhagalov, 2019). Glutamine can be transported in and out of the cells through a range of amino acid transports. ASCT2 and LAT1 have been reported as the main glutamine transporter in T lymphocytes (Hsu and Dzhagalov, 2019; Nakaya et al., 2014), however the family of SNAT, especially SNAT1 and SNAT2 were also shown be involved in glutamine transport in T cells (Hsu and Dzhagalov, 2019).

Expression of the glutamine transporters was measured by real time RT-qPCR and expanded MAIT cells expressed mRNA coding for ASCT2, SNAT1 and SNAT2 transporters (Figure 4.5 A-C). Upon stimulation with TCR Dynabeads, IL-12 and IL-18 for 4 hours, MAIT cells greatly increased their expression of *SLC1A5* (ASCT2) neutral amino transporter (Figure 4.5 A). More modest increases in the expression of *SLC38A1* (SNAT1) and *SLC38A2* (SNAT2) were observed (Figure 4.5 B&C). MAIT cells were shown to display a similar pattern in the expression of glutamine transporters to the reported expression of these transporters by T cells. In addition, upregulation of transcription of the glutamine transporter *SLC1A5* (ASCT2) this early into activation of MAIT cells suggests that the transport of glutamine into MAIT cells may play an important role in their activation and function.



Figure 4.5 MAIT cells express mRNA for glutamine transporters ASCT2 and SNAT1&2. (A-C) Scatter plots of normalized MAIT cell mRNA expression of glutamine transporters (A) *SLC1A5* (ASCT2) (n=9) and (B-C) *SLC38A1-2* (SNAT1-2) (n=11) by purified unstimulated MAIT cells and cells stimulated with TCR Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours in the control population sample. Statistical analysis performed using paired student's t-test, ns-not significant, *p≤0.05, **p≤0.01, **** p≤0.0001.

4.3.4. MAIT cells from individuals with obesity display no difference in expression of mRNA for glutamine transporters

We next decided to extend our investigations in obesity by investigating the mRNA expression pattern of ASCT2, SNAT1 and SNAT2 glutamine transporters on MAIT cells from people with obesity. Expanded MAIT cells from individuals with obesity were stimulated as per section 4.3.3. MAIT cells expressed *SLC1A5* (ASCT2), *SLC38A1*(SNAT1) and *SLC38A2* (SNAT2) mRNA at levels comparable to the control individuals (Figure 4.6 A-C). This suggests that similarly to the expression of mRNA for glucose transporters (Figure 4.4), MAIT cells from people with obesity, display no defect in the transcription of mRNA for glutamine transporters at 4 hours post activation with TCR Dynabeads, IL-12 and IL-18.



Figure 4.6 MAIT cells from donors with obesity showed no difference in mRNA expression for SNAT glutamine transporters. (A-C) Scatter plot of normalized MAIT cell mRNA expression of glutamine transporters (A) *SLC1A5* (ASCT2), (B) *SLC38A1* (SNAT1) and (C) *SLC38A2* (SNAT2) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours by the controls and individuals with obesity (*SLC1A5* control n=7; obese n=3; *SLC38A1 &2* control n=11; obese n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-not significant.

4.3.5. MAIT cells express amino acid transporters LAT1- LAT4

We next decided to investigate another family of amino acid transporters; L-type amino acid transporters. Out of this family, LAT1 is the best studied amino acid transporter. LAT1 is expressed by T cells and becomes upregulated upon stimulation (Sinclair et al., 2013). LAT1 is defined as an anti-porter, as it imports neutral essential amino acids into the cells in the exchange for glutamine. This particular amino acid transporter was shown to play an important role in mTORC1 activation, as it facilitates the influx of leucine, which in turn facilitates activation of mTORC1, a master regulator of the cells (Sancak et al., 2008; Wolfson et al., 2016). Functional LAT1 transporter is a heterodimer composed of CD98 heavy chain which was shown to be associated with LAT1. Stimulation of expanded MAIT cells with TCR Dynabeads, IL-12 and IL-18 for 4 hours resulted in a small but significant increase in *SLC3A2* (CD98) mRNA expression (Figure 4.7 A). Much larger increase in the transcription level in comparison to the resting state of the cell was observed with *SLC7A5* (LAT1), suggesting that this transporter may also play an important role in MAIT cell function (Figure 4.7 B).

We next assessed whether the increases in mRNA expression of LAT1 components are reflected at the protein level using flow cytometry and a CD98 specific antibody. CD98 protein was expressed on MAIT cells and was upregulated upon stimulation of PBMC with TCR Dynabeads, IL-12 and IL-18 cytokines for 18 hours (Figure 4.7 C&D). This data shows that the increase in the transcription of LAT1 (SLC7A5) & CD98 (SLC3A2) is reflected on the protein level, further reinforcing that this amino acid transporter may be central to MAIT cell function. Investigation of the expression of other LAT amino acid transporters showed a varied response. There was no significant difference in the expression of SLC7A8 (LAT2) mRNA upon stimulation, whereas a significant increase in the transcription of SLC43A1 (LAT3) was observed (Figure 4.7 E&F). SLC43A2 (LAT4) expression was significantly downregulated upon stimulation in comparison to the transcription in the resting cells (Figure 4.7 G). These results are in line with previously published data, which demonstrated upregulation of LAT1 expression upon stimulation by T cells. This suggests that amino acids transported via LAT1 are of particular importance for not only for T cell function but also may play a role in supporting function of MAIT cells.



Figure 4.7. MAIT cells express mRNA and protein for CD98 subunit of LAT1 amino acid transporter and mRNA for LAT2-4. (A) Scatter plot of normalized MAIT cell mRNA expression of *SLC3A2* (CD98) and (B) *SLC7A5* (LAT1) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours in the control population sample (n=11). (C) Scatter plot of surface MFI CD98 expression by MAIT cells from a PBMC culture and (D) representative histogram of two control samples showing MFI CD98 surface expression; unstimulated or stimulated with TCR microbeads at 25ng/mL (Miltenyi), IL-12 and IL-18 (at 50ng/mL each) for 18 hours in the control population sample (n=7). (E-G) Scatter plot of normalized MAIT cell mRNA expression of amino acid transporters *SLC7A5, SLC43A1, SLC43A2* (LAT2-4) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours in the control population sample (n=7). Statistical analysis performed using paired student's t-test, ns-not significant, *p≤0.05, ** p≤0.01, **** p≤0.001.

4.3.6. MAIT cells in obesity express lower level of LAT1, but show no significant difference in transcription of other amino acid transporters

We again extended our investigations into obesity. Stimulation of expanded MAIT cells from people with obesity as per section 4.3.5 led to modest upregulation of SLC3A2 (CD98) mRNA transcription and there was no difference noted between the two cohorts (Figure 4.8 A). However, the expression of SLC7A5 mRNA was diminished in people with obesity (Figure 4.8 B). We also noted decreased expression of CD98 protein on the cell surface of MAIT cells from people with obesity in comparison to the control individuals (Figure 4.8 C&D). Next, we investigated whether MAIT cells from people with obesity have altered transcription of the other three LAT transporters. Stimulation of the expanded MAIT cells as per section 4.3.5 showed no difference in the expression of mRNA for LAT2 (SLC7A8) or LAT3 (SLC43A1) between two cohorts (Figure 4.9 A&B). Expression of SLC43A2 (LAT4) mRNA showed a similar trend of downregulation of transcription upon cell activation, however MAIT cells from people with obesity displayed an impaired expression of the LAT4 transcription in comparison to the cohort of control individuals (Figure 4.9 C). Collectively, this data shows that MAIT cells from people with obesity show a defect in the expression of LAT1 amino acid transporter, which may have downstream impact on the transport of amino acids into the cells and thus may lead to their impaired function.







Figure 4.9 MAIT cells from obese donors express similar levels of LAT2-4 mRNA level to the control samples. (A-C) Scatter plot of normalized MAIT cell mRNA expression of glucose transporters *SLC7A5*, *SLC43A1*, *SLC43A2* (LAT2-4) by purified unstimulated MAIT cells and cells stimulated with Dynabeads (cell to bead ratio 1:1), IL-12 and IL-18 (at 50ng/mL each) for 4 hours from controls and individuals with obesity (lean n=11; obese n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-not significant, *p≤0.05.

4.3.7. Stimulation of MAIT cells increases their size

Having determined the expression of the major nutrient transporters on MAIT cells and the impact of activation, we next aimed to define the metabolic profile of MAIT cells. A study by Howden et al., which investigated the proteome of CD4⁺ and CD8⁺ T cells found that naïve CD8⁺ had a higher protein content than naïve CD4⁺ cells. This was reflected in the FSC measurement, where CD8⁺ cells were larger than CD4⁺ cells (Howden et al., 2019). The authors suggested the size of the cells was proportional to the protein content of that cell. Upon cell activation the protein content increases, of which a substantial proportion of the proteins are related to metabolism (Howden et al., 2019). Therefore, this data suggests that the size of a cell can be used to estimate the metabolic activity. The size of immune cells can be easily evaluated using the flow cytometry by measuring the FSC. Based on this observation from Howden et al., we assessed whether the size (therefore metabolic activity) of MAIT cells was altered upon stimulation with 5-ARU and MG for five days and observed an increase in MAIT cell size (Figure 4.10 A&B). This indicates that MAIT cells upregulate the synthesis of a large amount of proteins upon stimulation, suggesting high engagement in metabolic activity.



Figure 4.10 MAIT size increases upon extended stimulation. (A) Scatter plot and mean MAIT cell size (FSC-A) basally and after 5 days of stimulation. PBMC were unstimulated (basal) or stimulated (stim) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 (n=17). (B) Representative histograms from 2 control subjects of MAIT cell size after 5 days of culture. Statistical analysis performed using paired student's t-test, ****p≤0.0001.



4.3.8. MAIT cells upregulate glycolysis upon stimulation

As discussed in section 4.1, the role of glucose and glycolysis in T cell function is very well established (Cham et al., 2008; Chang et al., 2013; Jacobs et al., 2008; Shi et al., 2011). Having demonstrated the expression of glucose transporters by MAIT cells, we next investigated the level of glycolysis in resting and stimulated MAIT cells using extracellular flux analysis (Seahorse). Extracellular flux analysis allows for the measurement of glycolysis rate by measuring the ECAR. Upon transport of glucose into the cell, the molecule is metabolised through glycolysis to pyruvate and lactate, releasing protons during this process. As the protons are exported from the cell, this leads to acidification of the cell culture media. The concentration of these protons is measured during seahorse assay and is directly proportional to the rate of glycolysis. Expanded MAIT cells were unstimulated or stimulated with TCR microbeads, IL-12 and IL-18 for 18 hours prior to the seahorse analysis. Once placed on the seahorse instrument, data on the rates on MAIT cell glycolysis was collected by measuring ECAR over an hour. During the real-time measurement oligomycin, and ATP synthase inhibitor, was injected 15 minutes into the incubation to inhibit oxphos and forced the cells to fully switch to use glycolytic metabolism to support their function. At 35 minutes 2DG was injected to inhibit glycolysis and show that the ECAR measured in the experiments was due to the glycolysis and no other cell processes. Here, we demonstrated increased rates of ECAR by MAIT cells upon stimulation (Figure 4.11 A&B). This data shows that MAIT cells engage in glycolytic metabolism upon activation. In addition to ECAR rates, glycolytic capacity was measured by adding oligomycin, which inhibits ATP synthase and thus OxPhos. This forces the cell to use glycolysis to its maximal capacity in order to produce ATP to sustain cell function. There was no significant difference in the glycolytic capacity between basal and stimulated MAIT cells, however a trend towards higher glycolytic capacity upon stimulation was observed (Figure 4.11 C).



Figure 4.11 MAIT cells upregulate glycolysis upon stimulation. (A) Representative Seahorse trace and (B) scatter plot of ECAR rate as wells as (C) scatter plot of glycolytic capacity of MAIT cells basally or upon stimulation with TCR microbeads (Miltenyi 25ng/mL) and IL-12/IL-18 cytokines (at 50ng/mL each) for 18 hours (n=4). Statistical analysis performed using paired student's t-test, ns-not significant, **p \leq 0.01.

4.3.9. Stimulation of MAIT cells modulates expression of enzymes of glycolysis

With our observation that MAIT cells increase their glycolytic rate upon stimulation (Figure 4.11), we next aimed to confirm this finding by investigating the expression of key glycolytic enzymes, such as Hexokinase 2 (HK2), Pyruvate kinase (PKM) and Lactate dehydrogenase A (LDHA). Upon stimulation of expanded MAIT cells with TCR Dynabeads, IL-12 and IL-18 for 4 hours we observed increased mRNA expression of HK2 and PKM indicating enhanced glycolytic metabolism (Figure 4.12 A&B). To assess whether the increase in transcription of these enzymes is reflected on the protein level, we used flow cytometry and noted increased MAIT cell HK2 expression upon stimulation of PBMC with TCR Dynabeads, IL-12 and IL-18 for 18 hours (Figure 4.12 C&F). In contradiction to the molecular results, PKM2 expression at 18 hours post stimulation was shown to be unchanged (Figure 4.12 D&G). In addition, we investigated the protein level of LDHA in activated MAIT cells and found significantly enhanced expression of this enzyme (Figure 4.12 E&H). In summary this data suggests that MAIT cells upon stimulation rapidly engage a glycolytic metabolic programme.



4.3.10. MAIT cells from people with obesity display impaired glycolysis

Having demonstrated that glycolytic metabolism is central to MAIT cell activation (Figure 4.11 & 4.12), we next investigated the impact of obesity on MAIT cell metabolism. Stimulation of expanded MAIT cells from people with obesity with TCR Dynabeads, IL-12 and IL-18 for 18 hours did not increase their rates of glycolysis (Figure 4.13 A&B). MAIT cells form people with obesity also failed to increase their glycolytic capacity upon stimulation. In addition, glycolytic capacity of stimulated MAIT cells from people with obesity was significantly lower in comparison to the control group (Figure 4.13 C). To explore this failure to upregulate glycolysis in more detail, expression of mRNA for glycolytic enzymes was measured. In comparison to the control cohort, we noted no differences in the expression of either HK2 or PKM upon stimulation for 4 hours with TCR Dynabeads and cytokines (Figure 4.14 A&B). Despite the fact that MAIT cells from people with obesity displayed no impairment in the expression of mRNA for glucose transporters in comparison to the control individuals (Figure 4.4), this data indicates a clear defect in the rates of glycolysis. Collectively this data indicates that obesity not only impacts MAIT cells proliferation and cytokine production (chapter three), but also glycolytic metabolism.



Figure 4.13 MAIT cells from people with obesity don't upregulate glycolysis upon stimulation. (A) Representative Seahorse trace and (B) a scatter plot of MAIT cells ECAR rate as wells as (C) a scatter plot of glycolytic capacity of MAIT cells basally or upon stimulation with TCR microbeads (Miltenyi 25ng/mL) and IL-12/IL-18 cytokines (at 50ng/mL each) for 18 hours (control n=6; obese n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-not significant, *p≤0.05, **p≤0.01.


Figure 4.14 MAIT cells from people with obesity display a non-significant alteration mRNA expression of HKII and PKM2 enzymes. Scatter plot of (A) *HK2* (control n=13, obese n=6) and (B) *PKM* (control n=9, obese n=4) mRNA expression by purified MAITs from control and obese cohort, basally and upon stimulation with TCR Dynabeads (1:1 cell to bead ratio) and IL-12/IL-18 (50ng/mL each) cytokines for 4 hours. Statistical analysis performed using one-way ANOVA with Tukey's correction, ns- not significant.

4.3.11. MAIT cells actively use LAT1 amino acid transporters

We next investigated a role for LAT1 in MAIT cell metabolism, as it was the most abundantly expressed transporter on activated MAIT cells (Figure 4.7 A-D). Although LAT1 is known to transport multiple amino acids including leucine, kynurenine a metabolite of t-tryptophan is reported to be a substrate of LAT1 (Sinclair et al., 2018). Kynurenine was shown to be auto fluorescent, and this feature was used by Sinclair et al. to develop a kynurenine uptake assay, which allows for the measurement of kynurenine uptake via LAT-1 by immune cells using flow cytometry (Sinclair et al., 2018). In this study we used the kynurenine assay developed by Sinclair et al. to assess the activity of the LAT1 amino acid transporter in MAIT cells and the efficiency BCH to inhibit transport via LAT1 (Sinclair et al., 2018). We observed significant kynurenine uptake over 4-minute incubation by basal MAIT cells from PBMC samples (Figure 4.15 A&B) and expanded MAIT cells (Figure 4.15 C&D), which was significantly reduced upon incubation with LAT1 specific BCH inhibitor (Figure 4.15 A&B; Figure 4.15 C&D). This data indicates that LAT1 expressed by MAIT cells is active and capable of importing LAT1 specific substrates, whereas BCH was shown to specifically inhibit LAT1 mediated transport.



Figure 4.15 MAIT cells import kynurenine via LAT1 amino acid transporter. (A) Scatter plot and (B) representative histograms of basal uptake of kynurenine (200µM) in the presence or absence of BCH (20mM) by MAIT cells from PBMC of healthy individuals during 4 minutes of incubation (n=4) (C) Scatter bar and (D) representative histograms of basal uptake of kynurenine (200µM) by expanded MAIT cells from healthy individuals in the presence or absence of BCH (20mM) during 4 minutes (n=10). Statistical analysis performed using one-way ANOVA with Tukey's correction, **p≤0.01, ****p≤0.001.

4.3.12. MAIT cells from people with obesity display defective LAT1 mediated transport

Previously we had demonstrated a decrease in the expression of LAT1 transporter by MAIT cells from people with obesity (Figure 4.8). Therefore, we next investigated Kynurenine uptake by MAIT cells from people with obesity and demonstrated impaired kynurenine uptake in comparison to healthy controls (Figure 4.16 A&B). Collectively this data suggests that lower expression of LAT1, leads to this reduced uptake of kynurenine and potential large neutral amino acids.



4.3.13. MAIT cells require amino acid influx via LAT1 to activate mTORC1 and glycolysis

Published studies have shown that LAT1 is responsible for the influx of leucine into T cells, which was shown to activate mTOR, a master regulator of cellular metabolism. We first investigated if MAIT cells utilize mTOR by assessing mTORC1 activity as measured by the phosphorylation of S6, a downstream target of mTORC1. Upon stimulation of MAIT cells with TCR microbeads, IL-12 and IL-18 for 18 hours, phosphorylation of S6 was significantly increased (Figure 4.17 A&B). We next assessed the impact of LAT1 inhibition on mTORC1 activity and noted a non-significant reduction in the phosphorylation of S6 protein (Figure 4.17 C&D). As mTOR is known to orchestrate the metabolism of T cells, including the control of glycolysis, next we decided to investigate whether limiting amino acid influx will alter the rate of glycolysis. To address this question, we used the ECAR assay as described in section 4.3.8. Expanded MAIT cells were unstimulated or stimulated with TCR microbeads, IL-12 and IL-18 for 18 hours in the presence or absence of BCH before the cells were put on the Seahorse instrument for analysis. Limiting the amino acid influx using BCH resulted in the significant reduction in glycolysis as well as glycolytic capacity (Figure 4.18 A,B,C). This data suggests that amino acids imported via LAT1, most likely leucine, are required for activation of mTORC1, which in turn is required for upregulation of glycolysis.



Figure 4.17 Amino acid transport via LAT1 is important for mTOR activation. (A) Scatter plot and (B) representative histogram of pS6 expression by MAIT cells from PBMCs culture of control individuals. Cells unstimulated or stimulated with TCR microbeads at 25ng/mL (Milteneyi), IL-12 and IL-18 (at 50ng/mL each) for 18 hours (n=3). (C) Scatter plot and (D) representative histogram of pS6 expression by MAIT cells from PBMCs culture of control individuals. Cells stimulated with TCR microbeads at 25ng/mL (Milteneyi), IL-12 and IL-18 (at 50ng/mL each) for 18 hours (n=3). (C) Scatter plot and (D) representative histogram of pS6 expression by MAIT cells from PBMCs culture of control individuals. Cells stimulated with TCR microbeads at 25ng/mL (Milteneyi), IL-12 and IL-18 (at 50ng/mL each) for 18 hours with or without BCH inhibitor (50mM) (n=3). Statistical analysis performed using student's t-test, *p≤0.05.



Figure 4.18 Amino acid transport via LAT1 is necessary for upregulation of glycolysis. (A) Representative Seahorse trace and (B) a scatter plot of MAIT cells ECAR rate as wells as (C) a scatter plot of glycolytic capacity of MAIT cells basally or upon stimulation with TCR microbeads (Miltenyi 25ng/mL) and IL-12/IL-18 cytokines (at 50ng/mL each) in the presence or absence of BCH (50mM) for 18 hours. Cells cultured in media diluted 1:2 with HBSS (n=4). Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-not significant, **p≤0.01, ***p≤0.001.

4.4. Discussion

Over the past decade, research in the area of immunometabolism has massively increased. Elaborate studies have illustrated that T cell function heavily relies on the precisely orchestrated cell metabolism. This aspect of immune cell activity has been largely taken for granted and overlooked for many years, however it is now well established that altered cell metabolism can have detrimental effects on immune cell function. Therefore, in this study we aimed to elucidate for the first time, which metabolic pathways control MAIT cell activation.

We found that MAIT cells express mRNA for glucose transporters GLUT1, 3 and 4, the expression of which was modestly upregulated at an early point during their activation. The literature suggests that glucose transporters expression is controlled during transcription and protein synthesis, as stimulation of T cells increased the mRNA and protein levels (Cretenet et al., 2016; Frauwirth et al., 2002; Kavanagh Williamson et al., 2018). Our data indicates that the stimulation of MAIT cells also leads to the initiation of early metabolic reprogramming, to support the influx of glucose by increasing transcription of some of the GLUT transporters. Parallel work from our lab has shown that the activation of MAIT cells increases uptake of glucose after overnight stimulation (O'Brien et al. 2019; Figure 1D). Together these results illustrate an important role for glucose in supporting MAIT cell activation.

Studies in T cells have established that glycolysis is the dominant pathway used by effector T cells during activation (Chang et al., 2013; Gubser et al., 2013; Michalek et al., 2011; Procaccini et al., 2016). Our study shows that MAIT cells, a potent effector cell subset, also upregulate the rate of glycolysis upon stimulation, however there is no significant difference in the glycolytic capacity between basal and stimulated cells. This was elucidated by measuring the extracellular acidification rate, which assessed the glycolytic rate in MAIT cells in real time. Enhancement of glycolysis was accompanied by significant upregulation of hexokinase 2, the first enzyme of the glycolytic pathway. On the other hand, short-time stimulation of MAIT cells significantly increased the expression of transcripts for pyruvate kinase, the last enzyme of the glycolysis pathway. However, after overnight stimulation, the level of PKM2 enzyme was non-significantly lower in comparison to the resting cells. This suggested that the transcription of this enzyme is more important at earlier timepoints in MAIT cell activation. Expression of LDHA, an enzyme involved in the conversion of pyruvate into lactic acid was higher after overnight stimulation, suggesting that MAIT cells were still converting pyruvate into lactate. Collectively this data is in line with results published on effector T cell metabolism. It indicates that MAIT cells engage in a similar metabolic programme upon activation, during which they significantly increase glycolysis, likely to support the demand for ATP production and provision of intermediates required for the synthesis of proteins such as cytokines or lipids and nucleic acids required for cell division.

Recent research has highlighted the importance of leucine in controlling mTORC1 activation, which is responsible for control of glycolysis in T cells (Saxton et al., 2016; Sinclair et al., 2013; Wolfson et al., 2016), therefore next we decided to investigate how the import of this amino acid affects the glycolytic metabolism of MAIT cells. Leucine is imported into the cells via large amino acid transporters, of which LAT1 mediates is majority of the transport (Maimaiti et al., 2020; Scalise et al., 2018; Sinclair et al., 2013), therefore we measured this transporter expression by MAIT cells. Transcription of mRNA for both components of LAT1 heterodimer was significantly increased upon MAIT cell stimulation, which was reflected on the protein level. This is in line with the data reported by Sinclair et al. who demonstrated enhancement of LAT1 expression by conventional CD8⁺ T cells upon stimulation (Sinclair et al., 2013). Transcription of other members of the LAT family of transporters varied upon stimulation. The observed diversity in the expression pattern may occur as a result of the induction of different amino acid transporters at different timepoints. Although LAT1-3 presented a trend towards upregulation of transcription of those transporters, expression of LAT4 has been downregulated. LAT4 has been shown to transport a range of amino acids including essential and branched chained amino acids, suggesting that transport of these amino acids via LAT4 may not play a critical role in activated MAIT cells (Bodoy et al., 2005). As upregulation of LAT1 has been the most prominent, next we evaluated it's transport capacity using a kynurenine assay developed by the Cantrell lab (Sinclair et al., 2018). Kynurenine, a metabolite of tryptophan, is a substrate of LAT1. The fluorescent properties of kynurenine allowed for a rapid measurement of its uptake by the immune cells, and hence allowed for indirect assessment of the efficiency of

transport of amino acids via this amino acid transporter (Sinclair et al., 2018). In line with the data published by Sinclair *et al.*, our study found that LAT1 expressed by MAIT cells is functional and imports kynurenine into these cells. To assess the effect of amino acid transport via LAT1 on MAIT cell glycolytic metabolism, we limited the influx of amino acids using BCH, a LAT1 specific inhibitor. Despite the activation signals, restraining of the amino acid transport resulted in a significant reduction in the rate of glycolysis, which resembled the rate of unstimulated MAIT cells, as well as significant reduction of glycolytic capacity was observed. This data indicated that LAT1 mediated amino acid transport is indispensable for modulation of the glycolysis rate.

mTORC1 is the master regulator of metabolism, which controls multiple processes including glycolysis (Salmond, 2018). Direct inhibition of mTORC1 activity with rapamycin during T cell stimulation or deletion of mTORC1 component Rheb, has been shown to have a detrimental effect on glycolysis, leading to impairment in its upregulation (Hukelmann et al., 2016; Pollizzi et al., 2015). mTORC1 is sensitive to the influx of leucine, which is largely mediated by LAT1 amino acid transporter (Saxton et al., 2016; Sinclair et al., 2013; Wolfson et al., 2016). Sinclair et al. has elegantly shown that T cells absolutely require LAT1 for metabolic reprogramming, and control of mTORC1 activity (Sinclair et al., 2013). Here we hypothesised that blocking LAT1 during MAIT cell stimulation, may result in impairment in activation of mTORC1 complex. Indeed, activation of mTORC1 as measured by phosphorylation of S6 protein, was reduced upon MAIT cell activation in the presence of LAT1 inhibitor. This was confirmed in expanded MAIT cells (O'Brien et al. 2019 Figure 4I), where the inhibition of LAT1 mediated transport, resulted in significant reduction in mTORC1 activity. This is in line with study published by Loftus et al. on NK cells where the blocking of amino acid transport via LAT1, led to the reduction in pS6 protein expression and c-Myc activation (Loftus et al., 2018). Parallel work from our lab has also revealed that the inhibition of mTORC1 with rapamycin in MAIT cells, significantly impairs their glycolytic metabolism as shown through the measurement of extracellular acidification rate (O'Brien et al. 2019 Figure 3E). Similar reports were published by Hukelmann et al. and Donnely et al. in T cells and NK cells respectively, indicating mTORC1 is crucial for controlling of glycolysis in lymphocytes (Donnelly et al., 2014; Hukelmann et al., 2016).

Collectively this data shows that MAIT cells undergo metabolic reprogramming upon stimulation. Activation of MAIT cells increases the expression of LAT1 amino acid receptor on the cell surface, which facilitates the uptake of amino acids including leucine, to support the activated state of the cell. Detection of leucine by mTORC1 increases its activity and leads to upregulation of the glycolysis rate (Figure 4.19).

People with obesity display a significant defect in MAIT cell proliferation and cytokine production, however the underpinning mechanism responsible for those defects has not yet been described. Having established some of the metabolic pathways required for successful MAIT cell activation, next we investigated those in people with obesity. MAIT cells from people with obesity displayed a defect at each point of this metabolic pathway. In comparison to the control individuals, MAIT cells from people with obesity displayed diminished expression of LAT1. This most likely leads to the reduced uptake of kynurenine and hence other LAT1 substrates such as leucine. This results in lower accumulation of leucine in the cell, which is sensed by mTORC1. Lower levels of the amino acid concentration may mistakenly provide a signal to the cell indicating there are not enough nutrients available for the cell to engage in activation. The failure of leucine to reach the appropriate threshold required to induce mTORC1 activity leads to a failure to upregulate glycolysis (Figure 4.19). More research is required to elucidate why the expression of LAT1 amino acid transporter is lower in people with obesity. A defect in LAT1 expression may be due to only partial engagement of the cells in a signalling pathway which is activated upon cell stimulation to upregulate the expression of mRNA for LAT1 and failure to export fully formed LAT1 transporters onto the surface of the cell.

Alterations in the function and metabolism of immune cells from people with obesity are not limited to MAIT cells. Children with obesity were shown to have a defect in their cytotoxic activity and presented with an activated metabolic phenotype in comparison to their lean counterparts (Tobin et al., 2017). This study found that NK cells from children with obesity have enhanced basal glycolysis in comparison to lean children and failed to upregulate it any further upon stimulation. This was accompanied by higher activation of mTORC1 basally and failure to significantly upregulate its activity after 18 hours stimulation. These

results indicate that resting NK cells from children with obesity are highly glycolytic, but fail to enhance their metabolism upon activation, possibly leading to the alteration in their function (Tobin et al., 2017). The failure of NK cells to increase their glycolytic metabolism upon activation, in part resembles the findings from our study of MAIT cells, which are not capable of enhancing their glycolytic metabolism to the levels observed in the control subjects. Study by Michelet *et al.* suggested this alteration in NK cell metabolism in people with obesity is due to their intracellular accumulation of lipids, which were shown to significantly decrease their mTORC1 activity, glycolysis and impaired their cytotoxicity (Michelet *et al.*, 2018). In the future, lipid handling by MAIT cells and its impact on their metabolism should also be investigated. These studies further indicate that obesity has a significant impact on immune cells function and metabolism and highlights that immune cell metabolism is closely related to their function.



Figure 4.19 Summary of findings of chapter four. Upon stimulation of MAIT cells from control individuals (left), *IFNy* and *SLC7A5* mRNA is transcribed. Expression LAT1 transporter is upregulated, which allows for amino acid influx. Transport of amino acids such as leucine, activates mTORC1, which enhances glycolysis producing multiple metabolic precursors. This supports increased IFNy production. MAIT cells from people with obesity (right) upon stimulation display lower levels of *IFNY* and *SLC7A5* transcription. In addition, LAT1 expression and transport is also impaired, leading to defects in mTORC1 activation. This in turn results in failure to enhance glycolysis, leading to lower level of metabolic precursors and hence lower IFNy production.

Chapter Five

Results

Metabolic requirements for MAIT cell function

5.1 Introduction

5.1.1 Immunometabolism and cell function

Recognition of cellular metabolism as one of the crucial processes that impacts the activity of immune cells has led to substantial advancements in our understanding of mechanisms which control immune cell function. Understanding the complex web of interlinked metabolic pathways poses a great challenge in identifying their direct or indirect association with a particular cellular response. It is now known that the metabolism of immune cells doesn't stay constant during its lifespan. Metabolism itself is very flexible and is capable of adjusting to the energetic and biosynthetic demands of the cell at any given time (Almeida et al., 2016). Extensive research has identified glycolysis and OxPhos as two key pathways implicated in supporting the immune function of T cells. The metabolic state of quiescent or effector T cells were shown to be quite distinct. During quiescent state, cells primarily engage in Kreb's cycle and OxPhos, to produce high amounts of ATP, whereas stimulation results in rapid metabolic reprogramming, leading to the strong enhancement of glycolysis, which becomes the dominant metabolic pathway (O'Neill et al., 2016). This provides the cell with energy in the form of ATP and a range of metabolic intermediates, which directly or indirectly contribute to T cell effector function as described in chapter four. As a result, recent research of T cell metabolism has been largely focused on the consequences brought on by the disturbance of glycolysis and glycolysis-related pathways during cell activation. Indeed, glycolysis was described to support the production of cytokines by T cells and impact their proliferation. Glucose restriction or inhibition of glycolysis reduced the capability of T cells to produce IFNy (Jacobs et al., 2008; Renner et al., 2015). Similar effects were brought on by inhibition of mTORC1 or amino acid influx, which indirectly led to the restriction of glycolytic metabolism (Pollizzi et al., 2015; Sinclair et al., 2013; Yang et al., 2013). The relationship between the glycolytic metabolism and T cell cytokine production has been explored in detail in chapter four, therefore here we will focus on the association between T cell metabolism and their proliferation. The involvement of metabolism in the production of T cell effector molecules is beginning to unravel, however details of how metabolism affects T cell proliferation is still relatively poorly explored. T cell proliferation is dependent on a range of metabolic processes including glycolysis and OxPhos. As early as 1994, glucose was reported to be essential for T cell proliferation. (Greiner et al.,

1994). This finding has been revisited since by multiple researchers. Jacobs *et al.* showed that glucose is required for T cell proliferation, as a reduction of glucose below 0.5mM in the culture media, significantly affected their proliferation capacity (Jacobs et al., 2008). Chang *et al.* also demonstrated that activated T cells rely on glycolysis as their incubation in media without sugars, strongly inhibited their proliferation. Substitution of glucose with galactose, which forced the cells to switch from glycolysis to OxPhos to support their function, slowed down T cell proliferation, but didn't inhibit it (Chang et al., 2013), as it has been observed with direct treatment of T cells with 2DG (Cham et al., 2008). This is in line with an earlier study by Delgoffe *et al.* which investigated the impact of Rheb deficiency (mTOCR1) on T cell function in mice. This study showed that mTORC1 deficiency causes retardation of T cell proliferation, but not a complete inhibition (Delgoffe et al., 2011).

As cells have been shown to switch between glycolysis and OxPhos depending on their energetic and biosynthetic requirements, the role of OxPhos has been investigated in T cell proliferation. Blocking of OxPhos using complex I, III or ATP synthase inhibitors does not lead to complete obstruction of proliferation. In fact, Chang *et al.* this study has also presented that inhibition of OxPhos, lead to enhancement of glycolysis, likely to compensate for the loss of ATP and biosynthetic precursors induced by the inhibitors (Chang *et al.*, 2013). Study by Sena *et al.* has shown that CD4 T cells lacking complex III have retained their ability to proliferate during lymphopenia but failed to respond to TCR induced proliferation. Authors of this publication suggested this defect is likely due to the lack of mROS required for T cell activation, and not due to the inability to generate biosynthetic precursors or ATP (Sena *et al.*, 2013). This data suggests that T cells rely on both glycolysis and OxPhos to proliferate, however more research is required to elucidate the exact contribution of each of these pathways in supporting T cell proliferation.

Amino acids play a role in T cell proliferation. Deletion of SLC7A5, a component of LAT1 amino acid transporter in a mouse model severely impacted the proliferation of T cells *in vitro* (Sinclair et al., 2013). Based on the literature and results obtained in chapter 4, it can be hypothesised that deletion of SLC7A5 leads to lower influx of leucine, leading to lower activation of mTORC1 and lower rates of glycolysis. Again, this highlights the importance of glycolytic metabolism in T cell proliferation. The role of glutamine in T cell proliferation was also investigated. ASCT2 glutamine transporter was shown to be dispensable to T cell proliferation, as deletion of ASCT2, did not affect T cell proliferation (Nakaya et al., 2014). Although ASCT2 was not deemed indispensable, subsequent study by Sener *et al.* showed glutamine is required for proliferation, as media with no glutamine abrogated this process. In addition, inhibition of glutaminase, an enzyme which catalyses the metabolism of glutamine to glutamate, also led to significant impairment of T cell proliferation (Sener et al., 2016). Collectively these studies indicate that glutamine metabolism, similarly to glycolysis and OxPhos supports T cell proliferation.

MAIT cells, as a novel subset of T cells, are still very poorly understood in comparison to the conventional T cells. The past decade has led to a greater appreciation of this subset and their role in disease and health, however the mechanisms controlling MAIT cell functions remains largely unexplored. Using the knowledge gained during the investigation of MAIT cell metabolism in chapter four and the evidence presented in the literature discussed above on the associations between immune cell function and cell metabolism, here we decided to investigate whether MAIT cell metabolism also plays a role in supporting their effector functions.

5.2 Specific aims of this chapter

The specific aim of this chapter was to identify the metabolic requirements for MAIT cell functional responses.

Therefore to:

- Investigate how glucose availability and glycolysis affects is related to MAIT cell proliferation, expansion and IFNγ production
- Elucidate how amino acid influx affects MAIT cell proliferation, expansion and IFNy production
- Determine whether OxPhos is implicated in supporting MAIT cell proliferation and expansion

5.3 Results

5.3.1 MAIT cells use glucose to support proliferation

Having identified that MAIT cells engage in glycolytic metabolism upon activation, we next investigated the metabolic requirements for MAIT cell proliferation in the control individuals. We first cultured MAIT cells in culture media with varying levels of glucose (10mM, 5mM and 1mM) for 5 days and assessed MAIT cell proliferation via cell trace violet, CTV. MAIT cells proliferated readily at the standard concentration of 10mM glucose (Figure 5.1 A&B). Reduction of glucose to 5mM did not affect MAIT cell proliferation (Figure 5.1 C&D), however a reduction of glucose concentration in the media to 1mM slowed proliferation (Figure 5.1 C&D). As MAIT cell proliferation was dependent on IL-2 (Figure 3.5, 3.6, 3.8), next we investigated whether glucose restriction (1mM) impacts the levels of CD25. In glucose restricted conditions (1mM glucose), reduced IL-2R expression was noted in comparison to 10mM glucose (Figure 5.2 A&B). To confirm the requirement of glucose for MAIT cell proliferation and expansion, we substituted glucose in our culture media with galactose, which limits the rate of glycolysis in the cell. We demonstrate that in galactose culture media, MAIT cell expansion was greatly reduced in comparison to glucose culture media during the five-day incubation period (Figure 5.3 A&B). These results show that MAIT cells require an influx of appropriate amounts of glucose to facilitate their proliferation and expression of IL-2R, which allows them to respond to proliferative signals in the culture.



Figure 5.1 MAIT cell proliferation is restricted by glucose availability. (A) Scatter plot and (B) representative histograms of two control samples of MAIT cell proliferation as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM on day 0 and day 5 of culture. PBMCs were incubated in 10mM glucose full media and were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/mL) on day 1 followed by a high dose of IL-2 (33.3ng/mL) on day 4 (n=7). (C) Scatter plot of MAIT cell proliferation at 10mM, 5mM and 1mM glucose on day 5 of culture (n=7). (D) Representative comparison histograms of MAIT cell proliferation on day 5 with glucose at 10mM and 5 mM or 10mM and 1mM. Statistical analysis performed using paired student's t-test or one-way ANOVA with Tukey's correction, * p≤0.05, ****p<0.0001.



Figure 5.2 Restriction B^Pf¹¹ **BIUCOSCD2SuHGBH1DUCOSCD2SuHGBH1DUCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuH1CCD2SuD1CD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD2SuD1CD1CD2SuD1CD1CD2SuD1CD1CD1CD2SuD1CD1**



Figure 5.3 MAIT cells expansion is restricted by substitution of glucose with galactose. (A) Scatter plot and (B) representative dot plots of two control samples of MAIT cell expansion as measured by the percentage of CD3 cells on day 0 and day 7 of culture. PBMCs were incubated in 10mM glucose or 10mM glutamine media and were unstimulated (Day 0) or stimulated with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/mL) on day 1 followed by a high dose of IL-2 (33.3ng/mL) on day 4 (n=6). Statistical analysis performed using paired student's t-test or one-way ANOVA with Tukey's correction, * p≤0.05, ****p<0.0001.

5.3.2 MAIT cells require glycolysis for proliferation

As shown in section 5.3.1 glucose supports proliferation of MAIT cells and expression of IL-2R. Therefore, to further explore metabolic requirements for MAIT cell proliferation, we decided to investigate the significance of glycolysis in this process using a 2-deoxy-d-glucose inhibitor. 2DG is a glucose analogue, which enters glycolysis, just as glucose does, however 2DG inhibits hexokinase 2, the first enzyme of glycolysis and suppresses the entire pathway. PBMC were stained with CTV for analysis of MAIT cell proliferation or remained unstained for the assessment of MAIT cell expansion. Next cells were activated with 5-ARU-MG, in the presence or absence of 2DG for the first 18 hours of culture. Then cells were stimulated with IL-2 and cultured for a total of 5 days and analysed by flow cytometry. Here we demonstrated that glycolysis inhibition significantly reduced MAIT cell proliferation and expansion (Figure 5.4 A-D). As expression of IL-2R was shown to be sensitive to glucose concentration, next we investigated whether inhibition of glycolysis with 2DG also impacts the level of IL-2R present on MAIT cell surface. Similar to restriction of glucose (Figure 5.2 A&B), stimulation of MAIT cells as per section 5.3.1 in the presence or absence of the glycolysis inhibitor for 18 hours significantly reduced IL-2R expression upon stimulation (Figure 5.5 A&B). This data suggests that glycolysis is central to MAIT cell proliferation, as inhibition of this metabolic process halts expansion of MAITs cells, which in part may be due to the reduction in the expression of IL-2R.

5.3.3 MAIT cells require an influx of amino acids via LAT1 for proliferation Activated MAIT cells were shown to be highly dependent on the transport of amino acids via LAT1 transporter in order to enhance their rate of glycolysis (Figure 4.18). As glucose and glycolysis were shown to play a central role in MAIT cell proliferation (Figure 5.1, 5.3 & 5.4), we next investigated the requirement of amino acid transport via LAT1 for MAIT cells proliferation. PBMC were prepared and stimulated as described in section 5.3.2 in the presence or absence of BCH. Inhibition of amino acid transport by blocking the activity of LAT1 transporter using BCH, significantly reduced MAIT cell proliferation (Figure 5.6 A&B) and expansion (Figure 5.6 C&D). Collectively with data obtained in chapter four, these results highlight the importance of amino acid influx for MAIT cell function.



Figure 5.4 MAIT cells require glycolysis to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell proliferation over 5 days of culture with or without the addition of 2DG (10mM) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of 2DG (10mM) as measured by uptake and by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 in the presence or absence of 2DG (10mM) (n=4/5). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** $p \le 0.01$, **** $p \le 0.0001$.





Figure 5.6 MAIT cells require amino acid influx to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell proliferation over 5 days of culture with or without the addition of BCH (50mM) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of BCH (50mM) as measured by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 in the presence or absence of BCH (50mM) (n=6/4). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** p≤0.01, **** p≤0.0001.

5.3.4 MAIT cells require mTORC1 for proliferation

Influx of amino acids was shown to be important for the activation of mTORC1 in MAIT cells (Figure 4.17). mTORC1 is known to orchestrate metabolism of T cells. Indeed, previous studies in T cells have demonstrated that the activation of mTORC1 in part relies on the uptake of leucine, which is transported into the cells via LAT1. Activation of mTORC1, was in turn shown to be required for upregulation of glycolysis during activation. As both glycolysis and amino acids were shown to be required for MAIT cell proliferation, next we investigated the role of mTORC1 on MAIT cell proliferation. PBMC were stained with CTV for analysis of MAIT cell proliferation or remained unstained for the assessment of MAIT cell expansion. Then cells were stimulated with 5-ARU-MG, in the presence or absence of Rapamycin for the first 18 hours of culture. Then cells were supplemented with IL-2, cultured for a total of 5 days and analysed by flow cytometry. Rapamycin, a specific mTORC1 inhibitor, significantly limited MAIT cell proliferation (Figure 5.7 A&B) and expansion (Figure 5.7 C&D). This data indicates that activation of mTORC1 is required for progression of MAIT cell proliferation.



Figure 5.7 MAIT cells require mTORC1 activation to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell proliferation over 5 days of culture with or without the addition of Rapamycin (20nM) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of Rapamycin (20nM) as measured by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 in the presence or absence of Rapamycin (20nM) (n=6/4). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** $p \le 0.01$, **** $p \le 0.0001$.

5.3.5 MAIT cells require glutamine metabolism and OxPhos for proliferation

Having identified the dependency of MAIT cell proliferation on a LAT1-mTORC1-Glycolysis axis, we extended our investigations to other metabolic pathways. OxPhos has been reported to play a role in supporting conventional T cell proliferation (Bailis et al., 2019; Sena et al., 2013). Therefore, we decided to investigate the role of OxPhos in MAIT cell proliferation.

OxPhos is the most effective way of producing large amounts of energy in the form of ATP by the cell, and proliferation is a very energy and nutrient demanding process. In this series of experiments, we decided to explore whether the inhibition of particular components of the OxPhos machinery to limit ATP production impacts the rate of MAIT cell proliferation. Two inhibitors of the electron transport chain were used - oligomycin, an ATP synthase inhibitor or antimycin A, an inhibitor of complex III of electron transport chain (ETC). PBMC were prepared and stimulated as described in section 5.3.4 in the presence or absence of oligomycin or antimycin A for the first 18 hours of the experiment and then cells were analysed using flow cytometry. Inhibition of ATP synthase with oligomycin significantly reduced MAIT cell proliferation and expansion by day 5 of culture (Figure 5.8 A-D). Halt in the proliferation occurred early in proliferation, similar to what was observed following treatment with 2DG (Figure 5.4) and BCH (Figure 5.6), as signified by one to two discrete peaks shown on the representative histograms (Figure 5.8 B). Treatment of cells with antimycin A, the inhibitor of complex III of the ETC, slowed MAIT cell proliferation and expansion, however to a lesser extent (Figure 5.9 A-D). As depicted by the representative histograms in Figure 5.9 B, inhibition of complex III of ETC still allowed for MAIT cell expansion into multiple generations, shown by the multiple individual peaks on the graph.

This data suggests that the generation of ATP via the ETC is necessary for effective MAIT cell proliferation. Inhibition of the complex III of ETC has a negative impact on MAIT cell expansion, however it may still allow for ATP production. As previously mention in section 4.3.3, glutamine can feed into the Kreb's cycle and hence support OxPhos. In order for glutamine to enter the Kreb's cycle, it must undergo glutaminolysis, an enzymatic process during which glutamine is broken

down into α -ketoglutarate (α -KG), an intermediate of the Kreb's cycle (O'Neill et al., 2016). It has been shown that glutaminase (GLS), the first enzyme involved in this process, can be inhibited with a small molecule inhibitor CB-839 in leukaemia cells, resulting in their reduced proliferation and function. To assess the role of glutamine in supporting MAIT cell proliferation, we used a glutaminase inhibitor CB-839. PBMC were prepared and stimulated as described above in the presence or absence of CB-839 Treatment with CB-839 slowed MAIT cell proliferation (Figure 5.10 A&B), resulting in significantly reduced MAIT cell expansion (Figure 5.10 C&D). Overall, these results indicate a role for OxPhos in supporting MAIT cell proliferation, most likely through provision of ATP required to support the process of proliferation.



Figure 5.8 MAIT cells require active ATP synthase to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell proliferation over 5 days of culture with or without the addition of oligomycin (2µM) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of oligomycin (2µM) as measured by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 in the presence or absence of oligomycin (2µM) (n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** p≤0.01, **** p≤0.001.



Figure 5.9 MAIT cells require OxPhos complex III to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell_proliferation_over_5_days of culture with or without the addition of Antimycin A (A) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5µM, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of Antimycin A (4µM) as measured by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1µg/mL) and MG (100µL) on day 0 and received a low dose of IL-2 (6.75ng/µL) on day 1 followed by a high dose of IL-2 (33.3ng/µL) on day 4 in the presence or absence of Antimycin A (4µM) (n=6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** p≤0.01, **** p≤0.001.



Figure 5.10 MAIT cells require glutamine metabolism to proliferate in response to 5-ARU-MG and IL-2. (A) Scatter plot and (B) representative histograms of 2 control subjects showing MAIT cell proliferation over 5 days of culture with or without the addition of CB-839 (10 μ M) as measured by uptake and dilution of Cell Trace Violet (CTV) at 5 μ M, on day 0 and day 5 of culture. (C) Scatter plot and (D) representative dot plots of MAIT cell expansion over 5 days of culture with or without the addition of CB-839 (10 μ M) as measured by the percentage of CD3 population. For both experiments, PBMCs were unstimulated (Day 0) or stimulated (Day 5) with 5-ARU (1 μ g/mL) and MG (100 μ L) on day 0 and received a low dose of IL-2 (6.75ng/ μ L) on day 1 followed by a high dose of IL-2 (33.3ng/ μ L) on day 4 in the presence or absence of CB839 (10 μ M) (n=8/6). Statistical analysis performed using one-way ANOVA with Tukey's correction, ** p<0.01, **** p<0.0001.

5.3.6 Glycolysis is required for IFNy production

We next investigated the metabolic requirements for MAIT cell cytokine production, their primary effector function. Previous studies in T cells have shown that glycolysis is crucial for cytokine production. Renner *et al.* have shown that production of IFN γ is intrinsically dependent on glycolytic metabolism (Renner et al., 2015). To investigate expanded MAIT cells were stimulated with TCR Dynabeads, IL-12 and IL-18 in the presence or absence of 2DG for 24 hours and the supernatants were tested using ELISA. Here we demonstrate a strong reduction in IFN γ production by MAIT cells (Figure 5.11 A). This data shows that MAIT cells upregulate glycolysis upon cell activation, which is required for their production of IFN γ .




5.3.7 Amino acid transport is required for IFNy production

In chapter 4, we identified a LAT1-mTOR-glycolysis axis in activated MAIT cells. We next investigated if LAT1 activity was required for MAIT cell cytokine production. Previous studies have demonstrated that the inhibition of LAT1 function in T cells leads to a decrease in their cytokine production (Hayashi et al., 2013), however, this has not been investigated in MAIT cells. Using PBMC from healthy controls we show that upon activation with TCR microbeads, IL-12 and IL-18 for 18 hours MAIT cells increased their IFNy production (Figure 5.12 A&B), and with inhibition of the amino acid flux via LAT1, we observed a reduction in IFNy expression by MAIT cells (Figure 5.12 C&D). To confirm that the LAT1 inhibitor BCH was acting directly on MAIT cells, MAIT cells were expanded. Upon stimulation with TCR Dynabeads, IL-12 and IL-18 for 24 hours, MAIT cells produced high levels of IFNy (Figure 5.12 E), however IFNy levels were significantly reduced in the presence of the LAT1 inhibitor BCH (Figure 5.12 E). We next investigated whether the inhibition of amino acid influx had any impact on early transcription of mRNA coding for IFNy and T-bet mRNA. Expanded MAIT cells were pre-treated with BCH for 1 hour and then stimulated with TCR Dynabeads, IL-12 and IL-18 for further 4 hours. Here we demonstrate no impact on the transcription of mRNA for IFNy or T-bet (Figure 5.12 F&G). The mean fold increase of IFNG transcription was higher than transcription of IFNG in the absence of the inhibitor, however this difference was not significant (Figure 5.12 F). To investigate whether the inhibition of amino acids influx has any effect on the transcription of key glycolytic enzymes HKII and PKM, expression of mRNA for both of these enzymes was measured on expanded MAIT cells treated as described for Figure 5.12 F&G. Incubation of cells in the presence of BCH upon cell stimulation, revealed no difference in the transcription level of those two enzymes of glycolysis (Figure 5.13 A&B). In summary, this data shows that the transport amino acids via LAT1 is essential for IFNy production, but no difference in transcription was observed at 4 hours. Inhibition of amino acid flux had also no effect on the expression of key glycolytic enzymes at this timepoint.



Figure 5.12 Blocking amino acid transport by BCH reduces IFNy production by MAIT cells. (A) Scatter plot and (B) representative dot plots of IFNy production from PBMCs unstimulated or stimulated with TCR microbeads at 25ng/mL (Milteneyi), IL-12 and IL-18 (at 50ng/mL each) for 18 in the control individuals (n=3). (C) Scatter plot and (D) representative dot plots of IFNy production from PBMCs stimulated with TCR microbeads at 25ng/mL (Milteneyi), IL-12 and IL-18 (at 50ng/mL each) with or without addition of BCH (50mM) for 18 hours in the control population sample (n=3). (E) Scatter plot of IFNy production by in vitro expanded and purified MAIT cells upon stimulation with TCR Dynabeads (1:1 bead to cell ration) and IL-12/IL-18 (50ng/mL each) cytokines for 24 hours in the presence or absence of BCH (50mM) for 24 hours (n=7). (F) Scatter plot of IFNG (n=4) and (G) TBX21 (n=5) mRNA transcription by in vitro expanded and purified MAIT cells upon stimulation with TCR Dynabeads (1:1 bead to cell ration) and IL-12/IL-18 (50ng/mL each) cytokines for 4 hours in the presence or absence of BCH (50mM) for 1 hour prior to stimulation. Statistical analysis performed using student's t-test, or one-way ANOVA with Tukey's correction, ns- non-significant, *p<0.05, **p<0.01, ****p<0.0001



Figure 5.13 Blocking amino acid transport using BCH doesn't affect the transcription of *HK2* or *PKM* by MAIT cells at an early stage during activation. (A) Scatter plot of *HK2* mRNA and (B) *PKM* transcription by *in vitro* expanded and purified MAIT cells upon stimulation with TCR Dynabeads (1:1 bead to cell ration) and IL-12/IL-18 (50ng/mL each) cytokines for 4 hours in the presence or absence of BCH for 1 hour prior to stimulation(50mM) (n=5). Statistical analysis performed using one-way ANOVA with Tukey's correction, ns-non-significant, *p<0.05.

5.3.8 mTORC1 is required for IFNy production

We next investigated inhibition of mTORC1 has any effect on IFN_γ production using the specific inhibitor rapamycin Expanded MAIT cells were stimulated as per section 5.3.6 in the presence in Rapamycin inhibitor for 24 hours and IFN_γ production was measured by ELISA. A significant decrease in the production of IFN_γ cytokine by MAIT cells in the presence of mTORC1 inhibitor was observed (Figure 5.14 A). Together this data shows that activity of mTORC1 is required for upregulation of glycolysis, which in turn is required for successful cytokine production.

5.3.9 Glutamine metabolism supports IFNy production

Inhibition of glutamine metabolism limited MAIT cell proliferation (Figure 5.10); however, no studies have been performed to investigate whether the inhibition of glutamine metabolism affects MAIT cell cytokine production. We decided to investigate the effect of GLS inhibition on MAIT cell IFNγ production. Expanded MAIT cells were stimulated as per section 5.3.6 in the presence in CB-839 inhibitor for 24 hours and IFNγ was measured by ELISA. Inhibition of GLS with CB-839 resulted in a modest reduction in IFNγ production (Figure 5.15 A). Although the effect of CB-839 was modest in comparison to inhibition of glycolysis or mTORC1 with 2DG or Rapamycin respectively (Figure 5.11, 5.14 & 5.15), our data suggests a role for glutamine in supporting the MAIT cell function. This is further supported by the observed upregulation of ASCT2 amino acid transporter mRNA expression during the early stages of MAIT cell activation, what may support uptake of glutamine into the cells



Figure 5.14 Inhibition of mTORC1 with rapamycin reduces IFNy production by MAIT cells. (A) Scatter plot of IFNy production by in vitro expanded and purified MAIT cells upon stimulation with TCR Dynabeads (1:1 bead to cell ration) and IL-12/IL-18 (50ng/mL each) cytokines for 24 hours in the presence or absence of Rapamycin (20nM) for 24 hours (n=13). Statistical analysis performed using one-way ANOVA with Tukey's correction, ****p<0.0001.



Figure 5.15 Inhibition of glutaminase with CB-839 reduces IFNy production by MAIT cells. (A) Scatter plot of IFNy production by in vitro expanded and purified MAIT cells upon stimulation with TCR Dynabeads (1:1 bead to cell ration) and IL-12/IL-18 (50ng/mL each) cytokines for 24 hours in the presence or absence of CB839 (10 μ M) for 24 hours (n=13). Statistical analysis performed using one-way ANOVA with Tukey's correction, ****p<0.0001.

5.4 Discussion

This study shows that MAIT cells engage in multiple interconnected metabolic pathways upon activation, suggesting that early metabolic reprogramming is required to facilitate their effector functions. Multiple studies in T cells and NK cells have shown that cell metabolism is in fact required for optimal proliferation and cytokine production (Bailis et al., 2019; Chang et al., 2013; Keating et al., 2016; Loftus et al., 2018; Renner et al., 2015). Therefore, we investigated the metabolic requirements for MAIT cell functions within this chapter.

Although the rate of metabolism can be controlled at multiple stages, nutrient and substrate availability is one of the key limiting steps for any enzymatic reaction. Here we show that by restricting glucose availability to 1mM, this leads to impaired MAIT cell proliferation and expansion. This is in line with Jacobs *et al.* who demonstrated that glucose restriction negatively affects T cell proliferation (Jacobs et al., 2008). To further explore the role of glucose on MAIT cell function, we substituted glucose for galactose. Supplying galactose as the only source of carbohydrates, it urges the cell to favour OxPhos over glycolysis (Chang et al., 2013; Weinberg et al., 2010). MAIT cells cultured in galactose containing media displayed delayed expansion in comparison to the cells cultured in media with glucose. Chang *et al.* have recorded very similar findings in T cells, where the incubation of T cells in the presence of galactose slowed down their proliferation.

To further investigate the role of glucose and its metabolism on MAIT cell function, we inhibited glycolysis using 2DG. The presence of 2DG in culture led to a very strong inhibition of MAIT cell proliferation and IFN_Y production in comparison to the cells cultured without the inhibitor. In line with these results, Procaccini *et al.* have reported a significant reduction in proliferation of T cells when activated in the presence of 2DG (Procaccini *et al.*, 2016), whereas others have reported that glycolysis and glucose is required for their IFN_Y production (Dimeloe et al., 2016; Renner et al., 2015; Yin et al., 2015). In addition, restriction of glucose or inhibition of glycolysis resulted in the reduction in IL-2 receptor expression by MAIT cells upon stimulation. As we showed (in chapter four) exogenous supplementation of IL-2 strongly enhances MAIT cell proliferation, we therefore hypothesised that the level of glycolysis affects the expression of IL-2 receptor and hence contribute to the impaired proliferation of MAIT cells.

Collectively, our study provides extensive data, which supports the hypothesis that glycolysis is central to MAIT cell function as inhibition of glycolysis, through the withdrawal or restriction of glucose, induces a significant defect in their proliferation and IFNy production. Here we demonstrate that an abundance of glucose and glycolysis are important at the early stages of MAIT cell activation, although more studies are required to assess what metabolic events occur at later timepoints. The role of augmented glycolysis in MAIT cell activation remains to be fully elucidated, however studies conducted on T cells, have shown that the metabolic intermediates derived from glycolysis are the primary reason for the metabolic switch. For example, the first intermediate of glycolysis, glucose-6phosphate, can enter the pentose phosphate pathway to synthesise new nucleic acids, essential for cell proliferation. Other intermediates of glycolysis can contribute to amino acids synthesis, which may play a role in the augmented production of cytokines during cell activation (O'Neill et al., 2016). Intermediates of glycolysis were shown to be directly related to IFNy production in T cells (Ho et al., 2015; Peng et al., 2016). In addition, studies of cell cycle and cell metabolism reported that cell cycle progression is sensitive to nutrient availability, where restriction of glucose or inhibition of glycolysis prevented progression of the cell cycle (Kalucka et al., 2015). This suggests that inadequate rate of glycolysis or extracellular glucose concentration, may provide a signal to the cell indicating that there are insufficient nutrients for cell activation to take place, hence preventing cell proliferation and cytokine production. This is of particular importance in the hypoglycaemic settings in vivo, such as tumour microenvironment. Restricted access of MAIT cells to glucose may hinder their potential anti-tumour responses (Godfrey et al., 2018; Vacchini et al., 2020).

As this study has shown that amino acid influx is critical for the upregulation of glycolysis upon MAIT cell activation, next we investigated the role of LAT1 mediated amino acid transport on MAIT cell function. Inhibition of LAT1 using BCH resulted in a restriction of proliferation and expansion of MAIT cells. This is in line with the results described in chapter four, which depicted the inhibition of amino acid influx via LAT1, leads to suppressed mTORC1 activation, that occurs most likely due to hindered transport of leucine. Low activity of mTORC1 prevents enhancement of glycolysis, which we show is crucial for MAIT cell function. Limitation of leucine import most likely contributes to the defect in MAIT cell

proliferation. Interruption in the import of methionine via LAT1, an amino acid, which is required for methylation of DNA, process required for T cell proliferation and differentiation may also fortify this effect (Sinclair et al., 2019). In our study, LAT1 amino acid transport was also required for IFNy protein production by MAIT cells. Blocking of amino acid influx upon MAIT cell stimulation didn't affect early transcription of mRNA coding for IFNy and T-bet or the transcription of glycolytic enzymes, HK2 or PKM. This suggests that despite suboptimal uptake of amino acids, MAIT cells become activated and initiate transcription of mRNA for the proteins required to support their effector functions. This suggests that impairment in upregulation of glycolysis and IFNy synthesis in the presence of BCH occurs due to the alterations which occur at translational level. Low intracellular concentration of leucine and other amino acids prevents MAIT cells from enhancing glycolysis via mTORC1, leading to the impairment in IFNy production. Low cytokine production may be in part due to low levels of amino acids required to fulfil translation of the mRNA of the effector proteins. This is in line with the literature, as the studies showed that T cells also require amino acids to support cytokine production (Hayashi et al., 2013; Ron-Harel et al., 2019; Sinclair et al., 2019, 2013).

Leucine has been described to be required to activate mTORC1, which in turn is responsible for the regulation of glycolysis. As amino acid transport was shown to be required for MAIT cell function, next we investigated how the inhibition of mTORC1 with rapamycin would affect MAIT cell proliferation and IFNγ production. Similar to the inhibition of glycolysis and amino acid influx, preventing mTORC1 activation, decreased the proliferative capacity and production of IFNγ by MAIT cells. This is in line with studies which have demonstrated a reduction in production of IFNγ and proliferation of T cells when incubated in the presence of mTORC1 inhibitor (Chaoul et al., 2015; Herrero-Sánchez et al., 2016; Kurebayashi et al., 2012), which has been further confirmed in *in vivo* studies of Rheb or Raptor deficient mice (Pollizzi et al., 2015; Yang et al., 2013). Collectively, this data shows that MAIT cells enhance their glycolytic metabolism during activation to support their effector functions. Any defect in this pathway, that controls the glycolytic rate in the cell, leads to severe impairment of MAIT cell proliferation and IFNγ production.

It is now widely accepted that T cells use OxPhos as their primary source of energy in their quiescent state but switch to glycolysis upon activation (O'Neill et al., 2016). Data published on T cell metabolism indicates that engagement of the cells in OxPhos during proliferation is important (Bailis et al., 2019; Sena et al., 2013). After establishing that MAIT cells rely on glycolytic metabolism for their cytokine production, proliferation and expansion, next we focused on the role of OxPhos in MAIT cell function. Study performed by O'Brien et al. in our laboratory and described in her thesis, has shown that OxPhos was not required for production of IFNy by MAIT cells upon stimulation. In contrast to these results, here we have shown that although OxPhos may not be critical to MAIT cell cytokine production, it plays a role in their proliferation and expansion. Blocking of ATP synthase using oligomycin, resulted in the inhibition of MAIT cell proliferation and expansion, whereas the inhibition of complex III of ETC, resulted in a retardation of proliferation and expansion, rather than inhibition. The differences observed between the inhibition of two different components in ETC was most likely due to the role they fulfil in this reaction. Inhibition of ATP synthase has a more severe impact as it is required for production of all ATP generated during the respiration. Inhibition of complex III most likely only slows down the reaction, but still produces ATP. As a result, ETC provides the cell with a limited amount of energy, leading to suboptimal proliferation. This is in line with the data published by Bailis et al. who reported significant defect in the proliferation of T cells after inhibiting complex III of ETC (Bailis et al., 2019). In addition, we investigated the importance of glutamine in supporting MAIT function, by inhibiting GLS with CB-839. Inhibition of glutamine metabolism slowed down MAIT cell expansion, proliferation and IFNy production, but had a modest effect in comparison to direct or indirect inhibition of glycolysis. This is in line with a study by Carr et al. which showed that glutamine is essential for proliferation and cytokine production (Carr et al., 2010). Johnson et al. have also shown that a lack of glutamine in media strongly halts T cell proliferation, whereas inhibition of GLS with CB-839 modestly slows it down (Johnson et al., 2018).

Collectively this data provides extensive evidence that MAIT cells undergo an early metabolic reprogramming upon activation to support cytokine production and proliferation. Glycolytic metabolism was shown to be indispensable for MAIT cell function. Inhibiting any of the steps required for the modulation of glycolytic metabolism or direct inhibition of glycolysis resulted in profound defects in cell proliferation and IFNγ production (Figure 5.16). In this study, we showed that MAIT cells from people with obesity have dysregulated glycolytic metabolism and fail to adjust glycolysis upon stimulation. We suggest that this deficiency underpins the defect in proliferation and IFNγ production by MAIT cells from people with obesity. In addition, OxPhos was also shown to support MAIT cell proliferation, therefore this also should be assessed in MAIT cells in people with obesity. Defects in both metabolic pathways may be responsible for the observed impairment in MAIT cell proliferation from people with obesity and thus lead to the reduced frequencies observed in the peripheral blood and adipose tissue (Carolan et al., 2015; Magalhaes et al., 2015). Although this study provides a plausible explanation for the reduction of MAIT cell frequencies observed in people with obesity, to elucidate this further activation induced cell death of MAIT cells in people with obesity should also be assessed to fully explain lower MAIT cells frequencies.



Figure 5.16 Summary of results of chapter five. MAIT cells from control individuals require glycolytic metabolism for proliferation and IFN γ production. Inhibition of any step in the pathway which controls the rate of glycolysis was shown to have a profound impact on MAIT cell function. OxPhos was shown to also support MAIT cell proliferation, however to a lower extent

Chapter Six

General Discussion

6. Discussion

Nearly thirty years after the discovery of MAIT cells, they still remain largely unexplored in comparison to other cells of the innate and adaptive immune system. Their high frequencies in multiple human tissues and pleotropic functions suggest they play an important role in the protection of their host and maintenance of tissue homeostasis.

Advancements in the research of MAIT cell biology and function were possible due to the development of monoclonal antibodies against V α 7.2 TCR and identification of MAIT cell antigens. This led to the development of tetramers and flow cytometric antibodies, which allowed for more detailed studies (Corbett et al., 2014; Martin et al., 2009; Rahimpour et al., 2015; Reantragoon et al., 2013). MAIT cell phenotype and function have been described in healthy and diseased individuals, and it has been shown they are susceptible to the alterations that may occur in their environment during the development of illnesses (Provine and Klenerman, 2020). In healthy individuals, MAIT cells respond to TCR dependent or independent stimulation, through the upregulation of a large number of effector proteins including cytokines, chemokines and cytotoxic molecules (Provine and Klenerman, 2020). Production of these mediators has been found to support the resolution of microbial infection (Chua et al., 2012; Meierovics and Cowley, 2016). Expression of chemokines induces recruitment of other immune cells to the site of inflammation, which are likely to become stimulated by cytokines such as IFNy, TNF- α and IL-17, produced by activated MAIT cells (Meierovics et al., 2013; Meierovics and Cowley, 2016; Provine and Klenerman, 2020). These cytokines play an important role in eradiation of the infection. IFNy acts on macrophages and enhancing phagocytic ability, whereas TNF- α and IL-17 induce the production of more pro-inflammatory cytokines and chemokines, other immune cells as well as epithelial and mesenchymal cells and hence exacerbate inflammation (Gaffen et al., 2014; Wu et al., 2014).

As MAIT cells were identified as potent producers of cytokines, the expression of them has been used as markers of their function in the state of disease. Dysregulation of MAIT cell function has been identified in multiple inflammatory and autoimmune diseases (Cho et al., 2014; Gracey et al., 2016; Willing et al., 2018). The first alteration observed in many diseases in which the phenotype of MAIT cells is changed, is a decrease in their frequencies in the peripheral blood of individuals with diseases such as RA, IBD or SLE (Chiba et al., 2017; Hayashi et al., 2016; Serriari et al., 2014). Furthermore, blood MAIT cells largely display a pro-inflammatory profile with enhanced IL-17 production and a variable IFNγ expression (Pisarska et al., 2020). In RA and IBD increased frequency of type 17 MAIT cells has been reported at the sites of inflammation in the synovial fluid, gut mucosa or sclerotic lesions respectively, suggesting their migration and possible contribution to the inflammatory process (Gracey et al., 2016; Tominaga et al., 2017).

Investigations of MAIT cell function in obesity demonstrated severe alterations (Carolan et al., 2015; Magalhaes et al., 2015; Toubal et al., 2020). Activation of peripheral blood MAIT cells from people with obesity *in vitro* revealed a significant impairment in their IFN_Y production as compared to the controls, whereas IL-17 levels were strongly enhanced (Carolan et al., 2015; Magalhaes et al., 2015). In addition, AT MAIT cells from people with obesity also displayed enhanced IL-17 but diminished IFN_Y production (Carolan et al., 2015; Magalhaes et al., 2015). A very similar phenotype of peripheral blood MAIT cells was observed in colorectal cancer patients, who displayed a reduction in peripheral blood MAIT cell frequencies with impaired IFN_Y but enhanced IL-17 production (Ling et al., 2016). IFN_Y production by MAIT cells has been reported to be diminished in the adenocarcinoma and hepatocellular carcinoma lesions (Duan et al., 2019; Melo et al., 2019; Sundström et al., 2015). This suggests that cancer and obesity induce similar changes in MAIT cell function.

In this study, we confirmed lower frequencies of MAIT cells in the peripheral blood in people in obesity in comparison to the control individuals, which also displayed altered functional phenotype. In line with previously published data, MAIT cells from people with obesity presented with impaired IFNγ production but enhanced IL-17 synthesis (Carolan et al., 2015; Magalhaes et al., 2015). Here, for the first time, we report that in addition to the alterations in the cytokine production, peripheral blood MAIT cells from people with obesity display impaired MAIT cell expansion upon activation. This suggests that the failure of MAIT cells to expand may be in part responsible for the lower frequencies observed in the peripheral blood of people with obesity. What's more, this data suggests that obesity has

even bigger consequences on immune cell function. In fact, MAIT cells are not the only population affected by excessive weight gain. Michelet *et al.* and Tobin *et al.* reported impaired cytokine production and killing capabilities of NK cells from people with obesity (Michelet et al., 2018; Tobin et al., 2017).

Although changes in MAIT cell biology have been described in multiple disease settings, no clear mechanism controlling their function has yet been established. Recent progression in the field of immunometabolism identified an extra layer of control points of immune cell activation. Immune cell function is closely related to cell metabolism, where the disruption to the glycolytic metabolism or OxPhos pathways results in the disruption of their activation or effector functions (O'Neill et al., 2016). Resting T cells have been shown to primarily use Kreb's cycle and OxPhos to support energy production and survival (O'Neill et al., 2016). However, stimulation of T cells was shown to induce early metabolic reprogramming and metabolic switch to the use of glycolytic metabolism, which has been hypothesised to be more beneficial to the cell upon activation, due to the generation of metabolic intermediates during glycolysis (O'Neill et al., 2016). These intermediates can be used to support other metabolic pathways in order to synthesise amino acids, fatty acids or other molecules required to facilitate the activated state of the cell (O'Neill et al., 2016). Inhibition of glycolysis or glycolysis related pathways or restriction of glucose was shown to significantly reduce IFNy production of stimulated T cells as well impair their differentiation (Chang et al., 2013; Hayashi et al., 2013; Renner et al., 2015; Sinclair et al., 2013). Here we have hypothesised that MAIT cells from people with obesity have impaired metabolism, which leads to altered MAIT cell function. Due to the lack of data on MAIT cell metabolism, we first aimed to characterise the metabolic requirements for MAIT activation in control individuals.

For the first time, we have shown that MAIT cells express a range of glucose and amino acid transporters, which synthesis level is modulated upon cell stimulation. This suggested that MAIT cells upregulate the expression of nutrient transporters to enhance the transport of glucose and amino acids to facilitate their activated state. Indeed, previous experiments performed by our lab have shown that the stimulation of MAIT increases the influx of nutrients, as measured by 2NBDG uptake (O'Brien, Loftus & Pisarska *et al.*, 2019; Figure 1 C&D). Potential

enhanced transport of glucose further suggested its role in supporting MAIT cell activation. Indeed, measurement of the rates of glycolysis, by assessing ECAR rates revealed significant upregulation in the glycolytic metabolism upon MAIT cell stimulation. This was further supported by the measurement of the expression of glycolytic enzymes, which were also enhanced upon activation. To test the hypothesis that metabolism of MAIT cells supports their function, we investigated the impact of disrupting glycolysis in MAIT cells during their activation. Restriction of glucose in the culture media or inhibition of glycolysis with 2DG significantly impaired MAIT cell proliferation and expansion. In addition, incubation of cells with 2DG significantly inhibited IFNy production. Collectively, this strongly indicates that MAIT cells heavily rely on glycolytic metabolism upon their activation to exert their effector function. These results are in line with the research performed on T cells, which illustrated that upon stimulation, T cells undergo metabolic reprogramming, switching from OxPhos to glycolytic metabolism to support their function (O'Neill et al., 2016). Stimulation of T cells was shown to upregulate their glycolysis rates, whereas the limitation of glucose in the culture media or direct disruption of the glycolysis led to significant impairment in T cell cytokine production and proliferation (Chang et al., 2013; Gubser et al., 2013; Renner et al., 2015). It still remains to be elucidated why glycolysis is strongly favoured over OxPhos during the activation of lymphocytes however some hypotheses have been put forward. Although glycolysis is much less energy efficient than OxPhos, it provides a wide range of intermediates which can be used to de novo synthesis of amino acids, nucleic acids and fatty acids (O'Neill et al., 2016). These molecules in turn are thought to play a role in the production effector molecules such as cytokines or in providing building blocks required for cell proliferation and division. On the other hand, components of glycolysis have been identified to have a very direct role in supporting IFNy synthesis. For example, in resting T cells, GAPDH enzyme was shown to bind to IFNy mRNA, preventing it from transcription. Activation of T cells was shown to recruit GAPDH to facilitate enhanced glycolysis, freeing IFNy mRNA for translation (Chang et al., 2013). More research is required to establish the direct relationship between the glycolysis and immune cell activation and function.

To explore MAIT cell metabolism further, we decided to investigate the role of amino acids in supporting MAIT cell function, as expression of LAT1 and ASCT2 transporters appeared to be largely enhanced early into their activation. A study of T cells by Sinclair et al. highlighted the importance of leucine transport in their glycolytic metabolism, therefore, we next focused on the LAT1 mediated transport of amino acids. Here, for the first time we have shown that LAT1 mediated transport of amino acids during MAIT cell activation is required for the upregulation of their glycolytic rate. Previous studies in T cells have established that leucine is required for the activation of mTORC1, which in turn is required for the upregulation of glycolysis (Finlay et al., 2012; Saxton et al., 2016). Therefore, we sought to investigate the impact of limiting amino acid influx on mTORC1 activation. Indeed, blocking of LAT1 diminished the activation of mTORC1. Collectively, this indicates that the stimulation of MAIT cells leads to the upregulation of LAT1, which facilitates the enhanced transport of amino acids, including leucine into the cell. Leucine supports the activation of mTORC1, which in turn enhances the rate of glycolysis. Parallel work from our lab has shown that mTORC1 activation is required for the upregulation of glycolysis upon MAIT cell stimulation, as inhibition of mTORC1 activity with rapamycin resulted in a failure of MAIT cells to modulate glycolysis rate upon stimulation (O'Brien, Loftus, Pisarska et al, 2019; Figure 3E&F). Inhibition of mTORC1 activity or amino acid influx had a strong negative impact on MAIT cell proliferation, expansion and IFNy production, which mimics the effect obtained with the direct inhibition of glycolysis. This further indicates that both of these factors control the glycolytic metabolism in MAIT cells.

Although studies of T cell activation strongly focused on glycolytic metabolism, OxPhos has been shown to support T cell proliferation (Chang et al., 2013). To assess whether it also plays a role in the proliferation of MAIT cells, we used inhibitors of ETC and found that ATP synthase plays a particularly important role for MAIT cell proliferation, suggesting ATP derived during OxPhos is involved in supporting MAIT cell proliferation. On the other hand, inhibition of complex III of ETC, allowed for MAIT cell proliferation albeit at a slower rate. Similarly, preventing metabolism of glutamine, which plays a role in replenishing intermediates of Kreb's cycle, and hence supports OxPhos, led to impairment of MAIT cell proliferation, expansion and IFNγ production, but to a much lower extent than the inhibition of glycolytic metabolism. Collectively, this data indicates that OxPhos supports MAIT cell expansion and proliferation, however, plays a less significant role than glycolysis in supporting MAIT cell function upon activation.

In this thesis, we have shown that glycolytic metabolism is central to MAIT cell activation and is absolutely required for successful MAIT cell proliferation, expansion and IFNy production. As MAIT cells from people with obesity presented with severe impairment of both of these functions, we decided to compare their cell metabolism to the metabolism of the control individuals to identify any potential flaws, which may lead to the impairment in their function. Indeed, we found that MAIT cells from people with obesity fail to upregulate glycolysis upon stimulation. In addition, parallel work from our lab revealed that the activation of MAIT cells from people with obesity increases mTORC1 activation, but to a lower extent than in the control individuals (O'Brien, Loftus, Pisarska et al, 2019; Figure 3J). This suggests that inability to control the activity of mTORC1 leads to failure to enhance the rate of glycolysis. To elucidate this further, we measured the expression of LAT1, and the transport capabilities of this amino acid transporter required for the influx of leucine to activate mTORC1. We found that the expression of LAT1 by MAIT cells is lower in people with obesity, which likely leads to the observed lower transport capabilities. Collectively, data derived from the control individuals and people with obesity indicates that a defect in glycolytic metabolism displayed by MAIT cells from people with obesity underpins their impaired function. In this study we show that MAIT cells from people with obesity have impaired expression of LAT1 transporter, which likely leads to lower influx of amino acids including leucine. Lower availability of leucine probably leads to impaired mTORC1 activation, which further prevents glycolysis from being upregulated. Low rate of glycolysis fails to support stimulated MAIT cells, limiting the production of biosynthetic precursors which are likely to be required to enhance cytokine production and proliferation.

Here, we propose a mechanism responsible for impaired MAIT cell function in people in obesity, which may be translated into multiple disease settings, where MAIT cells present with a defect in activation, cytokine production or proliferation. Identification of the mechanism(s) controlling some of the effector functions exerted by MAIT cells will allow for the identification of targets, which could be

used to ameliorate immunometabolism in MAIT cells and hence restore their function. This data provokes multiple questions regarding MAIT immunometabolism and their function in diseases. Future research should focus on elucidating what induces the alterations in MAIT cell glycolytic metabolism in people with obesity. Here we hypothesise that inadequate influx of leucine due to the reduction in LAT1 expression, leads to a cascade of events which eventually results in impaired MAIT cell function. To further confirm this, measurement of leucine uptake by MAIT cells from individuals with and without obesity should be performed. In addition, further research should also investigate why the expression of LAT1 is impaired in MAIT cells with people with obesity. This defect was shown to be present at the transcriptional level, therefore investigations should focus on identifying which signalling pathway is responsible for inducing LAT1 transcription and whether it is also impaired in obesity. Sinclair et al. has also shown that direct TCR activation of T cells increases expression of LAT1, but the mechanism behind this upregulation remains to be elucidated (Sinclair et al., 2013).

Independently of this research, there could be also other causes for impaired glycolytic metabolism in MAIT cells, which should be considered. In obesity, the microenvironment of the cells is often dramatically altered (Fuster et al., 2016; Quail and Dannenberg, 2019). As obesity is closely related to the development of T2DM, in multiple cases, immune cells may be exposed to hyperglycaemia if the individual is unaware, they have diabetes, or their diabetes is badly controlled. Study by Hu et al. showed that the incubation of peripheral blood mononuclear cells in media with high glucose concentrations impairs their cytokine production, suggesting excessive nutrient concentration may impact immune cell function (Hu et al., 2018). It has been shown that individuals with obesity display elevated FFA concentration (Arner and Rydén, 2015; Madak-Erdogan et al., 2019; Ni et al., 2015). An excellent paper by Michelet et al., revealed that obesity induced intracellular lipid accumulation in NK cells. Increased uptake of lipids led to serious impairment of their immunometabolism characterised by lower activation of mTORC1, defect in enhancement of glycolysis upon cell activation and production of cytotoxic granules and IFNy production. This resembled a very similar phenotype to the one of MAIT cells described in this study. Therefore, the

impact of other nutrients on MAIT cell metabolism and function should also be investigated in the future.

As obesity is associated with an elevation in multiple pro-inflammatory factors, long-term exposure to these may impact MAIT cell function and metabolism. Leptin concentration in the peripheral blood has been shown to be elevated in people in obesity (Adeyemi and Abdulle, 2000; Considine et al., 1996; Farr et al., 2015). The consequences of high leptin are yet to be fully understood; however, leptin was shown to have pro-inflammatory properties on the cells of the immune system and modulate cell function (Wrann et al., 2012). A study by Wrann *et al.* has shown that short term incubation of NK in the presence of leptin increased their IFNγ secretion and cytotoxic capabilities, whereas long-term exposure of NK cells to leptin impaired their function (Wrann et al., 2012). This suggests that long term exposure to leptin could be altering immune cell metabolism, by inducing a defect in glycolysis and hence lead to lower IFNγ production.

Although our study provides a very valuable insight into MAIT cell function and metabolism it has limitations, which must be taken into consideration when interpreting these results. This research was performed using solely only human samples. Although primary cells are a wonderful resource, as they allow us to investigate the biology of MAIT cells directly from the cohort of interest, they introduce a large amount of variation, which can be observed in some of the data presented in this thesis. Although MAIT cells are readily detected in the peripheral blood, they are very enriched in tissues, hence the profile of these cells may alter based on their location in the body. This study uses only peripheral blood samples; therefore, these results can only be interpreted in the context of blood. Although our obesity cohort is very well characterized clinically, our control population is less well defined. Although the fitness of each blood donor is assessed before blood donation, it cannot be assumed that these donors don't have any underlying conditions or excessive weight, which could skew the data. Another point of note is the degree of obesity within our cohort, the mean BMI is 47 which is firmly in the morbid/severely obese category. Therefore, our conclusions around the impact of obesity may not reflect BMIs in the 30-40 range.

All of the experiments were performed in vitro, which poses another limitation to this study. All in vitro studies aim to resemble the in vivo environment as much as possible, for example, the incubation of cells at 37°C or by providing them with an abundance of nutrients normally present in their environment. After conducting this study, I believe we should reconsider the culture media we use, especially for immunometabolism studies. Standard culture media contains 10mM (10mmol/L) glucose. This concentration of glucose in human blood would be considered as a hyperglycaemia, as normal fasting blood glucose is considered to be below 5.6 mmol/L (American Diabetes Association, 2020; WHO, 2020b). This means that all *in vitro* assays which investigate immune cell activation or in fact immunometabolism are performed in hyperglycaemic conditions, which we consider to be harmful in vivo. Immune restriction assays performed in this thesis, which used 10mM, 5mM and 1mM glucose concentrations, have shown that MAIT cell proliferation was not impacted at 5mM, which would resemble normal blood glucose. This suggests that 5mM concentration of glucose could be sufficient to support the activation of immune cells in vitro, providing the cells with conditions, which would be closer to their normal environment. It is important to consider than in vivo, under normal conditions, blood glucose is tightly controlled by the action of pancreas and insulin, keeping it at a constant concentration of around 5mM to maintain homeostasis. Setting up of the experiments in the media containing 5mM may lead to the deprivation of glucose during the experiment depending on the activity of the cells and glucose consumption rates and hence "hypoglycaemic" environment. More basic research should be performed to assess the most optimal conditions of culture, to obtain the most accurate results possible.

This study used primary human MAIT cells for all investigations. Murine models, especially genetically modified mice would help to conclusively validate our results. Unfortunately, mice display lower frequencies of MAIT cells than humans, suggesting they may not play as an essential role in their immunity as they do in humans (Chen et al., 2017; Kawachi et al., 2006; Rahimpour et al., 2015). Therefore, data obtained from mouse studies may not fully reflect the biology of MAIT cells in humans. In addition, mice display a homogenous population of subjects, which is a major difference between studies conducted on human samples, as each human is genetically different. This genetic variation is an

important factor, especially during development of drugs or treatments as well as the heterogenicity of the human population, may largely affect their responses. What's more, mouse studies are conducted in largely controlled environments, whereas humans cannot be controlled in such a way. The differences in the lifestyles of individuals, albeit introduces variations into the data, better reflects the studied cohorts. Mouse studies provide a very valuable tool when investigating biology of MAIT cells in the whole organism, however I believe studying MAIT cell metabolism and function in the primary cells provides an advantage over mouse models in this setting.

This study has advanced the knowledge of MAIT cell biology and function. Here we show that MAIT cell metabolism is an integral part of their activation. Stimulation of MAIT cells enhances their glycolytic metabolism, which is sensitive to amino acid influx via LAT1 amino acid transporters. Interruption of this metabolic pathway at any point, results in diminished IFNγ production as well as impaired expansion and proliferation of these cells. In this study we show that MAIT cells from people with obesity present with impaired glycolytic metabolism and effector functions, therefore we propose that a defect in glycolysis is responsible for the altered function of MAIT cells from obese individuals. This study describes a novel metabolic axis which controls MAIT cell function, which could be targeted in the future to restore MAIT cell function in people in obesity and other diseases which present with a similar MAIT cell phenotype.

Chapter Seven

Bibliography

- Abrahamsson, S. V., Angelini, D.F., Dubinsky, A.N., Morel, E., Oh, U., Jones, J.L., Carassiti, D., Reynolds, R., Salvetti, M., Calabresi, P.A., Coles, A.J., Battistini, L., Martin, R., Burt, R.K., Muraro, P.A., 2013. Non-myeloablative autologous haematopoietic stem cell transplantation expands regulatory cells and depletes IL-17 producing mucosal-associated invariant T cells in multiple sclerosis. Brain 136, 2888–2903. https://doi.org/10.1093/brain/awt182
- Acedo, S.C., Gambero, S., Cunha, F.G.P., Lorand-Metze, I., Gambero, A., 2013. Participation of leptin in the determination of the macrophage phenotype: An additional role in adipocyte and macrophage crosstalk. Vitr. Cell. Dev. Biol. - Anim. 49, 473–478. https://doi.org/10.1007/s11626-013-9629-x
- Adeyemi, E., Abdulle, A., 2000. A comparison of plasma leptin levels in obese and lean individuals in the United Arab Emirates. Nutr. Res. 20, 157–166. https://doi.org/10.1016/S0271-5317(99)00149-9
- Aguirre, V., Uchida, T., Yenush, L., Davis, R., White, M.F., 2000. The c-Jun NH2-terminal kinase promotes insulin resistance during association with insulin receptor substrate-1 and phosphorylation of Ser307. J. Biol. Chem. 275, 9047–9054. https://doi.org/10.1074/jbc.275.12.9047
- Alkanani, A.K., Hara, N., Gottlieb, P.A., Ir, D., Robertson, C.E., Wagner, B.D., Frank, D.N., Zipris, D., 2015. Alterations in intestinal microbiota correlate with susceptibility to type 1 diabetes. Diabetes 64, 3510–3520. https://doi.org/10.2337/db14-1847
- Almeida, L., Lochner, M., Berod, L., Sparwasser, T., 2016. Metabolic pathways in T cell activation and lineage differentiation. Semin. Immunol. https://doi.org/10.1016/j.smim.2016.10.009
- Altman, B.J., Stine, Z.E., Dang, C. V., 2016. From Krebs to clinic: Glutamine metabolism to cancer therapy. Nat. Rev. Cancer. https://doi.org/10.1038/nrc.2016.71
- American Diabetes Association, 2020. Diabtetes Overview. Diagnosis [WWW Document]. URL https://www.diabetes.org/a1c/diagnosis (accessed 12.24.20).
- Ammitzbøll, C., von Essen, M.R., Chow, H.H., McWilliam, O., Holm Hansen, R.,
 Sellebjerg, F., 2020. MAIT cell subtypes in multiple sclerosis. J.
 Neuroimmunol. 339. https://doi.org/10.1016/j.jneuroim.2019.577117

- Annibali, V., Ristori, G., Angelini, D.F., Serafini, B., Mechelli, R., Cannoni, S., Romano, S., Paolillo, A., Abderrahim, H., Diamantini, A., Borsellino, G., Aloisi, F., Battistini, L., Salvetti, M., 2011. CD161(high)CD8+T cells bear pathogenetic potential in multiple sclerosis. Brain 134, 542–54. https://doi.org/10.1093/brain/awq354
- Arner, P., Rydén, M., 2015. Fatty Acids, Obesity and Insulin Resistance. Obes. Facts 8, 147–155. https://doi.org/10.1159/000381224
- Avgerinos, K.I., Spyrou, N., Mantzoros, C.S., Dalamaga, M., 2019. Obesity and cancer risk: Emerging biological mechanisms and perspectives. Metabolism 92, 121–135. https://doi.org/10.1016/j.metabol.2018.11.001
- Babu, E., Kanai, Y., Chairoungdua, A., Kim, D.K., Iribe, Y., Tangtrongsup, S.,
 Jutabha, P., Li, Y., Ahmed, N., Sakamoto, S., Anzai, N., Nagamori, S.,
 Endou, H., 2003. Identification of a Novel System L Amino Acid Transporter
 Structurally Distinct from Heterodimeric Amino Acid Transporters. J. Biol.
 Chem. 278, 43838–43845. https://doi.org/10.1074/jbc.M305221200
- Bailis, W., Shyer, J.A., Zhao, J., Canaveras, J.C.G., Al Khazal, F.J., Qu, R.,
 Steach, H.R., Bielecki, P., Khan, O., Jackson, R., Kluger, Y., Maher, L.J.,
 Rabinowitz, J., Craft, J., Flavell, R.A., 2019. Distinct modes of
 mitochondrial metabolism uncouple T cell differentiation and function.
 Nature 571, 403–407. https://doi.org/10.1038/s41586-019-1311-3
- Barathan, M., Mohamed, R., Vadivelu, J., Chang, L.Y., Saeidi, A., Yong, Y.K., Ravishankar Ram, M., Gopal, K., Velu, V., Larsson, M., Shankar, E.M., 2016. Peripheral loss of CD8+CD161++TCRVα7·2+ mucosal-associated invariant T cells in chronic hepatitis C virus-infected patients. Eur. J. Clin. Invest. 46, 170–180. https://doi.org/10.1111/eci.12581
- Beckman Coulter, 2020. Innate versus Adaptive Immune Response [WWW Document]. URL https://www.beckman.com/reagents/coulter-flowcytometry/cell-health-research-assays/research-assays/myeloid-activationantibody-cocktail/innate-versus-adaptive (accessed 1.9.21).
- Bel-Serrat, S., Heinen, M., Murrin, C., Leslie, D., Mehegan, J., Concannon, M.,
 Flood, C., Farrell, D., O'Briend Sarah, Eldin, N., Cecily, K., 2017. The
 Childhood Obesity Surveillance Initiative (COSI) in the Republic of Ireland:
 Findings from 2008, 2010, 2012 and 2015. Health Service Executive,
 Dublin.

Bello, N.A., Cheng, S., Claggett, B., Shah, A.M., Ndumele, C.E., Roca, G.Q.,

Santos, A.B.S., Gupta, D., Vardeny, O., Aguilar, D., Folsom, A.R., Butler, K.R., Kitzman, D.W., Coresh, J., Solomon, S.D., 2016. Association of weight and body composition on cardiac structure and function in the ARIC study (Atherosclerosis Risk in Communities). Circ. Hear. Fail. 9. https://doi.org/10.1161/CIRCHEARTFAILURE.115.002978

- Bernstein, L.E., Berry, J., Kim, S., Canavan, B., Grinspoon, S.K., 2006. Effects of etanercept in patients with the metabolic syndrome. Arch. Intern. Med. 166, 902–908. https://doi.org/10.1001/archinte.166.8.902
- Bhaskaran, K., Douglas, I., Forbes, H., Dos-Santos-Silva, I., Leon, D.A.,
 Smeeth, L., 2014. Body-mass index and risk of 22 specific cancers: A population-based cohort study of 5·24 million UK adults. Lancet 384, 755–765. https://doi.org/10.1016/S0140-6736(14)60892-8
- Bissonnette, R., Luger, T., Thaçi, D., Toth, D., Lacombe, A., Xia, S., Mazur, R., Patekar, M., Charef, P., Milutinovic, M., Leonardi, C., Mrowietz, U., 2018.
 Secukinumab demonstrates high sustained efficacy and a favourable safety profile in patients with moderate-to-severe psoriasis through 5 years of treatment (SCULPTURE Extension Study). J. Eur. Acad. Dermatology Venereol. 32, 1507–1514. https://doi.org/10.1111/jdv.14878
- Bodoy, S., Martín, L., Zorzano, A., Palacín, M., Estévez, R., Bertran, J., 2005.
 Identification of LAT4, a novel amino acid transporter with system L activity.
 J. Biol. Chem. 280, 12002–12011. https://doi.org/10.1074/jbc.M408638200
- Bonam, S.R., Partidos, C.D., Halmuthur, S.K.M., Muller, S., 2017. An Overview of Novel Adjuvants Designed for Improving Vaccine Efficacy. Trends Pharmacol. Sci. https://doi.org/10.1016/j.tips.2017.06.002
- Boulangé, C.L., Neves, A.L., Chilloux, J., Nicholson, J.K., Dumas, M.E., 2016.
 Impact of the gut microbiota on inflammation, obesity, and metabolic disease. Genome Med. https://doi.org/10.1186/s13073-016-0303-2
- Bray, G.A., Heisel, W.E., Afshin, A., Jensen, M.D., Dietz, W.H., Long, M.,
 Kushner, R.F., Daniels, S.R., Wadden, T.A., Tsai, A.G., Hu, F.B., Jakicic,
 J.M., Ryan, D.H., Wolfe, B.M., Inge, T.H., 2018. The science of obesity
 management: An endocrine society scientific statement. Endocr. Rev. 39,
 79–132. https://doi.org/10.1210/er.2017-00253
- Bray, G.A., Kim, K.K., Wilding, J.P.H., 2017. Obesity: a chronic relapsing progressive disease process. A position statement of the World Obesity Federation. Obes. Rev. 18, 715–723. https://doi.org/10.1111/obr.12551

- Brestoff, J.R., Kim, B.S., Saenz, S.A., Stine, R.R., Monticelli, L.A., Sonnenberg, G.F., Thome, J.J., Farber, D.L., Lutfy, K., Seale, P., Artis, D., 2015. Group 2 innate lymphoid cells promote beiging of white adipose tissue and limit obesity. Nature 519, 242–246. https://doi.org/10.1038/nature14115
- Canter, R.J., Aguilar, E., Wang, Z., Le, C., Khuat, L., Dunai, C., Rebhun, R., Tarantal, A., Blazar, B.R., Monjazeb, A., Murphy, W.J., 2018. Obesity results in higher PD-1-mediated T-cell suppression but greater T-cell effector functions following blockade. J. Clin. Oncol. 36, 65–65. https://doi.org/10.1200/jco.2018.36.5 suppl.65
- Cao, Y.L., Hu, C.Z., Meng, X., Wang, D.F., Zhang, J., 2008. Expression of TNFα protein in omental and subcutaneous adipose tissue in obesity. Diabetes Res. Clin. Pract. 79, 214–219. https://doi.org/10.1016/j.diabres.2007.08.030
- Carolan, E., Tobin, L.M., Mangan, B.A., Corrigan, M., Gaoatswe, G., Byrne, G., Geoghegan, J., Cody, D., O'Connell, J., Winter, D.C., Doherty, D.G., Lynch, L., O'Shea, D., Hogan, A.E., 2015. Altered Distribution and Increased IL-17 Production by Mucosal-Associated Invariant T Cells in Adult and Childhood Obesity. J. Immunol. 194, 5775–5780. https://doi.org/10.4049/jimmunol.1402945
- Carr, E.L., Kelman, A., Wu, G.S., Gopaul, R., Senkevitch, E., Aghvanyan, A., Turay, A.M., Frauwirth, K.A., 2010. Glutamine Uptake and Metabolism Are Coordinately Regulated by ERK/MAPK during T Lymphocyte Activation. J. Immunol. 185, 1037–1044. https://doi.org/10.4049/jimmunol.0903586
- Catrysse, L., van Loo, G., 2018. Adipose tissue macrophages and their polarization in health and obesity. Cell. Immunol. 330, 114–119. https://doi.org/10.1016/j.cellimm.2018.03.001
- CDC, 2020. Defining Adult Overweight and Obesity [WWW Document]. URL https://www.cdc.gov/obesity/adult/defining.html (accessed 12.29.20).
- Cerhan, J.R., Moore, S.C., Jacobs, E.J., Kitahara, C.M., Rosenberg, P.S.,
 Adami, H.O., Ebbert, J.O., English, D.R., Gapstur, S.M., Giles, G.G., Horn-Ross, P.L., Park, Y., Patel, A. V., Robien, K., Weiderpass, E., Willett, W.C.,
 Wolk, A., Zeleniuch-Jacquotte, A., Hartge, P., Bernstein, L., De Gonzalez,
 A.B., 2014. A pooled analysis of waist circumference and mortality in
 650,000 adults. Mayo Clin. Proc. 89, 335–345.
 https://doi.org/10.1016/j.mayocp.2013.11.011

- Cham, C.M., Driessens, G., O'Keefe, J.P., Gajewski, T.F., 2008. Glucose deprivation inhibits multiple key gene expression events and effector functions in CD8+ T cells. Eur. J. Immunol. 38, 2438–2450. https://doi.org/10.1002/eji.200838289
- Chang, C.H., Curtis, J.D., Maggi, L.B., Faubert, B., Villarino, A. V., O'Sullivan,
 D., Huang, S.C.C., Van Der Windt, G.J.W., Blagih, J., Qiu, J., Weber, J.D.,
 Pearce, E.J., Jones, R.G., Pearce, E.L., 2013. Posttranscriptional control of
 T cell effector function by aerobic glycolysis. Cell 153, 1239.
 https://doi.org/10.1016/j.cell.2013.05.016
- Chantranupong, L., Wolfson, R.L., Orozco, J.M., Saxton, R.A., Scaria, S.M., Bar-Peled, L., Spooner, E., Isasa, M., Gygi, S.P., Sabatini, D.M., 2014. The sestrins interact with gator2 to negatively regulate the amino-acid-sensing pathway upstream of mTORC1. Cell Rep. 9, 1–8. https://doi.org/10.1016/j.celrep.2014.09.014
- Chaoul, N., Fayolle, C., Desrues, B., Oberkampf, M., Tang, A., Ladant, D., Leclerc, C., 2015. Rapamycin impairs antitumor CD8p T-cell responses and vaccine-induced tumor eradication. Cancer Res. 75, 3279–3291. https://doi.org/10.1158/0008-5472.CAN-15-0454
- Chehimi, M., Vidal, H., Eljaafari, A., 2017. Pathogenic Role of IL-17-Producing Immune Cells in Obesity, and Related Inflammatory Diseases. J. Clin. Med.
 6, 68. https://doi.org/10.3390/jcm6070068
- Chen, L., Deng, H., Cui, H., Fang, J., Zuo, Z., Deng, J., Li, Y., Wang, X., Zhao, L., 2018. Inflammatory responses and inflammation-associated diseases in organs. Oncotarget. https://doi.org/10.18632/oncotarget.23208
- Chen, P., Deng, W., Li, D., Zeng, T., Huang, L., Wang, Q., Wang, Jinli, Zhang, W., Yu, X., Duan, D., Wang, Jinle, Xia, H., Chen, H., Huang, W., Li, J., Zhang, D., Zhong, X.-P., Gao, J., 2019. Circulating Mucosal-Associated Invariant T Cells in a Large Cohort of Healthy Chinese Individuals From Newborn to Elderly. Front. Immunol. 10, 260. https://doi.org/10.3389/fimmu.2019.00260
- Chen, Z., Wang, H., D'Souza, C., Sun, S., Kostenko, L., Eckle, S.B.G., Meehan,
 B.S., Jackson, D.C., Strugnell, R.A., Cao, H., Wang, N., Fairlie, D.P., Liu,
 L., Godfrey, D.I., Rossjohn, J., McCluskey, J., Corbett, A.J., 2017. Mucosalassociated invariant T-cell activation and accumulation after in vivo infection depends on microbial riboflavin synthesis and co-stimulatory

signals. Mucosal Immunol. 10, 58–68. https://doi.org/10.1038/mi.2016.39

- Chiba, A., Tajima, R., Tomi, C., Miyazaki, Y., Yamamura, T., Miyake, S., 2012. Mucosal-associated invariant T cells promote inflammation and exacerbate disease in murine models of arthritis. Arthritis Rheum. 64, 153–161. https://doi.org/10.1002/art.33314
- Chiba, A., Tamura, N., Yoshikiyo, K., Murayama, G., Kitagaichi, M., Yamaji, K., Takasaki, Y., Miyake, S., 2017. Activation status of mucosal-associated invariant T cells reflects disease activity and pathology of systemic lupus erythematosus. Arthritis Res. Ther. 19, 58. https://doi.org/10.1186/s13075-017-1257-5
- Cho, C.-H., Koh, Y.J., Han, J., Sung, H.-K., Lee, H.J., Morisada, T.,
 Schwendener, R.A., Brekken, R.A., Kang, G., Oike, Y., Choi, T.-S., Suda,
 T., Yoo, O.-J., Koh, Y., 2007. Angiogenic Role of LYVE-1-Positive
 Macrophages in Adipose Tissue.
 https://doi.org/10.1161/01.RES.0000259564.92792.93
- Cho, Y.-N., Kee, S.-J., Kim, T.-J., Jin, H.M., Kim, M.-J., Jung, H.-J., Park, K.-J., Lee, S.-J., Lee, S.-S., Kwon, Y.-S., Kee, H.J., Kim, N., Park, Y.-W., 2014.
 Mucosal-Associated Invariant T Cell Deficiency in Systemic Lupus Erythematosus. J. Immunol. 193, 3891–3901.
 https://doi.org/10.4049/jimmunol.1302701
- Chua, W.J., Truscott, S.M., Eickhoff, C.S., Blazevic, A., Hoft, D.F., Hansen, T.H., 2012. Polyclonal mucosa-associated invariant T cells have unique innate functions in bacterial infection. Infect. Immun. 80, 3256–3267. https://doi.org/10.1128/IAI.00279-12
- Coico, R., Sunshine, G., 2015. Immunology: A short course, 7th ed. John Wiley & Sons, Incorporated, New York.
- Collaboration(NCD-RisC), N.R.F., 2016. Trends in adult body-mass index in 200 countries from 1975 to 2014: A pooled analysis of 1698 populationbased measurement studies with 19.2 million participants. Lancet 387, 1377–1396. https://doi.org/10.1016/S0140-6736(16)30054-X
- Considine, R. V., Sinha, M.K., Heiman, M.L., Kriauciunas, A., Stephens, T.W., Nyce, M.R., Ohannesian, J.P., Marco, C.C., Mckee, L.J., Bauer, T.L., Caro, J.F., 1996. Serum immunoreactive-leptin concentrations in normal-weight and obese humans. N. Engl. J. Med. 334, 292–295. https://doi.org/10.1056/NEJM199602013340503

- Corbett, A.J., Eckle, S.B.G., Birkinshaw, R.W., Liu, L., Patel, O., Mahony, J., Chen, Z., Reantragoon, R., Meehan, B., Cao, H., Williamson, N.A., Strugnell, R.A., Van Sinderen, D., Mak, J.Y.W., Fairlie, D.P., Kjer-Nielsen, L., Rossjohn, J., McCluskey, J., 2014. T-cell activation by transitory neoantigens derived from distinct microbial pathways. Nature 509, 361–365. https://doi.org/10.1038/nature13160
- Cosgrove, C., Ussher, J.E., Rauch, A., Gärtner, K., Kurioka, A., Hühn, M.H.,
 Adelmann, K., Kang, Y.H., Fergusson, J.R., Simmonds, P., Goulder, P.,
 Hansen, T.H., Fox, J., Günthard, H.F., Khanna, N., Powrie, F., Steel, A.,
 Gazzard, B., Phillips, R.E., Frater, J., Uhlig, H., Klenerman, P., 2013. Early
 and nonreversible decrease of CD161++/MAIT cells in HIV infection. Blood
 121, 951–961. https://doi.org/10.1182/blood-2012-06-436436
- Cretenet, G., Clerc, I., Matias, M., Loisel, S., Craveiro, M., Oburoglu, L., Kinet, S., Mongellaz, C., Dardalhon, V., Taylor, N., 2016. Cell surface Glut1 levels distinguish human CD4 and CD8 T lymphocyte subsets with distinct effector functions. Sci. Rep. 6, 1–13. https://doi.org/10.1038/srep24129
- D'Souza, C., Pediongco, T., Wang, H., Scheerlinck, J.-P.Y., Kostenko, L.,
 Esterbauer, R., Stent, A.W., Eckle, S.B.G., Meehan, B.S., Strugnell, R.A.,
 Cao, H., Liu, L., Mak, J.Y.W., Lovrecz, G., Lu, L., Fairlie, D.P., Rossjohn,
 J., McCluskey, J., Every, A.L., Chen, Z., Corbett, A.J., 2018. MucosalAssociated Invariant T Cells Augment Immunopathology and Gastritis in
 Chronic Helicobacter pylori Infection . J. Immunol. 200, ji1701512.
 https://doi.org/10.4049/jimmunol.1701512
- Daousi, C., Casson, I.F., Gill, G. V., MacFarlane, I.A., Wilding, J.P.H., Pinkney, J.H., 2006. Prevalence of obesity in type 2 diabetes in secondary care: Association with cardiovascular risk factors. Postgrad. Med. J. 82, 280–284. https://doi.org/10.1136/pmj.2005.039032
- De La Serre, C.B., Ellis, C.L., Lee, J., Hartman, A.L., Rutledge, J.C., Raybould, H.E., 2010. Propensity to high-fat diet-induced obesity in rats is associated with changes in the gut microbiota and gut inflammation. Am. J. Physiol. -Gastrointest. Liver Physiol. 299. https://doi.org/10.1152/ajpgi.00098.2010
- DeBerardinis, R.J., Chandel, N.S., 2020. We need to talk about the Warburg effect. Nat. Metab. https://doi.org/10.1038/s42255-020-0172-2
- DeFronzo, R.A., Ferrannini, E., Groop, L., Henry, R.R., Herman, W.H., Holst, J.J., Hu, F.B., Kahn, C.R., Raz, I., Shulman, G.I., Simonson, D.C., Testa,

M.A., Weiss, R., 2015. Type 2 diabetes mellitus. Nat. Rev. Dis. Prim. 1. https://doi.org/10.1038/nrdp.2015.19

- Delgoffe, G.M., Kole, T.P., Zheng, Y., Zarek, P.E., Matthews, K.L., Xiao, B., Worley, P.F., Kozma, S.C., Powell, J.D., 2009. The mTOR Kinase Differentially Regulates Effector and Regulatory T Cell Lineage Commitment. Immunity 30, 832–844. https://doi.org/10.1016/j.immuni.2009.04.014
- Delgoffe, G.M., Pollizzi, K.N., Waickman, A.T., Heikamp, E., Meyers, D.J., Horton, M.R., Xiao, B., Worley, P.F., Powell, J.D., 2011. The kinase mTOR regulates the differentiation of helper T cells through the selective activation of signaling by mTORC1 and mTORC2. Nat. Immunol. 12, 295–304. https://doi.org/10.1038/ni.2005
- Delves, P., Martin, S., Burton, D., Roitt, I., 2011. Roitt's Essential Immunology, 12th ed. John Wiley & Sons, Incorporated, Hoboken.
- Department of Health, MRBI IPSOS, 2015. Healthy Ireland Survey 2015. Summary of Findings. Dublin.
- Dias, J., Sobkowiak, M.J., Sandberg, J.K., Leeansyah, E., 2016. Human MAITcell responses to Escherichia coli : activation, cytokine production, proliferation, and cytotoxicity . J. Leukoc. Biol. 100, 233–240. https://doi.org/10.1189/jlb.4ta0815-391rr
- Dimeloe, S., Burgener, A.V., Grählert, J., Hess, C., 2017. T-cell metabolism governing activation, proliferation and differentiation; a modular view. Immunology. https://doi.org/10.1111/imm.12655
- Dimeloe, S., Mehling, M., Frick, C., Loeliger, J., Bantug, G.R., Sauder, U.,
 Fischer, M., Belle, R., Develioglu, L., Tay, S., Langenkamp, A., Hess, C.,
 2016. The Immune-Metabolic Basis of Effector Memory CD4 + T Cell
 Function under Hypoxic Conditions . J. Immunol. 196, 106–114.
 https://doi.org/10.4049/jimmunol.1501766
- Doheny, K., 2015. Obesity, Cancer: Evidence Grows Supporting the Link [WWW Document]. URL

https://www.endocrineweb.com/professional/obesity/obesity-cancerevidence-grows-supporting-link (accessed 1.9.21).

Dominguez, H., Storgaard, H., Rask-Madsen, C., Hermann, T.S., Ihlemann, N., Nielsen, D.B., Spohr, C., Kober, L., Vaag, A., Torp-Pedersen, C., 2005. Metabolic and vascular effects of tumor necrosis factor-α blockade with etanercept in obese patients with type 2 diabetes. J. Vasc. Res. 42, 517– 525. https://doi.org/10.1159/000088261

- Donnelly, R.P., Loftus, R.M., Keating, S.E., Liou, K.T., Biron, C.A., Gardiner, C.M., Finlay, D.K., 2014. mTORC1-Dependent Metabolic Reprogramming Is a Prerequisite for NK Cell Effector Function. J. Immunol. 193, 4477–4484. https://doi.org/10.4049/jimmunol.1401558
- Duan, M., Goswami, S., Shi, J.Y., Wu, L.J., Wang, X.Y., Ma, J.Q., Zhang, Z., Shi, Y., Ma, L.J., Zhang, S., Xi, R. Bin, Cao, Y., Zhou, J., Fan, J., Zhang, X.M., Gao, Q., 2019. Activated and exhausted MAIT cells foster disease progression and indicate poor outcome in hepatocellular carcinoma. Clin. Cancer Res. 25, 3304–3316. https://doi.org/10.1158/1078-0432.CCR-18-3040
- Duncan, M.S., Vasan, R.S., Xanthakis, V., 2019. Trajectories of Blood Lipid Concentrations Over the Adult Life Course and Risk of Cardiovascular Disease and All-Cause Mortality: Observations From the Framingham Study Over 35 Years. J. Am. Heart Assoc. 8, e011433. https://doi.org/10.1161/JAHA.118.011433
- Dusseaux, M., Martin, E., Serriari, N., Péguillet, I., Premel, V., Louis, D., Milder, M., Le Bourhis, L., Soudais, C., Treiner, E., Lantz, O., 2011. Human MAIT cells are xenobiotic-resistant, tissue-targeted, CD161 hi IL-17-secreting T cells. Blood 117, 1250–1259. https://doi.org/10.1182/blood-2010-08-303339
- Eberhard, J.M., Hartjen, P., Kummer, S., Schmidt, R.E., Bockhorn, M.,
 Lehmann, C., Balagopal, A., Hauber, J., Van Lunzen, J., Wiesch, J.S. Zur,
 2014. CD161+ MAIT Cells Are Severely Reduced in Peripheral Blood and
 Lymph Nodes of HIV-Infected Individuals Independently of Disease
 Progression. PLoS One 9. https://doi.org/10.1371/journal.pone.0111323
- Emmer, C., Bosnjak, M., Mata, J., 2020. The association between weight stigma and mental health: A meta-analysis. Obes. Rev. 21. https://doi.org/10.1111/obr.12935
- Evensen, E., Wilsgaard, T., Furberg, A.S., Skeie, G., 2016. Tracking of overweight and obesity from early childhood to adolescence in a population-based cohort - the Tromsø Study, Fit Futures. BMC Pediatr. 16. https://doi.org/10.1186/s12887-016-0599-5

Fabbrini, E., Cella, M., McCartney, S.A., Fuchs, A., Abumrad, N.A., Pietka, T.A.,

Chen, Z., Finck, B.N., Han, D.H., Magkos, F., Conte, C., Bradley, D., Fraterrigo, G., Eagon, J.C., Patterson, B.W., Colonna, M., Klein, S., 2013. Association between specific adipose tissue CD4+ T-cell populations and insulin resistance in obese individuals. Gastroenterology 145, 366. https://doi.org/10.1053/j.gastro.2013.04.010

- Farr, O.M., Gavrieli, A., Mantzoros, C.S., 2015. Leptin applications in 2015: What have we learned about leptin and obesity? Curr. Opin. Endocrinol. Diabetes Obes. https://doi.org/10.1097/MED.000000000000184
- Felsenreich, D.M., Langer, F.B., Kefurt, R., Panhofer, P., Schermann, M., Beckerhinn, P., Sperker, C., Prager, G., 2016. Weight loss, weight regain, and conversions to Roux-en-Y gastric bypass: 10-year results of laparoscopic sleeve gastrectomy. Surg. Obes. Relat. Dis. 12, 1655–1662. https://doi.org/10.1016/j.soard.2016.02.021
- Finlay, D.K., Rosenzweig, E., Sinclair, L. V., Carmen, F.C., Hukelmann, J.L., Rolf, J., Panteleyev, A.A., Okkenhaug, K., Cantrell, D.A., 2012. PDK1 regulation of mTOR and hypoxia-inducible factor 1 integrate metabolism and migration of CD8+ T cells. J. Exp. Med. 209, 2441–2453. https://doi.org/10.1084/jem.20112607
- Franz, M.J., VanWormer, J.J., Crain, A.L., Boucher, J.L., Histon, T., Caplan, W., Bowman, J.D., Pronk, N.P., 2007. Weight-Loss Outcomes: A Systematic Review and Meta-Analysis of Weight-Loss Clinical Trials with a Minimum 1-Year Follow-Up. J. Am. Diet. Assoc. 107, 1755–1767. https://doi.org/10.1016/j.jada.2007.07.017
- Frauwirth, K.A., Riley, J.L., Harris, M.H., Parry, R. V., Rathmell, J.C., Plas, D.R., Elstrom, R.L., June, C.H., Thompson, C.B., 2002. The CD28 signaling pathway regulates glucose metabolism. Immunity 16, 769–777. https://doi.org/10.1016/S1074-7613(02)00323-0
- Fuster, J.J., Ouchi, N., Gokce, N., Walsh, K., 2016. Obesity-induced changes in adipose tissue microenvironment and their impact on cardiovascular disease. Circ. Res. https://doi.org/10.1161/CIRCRESAHA.115.306885
- Gaffen, S.L., Jain, R., Garg, A. V., Cua, D.J., 2014. The IL-23-IL-17 immune axis: From mechanisms to therapeutic testing. Nat. Rev. Immunol. https://doi.org/10.1038/nri3707
- Garyu, J.W., Uduman, M., Stewart, A., Rui, J., Deng, S., Shenson, J., Staron, M.M., Kaech, S.M., Kleinstein, S.H., Herold, K.C., 2016. Characterization of

diabetogenic CD8+ T cells: Immune therapy with metabolic blockade. J. Biol. Chem. 291, 11230–11240. https://doi.org/10.1074/jbc.M115.713362

- Gerriets, V.A., Danzaki, K., Kishton, R.J., Eisner, W., Nichols, A.G., Saucillo, D.C., Shinohara, M.L., MacIver, N.J., 2016. Leptin directly promotes T-cell glycolytic metabolism to drive effector T-cell differentiation in a mouse model of autoimmunity. Eur. J. Immunol. 46, 1970–1983. https://doi.org/10.1002/eji.201545861
- Gherardin, N.A., Liyen, L., Lorenztino, A., Davenport, A.J., Richardson, K.,
 Rogers, A., Darcy, P.K., Jenkins, M.R., Prince, H.M., Harrison, S.J., Quach,
 H., Fairlie, D.P., Kedzierska, K., McCluskey, J., Uldrich, A.P., Neeson, P.J.,
 Ritchie, D.S., Godfrey, D.I., 2018a. Enumeration, functional responses and
 cytotoxic capacity of MAIT cells in newly diagnosed and relapsed multiple
 myeloma. Sci. Rep. 8. https://doi.org/10.1038/s41598-018-22130-1
- Gherardin, N.A., Souter, M.N., Koay, H.-F., Mangas, K.M., Seemann, T.,
 Stinear, T.P., Eckle, S.B., Berzins, S.P., d'Udekem, Y., Konstantinov, I.E.,
 Fairlie, D.P., Ritchie, D.S., Neeson, P.J., Pellicci, D.G., Uldrich, A.P.,
 McCluskey, J., Godfrey, D.I., 2018b. Human blood MAIT cell subsets
 defined using MR1 tetramers. Immunol. Cell Biol. 96, 507–525.
 https://doi.org/10.1111/imcb.12021
- Gibbs, A., Leeansyah, E., Introini, A., Paquin-Proulx, D., Hasselrot, K.,
 Andersson, E., Broliden, K., Sandberg, J.K., Tjernlund, A., 2017. MAIT cells reside in the female genital mucosa and are biased towards IL-17 and IL-22 production in response to bacterial stimulation. Mucosal Immunol. 10, 35–45. https://doi.org/10.1038/mi.2016.30

Gnanaprakasam, J.N.R., Wang, R., 2017. MYC in regulating immunity: Metabolism and beyond. Genes (Basel). https://doi.org/10.3390/genes8030088

- Godfrey, D.I., Rô Me Le Nours, J., Andrews, D.M., Uldrich, A.P., Rossjohn, J.,
 2018. Review Unconventional T Cell Targets for Cancer Immunotherapy.
 Immunity 48, 453–473. https://doi.org/10.1016/j.immuni.2018.03.009
- Godfrey, D.I., Uldrich, A.P., Mccluskey, J., Rossjohn, J., Moody, D.B., 2015. The burgeoning family of unconventional T cells. Nat. Immunol. https://doi.org/10.1038/ni.3298
- Gold, M.C., Cerri, S., Smyk-Pearson, S., Cansler, M.E., Vogt, T.M., Delepine, J., Winata, E., Swarbrick, G.M., Chua, W.-J., Yu, Y.Y.L., Lantz, O., Cook,

M.S., Null, M.D., Jacoby, D.B., Harriff, M.J., Lewinsohn, D.A., Hansen, T.H., Lewinsohn, D.M., 2010. Human Mucosal Associated Invariant T Cells Detect Bacterially Infected Cells. PLoS Biol. 8, e1000407. https://doi.org/10.1371/journal.pbio.1000407

- Gold, M.C., Eid, T., Smyk-Pearson, S., Eberling, Y., Swarbrick, G.M., Langley, S.M., Streeter, P.R., Lewinsohn, D.A., Lewinsohn, D.M., 2013. Human thymic MR1-restricted MAIT cells are innate pathogen-reactive effectors that adapt following thymic egress HHS Public Access. Mucosal Immunol 6, 35–44. https://doi.org/10.1038/mi.2012.45
- González-Muniesa, P., Mártinez-González, M.A., Hu, F.B., Després, J.P.,
 Matsuzawa, Y., Loos, R.J.F., Moreno, L.A., Bray, G.A., Martinez, J.A.,
 2017. Obesity. Nat. Rev. Dis. Prim. 3. https://doi.org/10.1038/nrdp.2017.34
- Goodarzi, M.O., 2018. Genetics of obesity: what genetic association studies have taught us about the biology of obesity and its complications. Lancet Diabetes Endocrinol. https://doi.org/10.1016/S2213-8587(17)30200-0
- Gordon-Larsen, P., Adair, L.S., Nelson, M.C., Popkin, B.M., 2004. Five-year obesity incidence in the transition period between adolescence and adulthood: the National Longitudinal Study of Adolescent Health. Am J Clin Nutr 80, 569–75.
- Gracey, E., Qaiyum, Z., Almaghlouth, I., Lawson, D., Karki, S., Avvaru, N.,
 Zhang, Z., Yao, Y., Ranganathan, V., Baglaenko, Y., Inman, R.D., 2016. IL7 primes IL-17 in mucosal-associated invariant T (MAIT) cells, which
 contribute to the Th17-axis in ankylosing spondylitis. Ann. Rheum. Dis. 75,
 2124–2132. https://doi.org/10.1136/annrheumdis-2015-208902
- Greiner, E., Guppy, M., Brand, K., 1994. Glucose is essential for proliferation and the glycolytic enzyme induction that provokes a transition to glycolytic energy production. J Biol Chem 269, 31484–90.
- Gubser, P.M., Bantug, G.R., Razik, L., Fischer, M., Dimeloe, S., Hoenger, G.,
 Durovic, B., Jauch, A., Hess, C., 2013. Rapid effector function of memory
 CD8+ T cells requires an immediate-early glycolytic switch. Nat. Immunol.
 14, 1064–1072. https://doi.org/10.1038/ni.2687
- Guilherme, A., Virbasius, J. V., Puri, V., Czech, M.P., 2008. Adipocyte dysfunctions linking obesity to insulin resistance and type 2 diabetes. Nat. Rev. Mol. Cell Biol. https://doi.org/10.1038/nrm2391

Guo, Q., Wang, Y., Xu, D., Nossent, J., Pavlos, N.J., Xu, J., 2018. Rheumatoid

arthritis: Pathological mechanisms and modern pharmacologic therapies. Bone Res. https://doi.org/10.1038/s41413-018-0016-9

- Gutierrez-Arcelus, M., Teslovich, N., Mola, A.R., Polidoro, R.B., Nathan, A., Kim, H., Hannes, S., Slowikowski, K., Watts, G.F.M., Korsunsky, I., Brenner, M.B., Raychaudhuri, S., Brennan, P.J., 2019. Lymphocyte innateness defined by transcriptional states reflects a balance between proliferation and effector functions. Nat. Commun. 10, 1–15. https://doi.org/10.1038/s41467-019-08604-4
- Haga, K., Chiba, A., Shibuya, T., Osada, T., Ishikawa, D., Kodani, T., Nomura,
 O., Watanabe, S., Miyake, S., 2016. MAIT cells are activated and
 accumulated in the inflamed mucosa of ulcerative colitis. J. Gastroenterol.
 Hepatol. 31, 965–972. https://doi.org/10.1111/jgh.13242
- Hanley, A.J.G., Bowden, D., Wagenknecht, L.E., Balasubramanyam, A.,
 Langfeld, C., Saad, M.F., Rotter, J.I., Guo, X., Chen, Y.D.I., Bryer-Ash, M.,
 Norris, J.M., Haffner, S.M., 2007. Associations of adiponectin with body fat
 distribution and insulin sensitivity in nondiabetic Hispanics and AfricanAmericans. J. Clin. Endocrinol. Metab. 92, 2665–2671.
 https://doi.org/10.1210/jc.2006-2614
- Harman-Boehm, I., Blüher, M., Redel, H., Sion-Vardy, N., Ovadia, S., Avinoach, E., Shai, I., Klöting, N., Stumvoll, M., Bashan, N., Rudich, A., 2007.
 Macrophage Infiltration into Omental *Versus* Subcutaneous Fat across
 Different Populations: Effect of Regional Adiposity and the Comorbidities of
 Obesity. J. Clin. Endocrinol. Metab. 92, 2240–2247.
 https://doi.org/10.1210/jc.2006-1811
- Harms, R.Z., Lorenzo-Arteaga, K.M., Ostlund, K.R., Smith, V.B., Smith, L.M., Gottlieb, P., Sarvetnick, N., 2018. Abnormal T cell frequencies, including cytomegalovirus - Associated expansions, distinguish seroconverted subjects at risk for type 1 diabetes. Front. Immunol. 9, 2332. https://doi.org/10.3389/fimmu.2018.02332
- Harms, R.Z., Lorenzo, K.M., Corley, K.P., Cabrera, M.S., Sarvetnick, N.E.,
 2015. Altered CD161bright CD8+ Mucosal Associated Invariant T (MAIT)Like Cell Dynamics and Increased Differentiation States among Juvenile
 Type 1 Diabetics. PLoS One 10, e0117335.
 https://doi.org/10.1371/journal.pone.0117335

Hayashi, E., Chiba, A., Tada, K., Haga, K., Kitagaichi, M., Nakajima, S., Kusaoi,
M., Sekiya, F., Ogasawara, M., Yamaji, K., Tamura, N., Takasaki, Y., Miyake, S., 2016. Involvement of mucosal-associated invariant T cells in ankylosing spondylitis. J. Rheumatol. 43, 1695–1703. https://doi.org/10.3899/jrheum.151133

- Hayashi, K., Jutabha, P., Endou, H., Sagara, H., Anzai, N., 2013. LAT1 Is a Critical Transporter of Essential Amino Acids for Immune Reactions in Activated Human T Cells. J. Immunol. 191, 4080–4085. https://doi.org/10.4049/jimmunol.1300923
- Hegde, P., Weiss, E., Paradis, V., Wan, J., Mabire, M., Sukriti, S., Rautou, P.E.,
 Albuquerque, M., Picq, O., Gupta, A.C., Ferrere, G., Gilgenkrantz, H., Kiaf,
 B., Toubal, A., Beaudoin, L., Lettéron, P., Moreau, R., Lehuen, A.,
 Lotersztajn, S., 2018. Mucosal-associated invariant T cells are a
 profibrogenic immune cell population in the liver. Nat. Commun. 9.
 https://doi.org/10.1038/s41467-018-04450-y
- Hengst, J., Strunz, B., Deterding, K., Ljunggren, H.G., Leeansyah, E., Manns, M.P., Cornberg, M., Sandberg, J.K., Wedemeyer, H., Björkström, N.K., 2016. Nonreversible MAIT cell-dysfunction in chronic hepatitis C virus infection despite successful interferon-free therapy. Eur. J. Immunol. 46, 2204–2210. https://doi.org/10.1002/eji.201646447
- Herrero-Sánchez, M.C., Rodríguez-Serrano, C., Almeida, J., San-Segundo, L., Inogés, S., Santos-Briz, Á., García-Briñón, J., SanMiguel, J.F., Del Cañizo, C., Blanco, B., 2016. Effect of mTORC1/mTORC2 inhibition on T cell function: potential role in graft- *versus* -host disease control. Br. J. Haematol. 173, 754–768. https://doi.org/10.1111/bjh.13984
- Hiejima, E., Kawai, T., Nakase, H., Tsuruyama, T., Morimoto, T., Yasumi, T., Taga, T., Kanegane, H., Hori, M., Ohmori, K., Higuchi, T., Matsuura, M., Yoshino, T., Ikeuchi, H., Kawada, K., Sakai, Y., Kitazume, M.T., Hisamatsu, T., Chiba, T., Nishikomori, R., Heike, T., 2015. Reduced Numbers and Proapoptotic Features of Mucosal-associated Invariant T Cells as a Characteristic Finding in Patients with Inflammatory Bowel Disease. https://doi.org/10.1097/MIB.00000000000397
- Hinks, T.S.C., Marchi, E., Jabeen, M., Olshansky, M., Kurioka, A., Pediongco,
 T.J., Meehan, B.S., Kostenko, L., Turner, S.J., Corbett, A.J., Chen, Z.,
 Klenerman, P., McCluskey, J., 2019. Activation and In Vivo Evolution of the
 MAIT Cell Transcriptome in Mice and Humans Reveals Tissue Repair

Functionality. Cell Rep. 28, 3249-3262.e5. https://doi.org/10.1016/j.celrep.2019.07.039

- Hinks, T.S.C., Wallington, J.C., Williams, A.P., Djukanovic, R., Staples, K.J.,
 Wilkinson, T.M.A., 2016. Steroid-induced deficiency of mucosal-associated invariant T cells in the chronic obstructive pulmonary disease lung implications for nontypeable haemophilus influenzae infection. Am. J.
 Respir. Crit. Care Med. 194, 1208–1218.
 https://doi.org/10.1164/rccm.201601-0002OC
- Hinks, T.S.C., Zhou, X., Staples, K.J., Dimitrov, B.D., Manta, A., Petrossian, T., Lum, P.Y., Smith, C.G., Ward, J.A., Howarth, P.H., Walls, A.F., Gadola, S.D., Djukanović, R., 2015. Innate and adaptive T cells in asthmatic patients: Relationship to severity and disease mechanisms. J. Allergy Clin. Immunol. 136, 323–333. https://doi.org/10.1016/j.jaci.2015.01.014
- Ho, P.C., Bihuniak, J.D., MacIntyre, A.N., Staron, M., Liu, X., Amezquita, R., Tsui, Y.C., Cui, G., Micevic, G., Perales, J.C., Kleinstein, S.H., Abel, E.D., Insogna, K.L., Feske, S., Locasale, J.W., Bosenberg, M.W., Rathmell, J.C., Kaech, S.M., 2015. Phosphoenolpyruvate Is a Metabolic Checkpoint of Anti-tumor T Cell Responses. Cell 162, 1217–1228. https://doi.org/10.1016/j.cell.2015.08.012
- Hotamisligil, G.S., 2017. Inflammation, metaflammation and immunometabolic disorders. Nat. Publ. Gr. 542. https://doi.org/10.1038/nature21363
- Hotamisligil, G.S., Arner, P., Caro, J.F., Atkinson, R.L., Spiegelman, B.M., 1995. Increased adipose tissue expression of tumor necrosis factor-α in human obesity and insulin resistance. J. Clin. Invest. 95, 2409–2415. https://doi.org/10.1172/JCI117936
- Hotamisligil, G.S., Shargill, N.S., Spiegelman, B.M., 1993. Adipose expression of tumor necrosis factor-α: Direct role in obesity-linked insulin resistance.
 Science (80-.). 259, 87–91. https://doi.org/10.1126/science.7678183
- Howden, A.J.M., Hukelmann, J.L., Brenes, A., Spinelli, L., Sinclair, L. V., Lamond, A.I., Cantrell, D.A., 2019. Quantitative analysis of T cell proteomes and environmental sensors during T cell differentiation. Nat. Immunol. 20, 1542–1554. https://doi.org/10.1038/s41590-019-0495-x
- Howson, L.J., Napolitani, G., Shepherd, D., Ghadbane, H., Kurupati, P.,Preciado-Llanes, L., Rei, M., Dobinson, H.C., Gibani, M.M., Teng, K.W.W.,Newell, E.W., Veerapen, N., Besra, G.S., Pollard, A.J., Cerundolo, V.,

2018. MAIT cell clonal expansion and TCR repertoire shaping in human volunteers challenged with Salmonella Paratyphi A. Nat. Commun. 9, 1–11. https://doi.org/10.1038/s41467-017-02540-x

- Hsu, C.-L., Dzhagalov, I.L., 2019. Metabolite Transporters—The Gatekeepers for T Cell Metabolism. Immunometabolism 1, e190012. https://doi.org/10.20900/immunometab20190012
- Hu, R., Xia, C.Q., Butfiloski, E., Clare-Salzler, M., 2018. Effect of high glucose on cytokine production by human peripheral blood immune cells and type I interferon signaling in monocytes: Implications for the role of hyperglycemia in the diabetes inflammatory process and host defense against infection. Clin. Immunol. 195, 139–148. https://doi.org/10.1016/j.clim.2018.06.003
- Hukelmann, J.L., Anderson, K.E., Sinclair, L. V., Grzes, K.M., Murillo, A.B., Hawkins, P.T., Stephens, L.R., Lamond, A.I., Cantrell, D.A., 2016. The cytotoxic T cell proteome and its shaping by the kinase mTOR. Nat. Immunol. 17, 104–112. https://doi.org/10.1038/ni.3314
- Iglay, K., Hannachi, H., Howie, P.J., Xu, J., Li, X., Engel, S.S., Moore, L.M., Rajpathak, S., 2016. Prevalence and co-prevalence of comorbidities among patients with type 2 diabetes mellitus. Curr. Med. Res. Opin. 32, 1243–1252. https://doi.org/10.1185/03007995.2016.1168291
- Illés, Z., Shimamura, M., Newcombe, J., Oka, N., Yamamura, T., 2004. Accumulation of Vα7.2-Jα33 invariant T cells in human autoimmune inflammatory lesions in the nervous system. Int. Immunol. 16, 223–230. https://doi.org/10.1093/intimm/dxh018
- Jacobs, S.R., Herman, C.E., MacIver, N.J., Wofford, J.A., Wieman, H.L.,
 Hammen, J.J., Rathmell, J.C., 2008. Glucose Uptake Is Limiting in T Cell
 Activation and Requires CD28-Mediated Akt-Dependent and Independent
 Pathways. J. Immunol. 180, 4476–4486.
 https://doi.org/10.4049/jimmunol.180.7.4476

Jeon, J., Jung, K.J., Jee, S.H., 2019. Waist circumference trajectories and risk of type 2 diabetes mellitus in Korean population: The Korean genome and epidemiology study (KoGES). BMC Public Health 19.

https://doi.org/10.1186/s12889-019-7077-6

Jiang, J., Yang, B., An, H., Wang, X., Liu, Y., Cao, Z., Zhai, F., Wang, R., Cao,
Y., Cheng, X., 2016. Mucosal-associated invariant T cells from patients
with tuberculosis exhibit impaired immune response. J. Infect. 72, 338–352.

https://doi.org/10.1016/j.jinf.2015.11.010

- Johnson, M.O., Wolf, M.M., Madden, M.Z., Andrejeva, G., Sugiura, A.,
 Contreras, D.C., Maseda, D., Liberti, M. V., Paz, K., Kishton, R.J., Johnson,
 M.E., de Cubas, A.A., Wu, P., Li, G., Zhang, Y., Newcomb, D.C., Wells,
 A.D., Restifo, N.P., Rathmell, W.K., Locasale, J.W., Davila, M.L., Blazar,
 B.R., Rathmell, J.C., 2018. Distinct Regulation of Th17 and Th1 Cell
 Differentiation by Glutaminase-Dependent Metabolism. Cell 175, 17801795.e19. https://doi.org/10.1016/j.cell.2018.10.001
- Kalucka, J., Missiaen, R., Georgiadou, M., Schoors, S., Lange, C., De Bock, K., Dewerchin, M., Carmeliet, P., 2015. Metabolic control of the cell cycle. Cell Cycle 14, 3379–3388. https://doi.org/10.1080/15384101.2015.1090068
- Katsarou, A., Gudbjörnsdottir, S., Rawshani, A., Dabelea, D., Bonifacio, E.,
 Anderson, B.J., Jacobsen, L.M., Schatz, D.A., Lernmark, A., 2017. Type 1
 diabetes mellitus. Nat. Rev. Dis. Prim. 3, 1–17.
 https://doi.org/10.1038/nrdp.2017.16
- Kavanagh Williamson, M., Coombes, N., Juszczak, F., Athanasopoulos, M.,
 Khan, M.B., Eykyn, T.R., Srenathan, U., Taams, L.S., Dias Zeidler, J., Da
 Poian, A.T., Huthoff, H., 2018. Upregulation of Glucose Uptake and
 Hexokinase Activity of Primary Human CD4+ T Cells in Response to
 Infection with HIV-1. Viruses 10, 114. https://doi.org/10.3390/v10030114
- Kawachi, I., Maldonado, J., Strader, C., Gilfillan, S., 2006. MR1-Restricted
 Vα19 i Mucosal-Associated Invariant T Cells Are Innate T Cells in the Gut
 Lamina Propria That Provide a Rapid and Diverse Cytokine Response . J.
 Immunol. 176, 1618–1627. https://doi.org/10.4049/jimmunol.176.3.1618
- Keating, S.E., Zaiatz-Bittencourt, V., Loftus, R.M., Keane, C., Brennan, K., Finlay, D.K., Gardiner, C.M., 2016. Metabolic Reprogramming Supports IFN-γ Production by CD56 bright NK Cells . J. Immunol. 196, 2552–2560. https://doi.org/10.4049/jimmunol.1501783
- Keaver, L., Webber, L., Dee, A., Shiely, F., Marsh, T., Balanda, K., Perry, I., 2013. Application of the UK Foresight Obesity Model in Ireland: The Health and Economic Consequences of Projected Obesity Trends in Ireland. PLoS One 8, e79827. https://doi.org/10.1371/journal.pone.0079827
- Keller, A.N., Eckle, S.B.G., Xu, W., Liu, L., Hughes, V.A., Mak, J.Y.W., Meehan,B.S., Pediongco, T., Birkinshaw, R.W., Chen, Z., Wang, H., D'Souza, C.,Kjer-Nielsen, L., Gherardin, N.A., Godfrey, D.I., Kostenko, L., Corbett, A.J.,

Purcell, A.W., Fairlie, D.P., McCluskey, J., Rossjohn, J., 2017. Drugs and drug-like molecules can modulate the function of mucosal-associated invariant T cells. Nat. Immunol. 18, 402–411. https://doi.org/10.1038/ni.3679

- Kelly, J., Minoda, Y., Meredith, T., Cameron, G., Philipp, M.S., Pellicci, D.G., Corbett, A.J., Kurts, C., Gray, D.H.D., Godfrey, D.I., Kannourakis, G., Berzins, S.P., 2019. Chronically stimulated human MAIT cells are unexpectedly potent IL-13 producers. Immunol. Cell Biol. 97, 689–699. https://doi.org/10.1111/imcb.12281
- Khan, S.S., Ning, H., Wilkins, J.T., Allen, N., Carnethon, M., Berry, J.D., Sweis, R.N., Lloyd-Jones, D.M., 2018. Association of body mass index with lifetime risk of cardiovascular disease and compression of morbidity. JAMA Cardiol. 3, 280–287. https://doi.org/10.1001/jamacardio.2018.0022
- Kim, J.S., Ro, S.H., Kim, M., Park, H.W., Semple, I.A., Park, H., Cho, U.S., Wang, W., Guan, K.L., Karin, M., Lee, J.H., 2015. Sestrin2 inhibits mTORC1 through modulation of GATOR complexes. Sci. Rep. 5, 1–10. https://doi.org/10.1038/srep09502
- Kim, M., Yoo, S.J., Kang, S.W., Kwon, J., Choi, I., Lee, C.H., 2017. TNFα and IL-1β in the synovial fluid facilitate mucosal-associated invariant T (MAIT) cell migration. Cytokine 99, 91–98. https://doi.org/10.1016/j.cyto.2017.07.007
- Kimball, A.B., Okun, M.M., Williams, D.A., Gottlieb, A.B., Papp, K.A., Zouboulis, C.C., Armstrong, A.W., Kerdel, F., Gold, M.H., Forman, S.B., Korman, N.J., Giamarellos-Bourboulis, E.J., Crowley, J.J., Lynde, C., Reguiai, Z., Prens, E.-P., Alwawi, E., Mostafa, N.M., Pinsky, B., Sundaram, M., Gu, Y., Carlson, D.M., Jemec, G.B.E., 2016. Two Phase 3 Trials of Adalimumab for Hidradenitis Suppurativa. N. Engl. J. Med. 375, 422–434. https://doi.org/10.1056/nejmoa1504370
- Kjer-Nielsen, L., Patel, O., Corbett, A.J., Le Nours, J., Meehan, B., Liu, L., Bhati, M., Chen, Z., Kostenko, L., Reantragoon, R., Williamson, N.A., Purcell, A.W., Dudek, N.L., McConville, M.J., O'Hair, R.A.J., Khairallah, G.N., Godfrey, D.I., Fairlie, D.P., Rossjohn, J., McCluskey, J., 2012. MR1 presents microbial vitamin B metabolites to MAIT cells. Nature 491, 717– 723. https://doi.org/10.1038/nature11605

Klop, B., Elte, J.W.F., Cabezas, M.C., 2013. Dyslipidemia in Obesity:

Mechanisms and Potential Targets. Nutrients. https://doi.org/10.3390/nu5041218

- Klysz, D., Tai, X., Robert, P.A., Craveiro, M., Cretenet, G., Oburoglu, L.,
 Mongellaz, C., Floess, S., Fritz, V., Matias, M.I., Yong, C., Surh, N., Marie,
 J.C., Huehn, J., Zimmermann, V., Kinet, S., Dardalhon, V., Taylor, N.,
 2015. Glutamine-dependent α-ketoglutarate production regulates the
 balance between T helper 1 cell and regulatory T cell generation. Sci.
 Signal. 8, ra97. https://doi.org/10.1126/scisignal.aab2610
- Koay, H.F., Gherardin, N.A., Enders, A., Loh, L., Mackay, L.K., Almeida, C.F., Russ, B.E., Nold-Petry, C.A., Nold, M.F., Bedoui, S., Chen, Z., Corbett, A.J., Eckle, S.B.G., Meehan, B., D'Udekem, Y., Konstantinov, I.E., Lappas, M., Liu, L., Goodnow, C.C., Fairlie, D.P., Rossjohn, J., Chong, M.M., Kedzierska, K., Berzins, S.P., Belz, G.T., McCluskey, J., Uldrich, A.P., Godfrey, D.I., Pellicci, D.G., 2016. A three-stage intrathymic development pathway for the mucosal-associated invariant T cell lineage. Nat. Immunol. 17, 1300–1311. https://doi.org/10.1038/ni.3565
- Koay, H.F., Su, S., Amann-Zalcenstein, D., Daley, S.R., Comerford, I., Miosge, L., Whyte, C.E., Konstantinov, I.E., D'Udekem, Y., Baldwin, T., Hickey, P.F., Berzins, S.P., Mak, J.Y.W., Sontani, Y., Roots, C.M., Sidwell, T., Kallies, A., Chen, Z., Nüssing, S., Kedzierska, K., Mackay, L.K., McColl, S.R., Deenick, E.K., Fairlie, D.P., McCluskey, J., Goodnow, C.C., Ritchie, M.E., Belz, G.T., Naik, S.H., Pellicci, D.G., Godfrey, D.I., 2019. A divergent transcriptional landscape underpins the development and functional branching of MAIT cells. Sci. Immunol. 4. https://doi.org/10.1126/sciimmunol.aay6039
- Kohlgruber, A.C., Gal-Oz, S.T., Lamarche, N.M., Shimazaki, M., Duquette, D., Nguyen, H.N., Mina, A.I., Paras, T., Tavakkoli, A., Von Andrian, U., Banks, A.S., Shay, T., Brenner, M.B., Lynch, L., 2018. γδ T cells producing interleukin-17A regulate adipose regulatory T cell homeostasis and thermogenesis /631/250/256 /631/250/2504 article. Nat. Immunol. 19, 464– 474. https://doi.org/10.1038/s41590-018-0094-2
- Koppejan, H., Jansen, D.T.S.L., Hameetman, M., Thomas, R., Toes, R.E.M., Van Gaalen, F.A., 2019. Altered composition and phenotype of mucosalassociated invariant T cells in early untreated rheumatoid arthritis. Arthritis Res. Ther. 21, 3. https://doi.org/10.1186/s13075-018-1799-1

- Kotchen, T.A., 2010. Obesity-Related Hypertension: Epidemiology,
 Pathophysiology, and Clinical Management. Am. J. Hypertens. 23, 1170– 1178. https://doi.org/10.1038/ajh.2010.172
- Kratz, M., Coats, B.R., Hisert, K.B., Hagman, D., Mutskov, V., Peris, E.,
 Schoenfelt, K.Q., Kuzma, J.N., Larson, I., Billing, P.S., Landerholm, R.W.,
 Crouthamel, M., Gozal, D., Hwang, S., Singh, P.K., Becker, L., 2014.
 Metabolic dysfunction drives a mechanistically distinct proinflammatory
 phenotype in adipose tissue macrophages. Cell Metab. 20, 614–625.
 https://doi.org/10.1016/j.cmet.2014.08.010
- Kurebayashi, Y., Nagai, S., Ikejiri, A., Ohtani, M., Ichiyama, K., Baba, Y.,
 Yamada, T., Egami, S., Hoshii, T., Hirao, A., Matsuda, S., Koyasu, S.,
 2012. PI3K-Akt-mTORC1-S6K1/2 Axis Controls Th17 Differentiation by
 Regulating Gfi1 Expression and Nuclear Translocation of RORγ. Cell Rep.
 1, 360–373. https://doi.org/10.1016/j.celrep.2012.02.007
- Kuric, E., Krogvold, L., Hanssen, K.F., Dahl-Jørgensen, K., Skog, O., Korsgren, O., 2018. SHORT COMMUNICATION No Evidence for Presence of Mucosal-Associated Invariant T Cells in the Insulitic Lesions in Patients Recently Diagnosed with Type 1 Diabetes. https://doi.org/10.1016/j.ajpath.2018.04.009
- Kurioka, A., Jahun, A.S., Hannaway, R.F., Walker, L.J., Fergusson, J.R., Sverremark-Ekström, E., Corbett, A.J., Ussher, J.E., Willberg, C.B., Klenerman, P., 2017. Shared and distinct phenotypes and functions of human cD161++ Vα7.2+ T cell subsets. Front. Immunol. 8. https://doi.org/10.3389/fimmu.2017.01031
- Kurioka, A., Ussher, J.E., Cosgrove, C., Clough, C., Fergusson, J.R., Smith, K., Kang, Y.H., Walker, L.J., Hansen, T.H., Willberg, C.B., Klenerman, P., 2015. MAIT cells are licensed through granzyme exchange to kill bacterially sensitized targets. Mucosal Immunol. 8, 429–440. https://doi.org/10.1038/mi.2014.81
- Kwon, Y.S., Cho, Y.N., Kim, M.J., Jin, H.M., Jung, H.J., Kang, J.H., Park, K.J.,
 Kim, T.J., Kee, H.J., Kim, N., Kee, S.J., Park, Y.W., 2015. Mucosalassociated invariant T cells are numerically and functionally deficient in patients with mycobacterial infection and reflect disease activity.
 Tuberculosis 95, 267–274. https://doi.org/10.1016/j.tube.2015.03.004

Lai, Z.W., Kelly, R., Winans, T., Marchena, I., Shadakshari, A., Yu, J., Dawood,

M., Garcia, R., Tily, H., Francis, L., Faraone, S. V., Phillips, P.E., Perl, A., 2018. Sirolimus in patients with clinically active systemic lupus erythematosus resistant to, or intolerant of, conventional medications: a single-arm, open-label, phase 1/2 trial. Lancet 391, 1186–1196. https://doi.org/10.1016/S0140-6736(18)30485-9

- Lal, K.G., Kim, D., Costanzo, M.C., Creegan, M., Leeansyah, E., Dias, J.,
 Paquin-Proulx, D., Eller, L.A., Schuetz, A., Phuang-ngern, Y., Krebs, S.J.,
 Slike, B.M., Kibuuka, H., Maganga, L., Nitayaphan, S., Kosgei, J.,
 Sacdalan, C., Ananworanich, J., Bolton, D.L., Michael, N.L., Shacklett, B.L.,
 Robb, M.L., Eller, M.A., Sandberg, J.K., 2020. Dynamic MAIT cell response
 with progressively enhanced innateness during acute HIV-1 infection. Nat.
 Commun. 11, 1–13. https://doi.org/10.1038/s41467-019-13975-9
- Lamichhane, R., Schneider, M., de la Harpe, S.M., Harrop, T.W.R., Hannaway, R.F., Dearden, P.K., Kirman, J.R., Tyndall, J.D.A., Vernall, A.J., Ussher, J.E., 2019. TCR- or Cytokine-Activated CD8+ Mucosal-Associated Invariant T Cells Are Rapid Polyfunctional Effectors That Can Coordinate Immune Responses. Cell Rep. 28, 3061-3076.e5. https://doi.org/10.1016/j.celrep.2019.08.054
- Landsberg, L., Aronne, L.J., Beilin, L.J., Burke, V., Igel, L.I., Lloyd-Jones, D., Sowers, J., 2013. Obesity-Related Hypertension: Pathogenesis, Cardiovascular Risk, and Treatment. J. Clin. Hypertens. 15, 14–33. https://doi.org/10.1111/jch.12049
- Lauby-Secretan, B., Scoccianti, C., Loomis, D., Grosse, Y., Bianchini, F., Straif, K., 2016. Body Fatness and Cancer-Viewpoint of the IARC Working Group.
- Le Bourhis, L., Dusseaux, M., Bohineust, A., Bessoles, S., Martin, E., Premel, V., Coré, M., Sleurs, D., Serriari, N.E., Treiner, E., Hivroz, C., Sansonetti, P., Gougeon, M.L., Soudais, C., Lantz, O., 2013. MAIT Cells Detect and Efficiently Lyse Bacterially-Infected Epithelial Cells. PLoS Pathog. 9. https://doi.org/10.1371/journal.ppat.1003681
- Le Bourhis, L., Martin, E., Péguillet, I., Guihot, A., Froux, N., Coré, M., Lévy, E., Dusseaux, M., Meyssonnier, V., Premel, V., Ngo, C., Riteau, B., Duban, L., Robert, D., Rottman, M., Soudais, C., Lantz, O., 2010. Antimicrobial activity of mucosal-associated invariant T cells. Nat. Immunol. 11, 701–708. https://doi.org/10.1038/ni.1890

Ledderose, C., Heyn, J., Limbeck, E., Kreth, S., 2011. Selection of reliable

reference genes for quantitative real-time PCR in human T cells and neutrophils. BMC Res. Notes 4, 427. https://doi.org/10.1186/1756-0500-4-427

- Lee, C.H., Zhang, H.H., Singh, S.P., Koo, L., Kabat, J., Tsang, H., Singh, T.P., Farber, J.M., 2018. C/EBPδ drives interactions between human MAIT cells and endothelial cells that are important for extravasation. Elife 7. https://doi.org/10.7554/eLife.32532
- Lee, Y.S., Kim, J.W., Osborne, O., Oh, D.Y., Sasik, R., Schenk, S., Chen, A., Chung, H., Murphy, A., Watkins, S.M., Quehenberger, O., Johnson, R.S., Olefsky, J.M., 2014. Increased adipocyte O2 consumption triggers HIF-1α, causing inflammation and insulin resistance in obesity. Cell 157, 1339– 1352. https://doi.org/10.1016/j.cell.2014.05.012
- Lee, Y.S., Wollam, J., Olefsky, J.M., 2018. An Integrated View of Immunometabolism. Cell. https://doi.org/10.1016/j.cell.2017.12.025
- Leeansyah, E., Svärd, J., Dias, J., Buggert, M., Nyström, J., Quigley, M.F., 2015. Arming of MAIT Cell Cytolytic Antimicrobial Activity Is Induced by IL-7 and Defective in HIV-1 Infection. PLoS Pathog 11, 1005072. https://doi.org/10.1371/journal.ppat.1005072
- Leng, T., Akther, H.D., Hackstein, C.P., Powell, K., King, T., Friedrich, M., Christoforidou, Z., McCuaig, S., Neyazi, M., Arancibia-Cárcamo, C. V., Hagel, J., Powrie, F., Peres, R.S., Millar, V., Ebner, D., Lamichhane, R., Ussher, J., Hinks, T.S.C., Marchi, E., Willberg, C., Klenerman, P., 2019. TCR and Inflammatory Signals Tune Human MAIT Cells to Exert Specific Tissue Repair and Effector Functions. Cell Rep. 28, 3077-3091.e5. https://doi.org/10.1016/j.celrep.2019.08.050
- Lepore, M., Kalinicenko, A., Colone, A., Paleja, B., Singhal, A., Tschumi, A., Lee, B., Poidinger, M., Zolezzi, F., Quagliata, L., Sander, P., Newell, E., Bertoletti, A., Terracciano, L., De Libero, G., Mori, L., 2014. Parallel T-cell cloning and deep sequencing of human MAIT cells reveal stable oligoclonal TCRβ 2 repertoire. Nat. Commun. 5. https://doi.org/10.1038/ncomms4866

Lezmi, G., Abou-Taam, R., Garcelon, N., Dietrich, C., Machavoine, F., Delacourt, C., Adel-Patient, K., Leite-de-Moraes, M., 2019. Evidence for a MAIT-17–high phenotype in children with severe asthma. J. Allergy Clin. Immunol. 144, 1714-1716.e6. https://doi.org/10.1016/j.jaci.2019.08.003

Lezmi, G., Abou Taam, R., Dietrich, C., Chatenoud, L., de Blic, J., Leite-de-

Moraes, M., 2018. Circulating IL-17-producing mucosal-associated invariant T cells (MAIT) are associated with symptoms in children with asthma. Clin. Immunol. 188, 7–11. https://doi.org/10.1016/j.clim.2017.11.009

- Li, J., Reantragoon, R., Kostenko, L., Corbett, A.J., Varigos, G., Carbone, F.R., 2017. The frequency of mucosal-associated invariant T cells is selectively increased in dermatitis herpetiformis. Australas. J. Dermatol. 58, 200–204. https://doi.org/10.1111/ajd.12456
- Li, Y., Woods, K., Parry-Strong, A., Anderson, R.J., Capistrano, C., Gestin, A., Painter, G.F., Hermans, I.F., Krebs, J., Gasser, O., 2020. Distinct Dysfunctional States of Circulating Innate-Like T Cells in Metabolic Disease. Front. Immunol. 11, 448. https://doi.org/10.3389/fimmu.2020.00448
- Li, Z., Nie, L., Chen, L., Sun, Y., Li, G., 2019. Rapamycin relieves inflammation of experimental autoimmune encephalomyelitis by altering the balance of Treg/Th17 in a mouse model. Neurosci. Lett. 705, 39–45. https://doi.org/10.1016/j.neulet.2019.04.035
- Libby, P., Buring, J.E., Badimon, L., Hansson, G.K., Deanfield, J., Bittencourt, M.S., Tokgözoğlu, L., Lewis, E.F., 2019. Atherosclerosis. Nat. Rev. Dis. Prim. 5, 1–18. https://doi.org/10.1038/s41572-019-0106-z
- Ling, L., Lin, Y., Zheng, W., Hong, S., Tang, X., Zhao, P., Li, M., Ni, J., Li, C., Wang, L., Jiang, Y., 2016. Circulating and tumor-infiltrating mucosal associated invariant T (MAIT) cells in colorectal cancer patients. Nat. Publ. Gr. https://doi.org/10.1038/srep20358
- Liu, R., Nikolajczyk, B.S., 2019. Tissue immune cells fuel obesity-associated inflammation in adipose tissue and beyond. Front. Immunol. https://doi.org/10.3389/fimmu.2019.01587
- Liu, R.T., Zhang, M., Yang, C.L., Zhang, P., Zhang, N., Du, T., Ge, M.R., Yue, L.T., Li, X.L., Li, H., Duan, R.S., 2018. Enhanced glycolysis contributes to the pathogenesis of experimental autoimmune neuritis. J. Neuroinflammation 15, 51. https://doi.org/10.1186/s12974-018-1095-7
- Loftus, R.M., Assmann, N., Kedia-Mehta, N., O'Brien, K.L., Garcia, A., Gillespie,
 C., Hukelmann, J.L., Oefner, P.J., Lamond, A.I., Gardiner, C.M., Dettmer,
 K., Cantrell, D.A., Sinclair, L. V., Finlay, D.K., 2018. Amino acid-dependent
 cMyc expression is essential for NK cell metabolic and functional

responses in mice. Nat. Commun. 9, 1–15. https://doi.org/10.1038/s41467-018-04719-2

- Loh, L., Wang, Z., Sant, S., Koutsakos, M., Jegaskanda, S., Corbett, A.J., Liu,
 L., Fairlie, D.P., Crowe, J., Rossjohn, J., Xu, J., Doherty, P.C., McCluskey,
 J., Kedzierska, K., 2016. Human mucosal-associated invariant T cells
 contribute to antiviral influenza immunity via IL-18-dependent activation.
 Proc. Natl. Acad. Sci. U. S. A. 113, 10133–10138.
 https://doi.org/10.1073/pnas.1610750113
- Longo, M., Zatterale, F., Naderi, J., Parrillo, L., Formisano, P., Raciti, G.A., Beguinot, F., Miele, C., 2019. Adipose tissue dysfunction as determinant of obesity-associated metabolic complications. Int. J. Mol. Sci. 20. https://doi.org/10.3390/ijms20092358
- Lu, B., Liu, M., Wang, J., Fan, H., Yang, D., Zhang, L., Gu, X., Nie, J., Chen, Z., Corbett, A.J., Zhan, M.J., Zhang, S., Bryant, V.L., Lew, A.M., McCluskey, J., Luo, H. bin, Cui, J., Zhang, Y., Zhan, Y., Lu, G., 2020. IL-17 production by tissue-resident MAIT cells is locally induced in children with pneumonia. Mucosal Immunol. 1–12. https://doi.org/10.1038/s41385-020-0273-y
- Lumeng, C.N., Bodzin, J.L., Saltiel, A.R., 2007. Obesity induces a phenotypic switch in adipose tissue macrophage polarization. J. Clin. Invest. 117, 175– 184. https://doi.org/10.1172/JCI29881
- Luppino, F.S., De Wit, L.M., Bouvy, P.F., Stijnen, T., Cuijpers, P., Penninx, B.W.J.H., Zitman, F.G., 2010. Overweight, obesity, and depression: A systematic review and meta-analysis of longitudinal studies. Arch. Gen. Psychiatry. https://doi.org/10.1001/archgenpsychiatry.2010.2
- Lynch, L., Michelet, X., Zhang, S., Brennan, P.J., Moseman, A., Lester, C., Besra, G., Vomhof-Dekrey, E.E., Tighe, M., Koay, H.F., Godfrey, D.I., Leadbetter, E.A., Sant'Angelo, D.B., Von Andrian, U., Brenner, M.B., 2015. Regulatory iNKT cells lack expression of the transcription factor PLZF and control the homeostasis of T reg cells and macrophages in adipose tissue. Nat. Immunol. 16, 85–95. https://doi.org/10.1038/ni.3047
- Lynch, L., Nowak, M., Varghese, B., Clark, J., Hogan, A.E., Toxavidis, V., Balk, S.P., O'Shea, D., O'Farrelly, C., Exley, M.A., 2012. Adipose Tissue Invariant NKT Cells Protect against Diet-Induced Obesity and Metabolic Disorder through Regulatory Cytokine Production. Immunity 37, 574–587. https://doi.org/10.1016/j.immuni.2012.06.016

- Ma, E.H., Bantug, G., Griss, T., Condotta, S., Johnson, R.M., Samborska, B., Mainolfi, N., Suri, V., Guak, H., Balmer, M.L., Verway, M.J., Raissi, T.C., Tsui, H., Boukhaled, G., Henriques da Costa, S., Frezza, C., Krawczyk, C.M., Friedman, A., Manfredi, M., Richer, M.J., Hess, C., Jones, R.G., 2017. Serine Is an Essential Metabolite for Effector T Cell Expansion. Cell Metab. 25, 345–357. https://doi.org/10.1016/j.cmet.2016.12.011
- Macintyre, A.N., Gerriets, V.A., Nichols, A.G., Michalek, R.D., Rudolph, M.C., Deoliveira, D., Anderson, S.M., Abel, E.D., Chen, B.J., Hale, L.P., Rathmell, J.C., 2014. The glucose transporter Glut1 is selectively essential for CD4 T cell activation and effector function. Cell Metab. 20, 61–72. https://doi.org/10.1016/j.cmet.2014.05.004
- Madak-Erdogan, Z., Band, S., Zhao, Y.C., Smith, B.P., Kulkoyluoglu-Cotul, E., Zuo, Q., Casiano, A.S., Wrobel, K., Rossi, G., Smith, R.L., Kim, S.H., Katzenellenbogen, J.A., Johnson, M.L., Patel, M., Marino, N., Storniolo, A.M. V., Flaws11, J.A., 2019. Free fatty acids rewire cancer metabolism in obesity-associated breast cancer via estrogen receptor and mTOR signaling. Cancer Res. 79, 2494–2510. https://doi.org/10.1158/0008-5472.CAN-18-2849
- Magalhaes, I., Pingris, K., Poitou, C., Bessoles, S., Venteclef, N., Kiaf, B.,
 Beaudoin, L., Da Silva, J., Allatif, O., Rossjohn, J., Kjer-Nielsen, L.,
 McCluskey, J., Ledoux, S., Genser, L., Torcivia, A., Soudais, C., Lantz, O.,
 Boitard, C., Aron-Wisnewsky, J., Larger, E., Clément, K., Lehuen, A., 2015.
 Mucosal-associated invariant T cell alterations in obese and type 2 diabetic patients. J. Clin. Invest. 125, 1752–1762. https://doi.org/10.1172/JCI78941
- Maimaiti, M., Sakamoto, S., Yamada, Y., Sugiura, M., Rii, J., Takeuchi, N., Imamura, Y., Furihata, T., Ando, K., Higuchi, K., Xu, M., Sazuka, T., Nakamura, K., Kaneda, A., Kanai, Y., Kyprianou, N., Ikehara, Y., Anzai, N., Ichikawa, T., 2020. Expression of L-type amino acid transporter 1 as a molecular target for prognostic and therapeutic indicators in bladder carcinoma. Sci. Rep. 10. https://doi.org/10.1038/s41598-020-58136-x
- Mamtani, M., Kulkarni, H., Dyer, T.D., Almasy, L., Mahaney, M.C., Duggirala, R., Comuzzie, A.G., Blangero, J., Curran, J.E., 2013. Waist Circumference Independently Associates with the Risk of Insulin Resistance and Type 2 Diabetes in Mexican American Families. PLoS One 8. https://doi.org/10.1371/journal.pone.0059153

- Mannan, M., Mamun, A., Doi, S., Clavarino, A., 2016. Prospective associations between depression and obesity for adolescent males and females - A systematic review and meta-analysis of longitudinal studies. PLoS One 11. https://doi.org/10.1371/journal.pone.0157240
- Mantzoros, C.S., Magkos, F., Brinkoetter, M., Sienkiewicz, E., Dardeno, T.A., Kim, S.Y., Hamnvik, O.P.R., Koniaris, A., 2011. Leptin in human physiology and pathophysiology. Am. J. Physiol. - Endocrinol. Metab. https://doi.org/10.1152/ajpendo.00315.2011
- Martin, E., Treiner, E., Duban, L., Guerri, L., Laude, H., Toly, C., Premel, V.,
 Devys, A., Moura, I.C., Tilloy, F., Cherif, S., Vera, G., Latour, S., Soudais,
 C., Lantz, O., 2009. Stepwise Development of MAIT Cells in Mouse and
 Human. PLoS Biol. 7, e1000054.

https://doi.org/10.1371/journal.pbio.1000054

- Mayer-Davis, E.J., Lawrence, J.M., Dabelea, D., Divers, J., Isom, S., Dolan, L., Imperatore, G., Linder, B., Marcovina, S., Pettitt, D.J., Pihoker, C., Saydah, S., Wagenknecht, L., 2017. Incidence trends of type 1 and type 2 diabetes among youths, 2002-2012. N. Engl. J. Med. 376, 1419–1429. https://doi.org/10.1056/NEJMoa1610187
- McArdle, M.A., Finucane, O.M., Connaughton, R.M., McMorrow, A.M., Roche, H.M., 2013. Mechanisms of obesity-induced inflammation and insulin resistance: Insights into the emerging role of nutritional strategies. Front. Endocrinol. (Lausanne). https://doi.org/10.3389/fendo.2013.00052
- Meierovics, A., Yankelevich, W.J.C., Cowley, S.C., 2013. MAIT cells are critical for optimal mucosal immune responses during in vivo pulmonary bacterial infection. Proc. Natl. Acad. Sci. U. S. A. 110, E3119–E3128. https://doi.org/10.1073/pnas.1302799110
- Meierovics, A.I., Cowley, S.C., 2016. MAIT cells promote inflammatory monocyte differentiation into dendritic cells during pulmonary intracellular infection. J. Exp. Med. 213, 2793–2809. https://doi.org/10.1084/jem.20160637
- Melo, A.M., O'Brien, A.M., Phelan, J.J., Kennedy, S.A., Wood, N.A.W.,
 Veerapen, N., Besra, G.S., Clarke, N.E., Foley, E.K., Ravi, A., MacCarthy,
 F., O'Toole, D., Ravi, N., Reynolds, J. V., Conroy, M.J., Hogan, A.E.,
 O'Sullivan, J., Dunne, M.R., 2019. Mucosal-associated invariant T cells
 display diminished effector capacity in oesophageal adenocarcinoma.

Front. Immunol. 10. https://doi.org/10.3389/fimmu.2019.01580

- Michalek, R.D., Gerriets, V.A., Jacobs, S.R., Macintyre, A.N., Maclver, N.J.,
 Mason, E.F., Sullivan, S.A., Nichols, A.G., Rathmell, J.C., 2011. Cutting
 Edge: Distinct Glycolytic and Lipid Oxidative Metabolic Programs Are
 Essential for Effector and Regulatory CD4 + T Cell Subsets . J. Immunol.
 186, 3299–3303. https://doi.org/10.4049/jimmunol.1003613
- Michelet, X., Dyck, L., Hogan, A., Loftus, R.M., Duquette, D., Wei, K., Beyaz, S., Tavakkoli, A., Foley, C., Donnelly, R., O'Farrelly, C., Raverdeau, M., Vernon, A., Pettee, W., O'Shea, D., Nikolajczyk, B.S., Mills, K.H.G., Brenner, M.B., Finlay, D., Lynch, L., 2018. Metabolic reprogramming of natural killer cells in obesity limits antitumor responses. Nat. Immunol. 19, 1330–1340. https://doi.org/10.1038/s41590-018-0251-7
- Miyazaki, Y., Miyake, S., Chiba, A., Lantz, O., Yamamura, T., 2011. Mucosalassociated invariant T cells regulate Th1 response in multiple sclerosis. Int. Immunol. 23, 529–535. https://doi.org/10.1093/intimm/dxr047
- Molofsky, A.B., Nussbaum, J.C., Liang, H.E., Dyken, S.J.V., Cheng, L.E., Mohapatra, A., Chawla, A., Locksley, R.M., 2013. Innate lymphoid type 2 cells sustain visceral adipose tissue eosinophils and alternatively activated macrophages. J. Exp. Med. 210, 535–549. https://doi.org/10.1084/jem.20121964
- Montecucco, F., Bianchi, G., Gnerre, P., Bertolotto, M., Dallegri, F., Ottonello,
 L., 2006. Induction of neutrophil chemotaxis by leptin: Crucial role for p38 and Src kinases, in: Annals of the New York Academy of Sciences.
 Blackwell Publishing Inc., pp. 463–471.
 https://doi.org/10.1196/annals.1351.045
- Moulton, V.R., Suarez-Fueyo, A., Meidan, E., Li, H., Mizui, M., Tsokos, G.C.,
 2017. Pathogenesis of Human Systemic Lupus Erythematosus: A Cellular Perspective. Trends Mol. Med.

https://doi.org/10.1016/j.molmed.2017.05.006

Murayama, G., Chiba, A., Suzuki, H., Nomura, A., Mizuno, T., Kuga, T.,
Nakamura, S., Amano, H., Hirose, S., Yamaji, K., Suzuki, Y., Tamura, N.,
Miyake, S., 2019. A Critical Role for Mucosal-Associated Invariant T Cells as Regulators and Therapeutic Targets in Systemic Lupus Erythematosus.
Front. Immunol. 10, 2681. https://doi.org/10.3389/fimmu.2019.02681

Munford, H., Dimeloe, S., 2019. Intrinsic and Extrinsic Determinants of T Cell

Metabolism in Health and Disease. Front. Mol. Biosci. https://doi.org/10.3389/fmolb.2019.00118

- Muscogiuri, G., Cantone, E., Cassarano, S., Tuccinardi, D., Barrea, L.,
 Savastano, S., Colao, A., 2019. Gut microbiota: a new path to treat obesity.
 Int. J. Obes. Suppl. 9, 10–19. https://doi.org/10.1038/s41367-019-0011-7
- Nakaya, M., Xiao, Y., Zhou, X., Chang, J.H., Chang, M., Cheng, X., Blonska, M., Lin, X., Sun, S.C., 2014. Inflammatory T cell responses rely on amino acid transporter ASCT2 facilitation of glutamine uptake and mTORC1 kinase activation. Immunity 40, 692–705. https://doi.org/10.1016/j.immuni.2014.04.007
- Navale, A.M., Paranjape, A.N., 2016. Glucose transporters: physiological and pathological roles. Biophys. Rev. https://doi.org/10.1007/s12551-015-0186-2
- Nazare, J.A., Smith, J., Borel, A.L., Aschner, P., Barter, P., Van Gaal, L., Tan, C.E., Wittchen, H.U., Matsuzawa, Y., Kadowaki, T., Ross, R., Brulle-Wohlhueter, C., Alméras, N., Haffner, S.M., Balkau, B., Després, J.P., 2015. Usefulness of measuring both body mass index and waist circumference for the estimation of visceral adiposity and related cardiometabolic risk profile (from the INSPIRE ME IAA Study). Am. J. Cardiol. 115, 307–315. https://doi.org/10.1016/j.amjcard.2014.10.039
- Neurath, M.F., 2019. Targeting immune cell circuits and trafficking in inflammatory bowel disease. Nat. Immunol. https://doi.org/10.1038/s41590-019-0415-0
- Nguyen, L.S., Vautier, M., Allenbach, Y., Zahr, N., Benveniste, O., Funck-Brentano, C., Salem, J.E., 2019. Sirolimus and mTOR Inhibitors: A Review of Side Effects and Specific Management in Solid Organ Transplantation. Drug Saf. https://doi.org/10.1007/s40264-019-00810-9
- Ni, Y., Zhao, L., Yu, Haoyong, Ma, X., Bao, Y., Rajani, C., Loo, L.W.M., Shvetsov, Y.B., Yu, Herbert, Chen, T., Zhang, Y., Wang, C., Hu, C., Su, M., Xie, G., Zhao, A., Jia, Wei, Jia, Weiping, 2015. Circulating Unsaturated Fatty Acids Delineate the Metabolic Status of Obese Individuals. EBioMedicine 2, 1513–1522. https://doi.org/10.1016/j.ebiom.2015.09.004
- Nishimura, S., Manabe, I., Nagasaki, M., Eto, K., Yamashita, H., Ohsugi, M., Otsu, M., Hara, K., Ueki, K., Sugiura, S., Yoshimura, K., Kadowaki, T., Nagai, R., 2009. CD8+ effector T cells contribute to macrophage

recruitment and adipose tissue inflammation in obesity. Nat. Med. 15, 914– 920. https://doi.org/10.1038/nm.1964

- O'Brien, A., Kedia-Mehta, N., Tobin, L., Veerapen, N., Besra, G.S., Shea, D.O., Hogan, A.E., 2020. Targeting mitochondrial dysfunction in MAIT cells limits IL-17 production in obesity. Cell. Mol. Immunol. https://doi.org/10.1038/s41423-020-0375-1
- O'Neill, L.A.J., Kishton, R.J., Rathmell, J., 2016. A guide to immunometabolism for immunologists. Nat. Rev. Immunol. https://doi.org/10.1038/nri.2016.70
- Ofei, F., Hurel, S., Newkirk, J., Sopwith, M., Taylor, R., 1996. Effects of an Engineered Human Anti-TNF- Antibody (CDP571) on Insulin Sensitivity and Glycemic Control in Patients With NIDDM. Diabetes 45, 881–885. https://doi.org/10.2337/diab.45.7.881
- Ohashi, K., Parker, J.L., Ouchi, N., Higuchi, A., Vita, J.A., Gokce, N., Pedersen, A.A., Kalthoff, C., Tullin, S., Sams, A., Summer, R., Walsh, K., 2010. Owen, M.R., Doran, E., Halestrap, A.P., 2000. Evidence that metformin exerts its anti-diabetic effects through inhibition of complex 1 of the mitochondrial respiratory chain. Biochem. J. 348, 607–614. https://doi.org/10.1042/bj3480607Adiponectin promotes macrophage polarization toward an anti-inflammatory phenotype. J. Biol. Chem. 285, 6153–6160. https://doi.org/10.1074/jbc.M109.088708
- Oren, R., Farnham, A.E., Saito, K., Milofsy, E., Karnovsky, M.L., 1963.
 Metabolic patterns in three types of phagocytizing cells. J. Cell Biol. 17, 487–501. https://doi.org/10.1083/jcb.17.3.487
- Ouchi, N., Parker, J.L., Lugus, J.J., Walsh, K., 2011. Adipokines in inflammation and metabolic disease. Nat. Rev. Immunol. 11, 85–97. https://doi.org/10.1038/nri2921
- Owen, M.R., Doran, E., Halestrap, A.P., 2000. Evidence that metformin exerts its anti-diabetic effects through inhibition of complex 1 of the mitochondrial respiratory chain. Biochem. J. 348, 607–614. https://doi.org/10.1042/bj3480607
- Pachman, L.M., 1967. The carbohydrate metabolism and respiration of isolated small lymphocytes. In vitro studies of normal and phytohemagglutinin stimulated cells. Blood 30, 691–706.

https://doi.org/10.1182/blood.V30.6.691.691

Palmer, C.S., Hussain, T., Duette, G., Weller, T.J., Ostrowski, M., Sada-Ovalle,

I., Crowe, S.M., 2016. Regulators of Glucose Metabolism in CD4+ and CD8+ T Cells. Int. Rev. Immunol. 35, 477–488. https://doi.org/10.3109/08830185.2015.1082178

- Palmer, C.S., Ostrowski, M., Balderson, B., Christian, N., Crowe, S.M., 2015. Glucose metabolism regulates T cell activation, differentiation, and functions. Front. Immunol. https://doi.org/10.3389/fimmu.2015.00001
- Pantalone, K.M., Hobbs, T.M., Chagin, K.M., Kong, S.X., Wells, B.J., Kattan, M.W., Bouchard, J., Sakurada, B., Milinovich, A., Weng, W., Bauman, J., Misra-Hebert, A.D., Zimmerman, R.S., Burguera, B., 2017. Prevalence and recognition of obesity and its associated comorbidities: cross-sectional analysis of electronic health record data from a large US integrated health system. BMJ Open 7, 17583. https://doi.org/10.1136/bmjopen-2017-017583
- Paquin-Proulx, D., Costa, P.R., Silveira, C.G.T., Marmorato, M.P., Cerqueira, N.B., Sutton, M.S., O'Connor, S.L., Carvalho, K.I., Nixon, D.F., Kallas, E.G., 2018. Latent Mycobacterium tuberculosis infection is associated with a higher frequency of mucosal-associated invariant T and invariant natural killer T cells. Front. Immunol. 9, 19. https://doi.org/10.3389/fimmu.2018.01394
- Patsouris, D., Li, P.P., Thapar, D., Chapman, J., Olefsky, J.M., Neels, J.G., 2008. Ablation of CD11c-Positive Cells Normalizes Insulin Sensitivity in Obese Insulin Resistant Animals. Cell Metab. 8, 301–309. https://doi.org/10.1016/j.cmet.2008.08.015
- Peng, M., Yin, N., Chhangawala, S., Xu, K., Leslie, C.S., Li, M.O., 2016. Aerobic glycolysis promotes T helper 1 cell differentiation through an epigenetic mechanism. Science (80-.). 354, 481–484. https://doi.org/10.1126/science.aaf6284
- Perry, I., Dee, A., Staines, A., McVeigh, T., Rose Sweeney, M., O'Neill, C.,
 Callan, A., Doherty, E., JE, C., O'Dwyer, V., Kearns, K., Sharp, L., Kee, F.,
 Hughes, J., Balanda, K., 2012. The cost of overweight and obesity on the
 Island of Ireland. Cork.
- Pigeyre, M., Yazdi, F.T., Kaur, Y., Meyre, D., 2016. Recent progress in genetics, epigenetics and metagenomics unveils the pathophysiology of human obesity. Clin. Sci. 130, 943–986. https://doi.org/10.1042/CS20160136

- Pineda, E., Sanchez-Romero, L.M., Brown, M., Jaccard, A., Jewell, J., Galea, G., Webber, L., Breda, J., 2018. Forecasting Future Trends in Obesity across Europe: The Value of Improving Surveillance. Obes. Facts 11, 360– 371. https://doi.org/10.1159/000492115
- Pisarska, M.M., Dunne, M.R., O'Shea, D., Hogan, A.E., 2020. Interleukin-17 producing mucosal associated invariant T cells - emerging players in chronic inflammatory diseases? Eur. J. Immunol. 50, 1098–1108. https://doi.org/10.1002/eji.202048645
- Pollizzi, K.N., Patel, C.H., Sun, I.H., Oh, M.H., Waickman, A.T., Wen, J., Delgoffe, G.M., Powell, J.D., 2015. mTORC1 and mTORC2 selectively regulate CD8+ T cell differentiation. J. Clin. Invest. 125, 2090–2108. https://doi.org/10.1172/JCI77746

Porcelli, S., Yockey, C.E., Brenner, M.B., Balk, S.P., 1993. Analysis of T cell

antigen receptor (TCR) expression by human peripheral blood CD4-8- $\alpha\beta$

T cells demonstrates preferential use of several V β genes and an invariant

TCR α chain. J. Exp. Med. 178, 1–16. https://doi.org/10.1084/jem.178.1.1

Procaccini, C., Carbone, F., Di Silvestre, D., Brambilla, F., De Rosa, V.,
Galgani, M., Faicchia, D., Marone, G., Tramontano, D., Corona, M., Alviggi,
C., Porcellini, A., La Cava, A., Mauri, P., Matarese, G., 2016. The
Proteomic Landscape of Human Ex Vivo Regulatory and Conventional T
Cells Reveals Specific Metabolic Requirements [Immunity 44, 406-421,
(2016)]. Immunity. https://doi.org/10.1016/j.immuni.2016.02.022

Provine, N.M., Klenerman, P., 2020. MAIT Cells in Health and Disease. Annu. Rev. Immunol. 38, 203–228. https://doi.org/10.1146/annurev-immunol-080719-015428

Public Health England, 2014. Adult obesity and type 2 diabetes.

Quail, D.F., Dannenberg, A.J., 2019. The obese adipose tissue microenvironment in cancer development and progression. Nat. Rev. Endocrinol. https://doi.org/10.1038/s41574-018-0126-x

Rahimpour, A., Koay, H.F., Enders, A., Clanchy, R., Eckle, S.B.G., Meehan, B., Chen, Z., Whittle, B., Liu, L., Fairlie, D.P., Goodnow, C.C., McCluskey, J., Rossjohn, J., Uldrich, A.P., Pellicci, D.G., Godfrey, D.I., 2015. Identification of phenotypically and functionally heterogeneous mouse mucosalassociated invariant T cells using MR1 tetramers. J. Exp. Med. 212, 1095– 1108. https://doi.org/10.1084/jem.20142110

- Rao, R.R., Li, Q., Odunsi, K., Shrikant, P.A., 2010. The mTOR Kinase
 Determines Effector versus Memory CD8+ T Cell Fate by Regulating the
 Expression of Transcription Factors T-bet and Eomesodermin. Immunity
 32, 67–78. https://doi.org/10.1016/j.immuni.2009.10.010
- Raposo, B., Vaartjes, D., Ahlqvist, E., Nandakumar, K.-S., Holmdahl, R., 2015.
 System A amino acid transporters regulate glutamine uptake and attenuate antibody-mediated arthritis. Immunology 146, 607–617.
 https://doi.org/10.1111/imm.12531
- Ray, J.P., Staron, M.M., Shyer, J.A., Ho, P.C., Marshall, H.D., Gray, S.M.,
 Laidlaw, B.J., Araki, K., Ahmed, R., Kaech, S.M., Craft, J., 2015. The
 Interleukin-2-mTORc1 Kinase Axis Defines the Signaling, Differentiation,
 and Metabolism of T Helper 1 and Follicular B Helper T Cells. Immunity 43,
 690–702. https://doi.org/10.1016/j.immuni.2015.08.017
- Raychaudhuri, S.K., Abria, C., Mitra, A., Raychaudhuri, S.P., 2020. Functional significance of MAIT cells in psoriatic arthritis. Cytokine 125, 154855. https://doi.org/10.1016/j.cyto.2019.154855
- Reantragoon, R., Corbett, A.J., Sakala, I.G., Gherardin, N.A., Furness, J.B., Chen, Z., Eckle, S.B.G., Uldrich, A.P., Birkinshaw, R.W., Patel, O., Kostenko, L., Meehan, B., Kedzierska, K., Liu, L., Fairlie, D.P., Hansen, T.H., Godfrey, D.I., Rossjohn, J., McCluskey, J., Kjer-Nielsen, L., 2013. Antigen-loaded MR1 tetramers define T cell receptor heterogeneity in mucosal-associated invariant T cells. J. Exp. Med. 210, 2305–2320. https://doi.org/10.1084/jem.20130958
- Reis, B.S., Lee, K., Fanok, M.H., Mascaraque, C., Amoury, M., Cohn, L.B.,
 Rogoz, A., Dallner, O.S., Moraes-Vieira, P.M., Domingos, A.I., Mucida, D.,
 2015. Leptin Receptor Signaling in T Cells Is Required for Th17
 Differentiation. J. Immunol. 194, 5253–5260.
 https://doi.org/10.4049/jimmunol.1402996
- Renner, K., Geiselhöringer, A.-L., Fante, M., Bruss, C., Färber, S.,
 Schönhammer, G., Peter, K., Singer, K., Andreesen, R., Hoffmann, P.,
 Oefner, P., Herr, W., Kreutz, M., 2015. Metabolic plasticity of human T
 cells: Preserved cytokine production under glucose deprivation or
 mitochondrial restriction, but 2-deoxy-glucose affects effector functions.

Eur. J. Immunol. 45, 2504–2516. https://doi.org/10.1002/eji.201545473

- Ricciardi, S., Manfrini, N., Alfieri, R., Pagani, M., Abrignani, S., Biffo, S., 2018. The Translational Machinery of Human CD4 + T Cells Is Poised for Activation and Controls the Switch from Quiescence to Metabolic Remodeling. Cell Metab. 28, 895–906. https://doi.org/10.1016/j.cmet.2018.08.009
- Robson MacDonald, H., 1977. Energy metabolism and T-cell-mediated cytolysis: II. Selective inhibition of cytolysis by 2-deoxy-D-glucose*. J. Exp. Med. 146, 710–719. https://doi.org/10.1084/jem.146.3.710
- Rodríguez, A., Ezquerro, S., Méndez-Giménez, L., Becerril, S., Frühbeck, G., 2015. Revisiting the adipocyte: a model for integration of cytokine signaling in the regulation of energy metabolism. Am. J. Physiol. Metab. 309, E691– E714. https://doi.org/10.1152/ajpendo.00297.2015
- Rodríguez, L., Graniel, J., Ortiz, R., 2007. Effect of leptin on activation and cytokine synthesis in peripheral blood lymphocytes of malnourished infected children. Clin. Exp. Immunol. 148, 478–485. https://doi.org/10.1111/j.1365-2249.2007.03361.x
- Rodriguez, P.C., Quiceno, D.G., Ochoa, A.C., 2007. L-arginine availability regulates T-lymphocyte cell-cycle progression. Blood 109, 1568–1573. https://doi.org/10.1182/blood-2006-06-031856
- Rolf, J., Zarrouk, M., Finlay, D.K., Foretz, M., Viollet, B., Cantrell, D.A., 2013.
 AMPKα1: A glucose sensor that controls CD8 T-cell memory. Eur. J.
 Immunol. 43, 889–896. https://doi.org/10.1002/eji.201243008
- Ron-Harel, N., Ghergurovich, J.M., Notarangelo, G., LaFleur, M.W., Tsubosaka,
 Y., Sharpe, A.H., Rabinowitz, J.D., Haigis, M.C., 2019. T Cell Activation
 Depends on Extracellular Alanine. Cell Rep. 28, 3011-3021.e4.
 https://doi.org/10.1016/j.celrep.2019.08.034
- Ross, S.H., Cantrell, D.A., 2018. Signaling and Function of Interleukin-2 in T Lymphocytes. Annu. Rev. Immunol. 36, 411–433. https://doi.org/10.1146/annurev-immunol-042617-053352
- Rothman, K.J., 2008. BMI-related errors in the measurement of obesity. Int. J. Obes. 32, S56–S59. https://doi.org/10.1038/ijo.2008.87
- Rouxel, O., Da Silva, J., Beaudoin, L., Nel, I., Tard, C., Cagninacci, L., Kiaf, B.,Oshima, M., Diedisheim, M., Salou, M., Corbett, A., Rossjohn, J.,McCluskey, J., Scharfmann, R., Battaglia, M., Polak, M., Lantz, O.,

Beltrand, J., Lehuen, A., 2017. Cytotoxic and regulatory roles of mucosalassociated invariant T cells in type 1 diabetes. Nat. Immunol. 18, 1321– 1331. https://doi.org/10.1038/ni.3854

- Ryan, A.S., Berman, D.M., Nicklas, B.J., Sinha, M., Gingerich, R.L., Meneilly, G.S., Egan, J.M., Elahi, D., 2003. Plasma adiponectin and leptin levels, body composition, and glucose utilization in adult women with wide ranges of age and obesity. Diabetes Care 26, 2383–2388. https://doi.org/10.2337/diacare.26.8.2383
- Salmond, R.J., 2018. mTOR regulation of glycolytic metabolism in T cells. Front. Cell Dev. Biol. https://doi.org/10.3389/fcell.2018.00122
- Salou, M., Nicol, B., Garcia, A., Baron, D., Michel, L., Elong-Ngono, A., Hulin,
 P., Nedellec, S., Jacq-Foucher, M., Le Frère, F., Jousset, N., Bourreille, A.,
 Wiertlewski, S., Soulillou, J.P., Brouard, S., Nicot, A.B., Degauque, N.,
 Laplaud, D.A., 2016. Neuropathologic, phenotypic and functional analyses
 of Mucosal Associated Invariant T cells in Multiple Sclerosis. Clin. Immunol.
 166–167, 1–11. https://doi.org/10.1016/j.clim.2016.03.014
- Sancak, Y., Peterson, T.R., Shaul, Y.D., Lindquist, R.A., Thoreen, C.C., Bar-Peled, L., Sabatini, D.M., 2008. The rag GTPases bind raptor and mediate amino acid signaling to mTORC1. Science (80-.). 320, 1496–1501. https://doi.org/10.1126/science.1157535
- Sattler, A., Dang-Heine, C., Reinke, P., Babel, N., 2015. IL-15 dependent induction of IL-18 secretion as a feedback mechanism controlling human MAIT-cell effector functions. Eur. J. Immunol. 45, 2286–2298. https://doi.org/10.1002/eji.201445313
- Saxton, R.A., Knockenhauer, K.E., Wolfson, R.L., Chantranupong, L., Pacold, M.E., Wang, T., Schwartz, T.U., Sabatini, D.M., 2016. Structural basis for leucine sensing by the Sestrin2-mTORC1 pathway. Science (80-.). 351, 53–58. https://doi.org/10.1126/science.aad2087
- Saxton, R.A., Sabatini, D.M., 2017. mTOR Signaling in Growth, Metabolism, and Disease. Cell. https://doi.org/10.1016/j.cell.2017.02.004
- Sbarra AJ, Karnovsky M. L., 1959. The Biochemical Basis of Phanocytosis. I. Metabolic changes during the ingestion of particles by polymorphonuclear leukocytes. J. Biol. Chem. 234, 1355–1362.
- Scalise, M., Galluccio, M., Console, L., Pochini, L., Indiveri, C., 2018. The human SLC7A5 (LAT1): The intriguing histidine/large neutral amino acid

transporter and its relevance to human health. Front. Chem. https://doi.org/10.3389/fchem.2018.00243

- Seach, N., Guerri, L., Le Bourhis, L., Mburu, Y., Cui, Y., Bessoles, S., Soudais, C., Lantz, O., 2013. Double Positive Thymocytes Select Mucosal-Associated Invariant T Cells. J. Immunol. 191, 6002–6009. https://doi.org/10.4049/jimmunol.1301212
- Sena, L.A., Li, S., Jairaman, A., Prakriya, M., Ezponda, T., Hildeman, D.A.,
 Wang, C.R., Schumacker, P.T., Licht, J.D., Perlman, H., Bryce, P.J.,
 Chandel, N.S., 2013. Mitochondria Are Required for Antigen-Specific T Cell
 Activation through Reactive Oxygen Species Signaling. Immunity 38, 225–236. https://doi.org/10.1016/j.immuni.2012.10.020
- Sener, Z., Cederkvist, F.H., Volchenkov, R., Holen, H.L., Skålhegg, B.S., 2016. T Helper Cell Activation and Expansion Is Sensitive to Glutaminase Inhibition under Both Hypoxic and Normoxic Conditions. PLoS One 11, e0160291. https://doi.org/10.1371/journal.pone.0160291
- Serriari, N.-E., Eoche, M., Lamotte, L., Lion, J., Fumery, M., Marcelo, P., Chatelain, D., Barre, A., Nguyen-Khac, E., Lantz, O., Dupas, J.-L., Treiner, E., 2014. Innate mucosal-associated invariant T (MAIT) cells are activated in inflammatory bowel diseases. Clin. Exp. Immunol. 176, 266–274. https://doi.org/10.1111/cei.12277
- Shah, N.R., Braverman, E.R., 2012. Measuring Adiposity in Patients: The Utility of Body Mass Index (BMI), Percent Body Fat, and Leptin. PLoS One 7, e33308. https://doi.org/10.1371/journal.pone.0033308
- Shi, H., Kokoeva, M. V., Inouye, K., Tzameli, I., Yin, H., Flier, J.S., 2006. TLR4 links innate immunity and fatty acid-induced insulin resistance. J. Clin. Invest. 116, 3015–3025. https://doi.org/10.1172/JCI28898
- Shi, L.Z., Wang, R., Huang, G., Vogel, P., Neale, G., Green, D.R., Chi, H., 2011. HIF1α-dependent glycolytic pathway orchestrates a metabolic checkpoint for the differentiation of TH17 and Treg cells. J. Exp. Med. 208, 1367–1376. https://doi.org/10.1084/jem.20110278
- Simmonds, M., Llewellyn, A., Owen, C.G., Woolacott, N., 2016. Predicting adult obesity from childhood obesity: A systematic review and meta-analysis. Obes. Rev. 17, 95–107. https://doi.org/10.1111/obr.12334
- Sinclair, L. V., Howden, A.J.M., Brenes, A., Spinelli, L., Hukelmann, J.L., Macintyre, A.N., Liu, X., Thomson, S., Taylor, P.M., Rathmell, J.C.,

Locasale, J.W., Lamond, A.I., Cantrell, D.A., 2019. Antigen receptor control of methionine metabolism in T cells. Elife 8. https://doi.org/10.7554/eLife.44210

- Sinclair, L. V., Neyens, D., Ramsay, G., Taylor, P.M., Cantrell, D.A., 2018. Single cell analysis of kynurenine and System L amino acid transport in T cells. Nat. Commun. 9, 1–11. https://doi.org/10.1038/s41467-018-04366-7
- Sinclair, L. V., Rolf, J., Emslie, E., Shi, Y.B., Taylor, P.M., Cantrell, D.A., 2013. Control of amino-acid transport by antigen receptors coordinates the metabolic reprogramming essential for T cell differentiation. Nat. Immunol. 14, 500–508. https://doi.org/10.1038/ni.2556
- Siren, R., Eriksson, J.G., Vanhanen, H., 2012. Waist circumference a good indicator of future risk for type 2 diabetes and cardiovascular disease. BMC Public Health 12, 631. https://doi.org/10.1186/1471-2458-12-631
- Sobkowiak, M.J., Davanian, H., Heymann, R., Gibbs, A., Emgård, J., Dias, J.,
 Aleman, S., Krüger-Weiner, C., Moll, M., Tjernlund, A., Leeansyah, E.,
 Sällberg Chen, M., Sandberg, J.K., 2019. Tissue-resident MAIT cell
 populations in human oral mucosa exhibit an activated profile and produce
 IL-17. Eur. J. Immunol. 49, 133–143. https://doi.org/10.1002/eji.201847759
- Spaan, M., Hullegie, S.J., Beudeker, B.J.B., Kreefft, K., Van Oord, G.W.,
 Groothuismink, Z.M.A., Van Tilborg, M., Rijnders, B., De Knegt, R.J.,
 Claassen, M.A.A., Boonstra, A., 2016. Frequencies of circulating MAIT
 cells are diminished in chronic hCV, HIV and HCV/ HIV Co-Infection and do
 not recover during therapy. PLoS One 11.
 https://doi.org/10.1371/journal.pone.0159243
- Sugimoto, C., Hirotani, M., Yoshikiyo, K., Koshimizu, U., Wakao, R., Horinouchi, T., Mazaki, Y., Higashi, T., Fukazawa, T., Fujita, H., Sasaki, H., Wakao, H., 2016. The dynamics of mucosal-associated invariant T cells in multiple sclerosis. Springerplus 5, 1259. https://doi.org/10.1186/s40064-016-2923-9
- Suliman, S., Gela, A., Mendelsohn, S.C., Iwany, S.K., Tamara, K.L., Mabwe, S.,
 Bilek, N., Darboe, F., Fisher, M., Corbett, A.J., Kjer-Nielsen, L., Eckle,
 S.B.G., Huang, C.C., Zhang, Z., Lewinsohn, D.M., McCluskey, J.,
 Rossjohn, J., Hatherill, M., León, S.R., Calderon, R.I., Lecca, L., Murray,
 M., Scriba, T.J., Van Rhijn, I., Moody, D.B., 2020. Peripheral Blood
 Mucosal-Associated Invariant T Cells in Tuberculosis Patients and Healthy
 Mycobacterium tuberculosis-Exposed Controls. J. Infect. Dis. 222, 995–

1007. https://doi.org/10.1093/infdis/jiaa173

- Sun, K., Kusminski, C.M., Scherer, P.E., 2011. Adipose tissue remodeling and obesity. J. Clin. Invest. 121, 2094–2101. https://doi.org/10.1172/JCI45887
- Sundström, P., Ahlmanner, F., Akéus, P., Sundquist, M., Alsén, S., Yrlid, U.,
 Börjesson, L., Sjöling, Å., Gustavsson, B., Wong, S.B.J., Quiding-Järbrink,
 M., 2015. Human Mucosa-Associated Invariant T Cells Accumulate in
 Colon Adenocarcinomas but Produce Reduced Amounts of IFN-γ. J.
 Immunol. 195, 3472–3481. https://doi.org/10.4049/jimmunol.1500258
- Sung, H., Siegel, R.L., Torre, L.A., Pearson-Stuttard, J., Islami, F., Fedewa,
 S.A., Goding Sauer, A., Shuval, K., Gapstur, S.M., Jacobs, E.J.,
 Giovannucci, E.L., Jemal, A., 2018. Global patterns in excess body weight and the associated cancer burden. CA. Cancer J. Clin. 69, caac.21499. https://doi.org/10.3322/caac.21499
- Surendar, J., Frohberger, S.J., Karunakaran, I., Schmitt, V., Stamminger, W., Neumann, A.-L., Wilhelm, C., Hoerauf, A., Hübner, M.P., 2019. Adiponectin Limits IFN-γ and IL-17 Producing CD4 T Cells in Obesity by Restraining Cell Intrinsic Glycolysis. Front. Immunol. 10, 2555. https://doi.org/10.3389/fimmu.2019.02555
- Swinburn, B.A., Sacks, G., Hall, K.D., McPherson, K., Finegood, D.T., Moodie, M.L., Gortmaker, S.L., 2011. Series Obesity 1 The global obesity pandemic: shaped by global drivers and local environments, Lancet. https://doi.org/10.1016/S0140-6736(11)60813-1
- Teunissen, M.B.M., Yeremenko, N.G., Baeten, D.L.P., Chielie, S., Spuls, P.I., De Rie, M.A., Lantz, O., Res, P.C.M., 2014. The IL-17A-producing CD8 + T-cell population in psoriatic lesional skin comprises mucosa-associated invariant t cells and conventional t cells. J. Invest. Dermatol. 134, 2898– 2907. https://doi.org/10.1038/jid.2014.261
- Thaçi, D., Blauvelt, A., Reich, K., Tsai, T.F., Vanaclocha, F., Kingo, K., Ziv, M., Pinter, A., Hugot, S., You, R., Milutinovic, M., 2015. Secukinumab is superior to ustekinumab in clearing skin of subjects with moderate to severe plaque psoriasis: CLEAR, a randomized controlled trial. J. Am. Acad. Dermatol. 73, 400–409. https://doi.org/10.1016/j.jaad.2015.05.013
- Tilloy, F., Treiner, E., Park, S.H., Garcia, C., Lemonnier, F., De La Salle, H., Bendelac, A., Bonneville, M., Lantz, O., 1999. An invariant T cell receptor α chain defines a novel TAP-independent major histocompatibility complex

class lb-restricted α/β T cell subpopulation in mammals. J. Exp. Med. 189, 1907–1921. https://doi.org/10.1084/jem.189.12.1907

- Tobin, L.M., Mavinkurve, M., Carolan, E., Kinlen, D., O'Brien, E.C., Little, M.A., Finlay, D.K., Cody, D., Hogan, A.E., O'Shea, D., 2017. NK cells in childhood obesity are activated, metabolically stressed, and functionally deficient. JCI Insight 2. https://doi.org/10.1172/jci.insight.94939
- Tominaga, K., Yamagiwa, S., Setsu, T., Kimura, N., Honda, H., Kamimura, H., Honda, Y., Takamura, M., Yokoyama, J., Suzuki, K., Wakai, T., Terai, S., 2017. Possible involvement of mucosal-associated invariant T cells in the progression of inflammatory bowel diseases. Biomed. Res. 38, 111–121. https://doi.org/10.2220/biomedres.38.111
- Toubal, A., Kiaf, B., Beaudoin, L., Cagninacci, L., Rhimi, M., Fruchet, B., da Silva, J., Corbett, A.J., Simoni, Y., Lantz, O., Rossjohn, J., McCluskey, J., Lesnik, P., Maguin, E., Lehuen, A., 2020. Mucosal-associated invariant T cells promote inflammation and intestinal dysbiosis leading to metabolic dysfunction during obesity. Nat. Commun. 11. https://doi.org/10.1038/s41467-020-17307-0
- Toussirot, É., Laheurte, C., Gaugler, B., Gabriel, D., Saas, P., 2018. Increased IL-22- and IL-17A-Producing Mucosal-Associated Invariant T Cells in the Peripheral Blood of Patients With Ankylosing Spondylitis. Front. Immunol. 9, 1610. https://doi.org/10.3389/fimmu.2018.01610
- Treiner, E., Duban, L., Bahram, S., Radosavljevic, M., Wanner, V., Tilloy, F., Affaticati, P., Gilfillan, S., Lantz, O., 2003. Selection of evolutionarily conserved mucosal-associated invariant T cells by MR1. Nature 422, 164– 169. https://doi.org/10.1038/nature01433
- Trøseid, M., Nestvold, T.K., Rudi, K., Thoresen, H., Nielsen, E.W., Lappegård, K.T., 2013. Plasma lipopolysaccharide is closely associated with glycemic control and abdominal obesity: Evidence from bariatric surgery. Diabetes Care 36, 3627–3632. https://doi.org/10.2337/dc13-0451
- Ussher, J.E., Bilton, M., Attwod, E., Shadwell, J., Richardson, R., de Lara, C., Mettke, E., Kurioka, A., Hansen, T.H., Klenerman, P., Willberg, C.B., 2014.
 CD161++CD8+ T cells, including the MAIT cell subset, are specifically activated by IL-12+IL-18 in a TCR-independent manner. Eur. J. Immunol. 44, 195–203. https://doi.org/10.1002/eji.201343509

Vacchini, A., Chancellor, A., Spagnuolo, J., Mori, L., De Libero, G., 2020. MR1-

Restricted T Cells Are Unprecedented Cancer Fighters. Front. Immunol. https://doi.org/10.3389/fimmu.2020.00751

Van Vuuren, C.L., Wachter, G.G., Veenstra, R., Rijnhart, J.J.M., Van Der Wal, M.F., Chinapaw, M.J.M., Busch, V., 2019. Associations between overweight and mental health problems among adolescents, and the mediating role of victimization. BMC Public Health 19, 612. https://doi.org/10.1186/s12889-019-6832-z

Van Wilgenburg, B., Scherwitzl, I., Hutchinson, E.C., Leng, T., Kurioka, A.,
Kulicke, C., De Lara, C., Cole, S., Vasanawathana, S., Limpitikul, W.,
Malasit, P., Young, D., Denney, L., Moore, M.D., Fabris, P., Giordani, M.T.,
Oo, Y.H., Laidlaw, S.M., Dustin, L.B., Ho, L.P., Thompson, F.M.,
Ramamurthy, N., Mongkolsapaya, J., Willberg, C.B., Screaton, G.R.,
Klenerman, P., Barnes, E., Ball, J., Burgess, G., Cooke, G., Dillon, J.,
Gore, C., Foster, G., Guha, N., Halford, R., Herath, C., Holmes, C., Howe,
A., Hudson, E., Irving, W., Khakoo, S., Koletzki, D., Martin, N., Mbisa, T.,
McKeating, J., McLauchlan, J., Miners, A., Murray, A., Shaw, P.,
Simmonds, P., Spencer, C., Targett-Adams, P., Thomson, E., Vickerman,
P., Zitzmann, N., 2016. MAIT cells are activated during human viral
infections. Nat. Commun. 7, 1–11. https://doi.org/10.1038/ncomms11653

Vignali, D., Cantarelli, E., Bordignon, C., Canu, A., Citro, A., Annoni, A.,
Piemonti, L., Monti, P., 2018. Detection and characterization of CD8+ autoreactive memory Stem T cells in patients with type 1 diabetes.
Diabetes 67, 936–945. https://doi.org/10.2337/db17-1390

Vishvanath, L., Gupta, R.K., 2019. Contribution of adipogenesis to healthy adipose tissue expansion in obesity. J. Clin. Invest. https://doi.org/10.1172/JCI129191

- Vorkas, C.K., Wipperman, M.F., Li, K., Bean, J., Bhattarai, S.K., Adamow, M.,
 Wong, P., Aubé, J., Juste, M.A.J., Bucci, V., Fitzgerald, D.W., Glickman,
 M.S., 2018. Mucosal-associated invariant and γδ T cell subsets respond to
 initial Mycobacterium tuberculosis infection. JCI insight 3.
 https://doi.org/10.1172/jci.insight.121899
- Walker, L.J., Kang, Y.H., Smith, M.O., Tharmalingham, H., Ramamurthy, N.,
 Fleming, V.M., Sahgal, N., Leslie, A., Oo, Y., Geremia, A., Scriba, T.J.,
 Hanekom, W.A., Lauer, G.M., Lantz, O., Adams, D.H., Powrie, F., Barnes,
 E., Klenerman, P., 2012. Human MAIT and CD8αα cells develop from a

pool of type-17 precommitted CD8 + T cells. Blood 119, 422–433. https://doi.org/10.1182/blood-2011-05-353789

- Wallington, J.C., Williams, A.P., Staples, K.J., Wilkinson, T.M.A., 2018. IL-12 and IL-7 synergize to control mucosal-associated invariant T-cell cytotoxic responses to bacterial infection. J. Allergy Clin. Immunol. 141, 2182-2195.e6. https://doi.org/10.1016/j.jaci.2017.08.009
- Wang, H., D'Souza, C., Lim, X.Y., Kostenko, L., Pediongco, T.J., Eckle, S.B.G., Meehan, B.S., Shi, M., Wang, N., Li, S., Liu, L., Mak, J.Y.W., Fairlie, D.P., Iwakura, Y., Gunnersen, J.M., Stent, A.W., Godfrey, D.I., Rossjohn, J., Westall, G.P., Kjer-Nielsen, L., Strugnell, R.A., McCluskey, J., Corbett, A.J., Hinks, T.S.C., Chen, Z., 2018. MAIT cells protect against pulmonary Legionella longbeachae infection. Nat. Commun. 9, 1–15. https://doi.org/10.1038/s41467-018-05202-8
- Wang, H., Kjer-Nielsen, L., Shi, M., D'Souza, C., Pediongco, T.J., Cao, H.,
 Kostenko, L., Lim, X.Y., Eckle, S.B.G., Meehan, B.S., Zhu, T., Wang, B.,
 Zhao, Z., Mak, J.Y.W., Fairlie, D.P., Teng, M.W.L., Rossjohn, J., Yu, D., De
 St Groth, B.F., Lovrecz, G., Lu, L., McCluskey, J., Strugnell, R.A., Corbett,
 A.J., Chen, Z., 2019. IL-23 costimulates antigen-specific MAIT cell
 activation and enables vaccination against bacterial infection. Sci. Immunol.
 4. https://doi.org/10.1126/sciimmunol.aaw0402
- Wang, J., Zhan, Q., Zhang, L., 2016. A systematic review on the efficacy and safety of Infliximab in patients with psoriasis. Hum. Vaccines Immunother.
 12, 431–437. https://doi.org/10.1080/21645515.2015.1081322
- Wang, L.Y., Chyen, D., Lee, S., Lowry, R., 2008. The Association Between Body Mass Index in Adolescence and Obesity in Adulthood. J. Adolesc. Heal. 42, 512–518. https://doi.org/10.1016/j.jadohealth.2007.10.010
- Wang, R., Dillon, C.P., Shi, L.Z., Milasta, S., Carter, R., Finkelstein, D.,
 McCormick, L.L., Fitzgerald, P., Chi, H., Munger, J., Green, D.R., 2011.
 The Transcription Factor Myc Controls Metabolic Reprogramming upon T
 Lymphocyte Activation. Immunity 35, 871–882.
 https://doi.org/10.1016/j.immuni.2011.09.021
- Wang, Y., An, H., Liu, T., Qin, C., Sesaki, H., Guo, S., Radovick, S., Hussain,
 M., Maheshwari, A., Wondisford, F.E., O'Rourke, B., He, L., 2019.
 Metformin Improves Mitochondrial Respiratory Activity through Activation of
 AMPK. Cell Rep. 29, 1511-1523.e5.

https://doi.org/10.1016/j.celrep.2019.09.070

- Wang, Z., Aguilar, E.G., Luna, J.I., Dunai, C., Khuat, L.T., Le, C.T., Mirsoian, A., Minnar, C.M., Stoffel, K.M., Sturgill, I.R., Grossenbacher, S.K., Withers, S.S., Rebhun, R.B., Hartigan-O'Connor, D.J., Méndez-Lagares, G., Tarantal, A.F., Isseroff, R.R., Griffith, T.S., Schalper, K.A., Merleev, A., Saha, A., Maverakis, E., Kelly, K., Aljumaily, R., Ibrahimi, S., Mukherjee, S., Machiorlatti, M., Vesely, S.K., Longo, D.L., Blazar, B.R., Canter, R.J., Murphy, W.J., Monjazeb, A.M., 2019. Paradoxical effects of obesity on T cell function during tumor progression and PD-1 checkpoint blockade. Nat. Med. 25, 141–151. https://doi.org/10.1038/s41591-018-0221-5
- Warburg, O., 1925. The metabolism of carcinoma cells 1. J. Cancer Res. 9, 148–163. https://doi.org/10.1158/jcr.1925.148
- Ward, Z.J., Long, M.W., Resch, S.C., Giles, C.M., Cradock, A.L., Gortmaker, S.L., 2017. Simulation of Growth Trajectories of Childhood Obesity into Adulthood. N. Engl. J. Med. 377, 2145–2153. https://doi.org/10.1056/NEJMoa1703860
- Weinberg, F., Hamanaka, R., Wheaton, W.W., Weinberg, S., Joseph, J., Lopez, M., Kalyanaraman, B., Mutlu, G.M., Budinger, G.R.S., Chandel, N.S., 2010.
 Mitochondrial metabolism and ROS generation are essential for Krasmediated tumorigenicity. Proc. Natl. Acad. Sci. U. S. A. 107, 8788–8793. https://doi.org/10.1073/pnas.1003428107
- Weisberg, S.P., McCann, D., Desai, M., Rosenbaum, M., Leibel, R.L., Ferrante, A.W., 2003. Obesity is associated with macrophage accumulation in adipose tissue. J. Clin. Invest. 112, 1796–1808. https://doi.org/10.1172/jci19246
- Wen, L., Ley, R.E., Volchkov, P.Y., Stranges, P.B., Avanesyan, L., Stonebraker, A.C., Hu, C., Wong, F.S., Szot, G.L., Bluestone, J.A., Gordon, J.I., Chervonsky, A. V., 2008. Innate immunity and intestinal microbiota in the development of Type 1 diabetes. Nature 455, 1109–1113. https://doi.org/10.1038/nature07336Altman, B.J., Stine, Z.E., Dang, C. V., 2016. From Krebs to clinic: Glutamine metabolism to cancer therapy. Nat. Rev. Cancer. https://doi.org/10.1038/nrc.2016.71
- Babu, E., Kanai, Y., Chairoungdua, A., Kim, D.K., Iribe, Y., Tangtrongsup, S.,Jutabha, P., Li, Y., Ahmed, N., Sakamoto, S., Anzai, N., Nagamori, S.,Endou, H., 2003. Identification of a Novel System L Amino Acid Transporter

Structurally Distinct from Heterodimeric Amino Acid Transporters. J. Biol. Chem. 278, 43838–43845. https://doi.org/10.1074/jbc.M305221200

- Bodoy, S., Martín, L., Zorzano, A., Palacín, M., Estévez, R., Bertran, J., 2005.
 Identification of LAT4, a novel amino acid transporter with system L activity.
 J. Biol. Chem. 280, 12002–12011. https://doi.org/10.1074/jbc.M408638200
- Dimeloe, S., Burgener, A.V., Grählert, J., Hess, C., 2017. T-cell metabolism governing activation, proliferation and differentiation; a modular view. Immunology. https://doi.org/10.1111/imm.12655
- Dimeloe, S., Mehling, M., Frick, C., Loeliger, J., Bantug, G.R., Sauder, U.,
 Fischer, M., Belle, R., Develioglu, L., Tay, S., Langenkamp, A., Hess, C.,
 2016. The Immune-Metabolic Basis of Effector Memory CD4 + T Cell
 Function under Hypoxic Conditions . J. Immunol. 196, 106–114.
 https://doi.org/10.4049/jimmunol.1501766
- Khader, S.A., Divangahi, M., Hanekom, W., Hill, P.C., Maeurer, M., Makar,
 K.W., Mayer-Barber, K.D., Mhlanga, M.M., Nemes, E., Schlesinger, L.S.,
 Van Crevel, R., Vankalayapati, R., Xavier, R.J., Netea, M.G., 2019.
 Targeting innate immunity for tuberculosis vaccination. J. Clin. Invest.
 https://doi.org/10.1172/JCI128877
- Munford, H., Dimeloe, S., 2019. Intrinsic and Extrinsic Determinants of T Cell Metabolism in Health and Disease. Front. Mol. Biosci. https://doi.org/10.3389/fmolb.2019.00118
- O'Neill, L.A.J., Kishton, R.J., Rathmell, J., 2016. A guide to immunometabolism for immunologists. Nat. Rev. Immunol. https://doi.org/10.1038/nri.2016.70
- Owen, M.R., Doran, E., Halestrap, A.P., 2000. Evidence that metformin exerts its anti-diabetic effects through inhibition of complex 1 of the mitochondrial respiratory chain. Biochem. J. 348, 607–614. https://doi.org/10.1042/bj3480607
- Wang, Y., An, H., Liu, T., Qin, C., Sesaki, H., Guo, S., Radovick, S., Hussain,
 M., Maheshwari, A., Wondisford, F.E., O'Rourke, B., He, L., 2019.
 Metformin Improves Mitochondrial Respiratory Activity through Activation of
 AMPK. Cell Rep. 29, 1511-1523.e5.
 https://doi.org/10.1016/j.celrep.2019.09.070
- Wheaton, W.W., Weinberg, S.E., Hamanaka, R.B., Soberanes, S., Sullivan,L.B., Anso, E., Glasauer, A., Dufour, E., Mutlu, G.M., Scott Budigner, G.R.,Chandel, N.S., 2014. Metformin inhibits mitochondrial complex I of cancer

cells to reduce tumorigenesis. Elife 2014. https://doi.org/10.7554/eLife.02242

- Yan, R., Zhou, J., Li, Y., Lei, J., Zhou, Q., 2020. Structural insight into the substrate recognition and transport mechanism of the human LAT2–4F2hc complex. Cell Discov. https://doi.org/10.1038/s41421-020-00207-4
- Yin, Y., Choi, S.C., Xu, Z., Perry, D.J., Seay, H., Croker, B.P., Sobel, E.S., Brusko, T.M., Morel, L., 2015. Normalization of CD4+ T cell metabolism reverses lupus. Sci. Transl. Med. 7, 274ra18. https://doi.org/10.1126/scitranslmed.aaa0835
- Wheaton, W.W., Weinberg, S.E., Hamanaka, R.B., Soberanes, S., Sullivan,
 L.B., Anso, E., Glasauer, A., Dufour, E., Mutlu, G.M., Scott Budigner, G.R.,
 Chandel, N.S., 2014. Metformin inhibits mitochondrial complex I of cancer
 cells to reduce tumorigenesis. Elife 2014.
 https://doi.org/10.7554/eLife.02242
- White, U., Ravussin, E., 2019. Dynamics of adipose tissue turnover in human metabolic health and disease. Diabetologia. https://doi.org/10.1007/s00125-018-4732-x
- WHO, 2020a. Obesity and Overweight [WWW Document]. Obes. overweight. URL https://www.who.int/news-room/fact-sheets/detail/obesity-andoverweight
- WHO, 2020b. Mean fasting blood glucose [WWW Document]. URL https://www.who.int/data/gho/indicator-metadata-registry/imr-details/2380 (accessed 12.24.20).
- WHO, 2017. Weight bias and obesity stigma: considerations for the WHO European Region. Copenhagen.
- Wieman, H.L., Wofford, J.A., Rathmell, J.C., 2007. Cytokine stimulation promotes glucose uptake via phosphatidylinositol-3 kinase/Akt regulation of Glut1 activity and trafficking. Mol. Biol. Cell 18, 1437–1446. https://doi.org/10.1091/mbc.E06-07-0593
- Wilgenburg, B. van, Loh, L., Chen, Z., Pediongco, T.J., Wang, H., Shi, M.,
 Zhao, Z., Koutsakos, M., Nüssing, S., Sant, S., Wang, Z., D'Souza, C., Jia,
 X., Almeida, C.F., Kostenko, L., Eckle, S.B.G., Meehan, B.S., Kallies, A.,
 Godfrey, D.I., Reading, P.C., Corbett, A.J., McCluskey, J., Klenerman, P.,
 Kedzierska, K., Hinks, T.S.C., 2018. MAIT cells contribute to protection
 against lethal influenza infection in vivo. Nat. Commun. 9, 1–9.

https://doi.org/10.1038/s41467-018-07207-9

- Willing, A., Jäger, J., Reinhardt, S., Kursawe, N., Friese, M.A., 2018. Production of IL-17 by MAIT Cells Is Increased in Multiple Sclerosis and Is Associated with IL-7 Receptor Expression. J. Immunol. 200, 974–982. https://doi.org/10.4049/jimmunol.1701213
- Willing, A., Leach, O.A., Ufer, F., Attfield, K.E., Steinbach, K., Kursawe, N., Piedavent, M., Friese, M.A., 2014. CD8 ⁺ MAIT cells infiltrate into the CNS and alterations in their blood frequencies correlate with IL-18 serum levels in multiple sclerosis. Eur. J. Immunol. 44, 3119–3128. https://doi.org/10.1002/eji.201344160
- Winer, S., Chan, Y., Paltser, G., Truong, D., Tsui, H., Bahrami, J., Dorfman, R., Wang, Y., Zielenski, J., Mastronardi, F., Maezawa, Y., Drucker, D.J., Engleman, E., Winer, D., Dosch, H.M., 2009. Normalization of obesityassociated insulin resistance through immunotherapy. Nat. Med. 15, 921– 929. https://doi.org/10.1038/nm.2001
- Wolf, A.M., Wolf, D., Rumpold, H., Enrich, B., Tilg, H., 2004. Adiponectin induces the anti-inflammatory cytokines IL-10 and IL-1RA in human leukocytes. Biochem. Biophys. Res. Commun. 323, 630–635. https://doi.org/10.1016/j.bbrc.2004.08.145
- Wolfson, R.L., Chantranupong, L., Saxton, R.A., Shen, K., Scaria, S.M., Cantor, J.R., Sabatini, D.M., 2016. Sestrin2 is a leucine sensor for the mTORC1 pathway. Science (80-.). 351, 43–48. https://doi.org/10.1126/science.aab2674
- Won, E.J., Ju, J.K., Cho, Y.N., Jin, H.M., Park, K.J., Kim, T.J., Kwon, Y.S., Kee, H.J., Kim, J.C., Kee, S.J., Park, Y.W., 2016. Clinical relevance of circulating mucosal-associated invariant T cell levels and their anti-cancer activity in patients with mucosal-associated cancer. Oncotarget 7, 76274–76290. https://doi.org/10.18632/oncotarget.11187
- Wong, E.B., Gold, M.C., Meermeier, E.W., Xulu, B.Z., Khuzwayo, S., Sullivan,
 Z.A., Mahyari, E., Rogers, Z., Kløverpris, H., Sharma, P.K., Worley, A.H.,
 Lalloo, U., Baijnath, P., Ambaram, A., Naidoo, L., Suleman, M., Madansein,
 R., McLaren, J.E., Ladell, K., Miners, K.L., Price, D.A., Behar, S.M.,
 Nielsen, M., Kasprowicz, V.O., Leslie, A., Bishai, W.R., Ndung'u, T.,
 Lewinsohn, D.M., 2019. TRAV1-2+ CD8+ T-cells including oligoconal
 expansions of MAIT cells are enriched in the airways in human

tuberculosis. Commun. Biol. 2. https://doi.org/10.1038/s42003-019-0442-2

Woods, C., Powell, C., Saunders, J., O'Brien, W., Murphy, M., C, D., O, F., A, J., S, C., S, B., 2018. The Children's Sport Participation and Physical Activity Study 2018 ICSPPA 2019).

Wrann, C.D., Laue, T., Hübner, L., Kuhlmann, S., Jacobs, R., Goudeva, L., Nave, H., 2012. Short-term and long-term leptin exposure differentially affect human natural killer cell immune functions. Am. J. Physiol. -Endocrinol. Metab. 302. https://doi.org/10.1152/ajpendo.00057.2011

- Wu, C., Xue, Y., Wang, P., Lin, L., Liu, Q., Li, N., Xu, J., Cao, X., 2014. IFN-γ
 Primes Macrophage Activation by Increasing Phosphatase and Tensin
 Homolog via Downregulation of miR-3473b. J. Immunol. 193, 3036–3044.
 https://doi.org/10.4049/jimmunol.1302379
- Wu, H., Ballantyne, C.M., 2020. Metabolic Inflammation and Insulin Resistance in Obesity. Circ. Res. 1549–1564. https://doi.org/10.1161/CIRCRESAHA.119.315896
- Wu, M.-Z., Chen, Y., Zou, Y., Zhen, Z., Yu, Y.-J., Liu, Y.-X., Yuen, M., Ho, L.M., Siu-Ling Lam, K., Tse, H.-F., Yiu, K.-H., 2019. Impact of Obesity on Longitudinal Changes to Cardiac Structure and Function in Patients With Type 2 Diabetes Mellitus. Eur Hear. J Cardiovasc Imaging 20, 816–827.
- Yamaguchi, H., Hashimoto, K., 2002. Association of MR1 protein, an MHC class I-related molecule, with β2-microglobulin. Biochem. Biophys. Res. Commun. 290, 722–729. https://doi.org/10.1006/bbrc.2001.6277
- Yan, R., Zhou, J., Li, Y., Lei, J., Zhou, Q., 2020. Structural insight into the substrate recognition and transport mechanism of the human LAT2–4F2hc complex. Cell Discov. https://doi.org/10.1038/s41421-020-00207-4
- Yang, K., Shrestha, S., Zeng, H., Karmaus, P.W.F., Neale, G., Vogel, P., Guertin, D.A., Lamb, R.F., Chi, H., 2013. T Cell Exit from Quiescence and Differentiation into Th2 Cells Depend on Raptor-mTORC1-Mediated Metabolic Reprogramming. Immunity 39, 1043–1056. https://doi.org/10.1016/j.immuni.2013.09.015
- Yin, Y., Choi, S.C., Xu, Z., Perry, D.J., Seay, H., Croker, B.P., Sobel, E.S., Brusko, T.M., Morel, L., 2015. Normalization of CD4+ T cell metabolism reverses lupus. Sci. Transl. Med. 7, 274ra18. https://doi.org/10.1126/scitranslmed.aaa0835

Zatterale, F., Longo, M., Naderi, J., Raciti, G.A., Desiderio, A., Miele, C.,

Beguinot, F., 2020. Chronic Adipose Tissue Inflammation Linking Obesity to Insulin Resistance and Type 2 Diabetes. Front. Physiol. 10, 1607. https://doi.org/10.3389/fphys.2019.01607

- Zhang, M., Ming, S., Gong, S., Liang, S., Luo, Y., Liang, Z., Cao, C., Lao, J., Shang, Y., Li, X., Wang, M., Zhong, G., Xu, L., Wu, M., Wu, Y., 2019.
 Activation-Induced Cell Death of Mucosal-Associated Invariant T Cells Is Amplified by OX40 in Type 2 Diabetic Patients. J. Immunol. 203, 2614– 2620. https://doi.org/10.4049/jimmunol.1900367
- Zorena, K., Jachimowicz-Duda, O., Ślęzak, D., Robakowska, M., Mrugacz, M., 2020. Adipokines and Obesity. Potential Link to Metabolic Disorders and Chronic Complications. Int. J. Mol. Sci. 21, 3570. https://doi.org/10.3390/ijms21103570

Appendix

Obesity Reduces mTORC1 Activity in Mucosal-Associated Invariant T Cells, Driving Defective Metabolic and Functional Responses

Aisling O'Brien,^{*,1} Roisin M. Loftus,^{†,1} Marta M. Pisarska,^{‡,§,1} Laura M. Tobin,^{*,‡} Ronan Bergin,[§] Nicole A. W. Wood,^{†,§} Cathriona Foley,[†] Arimin Mat,^{*} Frances C. Tinley,[§] Ciaran Bannan,[†] Gary Sommerville,[¶] Natacha Veerapen,[∥] Gurdyal S. Besra,[∥] Linda V. Sinclair,[#] Paul N. Moynagh,^{§,**} Lydia Lynch,[†] David K. Finlay,^{†,††,‡‡} Donal O'Shea,^{*,‡,§,2} and Andrew E. Hogan^{*,‡,§,2}

Obesity underpins the development of numerous chronic diseases, such as type II diabetes mellitus. It is well established that obesity negatively alters immune cell frequencies and functions. Mucosal-associated invariant T (MAIT) cells are a population of innate T cells, which we have previously reported are dysregulated in obesity, with altered circulating and adipose tissue frequencies and a reduction in their IFN-γ production, which is a critical effector function of MAIT cells in host defense. Hence, there is increased urgency to characterize the key molecular mechanisms that drive MAIT cell effector functions and to identify those which are impaired in the obesity setting. In this study, we found that MAIT cells significantly upregulate their rates of glycolysis upon activation in an mTORC1-dependent manner, and this is essential for MAIT cell IFN-γ production. Furthermore, we show that mTORC1 activation is dependent on amino acid transport via SLC7A5. In obese patients, using RNA sequencing, Seahorse analysis, and a series of in vitro experiments, we demonstrate that MAIT cells isolated from obese adults display defective glycolytic metabolism, mTORC1 signaling, and SLC7A5 aa transport. Collectively, our data detail the intrinsic metabolic pathways controlling MAIT cell cytokine production and highlight mTORC1 as an important metabolic regulator that is impaired in obesity, leading to altered MAIT cell responses. *The Journal of Immunology*, 2019, 202: 3404–3411.

besity is a global epidemic, impacting over 600 million adults and 150 million children (1). Obesity is linked to numerous chronic diseases, including type II diabetes, cardiovascular disease, and certain cancers (2). Underpinning the obesity-driven development of chronic disease is systemic inflammation and immune cell dysregulation (3, 4). We have previously reported the negative impact of obesity on several immune cell populations, including invariant NK T cells, NK cells, dendritic cells, and mucosal-associated invariant T (MAIT) cells (5–8).

¹These authors contributed equally to the manuscript.

Received for publication December 7, 2018. Accepted for publication April 12, 2019.

This work was supported by the National Children's Research Centre. A.E.H. and D.O. are funded by the Health Research Board (HRB) (Grant RA-POR-2014-593), and A.O. is funded by the HRB Centre for Health and Diet Research (Grant

MAIT cells are a population of non-MHC–restricted T cells that are important in the immune defense against bacterial and viral infections. MAIT cells are early-responding T cells that are capable of rapidly producing multiple cytokines upon activation, such as IFN- γ , TNF- α , and IL-17 (9). MAIT cells are activated when their invariant TCR recognize bacterial derivatives presented on the MHC-like molecule MR1 (9, 10). MAIT cells can also be activated in a TCR-independent manner via cytokine stimulation (11). Dysregulated MAIT cell cytokine profiles have been reported in several diseases, including obesity, arthritis, and viral

HRC2014/13), both of which are funded by the Irish HRB. M.M.P., N.A.W.W., and A.E.H. are supported by the National Children's Research Centre. The Seahorse XF-96 Analyzer is funded by Science Foundation Ireland Grant 16/RI/3399.

The sequences presented in this article have been submitted to the Gene Expression Omnibus under accession number GSE126169.

Address correspondence and reprint requests to Dr. Andrew E. Hogan, Institute of Immunology, Maynooth University, Maynooth, County Kildare, Ireland. E-mail address: Andrew.E.Hogan@mu.ie

The online version of this article contains supplemental material.

Abbreviations used in this article: BCH, 2-aminobicyclo- (2, 2, 1)-heptane-2-carboxylic acid; 2DG, 2-deoxyglucose; ECAR, extracellular acidification rate; FMO, flow minus-1; MAIT, mucosal-associated invariant T; OxPhos, oxidative phosphorylation; ps6, ribosomal S6 protein; RNA-Seq, RNA sequencing.

Copyright © 2019 by The American Association of Immunologists, Inc. 0022-1767/19/\$37.50

^{*}Obesity Immunology Group, Education and Research Centre, St. Vincent's University Hospital, University College Dublin, Dublin 4, Ireland, [†]School of Biochemistry and Immunology, Trinity Biomedical Sciences Institute, Trinity College, Dublin 2, Ireland; [†]National Children's Research Centre, Dublin 12, Ireland; [†]Department of Biology, Institute of Immunology, Maynooth University, Maynooth, County Kildare W23 F2K8, Ireland; [†]Dana Farber Cancer Institute, Molecular Biology Core Facilities, Boston, MA 02215; [#]School of Biosciences, University of Birmingham, Birmingham B15 2TT, United Kingdom; [#]Division of Cell Signaling and Immunology, School of Life Sciences, University of Dundee, Dundee DD1 5EH, United Kingdom; ^{*†}School of Medicine, Dentistry and Biomedical Sciences, Wellcome–Wolfson Institute for Experimental Medicine, Queen's University, Belfast BT9 7BL, United Kingdom; ^{††}School of Biochemistry and Immunology, Trinity Biomedical Sciences Institute, Trinity College, Dublin 4, Ireland; ^{**}School of Pharmacy and Pharmaceutical Sciences, Trinity College, Dublin 2, Ireland

²Joint senior authorship.

ORCIDs: 0000-0001-9137-6625 (R.M.L.); 0000-0003-4685-7176 (R.B.); 0000-0001-9046-2417 (C.F.); 0000-0001-9066-7957 (A.M.); 0000-0002-9288-8574 (C.B.); 0000-0003-4182-4236 (N.V.); 0000-0002-5605-0395 (G.S.B.); 0000-0003-1248-7189 (L.V.S.); 0000-0003-2716-6679 (D.K.F.).

A.O., R.M.L., M.M.P., L.M.T., N.A.W.W., F.C.T., R.B., and C.B. performed the experiments and carried out analysis and approved the final manuscript as submitted. A.M. recruited patients and controls. N.V. and G.S.B. provided MAIT cell reagents and aided in the design of MAIT cell activation experiments. G.S. and C.F. performed and analyzed the RNA sequencing. A.E.H., L.V.S., P.N.M., D.K.F., R.M.L., L.L., and D.O. conceptualized and designed the study, analyzed the data, drafted the manuscript as submitted.

infection (8, 12–15). The pathways controlling MAIT cell cytokine responses are still under investigation.

It is now well established that cellular metabolism plays a critical role in determining immune cell responses (16). Extrinsic and intrinsic signals result in immune cell metabolic reprogramming, which provides both the energy and biological intermediates required for a robust immune response. Although different lymphocyte subsets can adopt distinct metabolic signatures, in general, effector lymphocytes are characterized by elevated glucose metabolism with high rates of glycolysis and oxidative phosphorylation (OxPhos) (16).

Currently, there is a paucity of data regarding the metabolic profile of MAIT cells or the importance of cellular metabolism in driving MAIT cell effector functions. In this study, we show that MAIT cells use glycolytic metabolism upon activation. Importantly, we show the requirement of this elevated rate of MAIT cell glycolysis for cytokine production. Using RNA sequencing (RNA-Seq) and Seahorse analysis, we show impaired glycolytic metabolism in MAIT cells from obese individuals. We show that MAIT cell glycolysis is dependent on mTORC1 activation, and in obesity mTORC1 activation is reduced. Finally, we show that mTORC1 activation in MAIT cells is dependent on amino acid transport via SLC7A5. Collectively, our data show for the first time, to our knowledge, the intrinsic metabolic pathways controlling MAIT cell cytokine production. Furthermore, we implicate a novel amino acid-dependent regulatory mechanism that is concurrent with impaired MAIT cell functions in obesity.

Materials and Methods

Study cohorts

A cohort of 30 obese adults (mean body mass index 45.1) was enrolled into this study, along with an age- and sex-matched cohort of healthy lean controls (mean body mass index 24.7) (Supplemental Table I). Exclusion criteria included patients with comorbid inflammatory conditions outside of type II diabetes. Patients who smoke or are on anti-inflammatory medications, GLP-1 therapies, or sodium-glucose cotransporter 2 inhibitors were also excluded.

Preparation of PBMCs and flow cytometric analysis

PBMC samples were isolated by density centrifugation from fresh venous blood samples. MAIT cell staining (1×10^6 PBMC) was performed using specific surface mAbs (all Miltenyi Biotec), namely, CD3, CD161, CD8, and TCRV α 7.2 (Supplemental Fig. 3). Cell populations were acquired using a BD FACSCanto II and analyzed using FlowJo software (Tree Star). Results are expressed as a percentage of the parent population, as indicated, and determined using flow minus-1 (FMO) and unstained controls.

MAIT cell RNA-Seq and analysis

Highly pure MAIT cells were isolated from peripheral blood samples using FACS (MoFlo XDP Sorter). RNA was extracted from cells using the RNeasy Mini Isolation Kit (Qiagen) and analyzed for quality control and quantification purposes using the Agilent Bioanalyzer and RNA Pico assay (Agilent Technologies). RNA-Seq libraries were prepared using the SMART-Seq v4 Ultra Low Input RNA Kit (Clontech Laboratories), and next-generation sequencing was carried out using Illumina NextSeq 500 Single-End 75 bp (SE75) chemistry. Single-end RNA-Seq reads were aligned to the human hg38 reference genome using Tophat2 (17) and Bowtie (18) and the Ensembl gene annotation downloaded from the BioMart Web site. The Qlucore Omics Explorer 3.2 software package was used for visualization of RNA-Seq data. Qlucore Omics software converted imported bam files into read counts and then normalized these across the samples using the trimmed mean of M-values method (19). Principal component analysis was performed using the default setting: mean = 0, variance = 1 normalization based on 503 genes that passed filtering by variance set at the level of 0.25. Gene set enrichment analysis ranked genes based on the correlation between their expression, and the normalized enrichment score reflected the degree to which Hallmark gene sets from the Molecular Signatures Database were overrepresented from the ranked gene list (20). Differential gene expression analysis was implemented in the Qlucore, which performed a two-group comparison between lean and obese MAIT cells. Genes downregulated in obesity within the glycolysis and MTORC1 gene sets with a p value <0.05 and \log_2 fold change >1.5 were represented in the heat maps generated in Qlucore. https://www.ncbi.nlm.nih.gov/geo/query/acc.cgi?acc=GSE126169.

MAIT cell cytokine production analysis

MAIT cell cytokine production was determined by intracellular flow cytometry. Briefly, PBMC (1×10^6) were cultured in the presence of a protein transport inhibitor in media alone (control) or stimulated with either TCR microbeads (Miltenyi Biotec), IL-12, IL-18, IL-12/18 (all BioLegend), or a combination of TCR microbeads and IL-12/IL-18 for 18 h at 37°C. Additionally, metabolic inhibitors were added as indicated: rapamycin (20 nM), 2-deoxyglucose (2DG) (10 μ M), oligomycin (2 μ M), or 2-aminobicyclo- (2, 2, 1)-heptane-2-carboxylic acid (BCH) (50 mM). For acute glucose restriction experiments, PBMC were stimulated in media containing less than 1 mM glucose for 18 h, after which cells were investigated for intracellular cytokine production by flow cytometry, as previously described.

MAIT cell cytokine production by ELISpot

MAIT cell cytokine production was also determined by ELISpot. Briefly, 96-well ELISpot plate(s) were precoated with mAbs to IFN- γ overnight at 4°C. Plates were washed in PBS to remove coating Ab and incubated at 37°C for 1 h in complete RPMI. Previously isolated MAIT cells (2 × 10⁵) were cultured on the precoated plates either basally or stimulated with IL-12/IL-18 and TCR microbeads at 37°C for 18 h. ELISpot plates were then washed with PBS/Tween for a minimum of six washes. Following this, 50 µl of biotinylated mouse anti-human IFN- γ was added and incubated at room temperature for 3 h. ELISpot plates were washed in wash buffer, and antibiotin-labeled Ab was added to the wells and incubated at room temperature for 2 h. Finally, prefiltered BCIP/NBT solution (50 µl) was added to each well and incubated at room temperature for 5 min or until distinct dark spots appeared in the positive-control wells. Plates were mash analyzed using a semiautomated ELISpot plate reader.

MAIT cell glucose consumption analysis

Fresh PBMC (1×10^6 ml) were activated as previously described for 18 h; cells were then washed and transferred to glucose-free media in the presence of the fluorescent glucose analogue 2NBDG (Life Technologies) for 2 h. Additionally, glut 1 expression was examined on basal and activated MAIT cells using an mAb specific for Glut-1 (R&D Systems). FMO and unstimulated PBMC were used as negative controls.

MAIT cell Seahorse analysis

For real-time analysis of the extracellular acidification rate (ECAR) and oxygen consumption rate of purified MAIT cells cultured under various conditions, a Seahorse XF-96 Analyzer (Seahorse Bioscience) was used. In brief, 200,000 purified, expanded MAIT cells were adhered to a CellTaq (BD Pharmingen)–coated 96-well XF Cell Culture Microplate (Seahorse Bioscience). Sequential measurements of ECAR and oxygen consumption rate following addition of the inhibitors (Sigma-Aldrich) oligomycin (2 μ M), FCCP (0.5 μ M), rotenone (100 nM) plus antimycin A (4 μ M), and 2DG, (30 mM) allowed for the calculation of basal glycolysis, glycolytic capacity, basal OxPhos, and maximal mitochondrial capacity. Each cell culture condition was evaluated in quadruplicate, and 14 measurements were made per sample.

MAIT cell mTOR analysis

PBMC (1 \times 10⁶ ml) were activated as previously described for 18 h. mTOR activity in MAIT cells and CD8⁺ T cells was examined via the phosphorylation of ribosomal S6 protein (pS6) (Cell Signaling Technology) by flow cytometry. Rapamycin treatment was used as a control for pS6 specificity and to determine the requirement of mTORC1 for IFN- γ production. FMO and unstimulated PBMC cells were used as negative controls.

MAIT cell amino acid uptake and BCH experiments

PBMC or MAIT cells were used to investigate kynurenine uptake as described in (21). Briefly, after surface Ab staining of samples, cells were resuspended in 200 ml of warmed HBSS (1×10^{6} cells) and incubated in a water bath at 37°C. Kynurenine (200 mM in HBSS), BCH (20 mM in HBSS), and HBSS were warmed to 37°C and added as appropriate. Uptake was stopped after 4 min by PFA (final concentration 1%) for 30 min at room temperature in the dark. After fixation, cells were washed twice in
PBS/0.5% BSA and resuspend in PBS/0.5% BSA prior to acquisition on flow cytometer. The 405-nm laser and 450/50 BP filter were used for kynurenine fluorescence detection. For SLC7A5 inhibition experiments, the concentration of amino acids in RPMI 1640 was diluted two-fold using HBSS (Invitrogen) in the presence or absence of BCH (50 mM; Sigma-Aldrich).

Statistics

Statistical analysis was completed using GraphPad Prism 6 Software. Data are expressed as SEM. We determined differences between two groups using Student *t* test and Mann–Whitney *U* test where appropriate. Analysis across three or more groups was performed using ANOVA. Correlations were determined using linear regression models and expressed using Pearson or Spearman rank correlation coefficient, as appropriate. The *p* values were expressed with significance set at <0.05.

Study approval

Ethical approval was granted by the Medical Research Ethics Committees at St. Vincent's University Hospital. All patients gave written informed consent prior to partaking in the study.

Results

Increased rates of glycolysis are essential for driving MAIT cell IFN- γ production

MAIT cells are potent producers of cytokine (IFN- γ and TNF- α among the most abundant). We show that in a stimulationdependent manner, a large proportion of MAIT cells produce IFN- γ and TNF- α but not IL-17, with the most robust response seen after stimulation with anti-CD3/CD28–coated beads (TCR beads) and the cytokines IL-12/IL-18 (Fig. 1A, Supplemental Figs. 1A, 3). The metabolic pathways used by MAIT cells during their cytokine production are currently undefined. We show that MAIT cells express the major glucose transporter Glut-1 and upregulate the uptake of the fluorescent 2DG analogue 2NBDG upon activation in a stimulus-dependent manner (Fig. 1B-D). These data suggested that MAIT cells upregulate glucose metabolism. Therefore, we next performed detailed metabolic analysis using the Seahorse extracellular flux analyzer. Interestingly, we found MAIT cells specifically upregulate their rates of glycolysis (Fig. 1E, 1F), but not OxPhos (Supplemental Fig. 2B), upon activation. In addition, the glycolytic capacity, ECAR following oligomycin injection, of MAIT cells also increased significantly upon stimulation, which suggests that these cells are increasing the expression of the glycolytic machinery (Fig. 1E). Therefore, overall there was a pronounced shift toward glycolytic metabolism in activated MAIT cells. To assess if cellular glycolysis was required for cytokine production by MAIT cells, we used low doses of the specific glycolytic inhibitor 2DG. The data showed that MAIT cell IFN- γ production was significantly reduced when the rate of glycolysis was limited with 2DG (Fig. 1G, 1H). Interestingly, 2DG treatment had no impact on TNF- α production by MAIT cells (Fig. 1I). In contrast, inhibition of mitochondrial respiration with the ATP synthase inhibitor oligomycin did not reduce IFN-y production by MAIT cells (Supplemental Fig. 2D, 2E) but did reduce TNF- α (Supplemental Fig. 2F). To further test the requirement for glucose metabolism in driving MAIT cell cytokine responses, we cultured MAIT cells under acute glucose restriction and showed a strong inhibition of IFN- γ , further



FIGURE 1. MAIT cells use glycolytic metabolism during IFN- γ production. (**A**) Scatter plot showing MAIT cell production of IFN- γ in a cohort of lean adults (unstimulated versus stimulated with TCR microbead/IL-12/IL-18). (**B**) Scatter plot showing Glut-1 expression (MFI) on either basal or stimulated MAIT cells from lean adults (n = 10). (**C** and **D**) Representative histograms and scatter plot showing 2NBDG uptake by MAIT cells stimulated with either TCR microbeads, IL-12, IL-18, both IL-12 and IL-18, or TCR microbeads and IL-12/IL-18 (n = 3). (**E** and **F**) Scatter plot and representative Seahorse traces displaying MAIT cells ECAR rates in either basal (open circles) or stimulated (gray circles) MAIT cells from lean adults. Data representative of six independent experiments. (**G**) Representative ELISpot wells and scatter plot displaying the effect of 2DG treatment on IFN- γ production by stimulated MAIT cells (n = 10). (**J**) Scatter plot displaying the effect of a adults (n = 5). Data representative of five independent experiments. Statistical comparisons using Student *t* test or ANOVA. Data representative of a minimum of 10 independent experiments unless otherwise stated. Error bars represent SEM. *p < 0.05, ** $p \le 0.01$, ***p < 0.001. MFI, mean fluorescence intensity.

supporting our findings that MAIT cell IFN- γ production is dependent on glycolytic metabolism (Fig. 1J).

Defective IFN- γ production by MAIT cells in obesity parallels defective cellular metabolism

We have previously reported defective MAIT cells in obese individuals, including impaired IFN- γ production, which we confirm in an expanded cohort of obese adults (Fig. 2A). Interestingly, in our obese cohort, we did not observe defective CD8⁺ T cell IFN- γ production (Fig. 2B). To investigate the mechanism(s) driving defective IFN- γ production by MAIT cells in obesity, we performed RNA-Seq of MAIT cells isolated from five lean and four obese individuals. Normal enrichment score analysis highlighted a reduction in the glycolytic metabolism pathway (Fig. 2C). To investigate the significance of the reduced glycolytic metabolism in obesity, Glut-1 expression and 2NBDG uptake assays were performed. The data showed that MAIT cells from obese adults show no significant difference in Glut-1 expression (Fig. 2D). However, there was a significant decrease in the uptake of 2NBDG in stimulated MAIT cells from obese individuals when compared with healthy adults (Fig. 2E). Detailed metabolic analysis was used to confirm the alterations in MAIT cell glycolytic metabolism between healthy and obese MAIT cells. In line with the reduced glycolytic gene expression and glucose uptake, activated MAIT cells from obese adults do not increase their rate of glycolysis (Fig. 2F, 2G). Indeed, MAIT cells from obese adults show reduced glycolytic capacity when compared with lean adults (Fig. 2H). Furthermore, acute glucose restriction had no additional impact/reduction on IFN- γ production in MAIT cells from obese individuals in comparison with the significant effect observed in MAIT cells from lean individuals (Fig. 2I). These data confirm that MAIT cells from obese adults are not performing increased rates of glycolysis during their activation.

mTORC1 activity is essential for driving MAIT cell glycolysis and is reduced in obesity

The signaling complex mTORC1 acts as both a nutrient sensor and a central metabolic regulator of effector immune cells. Studies have shown that mTORC1 is intrinsically linked with glycolytic metabolism in effector lymphocytes (22, 23). Strikingly, our RNA-Seq data revealed defective mTORC1 signaling. (Fig. 3A). To investigate a role for mTORC1 in driving MAIT cell metabolism, we measured mTORC1 activity in MAIT cells upon stimulation and demonstrate a robust increase as measured via the phosphorylation of pS6 (Fig. 3B, 3C). Importantly, these elevated levels of pS6 were prevented by the mTORC1 inhibitor rapamycin, demonstrating the specificity of this assay for mTORC1 signaling (Fig. 3D). We next investigated if mTORC1 signaling is required for increased rates of glycolysis in activated MAIT cells. Indeed, MAIT cells stimulated in the presence of rapamycin had significantly lower levels of glycolysis (Fig. 3E, 3F). As expected, rapamycin treatment had no significant impact on the ECAR of MAIT cells



FIGURE 2. MAIT cells from obese adults display defective glycolytic metabolism. (**A**) Scatter plot displaying MAIT cell production of IFN- γ in a cohort of lean (white circles) and obese (gray circles) adult (unstimulated versus stimulated with TCR microbead/IL-12/IL-18) (n = 15). (**B**) Scatter plot displaying CD8 T cell production of IFN- γ in a cohort of lean (white circles) and obese (gray circles) adult (unstimulated versus stimulated with TCR microbead/IL-12/IL-18) (n = 15). (**C**) Heat map depicting differentially downregulated genes (p value < 0.05) within the glycolysis gene set in MAIT cells isolated from lean and obese donors (n = 5 and 4, respectively). (**D**) Scatter plot showing Glut-1 MFI on either basal or stimulated MAIT cells from lean (white circles) or obese (gray circles) donors (n = 10). (**E**) Scatter plot showing 2NBDG MFI on either basal or stimulated MAIT cells from lean (white circles) or obese (gray circles) donors. (**F**) Representative Seahorse trace displaying MAIT cell ECAR rates in either basal (open circles) or stimulated (gray circles) or glycolytic capacity (ECAR post–oligomycin treatment) from lean (white circles) or obese (gray circles) donors (n = 5). (**C**) Production by stimulated MAIT cells from lean (white circles) or obese (gray circles) donors. (**F**) Representative stimulated lean donor (black circles). (**G** and **H**) Scatter plot displaying glycolysis levels (ECAR) or glycolytic capacity (ECAR post–oligomycin treatment) from lean (white circles) or obese (gray circles) donors (n = 6). (**I**) Scatter plot showing impact of a unimum of five independent experiments unless otherwise stated. Statistical comparisons using Student *t* test or ANOVA. Error bars represent SEM. *p < 0.05, ** $p \le 0.01$. MFI, mean fluorescence intensity.



FIGURE 3. MAIT cell glycolytic metabolism is dependent on mTORC1 activation, which displays defects in obesity. (**A**) Heat map depicting differentially downregulated genes (*p* value < 0.05) within the mTORC1 gene set in MAIT cells isolated from lean and obese donors (n = 5 and 4, respectively). (**B** and **C**) Representative histogram and scatter plot showing the activation of mTORC1 (phosphorylation of S6) in MAIT cells stimulated with either TCR microbeads, IL-12, IL-18, both IL-12 and IL-18, or TCR microbeads, IL-12 and IL-18 (n = 3). (**D**) Representative histogram showing the inhibition of S6 phosphorylation by rapamycin in stimulated MAIT cells. (**E**) Representative Seahorse traces displaying MAIT cell ECAR rates in either basal (open circles), stimulated (gray circles), or rapamycin-treated (open squares) in MAIT cells from lean donors. (**F**) Scatter plot displaying ECAR levels in stimulated obese MAIT cells. (**H**) Scatter plot detailing effect of rapamycin on ECAR rates in stimulated obese MAIT cells. (**H**) Scatter plot detailing effect of rapamycin on IFN- γ production by stimulated MAIT cells from lean donors. (**J**) Scatter plot detailing effect of state multiced MAIT cells from lean donors. (**J**) Scatter plot showing pS6 MFI in either basal or stimulated MAIT cells from lean (white circles) or obese (gray circles) donors. (**K**) Scatter plot showing pS6 MFI in either basal or stimulated comparisons using Student *t* test or ANOVA. Error bars represent SEM. *p < 0.05, ** $p \le 0.01$, ***p < 0.001.

from obese donors (Fig. 3G). We show that rapamycin treatment of activated MAIT cells from lean individuals inhibits IFN- γ , and this regulation occurs at an mRNA level (Fig. 3H, 3I). Together, these data demonstrate that mTORC1 is required for metabolic and functional MAIT cell responses. Indeed, the data show that activation-induced mTORC1 activity was significantly reduced in MAIT cells from obese adults (Fig. 3J). Similar to CD8⁺ T cell IFN- γ production, we observed no difference in CD8⁺ T cell mTORC1 activity between lean and obese donors (Fig. 3K). Collectively, these data highlight an important role for mTORC1 signaling in MAIT cell metabolism and highlight obesity-associated defects in mTORC1 as a possible driver of defective MAIT cell function.

mTORC1 activity is dependent on amino acid flux via SLC7A5, which is diminished in obesity

The data thus far highlight a crucial role for mTORC1 signaling in controlling MAIT cell metabolism and function. We next investigated the upstream regulators of mTOR activity in MAIT cells in the hopes that it may shed light on the cause behind the defective mTORC1 signaling observed in obese MAIT cells. Previous studies have highlighted the importance of amino acid transport through the system L amino acid transporter, SLC7A5, in controlling mTOR activity in primary T cells (24). To assess whether amino acid transport controls mTORC1 activity in MAIT cells, we first investigated amino acid transport in MAIT cells using a recently described flow cytometric–based assay (21). We show that MAIT cells readily uptake exogenous amino acids (as determined by active transport of the fluorescent system L transporter substrate kynurenine) (Fig. 4A, 4B). Furthermore, we show that treatment with the system L amino acid transport inhibitor BCH reduced kynurenine uptake by MAIT cells (Fig. 4C, 4D). We next investigated if MAIT cells from obese donors had altered amino acid uptake. Interestingly, we observed a significant decrease in uptake through system L transport compared with MAIT cells from lean donors (Fig. 4E, 4F). To investigate this reduction, we measured the expression of the predominant system L amino acid transporter expressed in T cells, SLC7A5, and show reduced mRNA expression in obese donors (Fig. 4G). Finally, using BCH treatment, we investigated the impact of this diminished amino acid uptake on mTORC1 activity and glycolytic metabolism in MAIT cells from healthy donors. We observed reduced mTORC1 activation and strong abrogation of glycolytic metabolism with BCH (Fig. 4H-K). As expected, we observed reduced IFN- γ production by MAIT cells treated with BCH (Fig. 4L). These data implicate for the first time, to our knowledge, a crucial role for amino acid availability in regulating MAIT cell metabolism and function in obesity.

Discussion

In this study, we show definitively that MAIT cells increase their rates of glucose uptake and glycolysis upon activation. We also show that this increased rate of MAIT cell glycolysis is essential for driving IFN- γ production. Furthermore, we identify mTORC1 as a key metabolic regulator that controls the rate of MAIT cell



FIGURE 4. mTORC1 activation in MAIT cells is dependent on amino acid uptake via SLC7A5, which is defective in obesity. (**A** and **B**) Representative histogram and scatter plot showing Kynurenine (Kyn) uptake by MAIT cells compared with HBSS control (no Kyn) (n = 10). (**C** and **D**) Representative histogram and scatter plot showing the effect of BCH (50 mM) on Kyn uptake by MAIT cells (n = 10). (**E** and **F**) Representative histogram and scatter plot showing SLC7A5 mRNA expression in MAIT cells from lean or obese donors (n = 5). (**H** and **I**) Representative histogram and scatter plot showing the effect of BCH (50 mM) on the activation of mTORC1 (pS6) in TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**J** and **K**) Representative Seahorse trace and scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12–/IL-18–activated MAIT cells (n = 4). (**L**) Scatter plot showing the effect of BCH (50 mM) on IFN- γ production by TCR-/IL-12– γ models (n = 4). Character plot showing five independent experiments unless otherwise stated. Statistical comparisons using Student *t* test or ANOVA. Error bars represent SEM. *p < 0.05, **p < 0.01, ***p < 0.001.

metabolism and thereby effector molecule production. We show that mTORC1 activation is under the control of amino acid uptake by the cell and expression of the system L amino acid transporter, SLC7A5. We previously reported that MAIT cells isolated from obese individuals are functionally defective. To understand the mechanisms at play, we used RNA-Seq to identify alterations in the gene expression of MAIT cells isolated from obese individuals. These analyses highlighted alterations in both glycolytic and mTORC1 pathways in MAIT cells isolated from obese individuals. Indeed, we confirm our sequencing data by showing reduced rates of glycolysis and mTORC1 activity in MAIT cells from obese adults, which parallel the loss of IFN- γ production reported in obesity (8). Finally, we provide evidence that defects in amino acid transport contribute to the described defective MAIT cell metabolism in obese patients.

MAIT cells are potent effector T cells, capable of rapidly producing multiple cytokines (9). During viral infections, activation of the MAIT cell is independent of the TCR but dependent on IL-18. Data from Slichter and colleagues (11) show that MAIT cells respond weakly to TCR stimulation in the absence of secondary stimulation by cytokine. Our data support this finding, showing that MAIT cells respond most strongly when stimulated via the TCR and cytokine receptors, specifically IL-18 and IL-12.

The pathways regulating MAIT cell cytokine production are poorly understood; however, a recent study by Zinser and colleagues (25) provided evidence that MAIT cell production of granzyme B was dependent on glycolytic metabolism. In other effector lymphocytes, it is well established that the cytokine profiles are regulated by intrinsic cellular metabolism (26–28). Activated T cells have been shown to have a large glucose requirement, especially Th 1 and 17 subsets along with CD8⁺ cells (27). We show that MAIT cells express the glucose transporter Glut-1 and upon cytokine stimulation increase their uptake of the fluorescent 2DG analogue 2NBDG. Like the cytokine responses, MAIT cells increased 2NBDG uptake with TCR stimulation; however, the greatest increases in glucose uptake were observed with TCR–cytokine costimulation. This is in keeping with the requirement for greater amounts of energy and biological intermediates following TCR/cytokine stimulation.

Studies in effector populations such as CD8⁺ T cells and NK cells have demonstrated that the cytokine production by these cells is dependent on glycolytic metabolism (20, 27). We investigated the metabolic pathways used by MAIT cells using the Seahorse extracellular flux analyzer and show that MAIT cells are glycolytic cells, increasing their rates of glycolysis upon activation. Interestingly, in contrast to NK cells, MAIT cells do not display parallel increases in OxPhos (28). To link MAIT cell glycolytic metabolism to cytokine production, we used the glycolytic pathway inhibitor 2DG and show that at low doses it strongly inhibits MAIT cell production of IFN-y. We also show that acute restriction of glucose during MAIT cell activation significantly impacted their activation, confirming their glycolytic requirements. Interestingly, 2DG treatment did not impact TNF-α production by MAIT cells, whereas treatment with oligomycin reduced TNF- α production but not IFN-y. Previous studies in macrophages have described the differential regulation of TNF- α , showing that 2DG blocks IL-1 β but not TNF-a. This suggests differential metabolic regulation of TNF- α production in MAIT cells.

We and others have previously reported that MAIT cells are defective in obese adults, although the mechanism(s) underpinning their dysregulation are unknown (8, 29). To elucidate the mechanisms driving the alterations in cytokine production, we performed RNA-Seq of MAIT cells isolated from lean and obese adults. We observed alterations in the expression over 400 genes between lean and obese donors, and principal component analysis

separated MAIT cells isolated from lean donors from MAIT cells isolated from obese donors. Pathway analysis revealed changes in several cellular metabolism pathways, including the downregulation of genes associated with glycolytic metabolism in MAIT cells isolated from obese individuals. To confirm our sequencing data, we performed a series of experiments, which show for the first time, to our knowledge, that glycolytic metabolism is defective in MAIT cells isolated from obese adults. This parallels our recent study showing defective glycolytic metabolism in NK cells from obese adults (30). In contrast to NK cells and the MAIT data from obese adults presented in this study, we previously reported that in a cohort of obese children there was altered NK cell metabolism with elevated rates of glycolysis, which may suggest that chronic activation of glycolytic machinery in childhood obesity results in a failure later in life (31).

The mTOR complex is highlighted both as an important signaling complex and regulator of metabolism in immune populations such as CD8⁺ T cells and NK cells (20, 26, 32). Our data identify mTORC1 as a key regulator of MAIT cell metabolism. In this study, we show that MAIT cells increase mTORC1 activity upon activation, which is essential for the elevated levels of MAIT cell metabolism upon activation. Finally, our sequencing data highlighted reductions in mTORC1 signaling in MAIT cells isolated from obese individuals. We show diminished mTORC1 activation in MAIT cells but not classical CD8⁺ T cells isolated from obese individuals as compared with lean controls. Previous studies show that inhibition of mTORC1 with rapamycin results in reduced effector immune cell functions (20, 33). Indeed, inhibition of mTORC1 activation with rapamycin inhibited IFN-y production, which is in line with reduced rate of glycolysis observed in rapamycin-treated MAIT cells and the requirement of glycolysis for MAIT cell IFN- γ production. These findings contribute to our understanding as to why MAIT cell functional responses are impaired in obesity.

To investigate why mTOR activity is diminished in MAIT cells from obese donors, we investigated the upstream regulators of mTOR. Previous studies by our collaborators have extensively shown that AKT is dispensable for T cell metabolism and NK cell metabolism but highlight amino acid transport as an important regulator of mTORC1 activity (24, 26, 34, 35). To this end, we show that MAIT cell activation of mTORC1 and subsequent upregulation of glycolytic metabolism and IFN- γ production is dependent on amino acid transport into the cell. We also show that MAIT cells from obese donors have reduced expression of the system L amino acid transporter SLC7A5 and thus reduced amino acid uptake. Numerous studies have highlighted the dysregulation of obesity on amino acid metabolism, in particular branched chained amino acids such as leucine, which is the candidate amino acid implicated in the activation of mTOR in T cells (36-41).

Collectively, our data show that MAIT cells use glycolytic metabolism during their production of IFN- γ and that this process is regulated by mTORC1, which in turn is dependent on amino acid flux into the cell. In obesity, RNA-Seq of MAIT cells revealed downregulation of genes involved in both glycolytic metabolism and mTORC1 signaling. We confirmed these alterations using a series of in vitro experiments. These findings provide a mechanism for the reported alterations in MAIT cell IFN- γ production in obesity and may help us to understand the alterations of MAIT cells in other human diseases such as hepatitis C virus, malignant myeloma, and colorectal cancer (42–44). Finally, understanding the metabolic pathways controlling MAIT cell activation may highlight novel targets for controlling their functions in disease (45).

Disclosures

The authors have no financial conflicts of interest.

References

- Gregg, E. W., and J. E. Shaw. 2017. Global health effects of overweight and obesity. N. Engl. J. Med. 377: 80–81.
- Calle, E. E., C. Rodriguez, K. Walker-Thurmond, and M. J. Thun. 2003. Overweight, obesity, and mortality from cancer in a prospectively studied cohort of U.S. adults. *N. Engl. J. Med.* 348: 1625–1638.
- Wellen, K. E., and G. S. Hotamisligil. 2005. Inflammation, stress, and diabetes. J. Clin. Invest. 115: 1111–1119.
- Ip, B. C., A. E. Hogan, and B. S. Nikolajczyk. 2015. Lymphocyte roles in metabolic dysfunction: of men and mice. *Trends Endocrinol. Metab.* 26: 91–100.
- Lynch, L., M. Nowak, B. Varghese, J. Clark, A. E. Hogan, V. Toxavidis, S. P. Balk, D. O'Shea, C. O'Farrelly, and M. A. Exley. 2012. Adipose tissue invariant NKT cells protect against diet-induced obesity and metabolic disorder through regulatory cytokine production. *Immunity* 37: 574–587.
- O'Shea, D., T. J. Cawood, C. O'Farrelly, and L. Lynch. 2010. Natural killer cells in obesity: impaired function and increased susceptibility to the effects of cigarette smoke. *PLoS One* 5: e8660.
- O'Shea, D., M. Corrigan, M. R. Dunne, R. Jackson, C. Woods, G. Gaoatswe, P. N. Moynagh, J. O'Connell, and A. E. Hogan. 2013. Changes in human dendritic cell number and function in severe obesity may contribute to increased susceptibility to viral infection. *Int. J. Obes.* 37: 1510–1513.
- Carolan, E., L. M. Tobin, B. A. Mangan, M. Corrigan, G. Gaoatswe, G. Byrne, J. Geoghegan, D. Cody, J. O'Connell, D. C. Winter, et al. 2015. Altered distribution and increased IL-17 production by mucosal-associated invariant T cells in adult and childhood obesity. *J. Immunol.* 194: 5775–5780.
- 9. Gapin, L. 2014. Check MAIT. J. Immunol. 192: 4475-4480.
- Kjer-Nielsen, L., O. Patel, A. J. Corbett, J. Le Nours, B. Meehan, L. Liu, M. Bhati, Z. Chen, L. Kostenko, R. Reantragoon, et al. 2012. MR1 presents microbial vitamin B metabolites to MAIT cells. *Nature* 491: 717–723.
- Slichter, C. K., A. McDavid, H. W. Miller, G. Finak, B. J. Seymour, J. P. McNevin, G. Diaz, J. L. Czartoski, M. J. McElrath, R. Gottardo, and M. Prlic. 2016. Distinct activation thresholds of human conventional and innate-like memory T cells. *JCI Insight* 1: e86292.
- Chiba, A., R. Tajima, C. Tomi, Y. Miyazaki, T. Yamamura, and S. Miyake. 2012. Mucosal-associated invariant T cells promote inflammation and exacerbate disease in murine models of arthritis. *Arthritis Rheum.* 64: 153–161.
- Cosgrove, C., J. E. Ussher, A. Rauch, K. Gärtner, A. Kurioka, M. H. Hühn, K. Adelmann, Y. H. Kang, J. R. Fergusson, P. Simmonds, et al. 2013. Early and nonreversible decrease of CD161++ /MAIT cells in HIV infection. *Blood* 121: 951–961.
- Leeansyah, E., A. Ganesh, M. F. Quigley, A. Sönnerborg, J. Andersson, P. W. Hunt, M. Somsouk, S. G. Deeks, J. N. Martin, M. Moll, et al. 2013. Activation, exhaustion, and persistent decline of the antimicrobial MR1-restricted MAIT-cell population in chronic HIV-1 infection. *Blood* 121: 1124–1135.
- 15. Kang, S. J., H. M. Jin, E. J. Won, Y. N. Cho, H. J. Jung, Y. S. Kwon, H. J. Kee, J. K. Ju, J. C. Kim, U. J. Kim, et al. 2016. Activation, impaired tumor necrosis factor-α production, and deficiency of circulating mucosal-associated invariant T cells in patients with scrub typhus. *PLoS Negl. Trop. Dis.* 10: e0004832.
- O'Neill, L. A., R. J. Kishton, and J. Rathmell. 2016. A guide to immunometabolism for immunologists. *Nat. Rev. Immunol.* 16: 553–565.
- Trapnell, C., L. Pachter, and S. L. Salzberg. 2009. TopHat: discovering splice junctions with RNA-Seq. *Bioinformatics* 25: 1105–1111.
- Langmead, B., C. Trapnell, M. Pop, and S. L. Salzberg. 2009. Ultrafast and memory-efficient alignment of short DNA sequences to the human genome. *Genome Biol.* 10: R25.
- Robinson, M. D., and A. Oshlack. 2010. A scaling normalization method for differential expression analysis of RNA-seq data. *Genome Biol.* 11: R25.
- Subramanian, A., P. Tamayo, V. K. Mootha, S. Mukherjee, B. L. Ebert, M. A. Gillette, A. Paulovich, S. L. Pomeroy, T. R. Golub, E. S. Lander, and J. P. Mesirov. 2005. Gene set enrichment analysis: a knowledge-based approach for interpreting genome-wide expression profiles. *Proc. Natl. Acad. Sci. USA* 102: 15545–15550.
- Sinclair, L. V., D. Neyens, G. Ramsay, P. M. Taylor, and D. A. Cantrell. 2018. Single cell analysis of kynurenine and System L amino acid transport in T cells. *Nat. Commun.* 9: 1981.
- Donnelly, R. P., R. M. Loftus, S. E. Keating, K. T. Liou, C. A. Biron, C. M. Gardiner, and D. K. Finlay. 2014. mTORC1-dependent metabolic reprogramming is a prerequisite for NK cell effector function. *J. Immunol.* 193: 4477–4484.
- Keating, S. E., V. Zaiatz-Bittencourt, R. M. Loftus, C. Keane, K. Brennan, D. K. Finlay, and C. M. Gardiner. 2016. Metabolic reprogramming supports IFN-γ production by CD56bright NK cells. J. Immunol. 196: 2552–2560.
- Sinclair, L. V., J. Rolf, E. Emslie, Y. B. Shi, P. M. Taylor, and D. A. Cantrell. 2013. Control of amino-acid transport by antigen receptors coordinates the metabolic reprogramming essential for T cell differentiation. *Nat. Immunol.* 14: 500–508.
- 25. Zinser, M. E., A. J. Highton, A. Kurioka, B. Kronsteiner, J. Hagel, T. Leng, E. Marchi, C. Phetsouphanh, C. B. Willberg, S. J. Dunachie, and P. Klenerman. 2018. Human MAIT cells show metabolic quiescence with rapid glucosedependent upregulation of granzyme B upon stimulation. *Immunol. Cell Biol.* 96: 666–674.

- Finlay, D. K., E. Rosenzweig, L. V. Sinclair, C. Feijoo-Carnero, J. L. Hukelmann, J. Rolf, A. A. Panteleyev, K. Okkenhaug, and D. A. Cantrell. 2012. PDK1 regulation of mTOR and hypoxia-inducible factor 1 integrate metabolism and migration of CD8+ T cells. J. Exp. Med. 209: 2441–2453.
- Buck, M. D., D. O'Sullivan, and E. L. Pearce. 2015. T cell metabolism drives immunity. J. Exp. Med. 212: 1345–1360.
- Assmann, N., K. L. O'Brien, R. P. Donnelly, L. Dyck, V. Zaiatz-Bittencourt, R. M. Loftus, P. Heinrich, P. J. Oefner, L. Lynch, C. M. Gardiner, et al. 2017. Srebp-controlled glucose metabolism is essential for NK cell functional responses. *Nat. Immunol.* 18: 1197–1206.
- Magalhaes, I., K. Pingris, C. Poitou, S. Bessoles, N. Venteclef, B. Kiaf, L. Beaudoin, J. Da Silva, O. Allatif, J. Rossjohn, et al. 2015. Mucosal-associated invariant T cell alterations in obese and type 2 diabetic patients. *J. Clin. Invest.* 125: 1752–1762.
- Michelet, X., L. Dyck, A. Hogan, R. M. Loftus, D. Duquette, K. Wei, S. Beyaz, A. Tavakkoli, C. Foley, R. Donnelly, et al. 2018. Metabolic reprogramming of natural killer cells in obesity limits antitumor responses. *Nat. Immunol.* 19: 1330–1340.
- Tobin, L. M., M. Mavinkurve, E. Carolan, D. Kinlen, E. C. O'Brien, M. A. Little, D. K. Finlay, D. Cody, A. E. Hogan, and D. O'Shea. 2017. NK cells in childhood obesity are activated, metabolically stressed, and functionally deficient. *JCI Insight* 2: e94939.
- 32. Viel, S., A. Marçais, F. S. Guimaraes, R. Loftus, J. Rabilloud, M. Grau, S. Degouve, S. Djebali, A. Sanlaville, E. Charrier, et al. 2016. TGF-β inhibits the activation and functions of NK cells by repressing the mTOR pathway. *Sci. Signal.* 9: ra19.
- Marçais, A., J. Cherfils-Vicini, C. Viant, S. Degouve, S. Viel, A. Fenis, J. Rabilloud, K. Mayol, A. Tavares, J. Bienvenu, et al. 2014. The metabolic checkpoint kinase mTOR is essential for IL-15 signaling during the development and activation of NK cells. *Nat. Immunol.* 15: 749–757.
- Macintyre, A. N., D. Finlay, G. Preston, L. V. Sinclair, C. M. Waugh, P. Tamas, C. Feijoo, K. Okkenhaug, and D. A. Cantrell. 2011. Protein kinase B controls transcriptional programs that direct cytotoxic T cell fate but is dispensable for T cell metabolism. *Immunity* 34: 224–236.
- Loftus, R. M., N. Assmann, N. Kedia-Mehta, K. L. O'Brien, A. Garcia, C. Gillespie, J. L. Hukelmann, P. J. Oefner, A. I. Lamond, C. M. Gardiner, et al.

2018. Amino acid-dependent cMyc expression is essential for NK cell metabolic and functional responses in mice. *Nat. Commun.* 9: 2341.

- Nicklin, P., P. Bergman, B. Zhang, E. Triantafellow, H. Wang, B. Nyfeler, H. Yang, M. Hild, C. Kung, C. Wilson, et al. 2009. Bidirectional transport of amino acids regulates mTOR and autophagy. *Cell* 136: 521–534.
- Adams, S. H. 2011. Emerging perspectives on essential amino acid metabolism in obesity and the insulin-resistant state. *Adv. Nutr.* 2: 445–456.
- Cheng, S., P. Wiklund, R. Autio, R. Borra, X. Ojanen, L. Xu, T. Törmäkangas, and M. Alen. 2015. Adipose tissue dysfunction and altered systemic amino acid metabolism are associated with non-alcoholic fatty liver disease. *PLoS One* 10: e0138889.
- Wiklund, P., X. Zhang, S. Pekkala, R. Autio, L. Kong, Y. Yang, S. Keinänen-Kiukaanniemi, M. Alen, and S. Cheng. 2016. Insulin resistance is associated with altered amino acid metabolism and adipose tissue dysfunction in normoglycemic women. *Sci. Rev.* 6: 24540.
- She, P., C. Van Horn, T. Reid, S. M. Hutson, R. N. Cooney, and C. J. Lynch. 2007. Obesity-related elevations in plasma leucine are associated with alterations in enzymes involved in branched-chain amino acid metabolism. *Am. J. Physiol. Endocrinol. Metab.* 293: E1552–E1563.
- Wolfson, R. L., and D. M. Sabatini. 2017. The dawn of the age of amino acid sensors for the mTORC1 pathway. *Cell Metab.* 26: 301–309.
- 42. Gherardin, N. A., L. Loh, L. Admojo, A. J. Davenport, K. Richardson, A. Rogers, P. K. Darcy, M. R. Jenkins, H. M. Prince, S. J. Harrison, et al. 2018. Enumeration, functional responses and cytotoxic capacity of MAIT cells in newly diagnosed and relapsed multiple myeloma. *Sci. Rep.* 8: 4159.
- Bolte, F. J., A. C. O'Keefe, L. M. Webb, E. Serti, E. Rivera, T. J. Liang, M. Ghany, and B. Rehermann. 2017. Intra-hepatic depletion of mucosalassociated invariant T cells in hepatitis C virus-induced liver inflammation. *Gastroenterology* 153: 1392–1403.e2.
- 44. Sundström, P., F. Ahlmanner, P. Akéus, M. Sundquist, S. Alsén, U. Yrlid, L. Börjesson, Å. Sjöling, B. Gustavsson, S. B. Wong, and M. Quiding-Järbrink. 2015. Human mucosa-associated invariant T cells accumulate in colon adenocarcinomas but produce reduced amounts of IFN-y. J. Immunol. 195: 3472–3481.
- Bettencourt, I. A., and J. D. Powell. 2017. Targeting metabolism as a novel therapeutic approach to autoimmunity, inflammation, and transplantation. J. Immunol. 198: 999–1005.



HIGHLIGHTS

REVIEW

Interleukin-17 producing mucosal associated invariant T cells - emerging players in chronic inflammatory diseases?

Marta M. Pisarska^{*1,2}, Margaret R. Dunne^{*3,4}, Donal O'Shea⁵ and Andrew E. Hogan^{1,2,5} \bigcirc

- ¹ Kathleen Lonsdale Institute for Human Health Research, Maynooth University, Kildare, Ireland
- ² National Children's Research Centre, Dublin, Ireland
- ³ Trinity Translational Medicine Institute, Department of Surgery, St James's Hospital, Dublin, Ireland
- ⁴ Trinity St James's Cancer Institute, St James's Hospital Dublin, Dublin, Ireland
- ⁵ Obesity Immunology Group, Education and Research Centre, St Vincent's University Hospital, University College, Dublin, Ireland

Mucosal associated invariant T (MAIT) cells are a population of evolutionarily conserved T cells, which express an invariant T cell receptor (TCR) and represent a significant subset of innate-like T cells in humans, yet their role in immunity is still emerging. Unlike conventional $\alpha\beta$ T cells, MAIT cells are not restricted by MHC molecules, but instead uniquely recognize microbially derived vitamin metabolites presented by the MHC-I like molecule MR1. MAIT cells are enriched in mucosal sites and tissues including liver and adipose tissue where they are thought to play an important role in immunosurveillance and immunity against microbial infection. In addition to their putative role in antimicrobial immunity, recent research on MAIT cells, in particular IL-17 producing MAIT cells, has demonstrated their involvement in numerous chronic inflammatory conditions. In this review, we give an overview of the work to date on the function and subsets of MAIT cells. We also examine the role of IL-17 producing MAIT cells in chronic inflammatory diseases ranging from autoimmune conditions, metabolic diseases to cancer. Furthermore, we discuss the most recent findings from the clinic that might help deepen our understanding about the biology of MAIT cells.

Keywords: chronic inflammation · IL-17 · mucosal-associated invariant T cells

Mucosal-associated invariant T cells

Mucosal associated invariant T (MAIT) cells comprise a unique population of evolutionarily conserved T cells, first described in humans by Porcelli and colleagues in the early 1990s, what was further confirmed by another landmark study by Tilloy et al. in 1999 [1, 2]. These cells are characterized by expression of

Correspondence: Dr. Andrew E. Hogan e-mail: Andrew.E.Hogan@mu.ie the invariant TCR- α chain, V α 7.2 (TRAV1-2-TRAJ33, TRAJ20, or TRAJ12) in human, or V α 19 (TRAV1-TRAJ33) in mice. This TCR chain is typically associated with a limited array of TCR- β chains, e.g., V β 2 or V β 13 in human, or V β 6 and V β 8 in mice [3]. This constrained gene usage of the TCR suggests the ability of these cells to target conserved antigens. MAIT cell antigen recognition is not restricted by MHC, instead the MHC class

^{*}These authors contributed equally to this work.

I-like molecule (MR1) is recognized by the MAIT TCR [3-5]. The MR1 complex typically presents metabolites derived from the biosynthesis of riboflavin (vitamin B2), a process mediated by many bacterial and yeast species, but not viruses or human cells [6]. This observation is consistent with the finding that MAIT cells are activated by riboflavin synthesizing bacteria, such as Salmonella and other Enterobactericaeae species [7]. In vitro experiments have identified a number of these riboflavin metabolites as MAIT cell antigens, including stimulatory antigens 5-(2-oxopropylideneamino)-6-D-ribitylaminouracil (5-OP-RU), 5-(2-oxoethylideneamino)-6-D-ribitylaminouracil (5-OE-RU), 6,7-dimethyl-8-D-ribityllumazine (RL-6,7-diMe), and 7-hydroxy-6-methyl-8-D-ribityllumazine (RL-6-Me-7-OH) [8]. The potency of these antigens varies, with a non-stimulatory MR1 ligand 6-formyl-pterin (6-FP), a natural breakdown by-product of folic acid, also described [6]. The most potent of these stimulating antigens are 5-OP-RU and 5-OE-RU, which are both generated from the riboflavin precursor 5-amino-6-D-ribitylaminouracil (5-A-RU) through non-enzymatic condensation with small ubiquitous metabolites methylglyoxal and glyoxal, respectively [8, 9]. Identification of these potent antigens has allowed for development of highly specific antigen-loaded MR1 tetramers, for improved identification of both human and mouse MAIT cells [10-13].

In humans, MAIT cells are detectable in cord blood and frequencies increase into adulthood before decreasing after the sixth decade of life [14, 15]. MAIT cells are widely distributed throughout the human body, comprising a substantial proportion of T cells in blood (1–10%), intestinal tissues [16, 17], lungs [18], skin [19], adipose tissue [20], and in particular the liver, where MAIT cells can account for up to 50% of all T cells [21]. In a study of human fetal tissues, MAIT cells were found to be rare in fetal thymus, spleen, and mesenteric lymph nodes but enriched in small intestine, liver, and lung [22]. MAIT cell development proceeds in a step-wise process, where thymic selection precedes peripheral expansion and is dependent on commensal microbes and MR1 expressing B cells [23, 24]. Subsequent studies showed that MAIT cell selecting thymocytes express MR1 and both CD4 and CD8, and that MR1 deficient mice lack MAIT cells [25]. Elaborate studies using MR1 loaded tetramers in mice showed that the MAIT cell compartment is composed of three prominent populations, defined by expression of CD24 and CD44 [26]. Of these populations, the CD24⁻/CD44⁺ subset was the only functionally active population and could be further subdivided into two distinct subsets of Tbet⁺ and RORyt⁺ MAIT cells. In humans, the stages of thymic maturation of MAIT cells were differentiated by CD27 and CD161 expression. Stage 1 was defined by lack of expression of CD161 and CD27, at stage 2 cells remained CD161⁻ but also gained expression of CD27, whereas at the most mature stage of thymic development MAIT cells were defined as CD161⁺ and CD27^{pos-lo}. In contrast to the stage 3 of murine MAIT cell maturation, human MAIT cells were observed to co-express Tbet+ and RORyt⁺. Upon in vitro stimulation, only a small subset of stage 3 thymic MAIT cells responded by cytokine production in comparison to the stage 3 MAIT cells found in blood, suggesting that



Figure 1. MAIT cell phenotype. In addition to their expression of the invariant TCRV α 7.2, MAIT cells can be defined by their expression of several surface molecules including CD161 and CD26. MAIT cells also express several cytokine and chemokine receptors on their surface, including IL-18R and CCR6. Also expressed on the surface of MAIT cells are the nutrient transporters GLUT-1 and SLC7A5, two key transporters in MAIT cell metabolism. MAIT cells can be further characterized by their expression of transcription factors such as PLZF, Tbet, and ROR γ t, which control the production of cytokines such as IFN- γ and IL-17 during activation. Finally, MAIT cells can also express lytic molecules including perforin and granzyme B.

further maturation of these cells takes place outside of the thymus [24].

MAIT cell phenotype and functions

Like other so-called unconventional T cells, such as $\gamma\delta$ T cells and invariant NK T (iNKT) cells, MAIT cells possess attributes of both innate and adaptive immunity, e.g., rapid effector function upon activation by conserved ligands and recognized by a classical, albeit semi-invariant, TCR. Human MAIT cells are classically defined by expression of an invariant Va7.2 TCR chain and high levels of the C-type lectin CD161 [3, 7]. MAIT cells can express CD8, either CD8 $\alpha\alpha$ or CD8 $\alpha\beta$ isoforms, CD4 or can be double negative for both CD4 and CD8, with CD8 expressing MAIT cells representing the majority [27, 28]. The CD4-/CD8- MAIT cell population has recently been described as a functionally distinct MAIT cell subset, producing higher levels of IL-17 and lower levels of IFN- γ [28]. Consistent with their presence in numerous tissue types, MAIT cells express several tissue homing molecules, including CCR5, CCR6, and CXCR6 [29]. MAIT cells also express multiple cytokine receptors, including IL-7R, IL-12R, and IL-18R [7, 30-32]. Several studies have outlined a requirement for cytokine priming or stimulation for full MAIT cell activation, in particular by cytokines IL-7, IL-12, and IL-18 [30, 33] (Fig. 1).

Similar to other unconventional T cell subsets, MAIT cells can be activated independently of their TCR, specifically by cytokines alone [34–36]. This means that MAIT cell activation is not solely reliant on TCR ligation alone, which results in a relatively weak



Figure 2. MAIT cell activation. MAIT cells can be activated in a MR1 dependent manner (1) where bacterial ligands are presented by antigen presenting cells (APC) on MR1 and recognized by the invariant V α 7.2 TCR resulting in the production of cytokine and lytic molecules. MAIT cells can also be activated in a MR1 independent manner (2) by inflammatory cytokine such as IL-12 and IL-18 that are produced by innate cells in the presence of viral or bacterial infection, again resulting in the production of effector molecules.

response. Therefore, these cells can respond to a wider set of stimuli than MR1 presented antigens alone, and in a synergistic manner where multiple signals are present [34, 36, 37] (Fig. 2).

Upon activation, MAIT cells can produce multiple cytokines and can be generally divided into MAIT cells that primarily produce Th1 type cytokines including IFN- γ and TNF- α (MAIT1 cells), and MAIT cells that predominantly produce Th17 type cytokines such as IL-17A (MAIT17 cells) [20, 29, 38, 39]. In the human protein atlas, it was noted that MAIT cells expressed among the highest levels of the protein coding gene RORC, which is the transcription factor responsible for IL-17 production [40]. Several other transcription factors have been detailed in MAIT cells including PLZF and Tbet [41]. In addition to rapid cytokine production, MAIT cells possess a full complement of cytolytic effector molecules such as granzymes (granzyme A and B) and perforin, which allow them to lyse bacteria-infected cells [42]. Upon activation, MAIT cells rapidly display this cytotoxic phenotype, which is heightened through TCR interactions and cytokines. The cytotoxic ability of these cells is tightly regulated however, and is ultimately dependent on MR1 ligation, which allows control of bacterial infection while reducing immune-mediated pathology [42]. Numerous recent studies have highlighted the importance of intrinsic metabolic pathways in controlling immune cell functions [43]. We have recently reported on the metabolic pathways controlling MAIT cell cytokine production [38]. Similar to CD8+ T cells and NK cells, MAIT cells increase their rate of glycolysis upon activation, which provides the biological intermediates needed for effector molecule production [38, 44]. MAIT cells express the major glucose transporter GLUT-1 and neutral amino acid transporter SLC7A5 upon stimulation increase their nutrient uptake [38]. Inhibition of glycolysis in MAIT cells reduced the capacity to produce IFN- γ and granzyme B, highlighting the importance of metabolism in MAIT cell effector function [38, 44]. Mammalian target of rapamycin (mTOR) is a well-characterized regulator of glycolysis in effector cells such as NK cells [45]. Upon activation, MAIT cells increase signaling through mTOR and inhibition with rapamycin results in reduced rates of glycolysis and cytokine production, further highlighting the importance of these intrinsic metabolic pathways in MAIT cells [38], (Fig. 3).

MAIT cells in host defense

The ability of MAIT cells to recognize bacterial metabolites via the highly conserved MR1 molecule suggests that these cells play a critical role in immune defense against microbial infections. MR1 KO mice show higher bacterial burden following injection with either *Escherichia coli* or *Mycobacterium abscessus*, thus implicating MAIT cells in immunity against these pathogens [7]. Furthermore, MR1 deficient mice exposed to low dose aerosol infection with *Mycobacterium bovis* demonstrated a higher bacterial burden compared to MR1 sufficient mice. This effect evens out by day 30 post-infection, however, suggesting that MAIT cells drive early containment of mycobacterial infections in the lung [35].



Figure 3. MAIT cell metabolism. Upon activation, MAIT cells increase their consumption of amino acids and glucose (1), which are transported into MAIT cells via SLC7A5 and GLUT-1, respectively (2-3). Upon sensing nutrients mTORC1 activation (4) leads to increased rates of glycolytic metabolism (5) that provides the intermediates for the production of molecules such as $IFN-\gamma$ and granzyme B (6).

MAIT cells are reduced in the blood but elevated in the lungs of humans with Mycobacterium tuberculosis (Mtb), and respond to Mtb-infected MR1 expressing lung epithelial cells [18]. In children with active Mtb infection, MAIT cell frequencies are decreased in the peripheral blood but unlike in Mtb infected adults, they do not accumulate in the lung [46]. MAIT cells have also been shown to respond in a TCR-dependent manner to Gram-negative bacteria such as Klebsiella pneumoniae, Pseudomonas aeruginosa, and Salmonella enterica, as well as Gram-positive bacteria including Streptococcus pneumoniae, Staphylococcus aureus, Staphylococcus epidermidis, and Clostridioides difficile, but do not respond to Streptococcus pyogenes or Enterococcus faecalis [7, 47-51]. A decrease in circulating MAIT cells has been reported in a number of bacterial infection settings, with lower MAIT cell presence correlating with severity of infection. This was observed in cystic fibrosis patients during Pseudomonas aeruginosa, active Mtb infection, in the miteborne disease scrub typhus, and in critically ill patients admitted to intensive care units with sepsis, who are also more susceptible to nosocomial infections [48, 52-55].

In addition to their broad antibacterial properties, a role for MAIT cells in viral infection has also emerged, detailing how MAIT cells become activated in a TCR-independent, IL-18-dependent manner in response to human viral infections including influenza, hepatitis C (HCV), and dengue virus [36, 56, 57]. Indeed, MAIT cell activation and IL-18 levels correlated with disease severity in acute dengue infection [56]. It was observed that MAIT cell activation was further enhanced in the presence of other proinflammatory cytokines such as IL-12 and type I IFNs [56]. This cytokine-driven mechanism of MAIT cell activation has also been demonstrated in an in vitro model of Zika virus infection, and resulted predominantly in IFN- γ production [57]. A similar IFN- γ response was observed in HCV infection, which resulted in a reduction of in vitro HCV replication [36]. MAIT cells were also shown to be important in the immune response against severe H1N1 influenza infection in mice, with MAIT cell deficient MR1^{-/-} mice showing higher mortality and greater weight loss than control groups [58].

MAIT cells in chronic inflammatory disease

Acute inflammation is a critical process in the protection and repair process, tightly regulated to resolve after injury or infection. However, numerous studies have provided empirical evidence that many diseases are caused by a systemic and non-resolving inflammation termed chronic inflammation [59]. These diseases are increasing in prevalence and dramatically impacting morbidity and mortality. Chronic inflammatory diseases include rheumatoid



Figure 4. MAIT17 cells in chronic inflammatory disease. Schematic outlining the body of works describing increased MAIT17 cells in chronic inflammatory conditions.

arthritis, non-alcoholic fatty liver, and type II diabetes mellitus. The role of MAIT17 cells in the development and pathogenesis of these conditions is summarized in Figure 4 and discussed herein. MAIT17 cells may be a major player in asthma pathology either directly or indirectly.

Respiratory disease

Asthma is a respiratory disease characterized by chronic airway inflammation with increased mucus production in the bronchioles, and is associated with activation of an allergen-specific T helper type 2 (T_H2) immune response [60]. Recently, IL-17 has emerged as a pathogenic player in asthma, in particular in severe asthma and immediate responses to bronchodilators [61]. The role of MAIT cells in asthma is unclear with reported reduced frequencies in blood, sputum, and lung biopsies of patients with asthma [62]. In this study, the authors conclude that a loss of MAIT cells in severe asthma might increase susceptibility to bacterial infection, which might impact asthma pathology, suggesting a protective role. However, in a study by Lezmi and colleagues, it was noted that MAIT17 cells were increased in the BALF of asthma patients, and were associated with exacerbations, suggesting a potential pathogenic role [63]. The exact role of MAIT cells in respiratory disease remains to be elucidated, however their relative abundance in humans and association with disease severity, suggests

Connective tissue disease

MAIT cells have been implicated in several rheumatoid diseases, including rheumatoid arthritis, systemic lupus erythematosus, and spondyloarthritis [64-68]. Rheumatoid arthritis (RA) is a chronic inflammatory autoimmune condition which affects the joints. Several immune cell populations have been implicated in the pathogenesis of disease including T cells, B cells, and macrophages [69]. Studies in RA patients revealed significant decreases in peripheral MAIT cells observed in comparison to healthy controls, and these decreases are correlated with disease activity scores [64]. MAIT cells have been shown to migrate into the synovial fluid in RA, mediated by IL-1 β and TNF- α driven expression of E-selectin at that site [70]. Chiba and colleagues demonstrated an improvement in disease severity in a murine model of collagen induced arthritis in MAIT cell deficient mice, which was paired with a reduction in IL-17 production [71]. Systemic lupus erythematosus (SLE) is an inflammatory autoimmune condition, which impacts several organs including skin, joints and the CNS [72]. Similar to RA, alterations in MAIT cell frequencies were associated with

disease severity in patients [64]. In a murine model of SLE, IL-17 producing MAIT cells were expanded and their deletion resulted in improved disease severity [73]. Spondyloarthritis is a group of chronic inflammatory conditions, which can affect the joints of the spine and pelvis. Several inflammatory cytokines have been implicated in the pathogenesis of the disease including TNF- α and IL-17 [74]. Three separate studies have investigated MAIT cells in ankylosing spondylitis (AS), with each observing reduced MAIT cell frequencies in the circulation [65, 66, 75]. In one of the studies, Gracey and colleagues reported an increase in MAIT cell numbers in the synovial fluid of patients with AS [65]. All three studies also reported that IL-17 production by MAIT cells was higher, suggesting that MAIT cells play a role in pathogenesis [76]. Sjögren syndrome (SjS) is a systemic autoimmune inflammatory disease that primarily affects the exocrine glands resulting in the severe dryness of mucosal surfaces, principally in the mouth and eyes [77]. MAIT cells were found to infiltrate the salivary glands of patients with SjS, displaying an altered phenotype and IFN- γ production, but no difference in IL-17 was reported [78]. In a subsequent report, Guggino and colleagues showed increased MAIT17 cells in salivary gland tissue of patients with SjS, and provided evidence for activation of the IL-7/IL-23 pathways in polarization of MAIT17 cells in SjS [79]. Based on the studies above, it appears that MAIT cells play an important role in the pathogenesis of connective tissue disease and may represent a novel therapeutic target.

Neurological disease

Multiple sclerosis (MS) is a chronic immune-mediated condition affecting the CNS in which T cells play a central pathogenic role in destruction of the myelin sheath [80]. In humans, CD8 T cells comprise the most abundant cell type in the CNS of MS patients in contrast to animal models [71] and MAIT cells account for approximately 5% of these CNS-infiltrating cells [81]. MAIT17 cells are increased in both the CNS and periphery of MS patients, which indicates that MAIT cells may be driving the inflammation associated with this autoimmune disorder [82]. A recent study by Willing et al. has suggested a role for IL-7 in augmentation of IL-17 production by MAIT cells in MS, where increased surface IL-7R expression on MAIT cells correlated with their IL-17 production [82]. Further evidence for MAIT17 cells in the pathogenesis of MS comes from the identification of an expanded population in MS patients of CCR6⁺, CD8⁺, CD161^{hi}, CD3⁺, IL-17⁺ cells, which bear all the hallmarks of MAIT cells [83]. The exact role for MAIT cells in the pathogenesis of MS remains to be elucidated but their increased production of IL-17 may highlight them as a potential therapeutic target.

Inflammatory skin disease

Recent studies have begun to elucidate a role for MAIT cells in psoriasis, one of the most common immune-mediated chronic inflammatory skin disorders [84]. Evidence suggests that IL-17A and T_H17-related cytokines are critical in the pathogenesis of this systemic disease [85]. A study by Teunissen et al. shows the presence of CD8⁺ MAIT cells in the dermis and epidermis of psoriatic plaques as well as healthy skin, with MAIT cells accounting for a proportion of IL-17A⁺ CD8⁺ T cells in plaques [19]. IL-17 producing MAIT cells were predominantly found in the psoriatic lesions and were almost absent in healthy skin. These results identify an additional source of IL-17 in psoriatic skin that may be contributing to the disease [19]. Psoriatic arthritis (PsA) is an inflammatory condition, which affects the joints of up to 25% of psoriasis patients. In the majority of cases, psoriasis precedes the onset of PsA [86]. MAIT cells are found to be enriched in the synovial fluid compared to blood of patients with PsA, and upon stimulation produced more IL-17 than in other rheumatoid conditions [68]. The contribution of MAIT cells to other IL-17-driven inflammatory skin diseases such as atopic dermatitis and hidradenitis suppurativa are unknown to date and require further investigation.

Inflammatory bowel disease

Inflammatory bowel disease (IBD) is a chronic inflammatory condition of the gastrointestinal tract encompassing two separate diseases: Crohn's disease (CD) and ulcerative colitis (UC) [87]. Both conditions are driven by over production of inflammatory mediators including TNF-α, IL-23, and IL-17 [88]. Serriari and colleagues investigated the frequencies of MAIT cells in cohorts of patients with either UC or CD and reported reduced frequencies in the circulation compared to healthy controls, which was paired with an increase of MAIT cells in the inflamed tissue [89]. In vitro, MAIT cells from both UC and CD produced more IL-17 [89]. These findings were supported by a subsequent study by Haga and colleagues, who reported reduced circulating MAIT cell frequencies and increased IL-17 production in UC patients [90]. Furthermore, increased MAIT cell levels in the inflamed mucosa correlated with disease activity, suggesting a possible pathogenic role for MAIT cells.

Metabolic disease

Obesity is emerging as the number one cause of preventable death in the Western world, underpinning the development of many chronic diseases, such as type 2 diabetes mellitus (T2DM), cardiovascular disease, and many cancers [91]. Obesity has also been identified as a confounding factor in other chronic diseases including psoriasis, arthritis, and cancer [92]. We and others have previously reported reduced MAIT cell frequencies in the periphery of adults living with obesity [20]. MAIT cells isolated from the periphery of these patients displayed a potent IL-17 phenotype, i.e., significantly higher IL-17 production and reduced IFN- γ production following stimulation compared to healthy controls [20, 93]. This predominant IL-17 phenotype may be contributing to the chronic inflammation seen in obesity and insulin resistance [94]. MAIT cells were found at higher frequencies in adipose tissue compared to blood, in both lean and obese individuals. MAIT cells in healthy adipose tissue produce IL-10, however this production is reduced in MAIT cells from the adipose tissue of obese patients. [20] MAIT cells may therefore play a regulatory role in healthy individuals and initiate a more inflammatory response through their pro-IL-17 phenotype in obese individuals. Following bariatric surgery (1-year post-surgery) MAIT cell levels were increased in blood compared to presurgical levels, however the levels detected were still lower than healthy controls [93]. The data associated with childhood obesity is slightly different. MAIT cell frequencies in the blood of obese children appear to be significantly higher when compared to their lean counterparts. However, as obese children progress into adulthood, their MAIT cell frequency declines [20]. This is not observed in the lean cohort, suggesting that MAIT cell numbers ultimately decline with the progression of obesity. This altered frequency is associated with both increased fasting insulin and insulin resistance, factors driving the development of type 2 diabetes [20]. Similar to adult obesity, increased MAIT cell IL-17 production was also seen in the pediatric obese cohort [20]. Recently, we have demonstrated that alterations in mitochondria ROS are linked to IL-17 production in MAIT cells from patients with obesity. Using several mitochondrial antioxidants, we were able to limit the production of IL-17 by MAIT cells from patients with obesity in vitro [95]. Non-alcoholic fatty liver disease (NAFLD) is characterized by the excess accumulation of fat in hepatocytes, which can drive fibrosis, progressing to cirrhosis [96]. In cohorts of patients with cirrhotic NAFLD, MAIT cell frequencies were reduced in both the periphery and the liver, but displayed elevated MAIT17 cells compared to controls [97]. Using an in vitro system, it was demonstrated that MAIT cells could drive a pro-inflammatory human hepatic myofibroblast phenotype in a IL-17/TNF-a dependent manner, suggesting MAIT cells are a pro-fibrogenic population [97]. The contribution of MAIT cells to the progression of metabolic disease remains to be elucidated but due to their abundance in humans, they may represent a novel therapeutic target.

Cancer

MAIT cells have been detected within many solid tumor types, including breast, lung, liver, thyroid, colorectal, kidney, brain, stomach, and esophagus [98–102] and in multiple myeloma [11]. These emerging studies have demonstrated that MAIT cells are capable of tumor cell lysis in vitro, however, the role of MAIT cells within cancer, and in particular MAIT17 cells, remains unclear. Ling and colleagues report elevated IL-17 production by MAIT cells in colorectal tumors, when compared to a healthy control group [99]. The authors also found increased IL-17 production by MAIT cells co-cultured in vitro with colorectal cancer cell lines, but also observed an MR1 dependent inhibition of the tumor cell cycle in this setting, suggesting a more complex role for MAIT17 cells in cancer [99]. In contrast, a recent study by Yan and colleagues provided evidence for the promotion of tumor metastasis

by MAIT cells, in particular MAIT17 cells [103]. Using two murine models, the authors showed increased tumor expression of MR1, which promoted MAIT cell activation and production of IL-17, which subsequently inhibited NK cell antitumor activity [103]. More work is required in order to define the role of MAIT cells in cancer, including in infection driven cancers, and to properly elucidate their immunotherapeutic potential.

Conclusion

MAIT cells represent a significant proportion of the human CD8⁺ T cell compartment, yet their place in immunity is only beginning to be understood. In recent years, the volume of studies into MAIT cell biology has exponentially increased, largely due to greater availability of antibodies and tetramers. Many of these studies have examined the role of MAIT cells in disease settings, giving new insights into the role played by MAIT cells in disease. The most striking observation common to many different disease states, is the significant alteration in MAIT cell frequency and cytokine profile. In particular, we have discussed the prevalence of MAIT17 cells in chronic inflammatory diseases. Currently, we do not fully understand the environmental signals and molecular pathways driving this expansion of MAIT17 cells. Several studies have implicated IL-7 as a licencing factor for MAIT17 cells [65, 79, 82], however, this was not specific to IL-17 as other studies detailed IL-7 licencing of IFN-y and Granzyme B [30, 104]. In addition to IL-7 priming of MAIT17 cells, some studies detailed additional cytokines such as IL-12 or IL-23 [79]. Further work will be required to establish the exact cytokine signals driving MAIT cells and if these are common or disease specific. As discussed, our group has highlighted the importance of immunometabolism in controlling MAIT cell cytokine responses, we highlighted altered mitochondria as a potential driver of IL-17 in MAIT cells from patients with obesity [38, 95]. Altered mitochondria haven been linked to IL-17 production in conventional T cells [105–107], however additional work will again be required to determine if this is a major mechanism driving IL-17 production in MAIT cells, and also to elucidate the environmental signals driving mitochondrial dysregulation. It is clear that despite the recent surge in MAIT cell studies, including numerous studies on MAIT17 cells, improved insight into the pathways controlling MAIT17 cells is required.

Acknowledgments: A.E.H. and D.O.S. are supported by the Health Research Board. A.E.H. and M.P. are supported by the National Children's Research Centre.

Conflict of interest: The authors declare no financial or commercial conflict of interest.

References

- 1 Tilloy, F., Treiner, E., Park, S. H., Garcia, C., Lemonnier, F., de la Salle, H., Bendelac, A. et al., An invariant T cell receptor alpha chain defines a novel TAP-independent major histocompatibility complex class Ibrestricted alpha/beta T cell subpopulation in mammals. J. Exp. Med. 1999. 189: 1907–1921.
- 2 Porcelli, S., Yockey, C. E., Brenner, M. B. and Balk, S. P., Analysis of T cell antigen receptor (TCR) expression by human peripheral blood CD4-8- alpha/beta T cells demonstrates preferential use of several V beta genes and an invariant TCR alpha chain. J. Exp. Med. 1993. 178: 1–16.
- 3 Treiner, E., Duban, L., Bahram, S., Radosavljevic, M., Wanner, V., Tilloy, F., Affaticati, P. et al., Selection of evolutionarily conserved mucosalassociated invariant T cells by MR1. Nature 2003. 422: 164–169.
- 4 Huang, S., Gilfillan, S., Cella, M., Miley, M. J., Lantz, O., Lybarger, L., Fremont, D. H. et al., Evidence for MR1 antigen presentation to mucosalassociated invariant T cells. J. Biol. Chem. 2005. 280: 21183–21193.
- 5 Huang, S., Gilfillan, S., Kim, S., Thompson, B., Wang, X., Sant, A. J., Fremont, D. H. et al., MR1 uses an endocytic pathway to activate mucosalassociated invariant T cells. J. Exp. Med. 2008. 205: 1201–1211.
- 6 Kjer-Nielsen, L., Patel, O., Corbett, A. J., Le Nours, J., Meehan, B., Liu, L., Bhati, M. et al., MR1 presents microbial vitamin B metabolites to MAIT cells. Nature 2012. **491**: 717–723.
- 7 Le Bourhis, L., Martin, E., Péguillet, I., Guihot, A., Froux, N., Coré, M., Lévy, E. et al., Antimicrobial activity of mucosal-associated invariant T cells. Nat. Immunol. 2010. **11**: 701–708.
- 8 Corbett, A. J., Eckle, S. B., Birkinshaw, R. W., Liu, L., Patel, O., Mahony, J., Chen, Z. et al., T-cell activation by transitory neo-antigens derived from distinct microbial pathways. *Nature* 2014. 509: 361–365.
- 9 Awad, W., Ler, G. J. M., Xu, W., Keller, A. N., Mak, J. Y. W., Lim, X. Y., Liu, L. et al., The molecular basis underpinning the potency and specificity of MAIT cell antigens. Nat. Immunol. 2020. 21: 400–411.
- 10 Reantragoon, R., Corbett, A. J., Sakala, I. G., Gherardin, N. A., Furness, J. B., Chen, Z., Eckle, S. B. et al., Antigen-loaded MR1 tetramers define T cell receptor heterogeneity in mucosal-associated invariant T cells. J. Exp. Med. 2013. 210: 2305–2320.
- 11 Gherardin, N. A., Loh, L., Admojo, L., Davenport, A. J., Richardson, K., Rogers, A., Darcy, P. K. et al., Enumeration, functional responses and cytotoxic capacity of MAIT cells in newly diagnosed and relapsed multiple myeloma. Sci. Rep. 2018. 8: 4159.
- 12 Rahimpour, A., Koay, H. F., Enders, A., Clanchy, R., Eckle, S. B., Meehan, B., Chen, Z. et al., Identification of phenotypically and functionally heterogeneous mouse mucosal-associated invariant T cells using MR1 tetramers. J. Exp. Med. 2015. 212: 1095–1108.
- 13 Gherardin, N. A., Souter, M. N., Koay, H. F., Mangas, K. M., Seemann, T., Stinear, T. P., Eckle, S. B. et al., Human blood MAIT cell subsets defined using MR1 tetramers. *Immunol. Cell Biol.* 2018. 96: 507–525.
- 14 Chen, P., Deng, W., Li, D., Zeng, T., Huang, L., Wang, Q., Wang, J. et al., Circulating mucosal-associated invariant T cells in a large cohort of healthy Chinese individuals from newborn to elderly. Front. Immunol. 2019. 10: 260.
- 15 Loh, L., Gherardin, N. A., Sant, S., Grzelak, L., Crawford, J. C., Bird, N. L., Koay, H. F. et al., Human mucosal-associated invariant T cells in older individuals display expanded TCR $\alpha\beta$ clonotypes with potent antimicrobial responses. J. Immunol. 2020. 204: 1119–1133.
- 16 Dunne, M. R., Elliott, L., Hussey, S., Mahmud, N., Kelly, J., Doherty, D. G. and Feighery, C. F., Persistent changes in circulating and intestinal γ^δ T cell subsets, invariant natural killer T cells and mucosal-associated

invariant T cells in children and adults with coeliac disease. PLoS One 2013. 8: e76008.

- 17 Tominaga, K., Yamagiwa, S., Setsu, T., Kimura, N., Honda, H., Kamimura, H., Honda, Y. et al., Possible involvement of mucosalassociated invariant T cells in the progression of inflammatory bowel diseases. *Biomed. Res.* 2017. **38**: 111–121.
- 18 Gold, M. C., Cerri, S., Smyk-Pearson, S., Cansler, M. E., Vogt, T. M., Delepine, J., Winata, E. et al., Human mucosal associated invariant T cells detect bacterially infected cells. PLoS Biol. 2010. 8: e1000407.
- 19 Teunissen, M. B., Yeremenko, N. G., Baeten, D. L., Chielie, S., Spuls, P. I., de Rie, M. A., Lantz, O. et al., The IL-17A-producing CD8+ Tcell population in psoriatic lesional skin comprises mucosa-associated invariant T cells and conventional T cells. J. Invest. Dermatol. 2014. 134: 2898–2907.
- 20 Carolan, E., Tobin, L. M., Mangan, B. A., Corrigan, M., Gaoatswe, G., Byrne, G., Geoghegan, J. et al., Altered distribution and increased IL-17 production by mucosal-associated invariant T cells in adult and childhood obesity. J. Immunol. 2015. 194: 5775–5780.
- 21 Kurioka, A., Walker, L. J., Klenerman, P. and Willberg, C. B., MAIT cells: new guardians of the liver. Clin Transl Immunology 2016. 5: e98.
- 22 Leeansyah, E., Loh, L., Nixon, D. F. and Sandberg, J. K., Acquisition of innate-like microbial reactivity in mucosal tissues during human fetal MAIT-cell development. Nat. Commun. 2014. 5: 3143.
- 23 Legoux, F., Bellet, D., Daviaud, C., El Morr, Y., Darbois, A., Niort, K., Procopio, E. et al., Microbial metabolites control the thymic development of mucosal-associated invariant T cells. *Science* 2019. 366: 494–499.
- 24 Martin, E., Treiner, E., Duban, L., Guerri, L., Laude, H., Toly, C., Premel, V. et al., Stepwise development of MAIT cells in mouse and human. PLoS Biol. 2009. 7: e54.
- 25 Seach, N., Guerri, L., Le Bourhis, L., Mburu, Y., Cui, Y., Bessoles, S., Soudais, C. et al., Double-positive thymocytes select mucosalassociated invariant T cells. J. Immunol. 2013. 191: 6002–6009.
- 26 Koay, H. F., Gherardin, N. A., Enders, A., Loh, L., Mackay, L. K., Almeida, C. F., Russ, B. E. et al., A three-stage intrathymic development pathway for the mucosal-associated invariant T cell lineage. Nat. Immunol. 2016. 17: 1300–1311.
- 27 Gapin, L., Check MAIT. J. Immunol. 2014. 192: 4475-4480.
- 28 Dias, J., Boulouis, C., Gorin, J. B., van den Biggelaar, R. H. G. A., Lal, K. G., Gibbs, A., Loh, L. et al., The CD4. Proc. Natl. Acad. Sci. U. S. A. 2018. 115: E11513-E11522.
- 29 Dusseaux, M., Martin, E., Serriari, N., Péguillet, I., Premel, V., Louis, D., Milder, M. et al., Human MAIT cells are xenobiotic-resistant, tissuetargeted, CD161hi IL-17-secreting T cells. Blood 2011. 117: 1250–1259.
- 30 Tang, X. Z., Jo, J., Tan, A. T., Sandalova, E., Chia, A., Tan, K. C., Lee, K. H. et al., IL-7 licenses activation of human liver intrasinusoidal mucosalassociated invariant T cells. J. Immunol. 2013. 190: 3142–3152.
- 31 Wallington, J. C., Williams, A. P., Staples, K. J. and Wilkinson, T. M. A., IL-12 and IL-7 synergize to control mucosal-associated invariant T-cell cytotoxic responses to bacterial infection. J. Allergy Clin. Immunol. 2017. 141: 2182–2195.e6.
- 32 Sattler, A., Dang-Heine, C., Reinke, P. and Babel, N., IL-15 dependent induction of IL-18 secretion as a feedback mechanism controlling human MAIT-cell effector functions. *Eur. J. Immunol.* 2015. 45: 2286–2298.
- 33 Slichter, C. K., McDavid, A., Miller, H. W., Finak, G., Seymour, B. J., McNevin, J. P., Diaz, G. et al., Distinct activation thresholds of human conventional and innate-like memory T cells. JCI Insight 2016. 1.
- 34 Ussher, J. E., Bilton, M., Attwod, E., Shadwell, J., Richardson, R., de Lara,
 C., Mettke, E. et al., CD161++ CD8+ T cells, including the MAIT cell

Eur. J. Immunol. 2020. 50: 1098-1108

subset, are specifically activated by IL-12+IL-18 in a TCR-independent manner. *Eur. J. Immunol.* 2014. **44**: 195–203.

- 35 Chua, W. J., Truscott, S. M., Eickhoff, C. S., Blazevic, A., Hoft, D. F. and Hansen, T. H., Polyclonal mucosa-associated invariant T cells have unique innate functions in bacterial infection. *Infect. Immun.* 2012. 80: 3256–3267.
- 36 van Wilgenburg, B., Scherwitzl, I., Hutchinson, E. C., Leng, T., Kurioka, A., Kulicke, C., de Lara, C. et al., MAIT cells are activated during human viral infections. Nat. Commun. 2016. 7: 11653.
- 37 Dias, J., Leeansyah, E. and Sandberg, J. K., Multiple layers of heterogeneity and subset diversity in human MAIT cell responses to distinct microorganisms and to innate cytokines. Proc. Natl. Acad. Sci. U. S. A. 2017. 114: E5434-E5443.
- 38 O'Brien, A., Loftus, R. M., Pisarska, M. M., Tobin, L. M., Bergin, R., Wood, N. A. W., Foley, C. et al., Obesity reduces mTORC1 activity in mucosalassociated invariant T cells, driving defective metabolic and functional responses. J. Immunol. 2019. 202: 3404–3411.
- 39 Dias, J., Sobkowiak, M. J., Sandberg, J. K. and Leeansyah, E., Human MAIT-cell responses to Escherichia coli: activation, cytokine production, proliferation, and cytotoxicity. J. Leukoc. Biol. 2016. 100: 233–240.
- 40 Uhlen, M., Karlsson, M. J., Zhong, W., Tebani, A., Pou, C., Mikes, J., Lakshmikanth, T. et al., A genome-wide transcriptomic analysis of proteincoding genes in human blood cells. *Science* 2019. 366: eaax9198.
- 41 Godfrey, D. I., Koay, H. F., McCluskey, J. and Gherardin, N. A., The biology and functional importance of MAIT cells. Nat. Immunol. 2019. 20: 1110–1128.
- 42 Kurioka, A., Ussher, J. E., Cosgrove, C., Clough, C., Fergusson, J. R., Smith, K., Kang, Y. H. et al., MAIT cells are licensed through granzyme exchange to kill bacterially sensitized targets. *Mucosal Immunol* 2015. 8: 429–440.
- 43 O'Neill, L. A., Kishton, R. J. and Rathmell, J., A guide to immunometabolism for immunologists. Nat. Rev. Immunol. 2016. 16: 553–565.
- 44 Zinser, M. E., Highton, A. J., Kurioka, A., Kronsteiner, B., Hagel, J., Leng, T., Marchi, E. et al., Human MAIT cells show metabolic quiescence with rapid glucose-dependent upregulation of granzyme B upon stimulation. *Immunol. Cell Biol.* 2018. **96**: 666–674.
- 45 Donnelly, R. P., Loftus, R. M., Keating, S. E., Liou, K. T., Biron, C. A., Gardiner, C. M. and Finlay, D. K., mTORC1-dependent metabolic reprogramming is a prerequisite for NK cell effector function. *J. Immunol.* 2014. 193: 4477–4484.
- 46 Malka-Ruimy, C., Ben Youssef, G., Lambert, M., Tourret, M., Ghazarian, L., Faye, A., Caillat-Zucman, S. et al., Mucosal-associated invariant T cell levels are reduced in the peripheral blood and lungs of children with active pulmonary tuberculosis. Front. Immunol. 2019. 10: 206.
- 47 Georgel, P., Radosavljevic, M., Macquin, C. and Bahram, S., The nonconventional MHC class I MR1 molecule controls infection by Klebsiella pneumoniae in mice. Mol. Immunol. 2011. 48: 769–775.
- 48 Smith, D. J., Hill, G. R., Bell, S. C. and Reid, D. W., Reduced mucosal associated invariant T-cells are associated with increased disease severity and Pseudomonas aeruginosa infection in cystic fibrosis. PLoS One 2014.
 9: e109891.
- 49 Howson, L. J., Napolitani, G., Shepherd, D., Ghadbane, H., Kurupati, P., Preciado-Llanes, L., Rei, M., Dobinson, H. C. et al., MAIT cell clonal expansion and TCR repertoire shaping in human volunteers challenged with Salmonella Paratyphi A. Nat. Commun. 2018. 9: 253.
- 50 Kurioka, A., van Wilgenburg, B., Javan, R. R., Hoyle, R., van Tonder, A. J., Harrold, C. L., Leng, T. et al., Diverse Streptococcus pneumoniae strains drive a mucosal-associated invariant T-cell response through

major histocompatibility complex class I-related molecule-dependent and cytokine-driven pathways. J. Infect. Dis. 2018. 217: 988–999.

- 51 Bernal, I., Hofmann, J. D., Bulitta, B., Klawonn, F., Michel, A. M., Jahn, D., Neumann-Schaal, M. et al., Activates human mucosal-associated invariant T cells. Front Microbiol 2018. 9: 2532.
- 52 Kang, S. J., Jin, H. M., Won, E. J., Cho, Y. N., Jung, H. J., Kwon, Y. S., Kee, H. J. et al., Activation, impaired tumor necrosis factor-α production, and deficiency of circulating mucosal-associated invariant T cells in patients with scrub typhus. PLoS Negl Trop Dis 2016. 10: e0004832.
- 53 Grimaldi, D., Le Bourhis, L., Sauneuf, B., Dechartres, A., Rousseau, C., Ouaaz, F., Milder, M. et al., Specific MAIT cell behaviour among innatelike T lymphocytes in critically ill patients with severe infections. *Intensive Care Med.* 2014. **40**: 192–201.
- 54 Kwon, Y. S., Cho, Y. N., Kim, M. J., Jin, H. M., Jung, H. J., Kang, J. H., Park, K. J. et al., Mucosal-associated invariant T cells are numerically and functionally deficient in patients with mycobacterial infection and reflect disease activity. *Tuberculosis (Edinb.)* 2015. 95: 267–274.
- 55 Pincikova, T., Paquin-Proulx, D., Moll, M., Flodström-Tullberg, M., Hjelte, L. and Sandberg, J. K., Severely impaired control of bacterial infections in a patient with cystic fibrosis defective in mucosalassociated invariant T cells. *Chest* 2018. 153: e93-e96.
- 56 Loh, L., Wang, Z., Sant, S., Koutsakos, M., Jegaskanda, S., Corbett, A. J., Liu, L. et al., Human mucosal-associated invariant T cells contribute to antiviral influenza immunity via IL-18-dependent activation. Proc. Natl. Acad. Sci. U. S. A. 2016. 113: 10133–10138.
- 57 Paquin-Proulx, D., Avelino-Silva, V. I., Santos, B. A. N., Silveira Barsotti, N., Siroma, F., Fernandes Ramos, J., Coracini Tonacio, A. et al., MAIT cells are activated in acute Dengue virus infection and after in vitro Zika virus infection. PLoS Negl Trop Dis 2018. 12: e0006154.
- 58 van Wilgenburg, B., Loh, L., Chen, Z., Pediongco, T. J., Wang, H., Shi, M., Zhao, Z. et al., MAIT cells contribute to protection against lethal influenza infection in vivo. Nat. Commun. 2018. 9: 4706.
- 59 Furman, D., Campisi, J., Verdin, E., Carrera-Bastos, P., Targ, S., Franceschi, C., Ferrucci, L. et al., Chronic inflammation in the etiology of disease across the life span. Nat. Med. 2019. 25: 1822–1832.
- 60 Holgate, S. T., Wenzel, S., Postma, D. S., Weiss, S. T., Renz, H. and Sly, P. D., Asthma. Nat. Rev. Dis. Primers 2015. 1: 15025.
- 61 Chesné, J., Braza, F., Mahay, G., Brouard, S., Aronica, M. and Magnan, A., IL-17 in severe asthma. Where do we stand? Am. J. Respir. Crit. Care Med. 2014. 190: 1094–1101.
- 62 Hinks, T. S., Zhou, X., Staples, K. J., Dimitrov, B. D., Manta, A., Petrossian, T., Lum, P. Y. et al., Innate and adaptive T cells in asthmatic patients: Relationship to severity and disease mechanisms. J. Allergy Clin. Immunol. 2015. 136: 323–333.
- 63 Lezmi, G., Abou-Taam, R., Garcelon, N., Dietrich, C., Machavoine, F., Delacourt, C., Adel-Patient, K. et al., Evidence for a MAIT-17-high phenotype in children with severe asthma. J. Allergy Clin. Immunol. 2019. 144: 1714–1716.e1716.
- 64 Cho, Y. N., Kee, S. J., Kim, T. J., Jin, H. M., Kim, M. J., Jung, H. J., Park, K. J. et al., Mucosal-associated invariant T cell deficiency in systemic lupus erythematosus. J. Immunol. 2014. 193: 3891–3901.
- 65 Gracey, E., Qaiyum, Z., Almaghlouth, I., Lawson, D., Karki, S., Avvaru, N., Zhang, Z. et al., IL-7 primes IL-17 in mucosal-associated invariant T (MAIT) cells, which contribute to the Th17-axis in ankylosing spondylitis. Ann. Rheum. Dis. 2016. **75**: 2124–2132.
- 66 Hayashi, E., Chiba, A., Tada, K., Haga, K., Kitagaichi, M., Nakajima, S., Kusaoi, M. et al., Involvement of Mucosal-associated Invariant T cells in Ankylosing Spondylitis. J. Rheumatol. 2016. 43: 1695–1703.

- 67 Chiba, A., Tamura, N., Yoshikiyo, K., Murayama, G., Kitagaichi, M., Yamaji, K., Takasaki, Y. et al., Activation status of mucosal-associated invariant T cells reflects disease activity and pathology of systemic lupus erythematosus. Arthritis Res. Ther. 2017. 19: 58.
- 68 Raychaudhuri, S. K., Abria, C., Mitra, A. and Raychaudhuri, S. P., Functional significance of MAIT cells in psoriatic arthritis. *Cytokine* 2020. 125: 154855.
- 69 Smolen, J. S., Aletaha, D., Barton, A., Burmester, G. R., Emery, P., Firestein, G. S., Kavanaugh, A. et al., Rheumatoid arthritis. Nat. Rev. Dis. Primers 2018. 4: 18001.
- 70 Kim, M., Yoo, S. J., Kang, S. W., Kwon, J., Choi, I. and Lee, C. H., TNFα and IL-1β in the synovial fluid facilitate mucosal-associated invariant T (MAIT) cell migration. Cytokine 2017. 99: 91–98.
- 71 Chiba, A., Tajima, R., Tomi, C., Miyazaki, Y., Yamamura, T. and Miyake, S., Mucosal-associated invariant T cells promote inflammation and exacerbate disease in murine models of arthritis. *Arthritis Rheum*. 2012. 64: 153–161.
- 72 Kaul, A., Gordon, C., Crow, M. K., Touma, Z., Urowitz, M. B., van Vollenhoven, R., Ruiz-Irastorza, G. et al., Systemic lupus erythematosus. Nat. *Rev. Dis. Primers* 2016. 2: 16039.
- 73 Murayama, G., Chiba, A., Suzuki, H., Nomura, A., Mizuno, T., Kuga, T., Nakamura, S. et al., A critical role for mucosal-associated invariant T cells as regulators and therapeutic targets in systemic lupus erythematosus. Front. Immunol. 2019. 10: 2681.
- 74 Sieper, J., Braun, J., Dougados, M. and Baeten, D., Axial spondyloarthritis. Nat. Rev. Dis. Primers 2015. 1: 15013.
- 75 Toussirot, É., Laheurte, C., Gaugler, B., Gabriel, D. and Saas, P., Increased IL-22- and IL-17A-producing mucosal-associated invariant T cells in the peripheral blood of patients with ankylosing spondylitis. Front. Immunol. 2018. 9: 1610.
- 76 Toussirot, E. and Saas, P., MAIT cells: potent major cellular players in the IL-17 pathway of spondyloarthritis? RMD Open 2018. 4: e000821.
- 77 Brito-Zerón, P., Baldini, C., Bootsma, H., Bowman, S. J., Jonsson, R., Mariette, X., Sivils, K. et al., Sjögren syndrome. Nat. Rev. Dis. Primers 2016. 2: 16047.
- 78 Wang, J. J., Macardle, C., Weedon, H., Beroukas, D. and Banovic, T., Mucosal-associated invariant T cells are reduced and functionally immature in the peripheral blood of primary Sjögren's syndrome patients. Eur. J. Immunol. 2016. 46: 2444–2453.
- 79 Guggino, G., Liberto, D. D., Pizzo, M. L., Saieva, L., Alessandro, R., Dieli, F., Triolo, G. et al., IL-17 polarization of MAIT cells is derived from the activation of two different pathways. *Eur. J. Immunol.* 2017. 47: 2002– 2003.
- 80 Filippi, M., Bar-Or, A., Piehl, F., Preziosa, P., Solari, A., Vukusic, S. and Rocca, M. A., Multiple sclerosis. Nat. Rev. Dis. Primers 2018. 4: 43.
- 81 Willing, A., Leach, O. A., Ufer, F., Attfield, K. E., Steinbach, K., Kursawe, N., Piedavent, M. et al., CD8⁺ MAIT cells infiltrate into the CNS and alterations in their blood frequencies correlate with IL-18 serum levels in multiple sclerosis. *Eur. J. Immunol.* 2014. 44: 3119–3128.
- 82 Willing, A., Jäger, J., Reinhardt, S., Kursawe, N. and Friese, M. A., Production of IL-17 by MAIT cells is increased in multiple sclerosis and is associated with IL-7 receptor expression. J. Immunol. 2018. 200: 974–982.
- 83 Annibali, V., Ristori, G., Angelini, D. F., Serafini, B., Mechelli, R., Cannoni, S., Romano, S. et al., CD161(high)CD8+T cells bear pathogenetic potential in multiple sclerosis. *Brain* 2011. **134**: 542–554.
- 84 Johnston, A. and Gudjonsson, J. E., Psoriasis and the MAITing game: a role for IL-17A+ invariant TCR CD8+ T cells in psoriasis? J. Invest. Dermatol. 2014. 134: 2864–2866.

- 85 Greb, J. E., Goldminz, A. M., Elder, J. T., Lebwohl, M. G., Gladman, D. D., Wu, J. J., Mehta, N. N. et al., Psoriasis. Nat. Rev. Dis. Primers 2016. 2: 16082.
- 86 Lloyd, P., Ryan, C. and Menter, A., Psoriatic arthritis: an update. Arthritis 2012. 2012: 176298.
- 87 de Souza, H. S. and Fiocchi, C., Immunopathogenesis of IBD: current state of the art. Nat. Rev. Gastroenterol. Hepatol. 2016. 13: 13–27.
- 88 Moschen, A. R., Tilg, H. and Raine, T., IL-12, IL-23 and IL-17 in IBD: immunobiology and therapeutic targeting. Nat. Rev. Gastroenterol. Hepatol. 2019. 16: 185–196.
- 89 Serriari, N. E., Eoche, M., Lamotte, L., Lion, J., Fumery, M., Marcelo, P., Chatelain, D. et al., Innate mucosal-associated invariant T (MAIT) cells are activated in inflammatory bowel diseases. *Clin. Exp. Immunol.* 2014. 176: 266–274.
- 90 Haga, K., Chiba, A., Shibuya, T., Osada, T., Ishikawa, D., Kodani, T., Nomura, O. et al., MAIT cells are activated and accumulated in the inflamed mucosa of ulcerative colitis. J. Gastroenterol. Hepatol. 2016. 31: 965–972.
- 91 Gregg, E. W. and Shaw, J. E., Global health effects of overweight and obesity. N. Engl. J. Med. 2017. 377: 80–81.
- 92 De Lorenzo, A., Gratteri, S., Gualtieri, P., Cammarano, A., Bertucci, P. and Di Renzo, L., Why primary obesity is a disease? J. Transl. Med. 2019. 17: 169.
- 93 Magalhaes, I., Pingris, K., Poitou, C., Bessoles, S., Venteclef, N., Kiaf, B., Beaudoin, L. et al., Mucosal-associated invariant T cell alterations in obese and type 2 diabetic patients. J. Clin. Invest. 2015. 125: 1752– 1762.
- 94 Zúñiga, L. A., Shen, W. J., Joyce-Shaikh, B., Pyatnova, E. A., Richards, A. G., Thom, C., Andrade, S. M. et al., IL-17 regulates adipogenesis, glucose homeostasis, and obesity. J. Immunol. 2010. 185: 6947–6959.
- 95 Brien, A. O., Kedia-Mehta, N., Tobin, L., Veerapen, N., Besra, G. S., Shea, D. O. and Hogan, A. E., Targeting mitochondrial dysfunction in MAIT cells limits IL-17 production in obesity. Cell Mol Immunol 2020.
- 96 Brunt, E. M., Wong, V. W., Nobili, V., Day, C. P., Sookoian, S., Maher, J. J., Bugianesi, E. et al., Nonalcoholic fatty liver disease. Nat. Rev. Dis. Primers 2015. 1: 15080.
- 97 Hegde, P., Weiss, E., Paradis, V., Wan, J., Mabire, M., Sukriti, S., Rautou, P. E. et al., Mucosal-associated invariant T cells are a profibrogenic immune cell population in the liver. Nat. Commun. 2018. 9: 2146.
- 98 Sundström, P., Ahlmanner, F., Akéus, P., Sundquist, M., Alsén, S., Yrlid, U., Börjesson, L. et al., Human mucosa-associated invariant T cells accumulate in colon adenocarcinomas but produce reduced amounts of IFNγ. J. Immunol. 2015. 195: 3472–3481.
- 99 Ling, L., Lin, Y., Zheng, W., Hong, S., Tang, X., Zhao, P., Li, M. et al., Circulating and tumor-infiltrating mucosal associated invariant T (MAIT) cells in colorectal cancer patients. Sci. Rep. 2016. 6: 20358.
- 100 Won, E. J., Ju, J. K., Cho, Y. N., Jin, H. M., Park, K. J., Kim, T. J., Kwon, Y. S. et al., Clinical relevance of circulating mucosal-associated invariant T cell levels and their anti-cancer activity in patients with mucosal-associated cancer. Oncotarget 2016. 7: 76274–76290.
- 101 Peterfalvi, A., Gomori, E., Magyarlaki, T., Pal, J., Banati, M., Javorhazy, A., Szekeres-Bartho, J. et al., Invariant Valpha7.2-Jalpha33 TCR is expressed in human kidney and brain tumors indicating infiltration by mucosal-associated invariant T (MAIT) cells. Int. Immunol. 2008. 20: 1517–1525.
- 102 Melo, A. M., O'Brien, A. M., Phelan, J. J., Kennedy, S. A., Wood, N. A. W., Veerapen, N., Besra, G. S. et al., Mucosal-associated invariant T cells display diminished effector capacity in oesophageal adenocarcinoma. Front. Immunol. 2019. 10: 1580.

- 103 Yan, J., Allen, S., McDonald, E., Das, I., Mak, J. Y. W., Liu, L., Fairlie, D. P. et al., MAIT cells promote tumor initiation, growth, and metastases via tumor MR1. *Cancer Discov.* 2020. **10**: 124–141.
- 104 Wallington, J. C., Williams, A. P., Staples, K. J. and Wilkinson, T. M. A., IL-12 and IL-7 synergize to control mucosal-associated invariant T-cell cytotoxic responses to bacterial infection. J. Allergy Clin. Immunol. 2018. 141: 2182–2195.e2186.
- 105 Bharath, L. P., Agrawal, M., McCambridge, G., Nicholas, D. A., Hasturk, H., Liu, J., Jiang, K. et al., Metformin enhances autophagy and normalizes mitochondrial function to alleviate aging-associated inflammation. *Cell Metab.* 2020.
- 106 Nicholas, D. A., Proctor, E. A., Agrawal, M., Belkina, A. C., Van Nostrand, S. C., Panneerseelan-Bharath, L., Jones, A. R. et al., Fatty acid metabolites combine with reduced β oxidation to activate Th17 inflammation in human type 2 diabetes. *Cell Metab.* 2019. **30**: 447–461.e5.
- 107 Zhang, D., Jin, W., Wu, R., Li, J., Park, S. A., Tu, E., Zanvit, P. et al., High glucose intake exacerbates autoimmunity through reactive-oxygen-species-mediated TGF- β cytokine activation. *Immunity* 2019. **51**: 671–681.e5.

Abbreviations: CD: Crohn's disease · IBD: inflammatory bowel disease · iNKT: invariant NK T cells · MAIT: mucosal associated invariant T cells · Mtb: Mycobacterium tuberculosis · MS: multiple sclerosis · mTOR: mammalian target of rapamycin · PsA: psoriatic arthritis · RA: rheumatoid arthritis · SjS: Sjögren syndrome · T2DM: type 2 diabetes mellitus · TCR: T cell receptor · UC: ulcerative colitis

Full correspondence:Dr Andrew Hogan, Institute of Immunology, Maynooth University, Maynooth, Co. Kildare, Ireland e-mail: Andrew.E.Hogan@mu.ie

Received: 20/3/2020 Revised: 25/5/2020 Accepted article online: 3/7/2020